AFIT/GOR/ENS/99M-16

A VALUE FOCUSED APPROACH TO DETERMINING THE TOP TEN HAZARDS IN ARMY AVIATION

Brian K. Sperling

AFIT/GOR/ENS/99M-16

Approved for public release; distribution unlimited

19990409 040

DIIC QUALITY INSPECTED 2

AFIT/GOR/ENS/99M-16

A VALUE FOCUSED APPROACH TO DETERMINING THE TOP TEN HAZARDS IN ARMY AVIATION

THESIS

Presented to the Faculty of the Graduate School of Engineering of the

Air Force Institute of Technology

Air University

In Partial Fulfillment of the

Requirements for the Degree of

Master of Science in Operations Research

Brian K. Sperling

March 1999

Approved for public release; distribution unlimited

THESIS APPROVAL

STUDENT: Brian K. Sperling

Advisor

CLASS: GOR-99M

THESIS TITLE: A Value Focused Approach to Determining the Top Ten Hazards in Army Aviation

DEFENSE DATE: 25 February 1999

COMMITTEE: NAME/TITLE/DEPARTMENT

Jack M. Kloeber Jr., Lieutenant Colonel, USA Assistant Professor of Operations Research Department of Operational Sciences

Richard F. Deckro, DBA Reader Professor of Operations Research Department of Operational Sciences

SIGNATURE

Jack M. How

Acknowledgments

I would like to express my sincere gratitude to my faculty advisor LTC Jack Kloeber. This effort would not have been possible without his patience and guidance. I would also like to thank my reader; Dr. Richard Deckro who always tried to keep me focused on my overall goal. Their insight and enthusiasm for this project was definitely appreciated.

In addition I would like to thank the entire Chain of Command at the U. S. Army Safety Center, particularly the Operations Research and Systems Analysis/Statistics Division. Their support for this undertaking was instrumental throughout the entire project.

Most importantly I want to thank my wife, JoAnna, and my son, Joshua. Without the emotional support that they provided at home this would have been a much more difficult and less enjoyable project. They were always there for me and I thank them from the bottom of my heart.

Acknowledg	gments	ii
List of Figu	res	V
List of Table	es	vii
Abstract		viii
1	Introduction	. 1-1
1.1	General Issue	. 1-1
1.2	Background	. 1-3
1.3	Problem Statement	. 1-5
1.4	Research Objectives	. 1-5
1.5	Research Approach	. 1-5
1.6	Scope of Problem	. 1-6
1.7	Overview	. 1-7
2	Literature Review	. 2-1
2.1	Introduction	2-1
2.2	Decision Analysis	
2.3	Value Focused Thinking	. 2-3
2.4	Multiattribute Preference Theory	2-7
2.5	Value Focused Thinking Application in Current Literature	2-8
3	Methodology	3-1
3.1	Introduction	. 3-1
3.2	Accident Model Value Hierarchy	3-3
3.3	Evaluation Measures	3-6
3.3.1	Degree of injury or illness:	3-7
3.3.2	Loss of or damage to equipment or property:	3-10
3.3.3	Environmental Damage	3-11
3.3.4	Other Mission-Impairing Factors	3-12
3.4	Single Dimensional Value Functions	3-16
3.4.1	Degree of injury or illness:	3-16
3.4.2	Loss of or damage to equipment or property	3-20
3.4.3	Environmental Damage	3-21
3.4.4	Other Mission Impairing Factors (Unit Readiness):	3-22
3.5	Assessing Weights	3-26
3.6	Converting Value (Severity) to Utility (Severity[U])	3-31
3.7	Establishing Accident Risk	3-34
3.8	Hazard to Accident Association	3-35
3.9	Summary	3-36
4	Results and Analysis	4-1
4.1	Introduction	4-1
4.2	Accident Severity	4-1
4.2.1	Degree of Injury/Illness Sensitivity	4-8
4.2.2	Property/Equipment Damage Sensitivity	4-10
4.2.3	Environmental Weight Sensitivity	4-11
4.2.4	Other Mission Factors Sensitivity	4-11
4.3	Accident Risk Ranking	4-12

Table of Contents

4.4	Hazard Severity Ranking	
4.4.1	Sensitivity of Criteria Weights on Hazard Severity	
4.5	Hazard Risk Ranking	
4.6	Summary	
5	Conclusions and Recommendations	
5.1	Conclusions	
5.2	Recommendations	
5.2.1	Value Functions	
5.2.2	Data Collection	
5.2.3	Environmental Data	
5.2.4	Hazards and Controls	
5.3	Contributions	

Appendicies:

Appendix	A: Accident and Hazard Codes	A-1
Appendix	B: Glossary of Safety Related Terms	B-1
Appendix	C: Accident Classification and Investigations	C-1
Appendix	D: Initial Data Set	D-1
Appendix	E: Probabilities and Hazard Severity Calculations	E-1
Appendix	F: Measure Scores	F-1
Appendix	G: Criteria Scores	G-1
Appendix	H: Sensitivity analysis Graphs	H-1
Bibliograp	ohy	Bib-1
~ ~	•	

List of Figures

Ŋ

Figure 2-1	A Decision Analysis Process Flowchart (Clemen, 1996: 6)	2-2
Figure 2-2	Value Hierarchy (Kloeber, Parnell: VFT Brief)	2-4
Figure 3-1	Top Level Criteria of Value Hierarchy	3-6
Figure 3-5-1	Criteria: Degree of Injury or Illness	3-7
Figure 3-3	Criteria: Loss/Damage to Equipment/Property	3-10
Figure 3-4	Criteria: Environmental Damage	3-11
Figure 3-5	Criteria: Other Mission Impairing Factors	3-12
Figure 3-6	SDVF: Lives Lost	3-17
Figure 3-7	SDVF: Partial Disabilities	3-18
Figure 3-8	SDVF: Total Disabilities	3-18
Figure 3-9	SDVF: Davs Hospitalized	3-19
Figure 3-10	SDVF: Injury Dollars	3-20
Figure 3-11	SDVF: Repair/Replace Dollars	3-21
Figure 3-12	SDVF: Fluid Spills	3-21
Figure 3-13	SDVF: Restricted Duty	3-22
Figure 3-14	SDVF: Lost Workdays	3-23
Figure 3-15	SDVF: Man-hours to Repair	3-24
Figure 3-16	SDVF: Total Loss	3-24
Figure 3-17	Complete Hierarchy	3-25
Figure 3-18	Relative Weight Assessment	3-28
Figure 3-19	First Tier Weights	3-28
Figure 3-20	Weights: Degree Injury/Illness	3-29
Figure 3-21	Weights: Damage to Property or Environment	3-29
Figure 3-22	Weights: Other Mission Impairing Factors	3-29
Figure 3-23	Weights: Global	3-30
Figure 3-24	Utility: Damage Cost	3-33
Figure 3-25	Utility: Man-hours to Repair	3-33
Figure 3-26	Utility: Injury Cost.	3-34
Figure 3-27	Methodology Framework	3-37
Figure 4-1	Accident Severity Chart	4-2
Figure 4-2	Accident Severity Composition	4-5
Figure 4-3	Correlation Regression.	4-7
Figure 4-4	Accident Sensitivity: Injury Weight	4-9
Figure 4-5	Accident Sensitivity: Property Damage Weight	4-10
Figure 4-6	Accident Sensitivity: Environmental Weight	4-12
Figure 4-7	Accident Sensitivity: Mission Factors Weight	4-14
Figure 4-8	Accident Severity versus Probability	4-16
Figure 4-9	Expected Hazard Severity	4-15
Figure 4-10	Hazard Sensitivity: Injury Weight	4-19
Figure 4-11	Hazard Severity versus Probability	4-21
Figure 4-12	Hazard Risk Ranking	4-22
Figure 4-13	Hazard Categories.	4-22
Figure 5-1	Environmental Damage Data	5-5

Figure H-	1 Accident Sensitivity: Injury Weight	H-1
Figure H-2	2 Accident Sensitivity: Property Damage Weight	H-1
Figure H-	3 Accident Sensitivity: Environment Weight	H-2
Figure H-	4 Accident Sensitivity: Other Mission Factors Weight	H-2
Figure H-:	5 Hazard Sensitivity: Injury Weight	Н-3
Figure H-	6 Hazard Sensitivity: Property Damage Weight	Н-3
Figure H-	7 Hazard Sensitivity: Environment Weight	H-4
Figure H-	8 Hazard Sensitivity: Other Mission Factors Weight	H-4
-	-	

List of Tables

Table 2-1	NASA's Objectives	
Table 3-1	Lottery Questions	
Table 3-2	Certainty Equivilant	
Table 4-1	Accident Severity Scores	
Table 4-2	Accident Severity (Top 13)	
Table 4-3	Accident Risk Ranking	
Table 4-4	Expected Hazard Scores	
Table 4-5	Accident/Hazard Relationship	
Table 4-6	Hazard Risk	4-20
Table 4-7	Hazard Categories	
Table 5-1	Top Ten Hazards	

Abstract

The United States Army Safety Center is challenged with identifying the top ten most severe hazards in Army Aviation. This research utilizes value-focused thinking and multiattribute preference theory concepts to produce a decision analysis model designed to aid decision-makers in their analysis process. The severity model is based on the Army's Risk Management doctrinal manual and has been tailored specifically for aviation related accidents and hazards. The model determines the severity and risk ranking for 65 categories of accidents and 24 existing hazards. A sensitivity analysis is conducted to examine the effects of variations in the weights of the top-level criteria for accident and hazard severity. Recommendations are presented for ways to use the information contained in this report to assist in developing risk reduction controls focused on force protection. The model provides the decision-makers with a decision analysis methodology that is consistent with Army doctrine and the values of the current chain of command at the Army Safety Center. Furthermore the model can be adjusted for different leadership levels or situations.

A VALUE FOCUSED APPROACH TO DETERMINING

THE TOP TEN HAZARDS IN ARMY AVIATION

1 Introduction

1.1 General Issue

Everyday as we respond to the nation's needs, we expose our soldiers to hazards in uncertain and complex environments. We do this with the full knowledge that there are inherent risks associated with any military operation. The nature of our profession will not allow for either complacency or a cavalier acceptance of risk. General Dennis J. Reimer Chief of Staff, Army

Over the years Army aviation has evolved tremendously from artillery observation to an integrated part of nearly every military operation of the combined arms team. In the last 10 years, largely due to increased capabilities and new technology, the mission of Army aviation has become more complex. Today's operational demands require Army aviation to not only perform with their inherent risks, but to do this in a variety of environments which increases that risk. To maintain their outstanding safety record, save soldiers lives and protect our force requires constant vigilance and the application of sound safety management techniques. One of those safety management techniques used is risk management, a decision-making process designed to minimize the severity and risks of the hazards associated with current military operations.

The United States Army Safety Center (ASC) is the official repository for Army accident data. A difficult problem for the ASC is the identification and ranking of

present and future hazards in Army Aviation. The current analytical methodologies are limited in scope to basic statistical comparisons to identify major safety related hazards risks and controls (FY99 HQDA Army Study Program Proposal, 1998). This tends to lead to analysis and results that are reactive rather than proactive. While it is a fundamental principal of command that *every* soldier's life is valued decisions affecting doctrine implementation, policy changes and training are made based on this type of analysis. Army leadership needs a proactive process to evaluate hazards, risks and controls in order to effectively perform risk management at all levels of the Army (FY99 HQDA Army Study Program Proposal, 1998).

Currently, hazards are ranked by one or a few criteria. For instance, the current Army Aviation database, the Risk Management Information System (RMIS), ranks hazards by the number of accidents which are caused by that hazard, total cost of the accidents or a myriad of other categories taken singularly. This is an ineffective and inconsistent manner for ranking hazards. The current system does not provide decision-makers with enough information to make key, necessary tradeoffs for certain risk reduction decisions nor does it provide a baseline to make comparisons between different accidents or hazards. The current top hazards are based on which criteria are important at a given period of time. These critical criteria may change from month to month or year to year based on public opinion, needs of the Army or other operational factors. The unstable prioritization of the evaluation criteria leads to an inconsistent and possibly unsupportable decision-making process.

1.2 Background

While Army Field Manual 100-14, the army's risk management doctrine, outlines specific guidelines for determining the severity of a hazard, these guidelines are not consistently followed when analyzing hazards (Moon, 1998). Furthermore, the senior leadership of the Safety Center and the Army have expressed concerns that the current methodology does not provide senior decision-makers with an accurate, multi dimensional representation of the most severe hazards that plague the field of aviation. This study was conducted in cooperation with the Safety Center in order to provide commanders, at all levels, with an additional tool to help assess risk with greater accuracy. Although risk management is a command responsibility, commanders must be equipped with the proper tools to analyze and evaluate risks in order to make wellinformed decisions. The goal is to make risk management a routine part of planning and executing all missions (Reimer, 1995).

To ensure the Safety Center's efforts to reduce risk remain correctly focused on aviation's most severe safety issues, the most problematic areas must be identified then their root causes addressed in order to decrease the associated risk. For the United States Army Safety Center hazards are the problematic areas. The senior leadership of the ASC has identified the need to develop a methodology to accurately rank the top hazards in the Army in order to allocate resources to the appropriate areas. The reduction or elimination of hazards falls under the umbrella of Risk Management. Army risk management applies across a wide range of military operations. Field Manual 100-14 explains the "principles, procedures and responsibilities to successfully apply the risk management process to conserve combat power and resources" (FM 100-14, 1998, pg. ii). Doctrinally, the risk

management process is incorporated into the planning of all military operations and training in order to assist in the identification and the assessment of hazards with greater precision. Traditionally, this process is implemented down the chain of command; yet, this process has not been adopted at some of the higher levels of decision-making and alternative evaluation (Warren, 1998). Specifically, the ASC does not account for the values discussed in FM 100-14 to incorporate the risk management process in their resource allocation procedures (Interview, Warren 1998). A result of an evaluation procedure consistent with the Army's values should lead to a more acceptable, consistent, and supportable ranking of hazards. Another application of these results may be a better methodology for allocating resources.

The process of identifying the objectives and associated measures involved an appropriate spectrum of people knowledgeable about Aviation hazards and resources available, including experts both inside and outside of the Army Safety Center. The output from this model is not limited to merely a list of hazards. The same value structure will be valuable for developing a portfolio of controls, designed to reduce hazard severity or risk, while limited by a budget. Several insights for the decision making process are available from the output of this model. The recommendations from this model do not take away from any decision-making authority from the Safety Center. In fact, the ASC's decisions will have a much stronger basis and authority (Keeney, 92) with the analysis actually being done up front. In addition, with "up front" analysis, decisions can be more quickly and more consistently with the values of the current chain of command and Army Doctrine.

1.3 Problem Statement

In order to integrate the Army's risk management process, as outlined the Army's field manual for risk management, and the Director of Army Safety's (DASAF) values into the Army Safety Center's resource allocation procedures, a methodology to more consistently evaluate Army Aviation hazards has been developed. Upon creating this rank order of hazards, resources may then be appropriately and optimally allocated.

1.4 Research Objectives

The objective of this research was to incorporate the values of the Army and its current decision-makers into a systematic, logical decision structure which analyzes existing hazards and develops a 'top ten" hazard recommendation list. A value hierarchy, discussed in Chapter Two, is the basis for developing alternatives and evaluating the worth of different solutions, such evaluation will facilitate making the tough tradeoffs by making them more explicit. Some of the associated sub-objectives are:

1) Structure a quantified model that represents Army Doctrine and the ASC's values with respect to aviation safety.

2) Identify the most severe aviation accidents.

3) Identify the highest risk accidents.

4) Prioritize by severity the hazards causing aviation accidents.

1.5 Research Approach

The foundation on which this research rests is the Director of Army Safety's FY98 Strategic Plan for the U.S. Army Safety Center. Specifically, objective number five: Develop a methodology and process for identifying and alerting the Army of the "top ten" hazards (FY98 Strategic Plan for the U.S. Army Safety Center, 1998).

Additionally, this analytical model incorporates the first three steps of the risk management process as outlined in Chapter Two of FM 100-14 (Risk Management).

Step 1. Identify hazards.

Step 2. Assess hazards to determine risks.

Note: The following steps will not be fully developed during this research project but they deserve to be mentioned here. Steps three and four will be addressed briefly and have been recommended as subjects for further development to the ASC. The methodology developed will promote improved controls and the generation of better controls. Controls are those steps taken or policies implemented to eliminate hazards or reduce their risk (FM 100-14, 1998).

Step 3. Develop controls and make risk decisions.

Step 4. Implement Controls (FM 100-14, 1998, p. 2-0)

Step 5. Supervise and Evaluate.

1.6 Scope of Problem

To accomplish the near term objectives of this study the scope of the research will be restricted in the following areas:

- 1. The study will be limited to analyzing hazards associated with rotary winged aircraft within Army Aviation.
- 2. Accident categories A, B and C will be the only accident categories used to gather data. Lower classes of accidents do not contain enough information to allow full evaluation.
- 3. Research will be limited to hazards which have produced mishaps, potential hazards will not be considered at this time.
- 4. The overall ranking of hazards will use data collected over the last eleven years. This time period best represents the tactics, techniques and aircraft types currently used in Army Aviation.

5. Currently, data is being transferred from the <u>Aviation Safety</u> <u>Management Information System (ASMIS) older database to a newer,</u> more user-friendly system the <u>Risk Management Information System</u> (RMIS). At this time, the upgrade does not contain the data necessary to complete this research; the research was therefore based on data taken from the ASMIS database.

1.7 Overview

Chapter Two briefly covers some of the fundamentals of decision analysis that will be applied throughout this research effort and presents some of the current related literature. Chapter Three describes the methodology used to develop the value hierarchy, to include its associated measures, functions and weights, in addition to associating hazards with accidents to determine their severity. Chapter Four discusses the data used and the results obtained along with presenting a sensitivity analysis of the criteria in the value hierarchy. Chapter Five discusses overall conclusions and recommendations for follow-on-work to be conducted at the Army Safety Center. Data used and pertinent definitions are also provided in the enclosed appendices.

2 Literature Review

2.1 Introduction

Sizing up opponents to determine victory, assessing dangers and distances is the proper course of action for military leaders. Sun Tzu, The Art of War

This chapter briefly introduces the reader to the decision analysis concepts used in this research. This discussion will include, at a minimum, the concepts and basis for value focused thinking which is used in this research extensively. Furthermore, some examples of literature that discuss the methods presented in this paper as applied to decision analysis problems in recent history will be presented.

2.2 Decision Analysis

While each individual decision has its own special set of issues, there are four basic sources of difficulty in making a decision. Decision analysis can help the decisionmaker with all of them. A decision may be considered difficult due to:

1. Complexity.

- 2. Uncertainty in the situation.
- 3. The decision-maker may be working toward multiple objectives.

4. A situation where different perspectives may lead to different conclusions

(Clemen, 1996: 3)

By using decision analysis and applying different techniques for different situations better decisions can be attained. Better decisions do not necessarily mean better outcomes; a well-structured and well thought out decision may lead to poor outcomes. Decision analysis provides "structure and guidance for systematic thinking in difficult situations" (Clemen 1996: 4). Decision analysis is not designed to do the decision-

maker's job for him or her, yet it should provide the necessary structure for a well-

informed decision. Derek Bunn, author of Applied Decision Analysis writes:

[t]he basic presumption of decision analysis is not at all to replace the decisionmaker's intuition, to relieve him or her of the obligations in facing the problem, or to be, worst of all a competitor to the decision maker's personal style of analysis, but to complement, augment, and generally work alongside the decision maker in exemplifying the nature of the problem. Ultimately, it is of most value if the decision-maker has actually learned something about the problem and his or her own decision-making attitude through the exercise (Bunn, 1984: 8).

Figure 2.1 depicts the systematic nature of the decision-making process.





Once the decision situation is identified and the objectives are understood, the problem can be structured. There are a series of example questions that might be asked during the different phases of the process and some expected deliverables. These are depicted in Figure 2.1. Proper time and care should be given to the formulation of the

problem to ensure the necessary values and alternatives are identified. A deterministic and probabilistic analysis should be conducted along with a sensitivity analysis to determine the effect that uncertainty has on final recommendations. At this point, a solution may be recommended or further analysis may be deemed necessary (Clemen, 1996: 6).

The process of decision-making, when dealing with multi objective problems is indeed complicated. This is due to the inherent complexity of these problems and because humans commonly rely on intuition to solve problems and make decisions. Intuition can fail when the decision-maker must make tradeoffs between conflicting objectives, account for uncertainties and dependencies between measures and possibly account for scaling problems within measures used (Kirkwood, 1997: 2). The Army places a great deal of confidence in the problem solving abilities of its senior decisionmakers. These individuals regularly deal with complex multi-objective problems. Intuition and past experience are used to balance choices, facts, available information and preferences to arrive at a logical decision for small or straightforward problems. As the problem and the decision context increase, certain decision-making theories and practices can assist these decision-makers in making logical and consistent decisions.

2.3 Value Focused Thinking

There are two basic thought processes that deal with the structure of decision analysis problems, these are alternative focused and value focused thinking. Alternative focused thinking, throughout the literature, has been deemed the traditional approach and is characterized as "reactive not proactive" (Keeney, 1992: 33). Alternative focused thinking develops alternatives to solve the problem by considering alternatives. Focusing

on the alternatives may sometimes cloud the issues at hand. A decision-maker expends his or her precious time evaluating alternatives rather than considering what is wanted, or "valued", from the decision. The focus remains on the choices not what the choices *should* be.

The second manner in which problems may be structured is value-focused thinking (Keeney: 1992). Value focused thinking provides the analyst with a means to present recommendations to a decision-maker that are logical and consistent with a certain set of values. That set of values should be based on the organization's or individual's goals and objectives. The set of values obtained is organized into a hierarchy with the overriding strategic objective, or goal, at the apex and supporting objectives (criteria), sub-objectives (sub-criteria) and measures of merit as depicted below.



Figure 2-2 Value Hierarchy (Kloeber, Parnell: VFT Brief)

This value hierarchy is usually obtained by one of two methods; the gold standard or the silver standard (Kloeber, Parnell, 1998: VFT Brief). The method used is not necessarily

a choice rather the decision context and information available guides you towards your

technique. The development of a value hierarchy is summarized below:

- Identify sub-objectives required to achieve objectives
- Identify measures of merit that quantify the value of achieving sub-objectives
- Assign weights to the sub-objectives and objectives

"The desirable properties for a value hierarchy should be completeness, non-redundancy,

decomposability, operability and small size" (Kirkwood, 1997: 16).

- Completeness: Each tier must adequately cover all concerns necessary to evaluate the overall objective.
- Nonredundancy: No two evaluation considerations, which are in the same tier, should overlap.
- Decomposability (Independence): The preference of one evaluation consideration should not depend on the level of another.
- Operability: The hierarchy must be understandable by the person or group that will use it.
- Small size: A small hierarchy is generally easier to use, this property should not interfere with maintaining the other properties.

The gold standard development is a deductive method that takes existing standards, visions, or objectives and expands them. These standards, visions, and objectives can be found in a variety of places. Steps certainly needed to be taken to develop the rest of the model, but receiving initial buy-in of the gold standard makes further development much easier to present to decision-makers. Each value will be clearly defined and broken down further until it is explicitly measurable.

This manner of developing a value model is not always possible. Military doctrine, standing procedures or an explicit vision statement may not exist for the

decision opportunity at hand. In such a case the silver standard development may be used. This is an inductive method where experts in the specific field under investigation develop a new value definition. This starts off with a brain storming session concentrating on the use verbs. This method provides new insights but requires many more participants and substantially more time to develop and establish support for. Once the brainstorming session is complete the experts form groups of like tasks and build their model from the bottom up instead of the top down (as in the gold standard development). The Air Force's 2025-value model was developed in this manner (AF 2020 An Operational Analysis, 1997)

The decision analysis cycle can be broken down into thee distinct phases: deterministic, probabilistic, and informational phase as depicted in Figure 2.1. In the first phase, deterministic, we are primarily concerned with developing our problem and our model. Here uncertainty is disregarded. By simplifying the form of our input data we can ensure that the structure of our problem will provide the output we want. That is not to say that we either agree or disagree with that output, we are more concerned that the output is of the correct form and provides meaningful information to the decision at hand. To ensure that an accurate model is developed, the value model must have a solid base with measures be well defined. Furthermore, although the value hierarchy specifically represents either the organization's or an individual decision-maker's internalized values, the range and utility of each measure must be accounted for. This can be accomplished through a series of interviews with the decision-maker. Misrepresentation of the value or utility curves of individual measures will skew the overall results. The model will still be consistent, yet may not accurately portray the best options. The effect of slight changes

in measurements may or may not have a prominent effect on the outcome; this can be examined in detail during the information phase with sensitivity analysis.

The probabilistic phase accounts for the uncertainty in the influential variables of our problem. The importance of this phase varies from problem to problem. Obviously the more uncertainty that is actually in the problem the more important the probabilistic phase. Some decisions can actually be made, with confidence in the recommendations, without incorporating uncertainty. This phase also allows the analyst to account for the risk tolerance of the decision-maker. By incorporating the uncertainty a wide range of analysis is available that is not possible in a purely deterministic model.

The last phase is the informational phase. Once model development is complete and uncertainty is included, it is time to organize the output to provide meaningful recommendations to the decision-maker. Sensitivity analysis plays a major role in this phase. By varying measures, weights, distributions or other variables that might provide insight, we can determine where a change in the decision policy should occur. Furthermore we can determine which measures the decision policy is sensitive to and decide which measures to gather more information on or probabilistically model. This becomes extremely valuable, especially if there is conflict surrounding the weighting of different values. In short, if the policy is not sensitive to a certain measure, the weight does not matter. Each one of these phases is used in this research and will be discussed in detail in Chapter Three.

2.4 Multiattribute Preference Theory

In order to compare alternatives using a value hierarchy (multiobjective value analysis) it is necessary to develop a model that combines all of the evaluation

considerations into on dimensionless score. Determining this model requires that single dimensional value functions be specified for each evaluation measure (Kirkwood, 1997: 53). These functions are determined through an interview with the decision-maker or his designated representative. The function should represent the way the decision-maker and/or the organization feels about different increments in value throughout the range of the evaluation measure. When incorporating uncertainty the functions are developed differently and are now referred to as utility functions. The difference in the formulation of the utility function is that the decision-maker now answers lottery type questions about each of the evaluation measures that contains uncertainty. The development of each of the functions for this research effort is discussed in Chapter Three.

The overall value function, which combines the values of all the single dimensional value functions, may be an additive value function. The additive value (or utility) function is a weighted average of all the single dimensional value (utility) functions within the value hierarchy (Kirkwood, 1997: 230,248).

2.5 Value Focused Thinking Application in Current Literature

This section provides some selected applications where value focused thinking was used to solve large-scale problems. Each problem is completely different in goals, stakeholders and specific problems. The common thread is that in each situation the fundamental objectives are identified and the decision analysis is not focused on the alternatives.

The National Aeronautics Space Administration was concerned with the selection of future space missions. The choice of space missions to accept is extremely complex.

There are numerous stakeholders involved with multiple uncertainties, competing objectives and numerous uncertainties. The foundation on which the mission selection process was developed was the objectives, or goals. NASA identified and prioritized their objectives as shown below.

Objective	Ranked	Relative
Enhance National Pride	1	100
Aid national defense	9	20
Promote international prestige	8	35
Foster international cooperation	7	40
Create economic benefits	5	50
Advance scientific knowledge	4	60
Promote education	6	45
Provide excitement and drama	2	90
Maintain fiscal responsibility	3	70

Table 2-1 NASA's Objectives (Keeney, 1998)

The possible mission alternatives were compared against the above criteria. The missions that were then ranked in priority order. Relative tradeoffs were discussed and paired comparisons were done between alternatives. One of the aspects of value focused thinking that was very valuable in this situation was how the multiple stakeholders were involved in the decisions. Furthermore, the stakeholders objectives were integrated into a combined hierarchy.

Multiattribute utility analysis has been used as a framework for public participation when dealing with environmental issues. In particular Sandia National Laboratories (SNL) used this approach to involve stakeholders within the local

community where a corrective action management unit (CAMU) was to be erected. The approach allowed technical experts to explain the essential technical considerations while allowing the local stakeholders to take part in establishing the value judgments made for the decision. This situation utilized a power-sharing concept and allowed free access to information. Furthermore this method was able to transfer technical competency to the public. The MUA approach enabled stake holders to participate effectively, even though they had limited understanding of all the technical details (Merkhofer, 1997: 838).

The Department of Energy has used value focused thinking extensively. Part of the Department of Energy's (DOE) program to evaluate alternatives for managing spent fuel was based on VFT (Keeney, 1998). Furthermore, in an AFIT thesis by Brian Grelk, <u>A CERCLA-Based Decision Support System For Environmental Remediation Strategy</u> <u>Selection</u>, the author develops a set CERCLA based evaluation measures and their corresponding single dimensional value functions to recommend spill remediation strategies. The uses of value focused thinking by the Department of Energy are clearly focused on public safety and regulations.

Numerous agencies throughout the Department of Defense have done quantitative and qualitative studies on the severity and probability of hazards and accidents. The ASMIS database was used by ANACAPA Sciences during a study of aircraft component failure. Analyses were conducted to determine the feasibility of extracting failure data from the existing database and calculating failure rates from that data. The analysis identified parts with relatively high failure rates. This data has been used to identify problematic components or parts, to monitor trends and to develop countermeasures to reverse part failure rates (ANACAPA, 1993:6).

The System Safety Risk Assessment Manual, developed at the Naval Safety Center, was prepared to assist system safety managers in making informed risk management decisions. The information was drawn from MIL-STD-882B, Navy Safety School Course Material and other information sources. It compiled several of the more common ways of defining safety risk. This report also introduced some new safety management concepts referred to as relative worth index and a safety performance baseline. The relative worth index takes five weighted factors to determine the relative magnitude of risk (or impact on the Navy) from the loss of a certain aircraft. A performance baseline establishes system and subsystem baselines, based on mishap rates. The research described in this report seems to touch on some of the concepts of value focused thinking but does not explain the theory behind its calculations nor does it cover the development or support of the evaluation criteria. The intended users for this report are the system safety and program managers (Kinzey, 1989:13, 24).

Based on the wide range of applications found in current literature, although no specific reference was found that explicitly describes the use of value focused thinking and multiattribute preference theory applied to aviation safety, the approach is certainly valid.

3 Methodology

3.1 Introduction

Risk Management is not an add-on feature to the decision-making process but rather a fully integrated element of planning and executing operations...Risk management helps us preserve combat power and retain the flexibility for bold and decisive action. Proper risk management is a combat multiplier that we can ill afford to squander.

> General Dennis J. Reimer Chief of Staff, Army 27 July 1995

A value focused thinking and multi attribute value theory approach to decision analysis is applicable to a wide range of problems. This is a valuable and structured manner by which a decision-maker can deal with problems involving various stakeholders with multiple competing objectives that require tradeoffs. Decision analysis is commonly used to make better tactical and strategic decisions that are not merely routine. Examples include business decisions involving long term and short term financial tradeoffs, balancing the tradeoffs between the efficiency of a manufacturing process versus the impact on the environment and even personal decisions involving career opportunities, employment, or purchases. Better analysis can be done and better decisions can be made using value focused thinking and multi attribute value theory. Furthermore, the decision-maker can explain the reasons for the decision easier and in a more structured fashion (Kirkwood, 1997: 1).

After determining the exact nature of the U. S. Army Safety Center's decision context and problem, and examining the analysis techniques available, a value-focused approach to the multiattribute preference problem seemed the most logical technique to use. When

examining the hazards and accidents there was no clear "winner" or "loser". No one accident or hazard, nor group of hazards, was easy to label as the most severe. The tradeoffs involved were numerous. Some of the areas of concern included cost incurred. soldier and civilian's lives lost, training time lost, and maintenance time allocated. In addition to these competing objectives there are numerous stakeholders that are affected by this ranking of hazards; the soldiers flying the aircraft, the unit deploying to war and the leadership of the Safety Center trying to implement risk management policies throughout the Army to mention a few. There is currently no structured method in place used to analyze hazards based the multiple objectives and various stakeholders mentioned above. Value focused thinking provides a base of evaluation to enable the Safety Center to make logical and consistent assessments of hazards using the current values of the Army. Involving the stakeholders at risk, and using the current values and doctrine of the Army to balance the tradeoffs between the various competing objectives, will make it easier to succeed at getting every one involved working toward a common goal and attaining a consensus on the outcome of the research.

The research effort was broken into three distinct phases: the deterministic, probabilistic and informational. In the deterministic phase, the accident value model is developed, data is collected and manipulated, and a dry run of the model is performed to see if the output from the model is the type of output required. During the probabilistic phase, accidents are associated with hazard occurrences to determine hazard severity and accident/hazard risk. In the informational phase it is time to organize the output to provide meaningful recommendations to the decision-maker. Sensitivity analysis plays a major role in this phase.

This chapter will cover in depth the development of the accident value hierarchy to include the evaluation measures and single dimension value functions, the methods of assessing weights and the application of the additive value function and power additive utility function. Furthermore, the accident severity scores will be combined together with their associated probabilities to attain a hazard severity and a hazard risk score

3.2 Accident Model Value Hierarchy

A hazard is "any actual or potential condition that can cause injury, illness, or death of personnel, damage to or loss of equipment, property or mission degradation" (FM 101-14, 1998: G-1). A hazard may or may not produce an accident (or mishap). Commonly, if a hazard is present in an operation, goes unchecked, and does not cause an accident, that hazard will go unnoticed and unrecorded. The information available for this research purely dealt with hazards that have resulted in accidents, for which the Safety Center maintains detailed records as well as the frequency of those accidents. The basis for evaluating a hazard rests in the severity of the accidents it causes. Therefore, the severity of an accident must be assessed to determine the severity, and risk, of the hazard that caused that accident.

In order to avoid confusion, the following definitions are provided in this section (as well as Appendix A).

- 1) Severity: The expected consequence of an event (hazardous incident) in terms of degree of injury, property damage, or other mission impairing factors that could occur (FM 101-14, 1998: G-3).
- 2) Risk: Chance of hazard or bad consequences; the probability of exposure to chance of injury or loss from a hazard; risk level is expressed in terms of hazard probability and severity (FM 101-14, 1998: G-2).

- 3) Accident: An unplanned event that causes personal injury or illness, or property damage (Army Regulation 385-40).
- Hazard: Any actual or potential condition that can cause injury, illness, or death of personnel, damage to or loss of equipment, property or mission degradation (FM 101-14, 1998: G-1).

Although the above definitions appear descriptive, during the development of this model there was much discussion with experts at the Safety Center concerning what constituted accidents and hazards. For the purpose of this study, "events" (hazardous incidents) that are contained in the Safety Center's current database are referred to as accidents and "system inadequacies" are referred to as hazards. Further research will be done to develop a better taxonomy to determine hazards.

In order to determine the severity of the accidents currently in the database, the attributes by which severity is measured needed to be determined. Value focused thinking is an organized and logical method to accomplish this and is the method used in this research. The general principle of value focused thinking is to discover all of the information useful in guiding one's decision" (Keeney, 1992: 23). A value hierarchy, developed properly, should "indicate everything [you] really care about in a decision context (Keeney, 1992: 23). The most preferred method of developing a value hierarchy is by utilizing a gold standard, if one exists. The gold standard utilizes existing values or standards of an organization and develops measures for those values in a top to bottom fashion. Army doctrine contains specific criteria by which the severity of a hazard should be assessed. FM 100-14 is the Army's official risk management manual and this set of criteria was established by the Safety Center as the standard criteria to be used to evaluate hazards. To use this set of criteria to determine hazard severity each category

had to be broken down into more explicit measurable parts. This was done by starting at the top objective and developing measures down to the lowest level. This method of moving from the top to the bottom is appropriate when the possible alternatives are unclear (Kirkwood, 1997: 21). In the case of this research, accidents, not alternatives, are evaluated. There is a wide range of accidents, the full range of which is unknown. A top down method is better suited for this type of situation. The Gold Standard method also makes attaining buy-in from the stakeholders in the organization much easier.

In the case of risk management, Army doctrine provides the gold standard. The U.S. Army relies on doctrine to set forth the standards and methods by which the officers and enlisted soldiers conduct their daily operations. The Field Manual 100-14: <u>*Risk*</u> <u>*Management*</u> printed in April of 1998 is the Army's doctrinal manual concerning Risk Management. This manual covers a wide range of Army operations and can be applied to almost all aspects of military life including Army Aviation. It explains the "principles, procedures, and responsibilities to successfully apply the risk management process to conserve combat power and resources"(FM 100-14, 1998: ii). This research focuses specifically on the second step of the risk management process; assess hazards to determine risks. FM 100-14 specifically outlines the terms in which the severity of an accident can be expressed:

- Degree of injury or illness
- Loss of or damage to equipment or property
- Environmental damage
- Other mission impairing factors such as loss of combat power

When assessing the degree of severity of an accident, doctrine confirms that the severity may be "based on knowledge of the results of similar past events" (FM 100-14, 1998: 2-9). This is the approach taken in this research effort. This set of criteria develops the first tier of the value hierarchy shown below.



Figure 3-1 Top Level Criteria of Value Hierarchy

The standards set forth in doctrine, interviews with risk management experts and accident investigation reports verified that these four criteria were collectively exhaustive and mutually exclusive. This means that the evaluation considerations in each layer, taken as a whole, include everything needed to evaluate each accident and nothing necessary to do the assessment are included in more than one criterion (Kirkwood, 1997: 17). Next, the evaluation criteria for each of the criteria needed to be developed. If there exists a good measure for the fundamental objective, that measure will be used. Otherwise, the fundamental objective will need to be broken down into more explicit measurable parts.

3.3 Evaluation Measures

This section will introduce the reader to the development of the Accident Severity Sub-Criteria and the metrics used to measure these sub-criteria. Some of the criteria, subcriteria, and measures are self-explanatory but they are all discussed below. The evaluation considerations are broken down further into the metrics that will be used to

measure each of the areas of concern. Measures can be classified as either one of the scales below:

- Natural: In general use with a common interpretation by everyone.
- Constructed: Developed for a particular decision problem to measure the degree of attainment of an objective.

They are further classified as one of the following:

- Direct: Directly measures the degree of attainment of an objective.
- Proxy: Reflects the degree of attainment of an associated objective(Kirkwood, 1997: 24).

Assessment of the severity of an accident does not account for the probability of occurrence. This is merely the expected consequence of an accident in terms of degree of injury, property/environmental damage, or other mission impairing factors that could occur (FM 101-14, 1998: 2-7). The considerations below have been agreed upon within the Safety Center, and are supported by doctrine, as proxy measurements for accident severity. To determine the severity of each accident the evaluation criteria presented in the following sections was used.

3.3.1 Degree of injury or illness:



Figure 3-2 Criteria: Degree of Injury or Illness

This criterion is designed to measures the severity of the injuries that occur in each type of accident. The considerations that were taken into account for the severity on an injury were focused on the impact on the individual, the unit and the Army as a whole. According to Army Regulation and Department of the Army Pamphlet 385-40, Field Manual 100-14, and the senior leadership of the USASC the evaluation considerations below adequately cover all the concerns necessary to evaluate the degree of the injury.

3.3.1.1 Loss of Life

This is a natural-direct measure scale that assesses the number of lives lost during an accident. All fatalities related to an accident are included, civilian and military personnel.

3.3.1.2 Permanent Disabilities

There are two categories of permanent disabilities accounted for in the ASMIS database and this study, they are discussed in the following two sections.

3.3.1.2.1 Total Disability

This is a natural-direct measure scale that assesses any nonfatal injury or occupational illness that, in the opinion of a competent medical authority, permanently and totally incapacitates a person to the extent that he or she cannot follow any gainful employment. (The loss or loss of use of both hands, feet, eyes, or any combination thereof as a result of a single accident will be considered as a permanent total disability (DA PAM 385-40, 1994: 143).
3.3.1.2.2 Partial Disability

This is a natural-direct measure scale that assesses any injury or occupational illness that does not result in death or permanent total disability but, in the opinion of competent medical authority, results in the loss or permanent impairment of any body part, with the following exceptions:

 $\underline{1}$ Loss of teeth.

<u>2</u> Loss of fingernails or toes.

 $\underline{3}$ Loss of tip of fingers or tip of toe without bone involvement.

<u>4</u> Inguinal hernia, if it is repaired.

<u>5</u> Disfigurement.

<u>6</u> Sprains or strains that do not cause permanent limitation of motion (DA PAM 385-40, 143).

3.3.1.3 Time Incapacitated

This is a natural-direct measure scale that assesses the number of days hospitalized, where hospitalization is defined as "admission to a hospital as an inpatient for medical treatment" (DA PAM 385-40, 1994: 143).

3.3.1.4 Injury Cost

This is a natural-direct measure scale that assesses the total cost of the injury to the government in dollars. An accident investigation board, in coordination with competent medical authority determines the cost. The cost determined by the board is the official and the only cost used for the accident investigation report. The cost figure

includes the cost of pay while away from work, medical treatment, hospitalization, dependent survival, unused training costs, gratuities, compensation, disability retirement, and burial (AR 385-40, 1994: 8).

3.3.2 Loss of or damage to equipment or property:



Figure 3-3 Criteria: Loss/Damage to Equipment/Property

There are some issues that need to be investigated further to determine whether the loss of secure equipment and certain types of equipment losses can be tracked and measured accurately. These proposed measures are presented at the conclusion of this paper. At this time this evaluation criteria contains only one measure that sufficiently represents the fundamental objective.

3.3.2.1 Repair/Replacement Cost

This is a natural-proxy measure scale that assesses the total cost of the accident with respect to equipment or property. This cost is determined by the accident investigation team and reported in an estimated cost of damage (ECOD) found on the final accident report. The cost computation criteria are found in Army Regulation 735-11 and include such things as actual costs of new or used parts or materials and labor costs (usually

estimated). When damaged equipment cannot be replaced, the cost reported will be the acquisition costs (AR 385-40, 8). The cost determined by the accident investigation board is the official and only cost used to determine the repair or replacement cost for Army equipment.

3.3.3 Environmental Damage





This area of concern considers the severity of the damage to the environment. This is the only evaluation consideration for this criterion. Although the Army's Risk Management doctrinal manual includes environmental damage as one of the top four criteria for determining the severity of an accident, there is little data available to facilitate an accurate measure of the damage caused by an accident. For this criteria a proxy measure scale was developed that used available data to determine some type of rating system for damage to the environment. This criteria contains three measurements: gallons of oil spilled, gallons of fuel spilled, and gallons of hydraulic fluid spilled. At this time, the measures for the severity of environmental damage are not collectively exhaustive. This research has been conducted using data from the past ten years; no additional data

concerning environmental damage was collected. Different data collection procedures may be initiated in the future by the Safety Center to collect more detailed information concerning the environment. While this is an area for future research, it is beyond the scope of the current study.

3.3.4 Other Mission-Impairing Factors



Figure 3-5 Criteria: Other Mission Impairing Factors

This criteria was designed to give the planner, or the person assessing the risk of a certain mission, the latitude to tailor the assessment specifically towards his mission. Mission-impairing factors for an infantry squad might be different than those of an aviation brigade although some may be the same. For example, heavy ground fog or heavy flooding may impair an infantry squad's river crossing but not an aviation brigade's deep attack on an enemy communication's sight. On the other hand unit morale is an area of concern for all levels of military operations. Hence, ground conditions may be included in the infantry squad's mission impairing factors and not in the aviation brigades while unit morale would be included in both value hierarchies.

This research is limited to Army Aviation Rotary Winged accidents. Unit readiness is the overall criteria. Unit morale and equipment replacement time are the two sub-criteria that are a proxy for the level of unit readiness. These two criteria combined represent how well the unit will perform and how well their equipment is prepared. Throughout this report, when referring to unit readiness, the unit is a company size element. A Company element ranges from approximately forty soldiers and 8 aircraft in an AH-64 equipped Attack Company to 100 soldiers and 24 aircraft in a UH-60 equipped Assault Company. For a company size element, with the data available, this evaluation consideration is collectively exhaustive and mutually exclusive. Recommendations have been made for further research to further develop this area of concern. Most notably, the Safety Center is concerned with the loss of Special Mission Aircraft but there is currently no way to measure the impact on a specific unit that looses one of these aircraft.

3.3.4.1 Unit Morale

The morale of a unit, historically, is correlated to the performance of a unit and therefore, unit readiness. As a soldier's time on restricted duty and lost workdays increase, a strain is put on the other members of the unit. For a short period of time the other members of the company can compensate for the loss of an individual, but as time goes on and people become tasked with unfamiliar jobs along with their normal duties, morale will tend to decrease. This measure assumes that all jobs are equal. Obviously, the loss of a Company Commander or a Senior Instructor Pilot will have a greater affect on the unit than a newly assigned pilot. Personal experience and interviews with various commanders and the leadership at the ASC determined that the measures below cover the

concerns to adequately evaluate unit morale. There are other indicators of unit moral (i.e. mission rate, commander's performance) but these are not associated with this study.

3.3.4.1.1 Days of Restricted Duty

This is a natural-direct measure scale that assesses the number of days that an individual (or all individuals involved) are unable to perform their normal duties (i.e. light duty, profile, grounded from flight) (DA PAM 385-40, 142)

3.3.4.1.2 Lost Workdays

This is a natural-direct measure scale that assesses cases in which an accident results in Army personnel missing one or more days of work. Days away from work are those workdays (consecutive or not) on which Army personnel would have worked but could not because of injury, occupational illness, or job relates physical deficiencies detected during medical surveillance examinations. Excluded are days that Army personnel would not have worked even though able to work (i.e. weekends or holidays) and the day of the injury or onset of the occupational illness (DA PAM 385-40, 141).

3.3.4.2 Equipment Replacement Time

Equipment readiness is a large contributor to unit readiness. The two measures below represent the time the unit will be without a specific piece of equipment, in this case an aircraft.

3.3.4.2.1 Man-Hours to Repair

This measure assesses the direct man-hours required to restore the aircraft to serviceable condition if it is economically repairable. A unit's operational readiness rate is

measured partially by the down time of aircraft within the unit, man-hours to repair the equipment is a proxy measurement for down time. A proxy measurement is required because aircraft down time while recorded and maintained at the unit level, was unavailable for this research. The hours are originally estimated using technical manual estimates. When work is complete, they are reported back on an Equipment Inspection and Maintenance Worksheet (DA Form 2404). The direct man-hours are a summation of the following:

 $\underline{1}$ The cumulative (estimated) man-hours required to remove, repair, and replace damaged aircraft assemblies, subassemblies, or components.

 $\underline{2}$ Man-hours expended in removing and replacing undamaged aircraft components in order to remove, repair, or replace damaged components.

 $\underline{3}$ Man-hours required to remove and replace a part that is not economically repairable.

4 Man-hours expended to determine damage amount (AR 385-40, 8).

3.3.4.2.2 Aircraft Total Loss

If the aircraft is deemed not economically repairable, it is determined to be a total loss. Depending on the type of aircraft and the number of that specific aircraft available to the unit, it may or may not be replaced. Regardless of the outcome, both cases normally result in a large time delay with regards to returning an aircraft back to the unit in a fully operational condition.

Doctrine, Army regulations and the Operations Research/Statistics

(ORSA/STATS) Division of the Army Safety Center support each of the evaluation

considerations presented. Aside from some concerns that have been discussed above, the

evaluation considerations have been developed in sufficient detail, are mutually exclusive.

3.4 Single Dimensional Value Functions

The previous section presented a summary of each evaluation measure; this section will cover the procedure for determining the piecewise linear single dimension value functions developed in this study. The range specified for each measure represents the range of the expected values of all accidents for that measure. To determine the value functions, relative value increments needed to be specified between each of the possible evaluation measure scores (Kirkwood, 62). The severity functions were developed during an interview with LTC Oren Hunsaker, the Operations Research and Statistics Division Chief at the U.S. Army Safety Center. LTC Hunsaker is an Army Aviator with over 18 years experience in the field of Army Aviation.

3.4.1 Degree of injury or illness:

3.4.1.1 Loss of Life

The range of the number of lives lost across all accidents varies from 0 to 17. As soon as an accident causes 1 fatality the severity function increases to a severity of .5. After the first fatality the function becomes linear which represents that the severity increment of each additional life is equal. The decision-maker expressed a zero tolerance for loss of life. Still, he felt there should be a substantial increase in the degree of injury when one fatality versus zero fatalities occurs. Therefore the severity increment given to an accident when one fatality occurs is more drastic than when the number of fatalities increases from one to two.



3.4.1.2 Permanent Disabilities

Aside from a fatality, the most serious type of injury is a permanent disability (AR 385-40). There are only two classes of permanent disabilities tracked by the Safety Center these are partial and total. The greatest severity increment is associated with the first occurrence of a partial disability. As represented by the Figure 3.7 below, the severity increment from zero partial disabilities to one partial disability is .25. After this initial jump the severity increment stays linear throughout the remainder of the range.

Figure 3.8 represents the total disabilities. The maximum total disabilities caused by any one accident is two. The decision-maker indicated that the severity increment for the occurrence of the first total disability is three times as great as the second. This does not indicate that the first soldier injured is more important, rather this indicates that the severity of the accident being evaluated will increase greater with the first occurrence of a disability than it will the second.



Figure 3-8 SDVF: Total Disabilities

3.4.1.3 Time Incapacitated

Other than cost, days hospitalized is the last measure of the severity of an injury. The function below represents the number of days an individual involved in the accident was hospitalized, in accordance with the definition in section 3.3. The range of scores from the current database is from 0 to 248 days (the maximum amount of days an accident victim was hospitalized).

The consideration taken into account during the interview was the impact on both the individual and the organization. Based on the military experience of the Chief of the ORSA/STATS Division at the Safety Center, he felt that the first two weeks of hospitalization causes the greatest relative impact on the victim, the family, and the unit. After the first two weeks the function represents an equal increase in severity for each additional day hospitalized.



3.4.1.4 Injury Cost

The last metric used to determine the degree of injury is the cost of the injuries caused by the accident (AR 385-40, 4). The basis for the severity increments depicted in the function below are the monetary increases that contribute to the classification of an Army accident. For a discussion of the Army mishap classification system refer to Appendix C. The decision-maker felt the severity increment from going from a Class B accident to a Class A accident was 5 times greater than the value increment moving from

a class C, D or E to a Class B. As the dollar amount nears one of the thresholds the slope

of the severity function begins to increase rapidly.

a) The accident classification increased from Class E to a Class B (\$200,000- \$1 Million) or

b) The accident classification increased from a Class B to a Class A (greater than \$1 Million)



Figure 3-10 SDVF: Injury Dollars

3.4.2 Loss of or damage to equipment or property

The severity function for the repair/replacement dollars was developed in a similar fashion to the injury cost above. The classification of accidents also depends on the cost of the damage as shown below:

1) <\$200,000: Class C and below

2) \$200,000-\$1 Million: Class B

3) Greater than \$1 Million: Class A

Due to the large range of damage cost, most of the thresholds occur near the lower end of the range but the severity continues to increase linearly as the cost increases once the Class A threshold has been surpassed. This linear increase enables the model to distinguish between different cost levels of Class A accidents.



Figure 3-11 SDVF: Repair/Replace Dollars

3.4.3 Environmental Damage

The Severity Function for gallons of fuel, oil, and hydraulic fluid spilled in an accident are all the same. The data available was categorized in bins:



Figure 3-12 SDVF: Fluid Spills

1) 0-1 gallons

2)>1-2 gallons

3) >2-10 gallons (at or below this level is unit level clean-up, minor spilled)

4) >10-20 gallons (Must be reported locally, local fire department clean-up; procedures vary depending on local Standing Operating Procedures)

5) >20 gallons (Intermediate to Major Spill, EPA involvement) (FM 20-401).

The functions are based the level of involvement of outside agencies, the amount of clean up required and the category of the spill (i.e. minor, intermediate, major).

3.4.4 Other Mission Impairing Factors (Unit Readiness):

3.4.4.1 Unit Morale



3.4.4.1.1 Days with Restricted Duty



The severity increments in this function were found by determining the impact on the organization when a soldier has restricted duties. Forty-five days is the mid-value, this is an aviation specific mid-value. The unit can usually compensate for a person on restricted duty for about two weeks. After two weeks the unit begins to suffer, and the severity increment starts to increase. Furthermore, many of the aviator currency requirements are 45 days, after which additional training is necessary and unit readiness begins to decline, therefore the slope of the function begins to increase.

3.4.4.1.2 Lost Workdays

The rationale for the severity increments for the number of days lost is basically the same as the restricted days. For the first two weeks, the unit can absorb the missing soldier's requirements by spreading out his tasks to other service members. After two weeks, unit morale, and therefore readiness begins to suffer.



Figure 3-14 SDVF: Lost Workdays

3.4.4.2 Equipment Replacement Time

3.4.4.2.1 Man-Hours to Repair

This severity measure focuses on the impact on maintenance and training. The range extends all the way to ten thousand man-hours. The decision-maker felt more

comfortable translating this range into a more useful scale. Therefore, to determine the

severity increments, hours were converted to weeks using the following conversion:

100 man-hours = 1 week * 2 Maintenance Personnel * 50 Hours/week

In these terms, the effect on a unit is as follows:

 $\underline{1}$ from 0-400 hrs (30 days) the unit maintenance status report is affected, the other aircraft available in the unit should be able to absorb the extra hours they will be required to fly.

 $\underline{2}$ From 400 hrs - 2400 hrs (6 months) the training of the unit starts to suffer in addition to the maintenance therefore, the slope of the function increases slightly.

 $\underline{3}$ After 6 months the organization's readiness continues to decline but at a slower rate, the unit will probably use outside aircraft support.



Figure 3-15 SDVF: Man-hours to Repair

3.4.4.2.2 Aircraft Total Loss



Figure 3-16 SDVF: Total Loss

The total loss of an aircraft is a binary severity function, either the aircraft s or is not a total loss.

Once all of the sub-criteria and measures were agreed upon and deemed complete (collectively exhaustive), non-redundant (mutually exclusive), independent, understandable and practical in size they can be formally presented as the approved severity hierarchy. Although there may be some additions and follow on research designed to improve some of the evaluation considerations, below is the complete, firstcut severity hierarchy developed with the information currently available.



As depicted above, the severity hierarchy consists of four accident severity criteria (the first tier), eight sub-criteria (second boxed tier) and thirteen measures. A "test for importance" (Kirkwood, 1997: p.19) is conducted during the sensitivity analysis. This

test indicates whether variations within the measure or evaluation consideration could change the preferred alternative.

3.5 Assessing Weights

Commonly, the single dimensional severity functions are designed so that each function equals zero for the least preferred level and one for the most preferred level (Kirkwood, 1997:68). This is not the case for the value functions described in this paper. By design, the accident that receives the highest score, when evaluated using this severity hierarchy, is the most severe. Therefore, the most severe, or the least preferred scores, will equal one and the most preferred will score zero.

Given these properties of the single dimensional value functions, "it follows that the weight for an evaluation measure is equal to the increment (or decrement) in value that is received from moving the score from its (most) preferred level to its (least) preferred level" (Kirkwood, 1997:68). This is the basis for determining the decisionmaker's weights for each criterion using the swing weight method. The assessment of the weights took place during an interview with COL Warren, Deputy Commander of the Army Safety Center. The weights were assessed in two ways. The first tier was assessed using swing weighting and the second tier and measures were assessed directly.

During the swing weighting procedure a base criterion was established, one by which the other criteria would be compared. This base criterion was the loss or damage to equipment or property. This was the easiest and most logical because the criterion contained only one measure. The first step was to determine the weights of the other three criteria relative to the base criterion. This was done by comparing the value the decision-maker gained as one criterion increased from the lowest possible scores in each

of it's measures to the highest possible scores in each measure to the severity gained from the same swing (low to high) in scores of the base criteria. The relative weights are shown in the Figure 3-18. The notation should be read as follows: The decision maker felt that the severity increase from "degree of injury or illness" was two times as much as the severity increase from "the loss/damage to equipment/property" when their measures were increased from the lowest to the highest scores possible. Each of the relative criteria weights was assessed in the same manner.

It is important to note that the weighting process occurred after the range of each measure was determined. If the decision-maker does not know how the score of each measure may vary prior to determining weights, the weights may be meaningless. For example, if the damage cost across all accidents varied very little, the decision-maker may not want to weight it as high as if the cost varied from 0 to nearly \$25 million (as was the case in with this criterion). If the range of the measure is small it may have very little effect on the overall outcome and therefore, should be determined and considered before the weighting procedure.

During the interview with Deputy Commander of the Safety Center it was important to establish preferential independence between the criteria. Developing criteria that are preferentially independent ensures that the level of one criterion does not affect how the decision-maker feels about increases in the other (Kirkwood, 1997: p.238). This was the case with all the criteria established in this research. The relative weights that were assessed by COL Warren for the first tier (Severity Criteria) are shown in Figure 3-18. Once these relative weights were determined, and knowing that the weights must sum to one (Kirkwood, 1997:70), the following relationship was established:



Figure 3-18 Relative Weight Assessment

x = weight of Loss/damage to equip/property

2x + x + .5x + .75x = 1

4.25x = 1

therefore, x = .236

Hence,



Figure 3-19 First Tier Weights

The weights of the sub-criteria and the measures were assessed using a points available or a "marble technique" (Kloeber, Lecture 1998). The decision-maker was told

that he had one hundred marbles and had to divide them up between the sub-criteria with respect to their value. Each of the sub-criteria is presented below.



Figure 3-20 Weights: Degree Injury/Illness



Figure 3-21 Weights: Damage to Property or Environment



Figure 3-22 Weights: Other Mission Impairing Factors



This results in the following global weights:

Figure 3-23 Weights: Global

3.6 Converting Value (Severity) to Utility (Severity[U])

Although all of the measures contain uncertainty, there were three measures that the decision-maker felt contained a level of uncertainty that he was uncomfortable with and wished to account for. These measures are: Number of man-hours to repair, damage cost, cost of injury. To account for the uncertainty in these measures the value functions are converted to utility functions. Once this conversion is done, all the scores will be referred to as severity(u) scores. This conversion is accomplished by establishing the decision-maker's risk aversion for the attributes in question. During the interview with ORSA/STATS Division Chief it was established that his risk aversion could be assumed to be constant. Therefore, the severity(u) function can be developed by establishing his risk aversion constant, ρ_1 and applying it to the exponential equation below:

$$U(x) = ((1 - \exp [-v(x)/\rho_i])/(1 - \exp(-1/\rho_i)), \text{ for } \rho_i \neq \text{Infinity (equation 1)}$$
$$U(x) = V(x), \qquad \qquad \text{for } \rho_i = \text{Otherwise (Kirkwood, 1997: p.164)}$$

The method that was used to establish the ρ_i was the "lottery question". The idea is to find a specific level that the decision-maker is indifferent about which side of the bet to take method (Clemen, 1996, p. 269). In order to do this the decision-maker was presented with a lottery that involved a 50% chance of receiving either the highest or the lowest score in each measure. The decision-maker then chose a value between those numbers that, if he was guaranteed to receive that value, he would be indifferent to the lottery or the guaranteed value. This value is referred to as the certainty equivalent (CE) or indifference point. The Table 2 represents the lottery questions asked to the decisionmaker for each measure and his final certainty equivalent.

Tab	ole 3-1 Lo		
	Man Hours to Repair	Repair/Replace Dollars	Injury Dollars
50% Chance Low	0	0.	0
50% Chance High	10,400	24.6 Million	5.8 Million
100% Chance (CE)	2,400	6 Million	1 Million

Because the severity functions are not linear, once the CE is determined its value must be established using the severity functions developed earlier in the chapter. The converted certainty equivalents are given below:

Table 3-2 Certainty Equivalent			
Measure	Raw Score (CE)	Severity (CE)	
Man-Hours	2,400	.5	
Damage Cost	\$ 6 Million	.5981	
Injury Cost	\$1 Million	.75	
i			

This CE provides a third point to develop an exponential severity(u) curve for each of the evaluation considerations. As discussed in section 3.5, for the functions in this research, the most preferred levels receive a score of zero, the least preferred levels receive a score of one and now a level of indifference has been developed. Given three points, and assuming the decision-maker's risk aversion is constant, an exponential conversion function can be used to convert severity functions into utility functions that account for risk aversion. The equation for the exponential utility function depends on the range of the evaluation measure and a constant. The range for each measure has been converted to

a value between zero and one; the constant is the exponential constant, represented by the Greek letter ρ (rho). When ρ is very large the utility function becomes a straight line and V(x) equals U(x). The shape of the utility functions represents the decision-maker's risk attitude. A concave shape indicates a risk averse decision-maker, a convex shape indicates a risk seeker, and a straight line indicates risk neutrality (Kirkwood, 1997: 138).



Figure 3-24 Utility: Damage Cost



Figure 3-25 Utility: Man-hours to Repair



Figure 3-26 Utility: Injury Cost

The severity of each accident can now be established. The rating of the severity of an accident does not account for the probability of an accident occurring, rather this is a measure of the expected consequence of an accident should it occur.

3.7 Establishing Accident Risk

To assess the risk of an accident one must account for the severity and the probability of occurrence (FM 100-14, 1998: p. G-2). The database that was used contained thirteen hundred and fifty six events, these events were categorized into sixty-five types of accidents. During an accident sequence more than one type of accident may occur. For example, an aircraft may experience a loss of hydraulic power at a hover which may lead to a tree strike. In this case, two types of accidents occurred. The accident investigation team determines the primary, secondary and tertiary events in the accident sequence. Only the primary event was used for this study. Therefore, the probability of occurrence of an accident is given by the equation below.

This research assumes that an accident has occurred, therefore the probabilities that were used were conditional probabilities. These conditional probabilities are listed in Appendix E.

3.8 Hazard to Accident Association

Individual hazards can result in various types of accidents. In order to establish the severity of a hazard_i, the severity of all the accidents that were a result of hazard_i must be summed up in the following manner.

Severity of Hazardi =
$$\sum_{j=1}^{94} P(Accident_j | Hazard_i) * Severity (Accident_j)$$

To determine the conditional probability that Accident_i occurred given that Hazard_j occurred, P(Accident_i | Hazard_j), the database had to be manipulated. A primary hazard was associated with each accident. Once the hazards and accidents were organized into a matrix format each conditional probability could be accessed. The law of total probability was used to confirm that these probabilities were correct. By viewing the sample space, S (Hazard_j), as a union of mutually exclusive subsets the law of total probability can be used: $S = B_1 \cup B_2 \cup ... \cup B_k$ where $P(B_j) > 0$, for i = 1, 2, ..., k and $B_i \cap B_j = \emptyset$ for $i \neq j$. Then for any event A

$$P(A) = \sum_{j=1}^{k} P(B_j)P(A|B_j).$$
 (Wackerly, 1995; p.61)

Where,

 $P(A|B_i) = P(Accident_i | Hazard_i)$ $P(B_i) = P(Hazard_i) \text{ (directly available from the database)}$ $P(A) = P(Accident_i) \text{ (directly available from the database)}$ Example, 24

 $P(\text{Accident}_{94}) = \sum_{j=1}^{24} P(\text{Accident}_{94} | \text{Hazard}_{j}) P(\text{Hazard}_{j}) = (.015152)(.097345) = .00147497$ This number agrees with the actual probability of Accident (94)

Once these probabilities were confirmed the severity of each hazard could easily be

computed. Utilizing the definition of risk as applied to hazards:

Hazard Risk = Hazard Severity * P(Hazard) (FM 100-14, 1998; 1-1)

A score for the overall risk of each hazard was calculated.

3.9 Summary

The following diagram summarizes the methodology followed throughout this effort. Furthermore, it was used throughout the research process to update key stakeholders on their roles and responsibilities and the output that they would receive from.

Chapter Four presents the results of this methodology and an analysis of the results to include a sensitivity analysis of the hazard severity criteria.



Figure 3-27 Methodology Framework

.

4 Results and Analysis

4.1 Introduction

This chapter presents the results from the analysis of the deterministic and probabilistic model that was developed in Chapter Three. There were sixty-five categories of accidents analyzed using the value-focused model. The total scores and scores categorized by criteria are presented. The deterministic view indicated the pure severity of accidents and hazards. The probabilistic portion focuses on the drastic change in the ranking of accidents and hazards when the probability of occurrence is accounted for. Finally, the sensitivity of each of the top-level criteria in the value hierarchy is analyzed for its effect on both accidents and hazard severity.

4.2 Accident Severity

Each of the thirteen hundred and fifty six accidents was categorized into one of sixty-five types of accidents. These categories are defined by AR 385-40 and presented in Appendix A. There are two approaches for determining the severity of each accident. If all of the severity functions are linear, the severity can be assessed by taking the average raw scores in each category then determining the value of that score. The severity functions in this study were not linear. They were a combination of piecewise linear and exponential functions and therefore, averages could not be used. Each individual occurrence of an accident had to be scored separately. The expected value for the severity of each accident was determined by calculating the severity for each single dimensional severity function then multiplying the expected values by their corresponding weights and finally summing the results (Kirkwood, 1997: 158). Hence,

the total severity of each accident data point was calculated. This became the individual accident severity score (presented in Appendix F). All severities in the same accident category were then averaged giving the expected value. A graphic representation of the span of the severity scores and a table of results in descending order are presented below.



Figure 4-1 Accident Severity Chart

		Normalized	Actual
Rank	Definition	Score	Score
1	SPIN/STALL	1.0000	0.3835
2	ELECTRICAL SYSTEM	0.8491	0.3256
3	INFLIGHT BREAKUP	0.8214	0.3150
4	INFLIGHT BREAKUP-MAST BUMPING	0.5341	0.2048
5	COLLISION WITH GROUND/WATER	0.5227	0.2004
6	ABANDONED AIRCRAFT	0.4752	0.1822
7	FIRE AND/OR EXPLOSION IN THE AIR	0.4492	0.1723
8	MULTIPLE AIRCRAFT EVENT	0.4100	0.1572
9	UNDERSHOOT	0.3601	0.1381
10	WIRE STRIKE	0.3523	0.1351
11	EXCESSIVE YAW/SPIN	0.3363	0.1290
12	BROWNOUT	0.3363	0.1290
13	FUEL STARVATION	0.3044	0.1167
14	MOC	0.1777	0.0681
15	ROTOR/PROP WASH	0.1572	0.0603
16	ENGINE FAILURE/OVERSPEED/OVERTEMP	0.1522	0.0584
17	FIRE AND/OR EXPLOSION ON THE GROUND	0.1520	0.0583
18	DITCHING	0.1491	0.0572
19	AIR TO GROUND COLLISION	0.1457	0.0559
20	FUEL EXHAUSTION	0.1276	0.0489
21	REFUELING OPERATIONS	0.1187	0.0455
22	POWER TRAIN	0.1129	0.0433
23	GROUND LOOP/SWERVE	0.1127	0.0432
24	AIRFRAME	0.1090	0.0418
25	INSTRUMENTS	0.0946	0.0363
26	PRECAUTIONARY LANDING	0.0921	0.0353
27	FORCED LANDING	0.0910	0.0349
28	WEAPONS RELATED	0.0812	0.0311
29	TREE STRIKE	0.0804	0.0308
30	ENGINE OVERTORQUE/OVERLOAD	0.0786	0.0301
31	OVERSTRESS	0.0769	0.0295
32	PERSONNEL HANDLING EVENT	0.0710	0.0272
33	HUMAN FACTOR EVENT	0.0617	0.0237

.

Table 4-1Accident Severity Scores

•

		Normalized	Actual
Rank	Definition	Score	Score
34	FLIGHT RELATED	0.0614	0.0235
35	HELOCASTING	0.0597	0.0229
36	OTHER COLLISION	0.0563	0.0216
37	OBJECT STRIKE	0.0557	0.0214
38	WHEELS UP LANDING	0.0556	0.0213
39	EQUIPMENT LOSS/DROPPED OBJECT	0.0447	0.0172
40	AIRCRAFT GROUND ACCIDENT	0.0436	0.0167
41	ROTOR/PROPELLERS	0.0406	0.0156
42	HARD LANDING	0.0385	0.0148
43	DRIVE TRAIN (EXCEPT XMSN)	0.0346	0.0133
44	MISSING AIRCRAFT	0.0310	0.0119
45	PNEUMATIC SYSTEM	0.0291	0.0112
46	TIEDOWN STRIKE	0.0265	0.0102
47	ANTENNA STRIKE	0.0193	0.0074
48	TRANSMISSION FAILURE	0.0184	0.0071
49	CARGO EVENT	0.0128	0.0049
50	LANDING GEAR COLLAPSE/RETRACTION	0.0119	0.0046
51	LIGHTNING STRIKE	0.0105	0.0040
52	ANIMAL STRIKE	0.0059	0.0023
53	SPIKE KNOCK	0.0049	0.0019
54	ENGINE OVERSPEED/OVERTEMP	0.0036	0.0014
55	ROTOR OVERSPEED	0.0017	0.0007
56	UTILITY/ENVIROMENTAL CONTROL SYSTEM	0.0011	0.0004
57	ABORTED TAKEOFF	0.0009	0.0004
58	BIRD STRIKE	0.0006	0.0002
59	SINGLE ENG LANDING	0.0006	0.0002
60	CONTRACTOR AIRCRAFT ACCIDENT	0.0005	0.0002
61	TAIL BOOM STRIKE	0.0003	0.0001
62	EXTERNAL STORES EVENT	0.0002	0.0001
63	FLIGHT CONTROL	0.0000	0.0000
64	STRUCTURAL ICING	0.0000	0.0000
65	MAST BUMPING	0.0000	0.0000

.

The scores above represent the expected severity of an accident, should that accident occur. A description of each accident is provided in Appendix A. Note from Figure 1 and Table 1 that there is a large drop off in the severity score after the thirteenth ranked accident. The goal of this study is to determine the top ten hazards. Although the rankings will change when the probabilities of occurrence are applied, we will concentrate on the top thirteen accidents at this point to identify trends. The figure below depicts the contribution of each fundamental objective for each of the top thirteen accidents.



■ Damage to Equip Prop ■ Degree Injury/Illness ■ Environmental Damage ■ Other MSN Factor

Figure 4-2 Accident Severity Composition

Rank	Description
1	SPIN/STALL
2	ELECTRICAL SYSTEM
3	INFLIGHT BREAKUP
4	INFLIGHT BREAKUP-MAST BUMPING
5	COLLISION WITH GROUND/WATER
6	ABANDONED AIRCRAFT
7	FIRE AND/OR EXPLOSION IN THE AIR
8	MULTIPLE AIRCRAFT EVENT
9	UNDERSHOOT
10	WIRE STRIKE
11	EXCESSIVE YAW/SPIN
12	BROWNOUT
13	FUEL STARVATION

Table 4-2Accident Severity (Top 13)

Examining the breakdown of the scores, one will note the lack of contribution that the environment criterion lends to the ranking of accident severity. As more and different data collection procedures are done in the area of the environment, this may increase. Currently, fluid spills are the only measures used to determine the severity of environmental damage, as the measures increase the decision-maker may increase the weight of this criterion. The other three categories seem to contribute consistently throughout the range of accidents. There was initial concern from the Safety Center that there was a high degree of correlation between the level of equipment/property damage and the degree of injury. The results do not suggest that this is true. Figure 4-3 is a plot of degree of injury/illness versus damage to equipment or property. The resulting linear regression is a weak regression and indicates a lack of correlation between the two criteria.





	Proper	ty Damage `	<u>Vs Injury</u>	/Illness		
RSquare			0.266553			
R	Square A	\di		0.254911		
Observations				65		
		Analysis of	<u>Variance</u>			
Source	DF	Sum of Squ	ares M	lean Square	F Ratio	
Model	1	0.01825	5969	0.018260	22.8958	
Error	63	0.05024	1324	0.000798	Prob>F	
C Total	64	0.06850	0292		<.0001	
	F	Parameter E	stimates			
Term	_	Estimate	Std Error	t Ratio	Prob>ltl	
injury/illness	;	0.5005116	0.104601	4.78	<.0001	

The R-Square value is low, the F- value is large along with the t-value. All these factors suggest that these two measures are probabilistically independent. The lack of correlation can also be seen by referring to Figure 4-2. The severity of accidents 52, 26, 39 and 48 (In-flight Breakup-Mast Bumping, Undershoot, Abandoned Aircraft and Brownout) are largely due to property damage and the degree of injury contributes almost
nothing while the severity of accidents 38 and 37 (Spin/Stall and In-flight Breakup) are the exact opposite.

4.2.1 Degree of Injury/Illness Sensitivity

Hazard severity is a linear combination of accident severities. Therefore, to conduct a quality sensitivity analysis for overall hazard severity it is necessary to conduct a sensitivity analysis on accident severity. This analysis will focus on fluctuations in the first tier criteria weights that were assigned by the decision-maker. The variation in accident severity was examined over the full range of possible weights (0-1) for each of the criteria; these results can be seen in Appendix H. The more important results are shown below. The weights for each criterion were varied throughout the feasible range. These feasible ranges of weights are the absolute minimum and maximum weights that the decision-maker would assign to each of the criteria. For example, although the degree of injury/illness was given a weight of .47 the decision-maker stated that he would never assign that criterion a weight of below .30 but he would also never go above .60. Therefore the feasible range for the degree of injury/illness is .30 to .60. As the weight of one measure decreases the others must increase to keep the same relative ratio between the other weights. As shown in the graph below, the overall severity score is relatively insensitive to fluctuations, within the feasible range, of the weight of degree of injury or illness.





The minimum points in the range occur when the weight of this criterion is .3. At the highest points the weight is .6 and the mark in the middle is the severity value at the current weights. As can be seen by the chart in Figure 4-4, the top thirteen accidents remain the top thirteen. As the weight is increased or decreased from the original .47, some variations occur in the in the order of the top thirteen, but they remain clearly above the remainder of the accidents.

If an accident is sensitive to fluctuations in weight of a criterion this indicates that the severity of that accident is skewed heavily towards one or more criteria. For example, accidents 52, 39, 26 and 48 have the largest range in severity score when injury weight is varied. As depicted in Figure 4-2, these accidents are primarily composed of two criteria, damage to equipment/property and other mission factors. As the criterion weight is varied, the remaining criteria increase or decrease in order to maintain their initial weight ratio. This insight can be utilized when developing controls to reduce accident severity. Controls directed towards the two criteria discussed above can drastically decrease the severity of those same accidents. An accident that remains constant throughout the weight fluctuations has an evenly distributed composition and may require a series of controls to significantly reduce its severity. The same four accidents react the same to fluctuations as each of the criterion weight is varied. These types of accidents will provide a large amount of reduction with controls focused on one area and may provide the "best bang for your buck".

4.2.2 Property/Equipment Damage Sensitivity

The weight of property/equipment damage was varied from 10 to 30, with the



Figure 4-5 Accident Sensitivity: Property Damage Weight

initial weight of .26. Once again we can see a drastic drop in the severity range after the thirteenth ranked accident. The same generalization can be made for fluctuations in the

weight of property damage. The top thirteen continue to remain the top thirteen, this indicates that these accidents are severe in each category.

4.2.3 Environmental Weight Sensitivity

The original weight for the environment was .117. The feasible range was determined to be 0 to .30. There is very little fluctuation in the overall severity within the feasible range of the weight for the environmental criterion.



Accident Rank

Figure 4-6 Accident Sensitivity: Environmental Weight

4.2.4 Other Mission Factors Sensitivity

The initial weight of other mission factors, which concentrates on unit readiness, was .177. The weight was varied from .10 to .30, the results are shown in Figure 4-7. Once again, the overall severity of each accident is relatively insensitive to minor fluctuations in the weight of other mission factors. Although the ranks of different accidents may have minor changes, the sensitivity analysis presented above indicates a stable rating system of accident severity. The following sections will develop the accident risk ranking and hazard severity and risk.



Accident Rank

Figure 4-7 Accident Sensitivity: Mission Factors Weight

4.3 Accident Risk Ranking

After applying the probability of a specific accident occurring there is a drastic change in the top accidents. Table 6 represents the normalized risk score of the top twenty accidents. The scores for the risk assessment have been normalized, by dividing each risk score by the largest risk score, in order to maintain a consistent scale from zero to one. Upon comparing accident severity and accident risk a few observations can be made. Only 5 of the top ten accidents remained in the top ten once the probabilities were applied. Furthermore, only 11 of the top 20 remained in the top 20 and no accident maintained its initial ranking.

		<u>Risk</u>	<u>Severity</u>
Event	<u>Risk</u>	Rank	Rank
COLLISION WITH GROUND/WATER	1.0000	1	5
ENGINE FAILURE/OVERSPEED/OTEMP	0.3286	2	16
WIRE STRIKE	0.2968	3	10
TREE STRIKE	0.2567	4	29
MULTIPLE AIRCRAFT EVENT	0.2519	5	8
EXCESSIVE YAW/SPIN	0.1712	6	11
FUEL STARVATION	0.1175	7	13
SPIN/STALL	0.0878	8	1
ENGINE OVERTORQUE/OVERLOAD	0.0869	9	30
INFLIGHT BREAKUP	0.0865	10	3
FLIGHT RELATED	0.0851	11	34
OTHER COLLISION	0.0702	12	36
AIRCRAFT GROUND ACCIDENT	0.0681	13	40
FIRE AND/OR EXPLOSION IN THE AIR	0.0552	14	7
HARD LANDING	0.0507	15	42
EQUIPMENT LOSS/DROPPED OBJECT	0.0471	16	39
OBJECT STRIKE	0.0420	17	37
MOC	0.0405	18	14
ELECTRICAL SYSTEM	0.0298	19	2
BROWNOUT	0.0295	20	12

Table 4-3Accident Risk Ranking

This indicates that although an accident may be severe, it may not be considered as great a risk to our soldiers if it occurs with a low probability. For example, an electrical system malfunction was clearly a severe accident (ranked number two in severity) yet over the last 11 years, given that an accident occurred, this type of malfunction occurred with a low probability of .001475. The ranking of this accident dropped drastically when probability was applied to determine the risk. Whereas the severity of a tree strike was ranked 29th out of 65 types of accidents yet, because tree strikes occurred with a probability of .134, it moved up to number four in the risk ranking. While all threats to our troops are important, a combination of severity and probability can help guide the allocation of limited resources to attain the greatest effect. The results presented in this section stress the concept that risk is a combination of the severity of the consequences and probability of occurrence. The relationship of each accident's severity versus the probability is summarized in the following figure.



The graph above is divided into four categories (A-D). Most of the accidents fall into the low-low category. The one accident that falls into the high-high category is Collision with Ground/Water, which is ranked number one in accident risk. This type of analysis can be used to direct efforts to reduce the risk of a certain accident, by either reducing the severity or the probability of occurrence.

4.4 Hazard Severity Ranking

Accident and risk have now been established. The remainder of this chapter focuses on hazard severity and risk. The calculations used to determine the severity and risk of a hazard do not directly include accident risk. Instead, each occurrence of an accident is associated with the hazards that caused the accident. This results in a conditional probability that is necessary for severity calculations, this probability is P(Accident_i | Hazard_j contributed to an accident). Appendix E contains the probabilities and actual calculations that were used to establish the severity of each specific hazard. A discussion of the methodology used to determine the hazard severity is also contained in Section 3.8. The accident severities and associated conditional probabilities were combined using the following relationship:

Expected Severity of Hazardi =
$$\sum_{j=1}^{94} P(Accident_j | Hazard_i) * Severity (Accident_j)$$

A graphic representation of the span of the expected hazard severity scores is shown in Figure 4-8 followed by a table of the results in descending order from the most severe to the least. In Table Four, the left column refers to system inadequacies as defined by AR 385-40. The system inadequacies represent hazards throughout this research effort, they are defined in numerical order in Appendix A. One of the system inadequacies described in Appendix A has been ignored through this paper. System Inadequacy 97 is insufficient information. This does not apply to the pilots flying the aircraft; this inadequacy comes from an accident investigation that concludes that there was insufficient information to determine the hazard that caused the accident. Figure 4-9 represents the percentage composition of each of the top ten hazards. This

chart does not account for the probability of a hazard contributing to an accident, merely the severity of the consequences if it is a contributing factor.

Expected Hazard Severity





Motivation/Mood

Fatigue

Effects of Alcohol, Drugs, lilness

Overconfidence/Complacency

Inadequate Supervision by Higher

Inadequate QC ,Manuf, Assembly Inadequate Supervision/Staff Off

Environmental Conditions

Inadequate Unit Training Inadequate Experience

Inadequate Attention

Habit Interference

Lack of Confidence

Inadequate School Training Material Improperly Designed Inadequate Facilities/Services Inadequate Written Procedure

Inadequate Maintenance Inspection

Insufficient Number/type personnel

Inadequate Composure

Hazard	Definition	Severity
SI18	Inadequate Motivation/Mood	0.079702329
SI20	Effects of Alcohol, Drugs, Illness	0.078789787
SI15	Inadequate Composure	0.077007882
SI03	Inadequate/Improper Supervision by Unit Com	0.073625832
SI16	Overconfidence/Complacency	0.067180833
SI19	Fatigue	0.063986499
SI01	Inadequate/Improper Supervision by Higher	0.058769655
SI21	Environmental Conditions	0.058752913
SI13	Inadequate QC, Manuf, Assembly	0.056988794
SI02	Inadequate/Improper Supervision by Staff Off	0.056283277
SI06	Inadequate Unit Training	0.056052383
SI07	Inadequate Experience	0.054184297
SI04	Inadequate/Improper Supervision by Dir Supervisor	0.05118832
SI99	Inadequate Attention	0.050760243
SI05	Inadequate School Training	0.050032107
SI11	Material Improperly Designed	0.049974531
SI10	Inadequate Facilities/Services	0.049040987
SI09	Inadequate Written Procedure	0.047979915
SI14	Inadequate Maintenance Inspection	0.046927051
SI08	Habit Interference	0.043605339
SI12	Insufficient Number/type personnel	0.042368548
SI17	Lack of Confidence	0.038541656

Table 4-4Expected Hazard Scores

Although when dealing with accidents, severity may be reduced significantly by focusing on one criterion, any control designed to significantly reduce the severity of a hazard may need to address all of the criteria. Because the severity of each hazard is a linear combination of all accident severities, the severity of individual accidents becomes convoluted in the overall hazard severity. The following table indicates the number of accidents that were involved in each of the ten most severe hazards.

Hazard	Definition	# Accidents
SI18	Inadequate Motivation/Mood	28
SI20	Effects of Alcohol, Drugs, Illness	7
SI15	Inadequate Composure	17
SI03	Inadequate/Improper Supervision by Unit Com	20
SI16	Overconfidence/Complacency	38
SI19	Fatigue	21
SI01	Inadequate/Improper Supervision by Higher	11
SI21	Environmental Conditions	34
SI13	Inadequate QC, Manuf, Assembly	22
SI02	Inadequate/Improper Supervision by Staff Off	10

 Table 4-5
 Accident/Hazard Relationship

Due to the wide range of accidents involved in each hazard, controls designed to reduce the severity of specific accidents may have little effect on a specific hazard's severity. No single accident occurred in all ten of the most severe hazards and only three accidents were involved in nine of the ten. These accidents are:

1) Collision with Ground/Water

2) Other Collision

3) Tree Strike

If a commonality could be identified between these types of accidents or other groups that occur often throughout the most severe hazards controls could be developed that might reduce severity in groups of hazards, not just one. Section 4.5 applies the probability of occurrence to the severities listed above. A different set of controls may be developed to reduce the probability of occurrence of hazard instead of the severity. The effectiveness of this type of control is discussed in Section 4.5.

4.4.1 Sensitivity of Criteria Weights on Hazard Severity

As seen in Section 4.2, as the weight of the criteria is varied some of the accidents increase in severity while some decrease. An individual hazard severity is a linear combination of all of the possible accidents that could have been caused by that hazard. The Figure 4-10 represents the range of hazard scores due to varying the weight of the degree of injury/illness. The X-axis contains the hazards ranked from most severe to least severe and the y-axis represents the severity of each hazard feasible range, the highest scores contain the minimum weight. The weight was only varied inside the feasible range for each criterion.



It is clear from the sensitivity graph above that the hazard ranking is insensitive to the weight of the degree of injury or illness within its feasible range. Out of the top ten hazards, hazard seven was the only one to change in rank order throughout the feasible range, and only changed from number seven to number eight. The other three criteria's

sensitivity graphs closely resemble this criterion, these additional graphs are presented in Appendix G: Sensitivity Analysis.

4.5 Hazard Risk Ranking

Whether a hazard has occurred cannot be tracked accurately. There are, most likely, numerous instances when hazards occur and do not cause accidents. The data used for this research assumes that a hazard has occurred and has contributed to an accident. It is logical to assume that the more accidents that have resulted from a particular hazard, the more likely that hazard has occurred.

Hazard	Definition	Severity	P(Hazard)	Hazard Risk	Normalized
SI16	Overconfidence/Complacency (5)	0.06718	0.20501	0.01377	1.00000
SI99	Inadequate Attention (14)	0.05076	0.22345	0.01134	0.82352
SI21	Environmental Conditions (8)	0.05875	0.14454	0.00849	0.61659
SI18	Inadequate Motivation/Mood (1)	0.07970	0.10103	0.00805	0.58466
SI07	Inadequate Experience(12)	0.05418	0.13864	0.00751	0.54543
S106	Inadequate Unit Training (11)	0.05605	0.11209	0.00628	0.45619
S109	Inadequate Written Procedure (18)	0.04798	0.12463	0.00598	0.43417
SI04	Inadequate/Improper Supervision by Direct Supervisor (13)	0.05119	0.10619	0.00544	0.39468
SI11	Material Improperly Designed (16)	0.04997	0.09735	0.00486	0.35321
SI13	Inadequate Quality Control,Manufactur, Assembly (9)	0.05699	0.04867	0.00277	0.20139
SI15	Inadequate Composure(3)	0.07701	0.03540	0.00273	0.19792
SI14	Inadequate Maintenance Inspection (15)	0.04693	0.05678	0.00266	0.19347
SI19	Fatigue(6)	0.06399	0.03761	0.00241	0.17473
S103	Inadequate/Improper Supervision by Unit Com (4)	0.07363	0.03171	0.00233	0.16952
SI10	Inadequate Facilities/Services (17)	0.04904	0.02876	0.00141	0.10241
S108	Habit Interference (20)	0.04361	0.02729	0.00119	0.08639
S105	Inadequate School Training (13)	0.05003	0.02286	0.00114	0.08305
S102	Inadequate/Improper Supervision by Staff Off (10)	0.05628	0.01549	0.00087	0.06329
SI17	Lack of Confidence (22)	0.03854	0.01622	0.00063	0.04540
SI01	Inadequate/Improper Supervision Higher (7)	0.05877	0.01032	0.00061	0.04405
SI20	Effects of Alcohol, Drugs, Illness (2)	0.07879	0.00664	0.00052	0.03797
SI12	Insufficient Number/type personnel (21)	0.04237	0.00664	0.00028	0.02042

Table 4-6Hazard Risk

The probabilities used to compute the hazard risk scores Table 4 are the probabilities, in the last eleven years, that hazard_i was a contributing factor to an accident. Note that the sum of all P(hazard_i) does not equal one. More than one hazard can be determined to be a contributing factor to an accident. For this specific set of data there are 2,398 hazard occurrences with only 1,356 accidents, this is an average of 1.8 hazards contributing to each accident. In Table Four the previous ranking of the hazard's severity is shown in parenthesis in the Hazard description column. Only four of the top ten severe accidents stayed in the top ten when the probabilities were considered. Figure 4-11 depicts hazard severity versus the probability of occurrence.



Figure 4-11 Hazard Severity versus Probability

This chart is divided into four regions. These regions are a quick reference to the distribution of risk when broken down into components of severity and probability. Hazards with the greatest risk are found around the outskirts of the chart. This is where the hazard is composed of a high severity, a high probability, or both. The data points of

the four most severe hazards are labeled. The distribution of the scores of the top ten hazards is depicted below in the Figure 4-12.



The categories of hazards (system inadequacies) are shown in Appendix A. The

ten most sever hazards are categorized below.



Figure 4-13 Hazard Categories

Category	Percentage	Hazard
Individual Failure	40	Overconfidence/Complacency
		Poor Motivation/Mood
		Inadequate Attention
(Environmental Conditions
Training Failure	20	Unit Training
		Inadequate Experience
Support Failure	20	Material Design
		Inadequate Quality Control
Leader Failure	10	Direct Supervisor
Standards Failure	10	Inadequate Written Procedures

Table 4-7Hazard Categories

4.6 Summary

This chapter has presented the results of this research effort and a one-way sensitivity analysis of the weights of the major criteria. The goal of this project was to determine the top ten hazards, while accomplishing this goal other additional information was obtained. Accident and hazard severity and risk have been presented and the relationship between severity and risk has been discussed. The development of risk from severity is the first step in reducing the risk of accidents and hazards.

If the information presented here is used to allocate resources, a portfolio analysis is recommended to determine which of the hazards can be reduced or managed with the resources available. The model developed is not restricted to the hazards in this paper yet it is susceptible to the accuracy of the data and assumptions. These assumptions are not limited to the research in this paper the accuracy of the data is also dependent on the quality of the investigation done by the accident investigation team.

5 Conclusions and Recommendations

5.1 Conclusions

Determining the severity and risk of hazards in Army Aviation is a complicated process. Although there are army regulations, doctrine and technical experts energized towards a common goal of reducing severity and risk of such hazards there is currently no consistent methodology for accomplishing this. Investigation into the relevant doctrine, careful delineation of the terms involved, and value focused thinking provide the basis for developing and consistently using a structured methodology to assist decision-makers in making logical decisions concerning the allocation of resources for risk management. Identifying the underlying values of the Army Safety Center and Army doctrine will also assist the risk management experts in developing alternatives in order to reduce that risk.

The methodology presented in this paper is theoretically sound and a method of modeling supported by multi-attribute preference theory. The model developed through this research is however, heavily dependent upon data collection procedures and safety investigation results. The hazards (System Inadequacies) and accidents (events) used in this research were agreed upon as data that would be useful to (1) validate the concept of using value focused thinking to rank accidents and hazards and (2) develop the cornerstone for a research effort to be continued at the Army Safety Center for a proposed five year plan. The continuation of the research will continue to improve the model by developing a better taxonomy to identify the existing hazards in Army Aviation and furthermore to incorporate ground-related accidents and hazards.

The ten hazards with the highest risk rating are:

Hazard	Definition	Severity	P(Hazard)	Hazard
				Risk
SI16	Overconfidence/Complacency (5)	0.0671	0.205	0.0137
SI99	Inadequate Attention (14)	0.0507	0.223	0.0113
SI21	Environmental Conditions (8)	0.0587	0.144	0.0084
SI18	Inadequate Motivation/Mood (1)	0.0797	0.101	0.0080
SI07	Inadequate Experience(12)	0.0541	0.138	0.0075
SI06	Inadequate Unit Training (11)	0.0560	0.112	0.0062
SI09	Inadequate Written Procedure (18)	0.0479	0.124	0.0059
SI04	Inadequate/Improper Supervision by	0.0511	0.106	0.0054
	Direct Supervisor (13)			
SI11	Material Improperly Designed (16)	0.0499	0.097	0.0048
SI13	Inadequate Quality Control,	0.0569	0.048	0.0027
	Manufacturing, Assembly (9)			

Table 5-1 Top Ten Hazards

The risk score can be reduced by either decreasing the severity of the hazard or by reducing the probability of occurrence. As shown in Section 4.4, the most direct approach to reducing the severity of a specific hazard is to reduce the probability of occurrence. This method may be more direct yet more difficult. There are some critical unknowns present when dealing with the probability or hazards specifically, how many times does a hazard occur before it produces an accident? Further research should be conducted to determine the actual probability of a hazard occurring, regardless of whether it contributed to an accident or not.

Figures 4-8 and 4-13 represent accident/hazard severity versus the probability of occurrences. The obvious accidents and hazards to focus on are the ones in area C, high severity and high probability. Others can also be reduced but this area is high severity and high probability. The most efficient way to reduce hazards is to take steps towards

reducing their probability. This will consequently reduce the probability of individual accidents occurring. When attempting to reduce accident risk controls should be developed to reduce their severity. These may be two drastically different approaches. To reduce the severity of an accident the senior leadership might look at more protection in the cockpit of the aircraft while to reduce the probability of occurrence of a hazard they would turn towards the pilot training program. For instance, the to hazard risk is overconfidence/complacency of the pilot. A chief contribution is the hazard's correlation with the severity of tree strikes. The probability that a tree strike occurred, given that the accident was caused partially by overconfidence or complacency, is slightly over 19%. Therefore, we can reduce the risk of this hazard by additional training in an attempt to reduce the probability of the occurrence of the hazard or possibly install a proximity-warning device to reduce the number of tree strikes. Both of these methods would reduce the overall risk of this hazard.

5.2 Recommendations

5.2.1 Value Functions

The severity functions developed in this research were developed over an eightmonth period in conjunction with the Army Safety Center and other safety experts. To expand this model for Army use, the single dimensional value functions may need to be adjusted in order to represent decision- maker's views at different command levels. For this research, experts in the field of aviation, maintenance, personnel and safety were used to develop the value functions. For future use this list of experts may have to be expanded down to the field units. Some of the concerns involve the preferences of senior

leadership of the Army on dollars expended and lives. Furthermore, the Safety Center is trying to initiate a program, based on value focused thinking to change the way the Army manages risk. This model is based on a gold standard taken from Army doctrine and developed with the Safety Center. In order to change the way risk management is performed; doctrine may have to be rewritten necessitating a change in the value hierarchy. At that time, a gold standard may not be the best method. A silver standard, bottom up approach, may be the appropriate methodology by establishing a new standard by involving experts from different facets of military life and developing a new hierarchy for the severity of accidents. Such a visionary style of leadership usually requires other methods of building value models, a silver standard is one and an (Platinum) interview method is another (Parnell, 1997).

5.2.2 Data Collection

The span of years over which data was collected for this project (11 years) was chosen in order to gather as much data as possible without incorporating data that may be outdated due to changes in tactics, techniques and procedures in Army Aviation. This represents a period of time when the Army introduced three relatively new aircraft into various units in the Army: the AH-64, the UH-60, and the OH-58 D. This was a time period where trends were set and tactics and techniques were being developed. The problem with this time period is that there were a number of accident types that had only one occurrence while other accidents occurred over 150 times. Using older data may lead to erroneous results. If the data collection period is too small the amount of data may not be sufficient and also lead to ill-conceived conclusions.

Given only one occurrence of an accident, there was not enough information to generalize results of similar accidents in the future. No standard deviation could be developed nor could a range be established. In order to alleviate this problem some of the accident categories may be able to be combined or associated with a like group of accidents. In that manner conclusions can be drawn with less uncertainty about a group of accidents as opposed to relying on one, or very few data points.

5.2.3 Environmental Data

Although Army safety doctrine dictates that one of the four criteria for establishing the severity of an accident or a hazard is the damage to the environment, there has been very little data collected on the environment during the accident investigation. Figure 5-1 represents some of the environmental areas that are represented



Figure 5-1 Environmental Damage Data

in the environmental protection matrix outlined in FM 20-400 / MCRP 4-11B, the <u>Military Environmental Protection Manual</u>. Further investigation could be conducted to establish measures that could adequately describe each of the evaluation considerations depicted in the dashed boxes below.

Currently the only information that accurately records damage to the environment in the database at the Safety Center is gallons of oil, fuel and hydraulic fluid spilled at the sight. Other factors such as fire and chemicals present are tracked but their effect on the environment cannot be derived. The database is built from the accident investigation forms that are filled out by a board assigned to the accident. More extensive guidelines could be established for the collection of data concerning the environment. Although it would take years to establish a meaningful database, adequate measures for the areas of concern mentioned above could be attained by the investigation board at the accident sight. Some example questions that could be asked at the sight are given below.

If a fire was present what was the extent of damage to the surrounding area?
 If the accident occurred overseas was there damage to civilian property, what type and how much?

3) What clean up was required by the local unit or civilian agency?

4) Was the accident near a water source?

5.2.4 Hazards and Controls

One of the problems encountered during this research was establishing what constituted a hazard. The Army Safety Center has taken the steps necessary to continue the work in the area of hazard severity, one of the major goals of their efforts is to develop a better hazard classification system and controls classification system. The

areas of concern and the measures used in this research should assist in that effort. The value hierarchy established here should be particularly helpful in developing controls to reduce the severity of accidents and hazards.

By Identifying the accident severity criteria that contributed the greatest to the severity of an accident or hazard, risk managers can focus their efforts to reduce those specific effects. Furthermore, if experts deem that the efforts required to reduce the severity are not feasible they can chose to focus on reducing the probability of the event occurring and therefore reducing the risk. In this manner, resources expended towards the reduction of risk can be directed more efficiently and effectively.

In addition to establishing controls, the approach used in this research to determine hazard severity can be used to identify those combinations of hazards that drastically increase risk. The following equation can be used to determine those combinations of hazards with the largest severities:

Severity of Hazard Combination_{jk} = $\sum_{i=1}^{65} P(Accident_i \mid Hazard_{jk}) * Severity (Accident_i)$

5.3 Contributions

The consequences of an accurate hazard severity ranking and appropriate allocation of resources include effects on soldier's lives, unit morale, scientific advancement, and technological progress, as well economic costs and benefits. A careful analysis of each hazard has high potential payoff for the Army safety Center, Aviation, and the Department of the Army as a whole. The benefits that will be derived are; lives saved due to the identification and remediation of the correct hazards and the possibility

of optimal resource allocation within a given budget. In addition, the methodology can be applied to other safety settings and issues.

Aside from identifying hazard severity and risk, the same measures were established for specific accidents. This type of information is valuable at the tactical level. Certain hazards are unavoidable do to mission requirements. If tactical commanders know the correlation of certain accidents to existing hazards they can take steps to reduce the chance or severity of those specific accidents.

The proposal, projected benefits and a summarized methodology for this research was submitted to the Department of the Army Studies Program. The research effort was selected and awarded \$98,000 for continued research in coordination with the Safety Center. A statement of work has since been developed to expand the scope of the project. The overriding goal is to revamp the manner in which hazards are identified and ranked and how controls are developed and implemented to reduce the severity and risk of hazards and accidents. The study will initially be centered on the Army Safety Center but will be developed for field unit implementation over a five-year period. A description of the continuation of work will be resubmitted each year to the studies program for additional funding.

Appendix A: Accident/Incident Event and System Inadequacies Codes Associated with Aircraft Accidents

A.1 Accident/Event Codes.

The following codes and explanations below are provided to categorize aviation accidents by the type of event(s) involved.

Code: 01 Title: Precautionary landing (PL).

Explanation: A landing resulting from unplanned events, occurring while the aircraft is in flight that makes further flight inadvisable. This event is to be used for PLs where no other event applies or in conjunction with other materiel failure events.

Code: 02 Title: Forced landing (FL).

Explanation: A landing caused by failure or malfunction of engines, systems, or components that makes continued flight impossible. This event is to be used in conjunction with other materiel failure/malfunction events.

Code: 03 Title: Aborted takeoff.

Explanation: An unplanned event that occurs before liftoff that interrupts a planned flight. This event is to be used for aborted takeoffs where no other event applies or in conjunction with other materiel failure events.

Code: 04 **Title:** Human factor event.

Explanation: A psychological, physiological, or pathological condition that occurs to personnel when intent for flight exists and results in interference with a crewmember's duties during aircraft operations or mission being delayed, diverted, or aborted.

Code: 05 Title: Cargo event.

Explanation: Injury or property damage resulting from cargo- related accident/incident; intentional or unintentional jettisoning of cargo hook load.

Code: 06 Title: Personnel handling event.

Explanation: Injury or property damage involving personnel handling errors or personnel handling.

Code: 07 Title: External stores event.

Explanation: Injury or property damage resulting from external stores handling errors or equipment failures.

Code: 08 Title: Multiple aircraft event.

Explanation: Injury or property damage resulting from the interactions of two or more aircraft. To qualify as a multiple aircraft event, two or more aircraft, with engines running, must be involved.

Code: 09 Title: Misappropriated aircraft.

Explanation: An aircraft accident that occurs during the operation of an Army aircraft that has been misappropriated, regardless of aircrew designation. Intent for flight must exist.

Code: 10 Title: Drone aircraft.

Explanation: Drone aircraft are identified by "Q" designator, and may be flown or operated by rated or nonrated personnel, or by remote control. When manned, they will be regarded as aircraft and reported accordingly. When unmanned, and operated by remote control, the accident will be reported using DA Form 285.

Code: 11 Title: Contractor aircraft accident.

Explanation: An aircraft accident that occurs as a result of a government contractor's operation in which there is damage to Army property or injury to Army personnel. Included is non-delivered equipment for which the Army has assumed responsibility.

Code: 12 Title: Aircraft ground accident.

Explanation: Injury or property damage involving an Army aircraft in which no intent for flight exists and the engines are in operation.

Code: 13 Title: Laser-induced/related.

Explanation: Property damage or personnel injury resulting from laser operations created. May be used in conjunction with other events.

Code: 14 Title: Fratricide.

Explanation: Persons killed or wounded, or equipment damaged, in military action, mistakenly or accidentally, by friendly forces actively engaged with the enemy, who are directing fire at hostile force or what is thought to be a hostile force.

Code: 15-19 (Reserved for future additions.)

Code: 20 Title: Refueling Accident.

Explanation: Damage incurred during refueling operations on the ground or in-flight.

Code: 21 Title: Midair Collision.

Explanation: Those accidents in which more than one aircraft collide in flight. Hover is considered in flight. Damage does not have to be done to both aircraft (will be used in addition to "08 multiple aircraft event").

Code: 22 Title: Helo-casting.

Explanation: Property damage or personnel injury occurring during helo-casting operations.

Code: 23 Title: Hard Landing.

Explanation: Damage incurred due to excessive sink rate on landing touchdown. Includes auto-rotation landings when skids are damaged; main rotor blade flexing into tail boom; tire blowing on touchdown; landing gear driven into fuselage; fuselage, wing, etc., buckling. Note: The landing area must be suitable for a probable successful landing.

Code: 24 Title: Wheels-Up Landing.

Explanation: Aircraft equipped with retractable landing gear in the wells. Includes intentional gear-up landings; crew forgetting to lower gear; gear does not extend when gear handle placed down.

Code: 25 Title: Landing Gear Collapse/Retraction.

Explanation: During takeoff, landing, or taxi, the gear collapses for any reason or the crew inadvertently retracts or retracts too soon on takeoff (does not include gear shearing due too hard landing).

Code: 26 Title: Undershoot.

Explanation: When an approach is being made to a prepared area of field and the aircraft touches down short of the suitable landing surface. (Does not include striking wires, trees, etc., on approach except an aircraft striking an airport boundary fence.)

Code: 27 Title: Overshoot or Overrun.

Explanation: Landing in which the aircraft runs off the end of the runway because of touchdown speed, too-short runway, touching down too long, or failure of brakes.

Code: 28 Title: Ditching.

Explanation: Landing in a controlled attitude in water. (Does not include creeks, streams, etc., or those landings to ships or barges in which the aircraft crashes in the water.)

Code: 29 Title: Ground Loop/Swerve.

Explanation: When aircraft damage is incurred because absolute directional control is not maintained (intentional or unintentional). Includes F/W ground loops; R/W autorotational landings; R/W running landings due to antitorque failures; aircraft running off side of runway.

Code: 30 Title: Collision with Ground/Water.

Explanation: Those accidents in which the aircraft strikes the ground or water unintentionally. Includes crashing into a mountain under IFR, IMC, or night; inadvertent flying into the ground or water, such as making a gun run and failing to pull up; low-level flight resulting in striking ground or water.

Code: 31 Title: Aircraft Collisions on the Ground.

Explanation: Accidents in which two or more aircraft collide on the ground. None of the aircraft can be in flight. (used in addition to '08' multiple aircraft event).

Code: 32 Title: Other Collisions.

Explanation: Accidents when an aircraft collides with something not accounted for by other type events listed.

Code: 33 Title: Rotor overspeed.

Explanation: Main rotor RPM exceeding the allowable limits for continued flight.

Code: 34 **Title:** Fire and/or Explosion on the Ground.

Explanation: Accidents that are initiated by a fire or explosion. The damage incurred must be prior to lift-off and/or after touchdown.

Code: 35 **Title:** Fire and/or Explosion in the Air.

Explanation: Same as on the ground except damage must be after lift-off and before touchdown.

Code: 36 Title: Equipment Loss or Dropped Object.

Explanation: Accidents in which some part of the aircraft or attached equipment is lost in flight, other than cargo and external stores.

Code: 37 Title: In-flight Breakup.

Explanation: Accidents in which aircraft begins to break up in flight. In these accidents,

any type of landing is not expected. Includes loss of main rotor blades; loss of wing.

Code: 38 Title: Spin or Stall.

Explanation: Fixed wing aircraft type accidents resulting in stalling and/or spinning due of loss of airspeed or excessive angle of attack.

Code: 39 Title: Abandoned Aircraft.

Explanation: Accidents in which all flight crew eject or parachute.

Code: 40 Title: Flight-Related Accident.

Explanation: Damage to property or injury to personnel without damage to aircraft.

Code: 41 Title: Instrument Meteorological Condition (IMC).

Explanation: Aircraft must be in IMC conditions when the accident/ emergency occurs.

This is a condition event and should not be used in the first position.

Code: 42 Title: Rappelling.

Explanation: Property damage or personnel injury occurring during rappelling operations.

Code: 43 Title: STABO.

Explanation: Property damage or personnel injury occurring during STABO operations.

Code: 44 Title: Overstress.

Explanation: Stress damage to aircraft as a result of operating aircraft outside the design limitations.

Code: 45 Title: Foreign Object Damage (FOD) Incident.

Explanation: Internal or external FOD damage confined to aircraft turbine engines only.

Code: 46 Title: Rotor/Prop Wash.

Explanation: Property damage or personnel injury resulting from rotor/prop wash (does not include damage incurred by event 75).

Code: 47 Title: Engine Overspeed/Overtemp.

Explanation: Engine RPM or temperature exceeding the allowable limits for continued operations.

Code: 48 Title: Brownout.

Explanation: Loss of visual reference to the ground or horizon caused by rotor wash swirling dust around the aircraft. This is a condition event and should not be used in the first position.

Code: 49 Title: Bird Strike.

Explanation: Accidents in which any part of the aircraft collides with a bird while in flight.

Code: 50 Title: Tree Strike.

Explanation: Accidents as a result of aircraft striking vegetation during any phase of flight.

Code: 51 Title: Wire Strike.

Explanation: Accidents as a result of the aircraft striking any kind of wires during any phase of flight.

Code: 52 Title: In-flight Breakup due to mast bumping.

Explanation: Accidents in which the main rotor separates as result of mast bumping.

Code: 53 Title: Missing Aircraft.

Explanation: Used when an aircraft does not return from a flight and is presumed to have crashed.

Code: 54 Title: FOD.

Explanation: Accident in which foreign object damage is the only damage incurred.

Code: 55 Title: Dynamic Rollover.

Explanation: Accident in which the main rotor blades strike the terrain as a result of exceeding the lateral CG limits, while the aircraft structure is still intact.

Code: 56 Title: Maintenance Operational Check (MOC).

Explanation: Accidents that occur during an MOC while the engine(s) is (are) in operation and/or rotors turning.

Code: 57 Title: Weapons Related.

Explanation: Accidents that result in property damage or injury to personnel.

Code: 58 Title: Lightning Strike.

Explanation: Damage to aircraft/injury to occupant because of lightning strike(s).

Code: 59 Title: Rescue operations.

Explanation: Property damage or personnel injury occurring during rescue operations.

Code: 60 Title: Object Strike.

Explanation: Aircraft/aircraft component struck objects other than ground, trees, or objects included in other events.

Code: 61 Title: Air to Ground Collision.

Explanation: Aircraft in the air collides with or strikes aircraft on the ground.

Code: 62 Title: Stump Strike.

Explanation: Aircraft contacts stump during routine landing.

Code: 63 Title: Antenna Strike.

Explanation: Aircraft damage caused by contact with an antenna.

Code: 64 Title: Engine Overtorque/Overload.

Explanation: Engines that have been subjected to torque loads beyond power limits specified, or engine loses rpm because of overload of aircraft for density altitude.

Code: 65 Title: Whiteout.

Explanation: Loss of visual reference to the ground or horizon caused by rotor wash swirling snow around the aircraft. This is a condition event and should not be used in the first position.

Code: 66 Title: Tiedown Strike.

Explanation: Damage to the aircraft caused by main rotor tiedown device attached to M/R rotor during engine start.

Code: 67 Title: Parachute.

Explanation: Accidents involving paradrop operations inside or still attached to the aircraft.

Code: 68 Title: Mast Bumping.

Explanation: Damage resulting from contact between the main rotor and mast but not resulting in rotor separation.

Code: 69 Title: Structural Icing.

Explanation: The formation of ice on aircraft structures to include the rotor systems.

Does not include carburetor, induction, or pitot static system icing.

Code: 70 Title: Engine Failure.

Explanation: Engine fails to develop sufficient power to maintain flight or internal failure of power plant. Excludes fuel starvation or fuel exhaustion and FOD.

Code: 71 Title: Transmission Failure.

Explanation: Internal failure of a main transmission.

Code: 72 Title: Vertical Fin Strike.

Explanation: Damage caused by the tail rotor blades coming in contract with the vertical fin on single rotor helicopters.

Code: 73 Title: Spike Knock.

Explanation: Damage occurred when the transmission spike contacts the striker plate with sufficient force to cause damage.

Code: 74 Title: Seatbelt/Restraint Harness Strike.

Explanation: Damage caused by unsecured seatbelts/restraint harnesses.

Code: 75 Title: Blade Flapping.

Explanation: Damage resulting from wind or rotor wash from other aircraft that causes the M/R blades to flap to the extent that damage occurs.

Code: 76 Title: Fuel Exhaustion.

Explanation: Power loss resulting from using all usable fuel aboard an aircraft.

Code: 77 Title: Fuel Starvation.

Explanation: The result of fuel ceasing to flow to the power plant while fuel is still on board the aircraft. Example: The pilot fails to switch tanks when one runs dry or blockage of fuel lines occurs because of contamination.

Code: 78 Title: Animal Strike.

Explanation: During takeoff, landing, etc., an animal is struck by any part of the aircraft.

Code: 79 Title: Battery Fire/Overheat.

Explanation: A fire in the battery compartment or overheated battery, usually resulting in electrical failure.

Code: 80 Title: Excessive Yaw/Spin.

Explanation: May occur on the ground or in the air (helicopter only). A maneuver where the aircraft yaws excessively or spins when power is added without adequate antitorque input or a loss of antitorque control occurs.

Code: 81 Title: Tail Boom Strike.

Explanation: Main rotor contacts tail boom on the ground due to wind conditions. Excludes hard landings and damage caused by rotor wash. In addition to events 70 and 71 listed above the following events are used to categorize materiel factor related mishap events. The event applies regardless of the cause of the failure/malfunction (FWT, maintenance, design or manufacture).

Code: 82 Title: Airframe.

Explanation: Failure/malfunction of any airframe structure to include doors windows, fairings, canopies, etc to include hardware.

Code: 83 Title: Landing Gear.

Explanation: Failure/malfunction of any landing gear part exclusive of the hydraulics.

Code: 84 Title: Power train.

Explanation: Failure/malfunction of any part/component of the power train except when events 47 or 70 applies.

Code: 85 Title: Drive Train.

Explanation: Failure/malfunction of any part/component of the drive train except when events 86 and 71 applies.

Code: 86 Title: Rotor/Propellers.

Explanation: Failure/malfunction of rotor/prop assemblies, hubs, blades, etc. Excludes other drive train part failures; e.g. gearboxes, mast etc.

Code: 87 Title: Hydraulics System.

Explanation: Failure/malfunction of any hydraulic part. The failure of other systems resulting from hydraulic initiated failures will be coded as hydraulic.

Code: 88 Title: Pneumatic System.

Explanation: Failure/malfunction of any pneumatic part. The failure of any other system resulting from pneumatic initiated failures will be coded as pneumatic.

A-12
Code: 89 Title: Instruments.

Explanation: Failure/malfunction of any part of the instrument system that results in a faulty instrument indication.

Code: 90 Title: Warning System.

Explanation: Failure/malfunction of any part of the warning system that results in an false indication of a failure/malfunction. Includes electrical components of the warning system.

Code: 91 Title: Electrical System.

Explanation: Failure/Malfunction of any part of the AC or DC electrical systems.

Includes current producing, transforming, converting and amplifying parts e.g. battery, generator, alternator, relay etc.

Code: 92 Title: Fuel System.

Explanation: Failure of any part of the fuel system. Does not include the fuel metering/fuel control unit that will be reported as part of the engine.

Code: 93 Title: Flight Control.

Explanation: Failure/malfunction of any part of the system. Excludes hydraulic part failures.

Code: 94 Title: Utility/Environmental Control System.

Explanation: Failure/malfunction of any part of the system.

Code: 95 Title: Avionics.

Explanation: Failure of any part of the radio navigation/communication equipment.

Code: 96 Title: Cargo Handling Equipment.

Explanation: Failure of the cargo handling equipment attached to the aircraft only.

Code: 97 Title: Armament.

Explanation: Failure of any part to include the aiming/firing system.

A.2 System Inadequacies/Hazards.

A. LEADER FAILURE

Code: 01 Title: Inadequate/improper supervision by higher command.

Code: 02 Title: Inadequate/improper supervision by staff officer.

Code: 03 Title: Inadequate/improper supervision by unit command.

Code: 04 **Title:** Inadequate/improper supervision by direct supervisor/noncommissioned officer in charge/platoon leader/instructor.

Explanation: NOTE: Inadequate supervision becomes a root cause when it leads to accident-causing personnel mistakes or materiel failure/malfunctions. Inadequate supervision is more clearly identifiable at the immediate-supervisor level.

B. TRAINING FAILURE

Code: 05 Title: Inadequate school training.

Explanation: School training becomes a root cause when people make accident-causing mistakes because the school training was inadequate in content or amount.

Code: 06 Title: Inadequate unit/on-the-job training.

Explanation: Unit/on-the-job training becomes a root cause when people make accidentcausing mistakes because the training provided was inadequate in content or amount.

Code: 07 Title: Inadequate experience.

Explanation: Supervised on-the-job experience is the follow-up to school and unit training programs. Experience becomes a root cause when people make accident-causing mistakes because the experience provided was inadequate in content or amount.

Code: 08 Title: Habit interference.

Explanation: Habit interference becomes a root cause when a person makes an accidentcausing error because task performance was interfered with either the way he usually performs similar tasks, or the way he usually performs the same task under different conditions or with different equipment.

C. STANDARDS FAILURE

Code: 09 **Title:** Inadequate written procedures for operation under normal or abnormal/emergency conditions.

Explanation: Inadequate written procedures (AR, TM, FM, SOP, written directives) become the root causes when they lead to accident-causing mistakes or materiel failures/malfunctions.

D. SUPPORT FAILURE

Code: 10 Title: Inadequate facilities/services.

Explanation: Inadequate facilities or services become root causes when the maintenance, space and/or support provided for personnel and materiel to accomplish their functions cause mistakes or failures/malfunctions that lead to accidents. (Examples of facilities or services are recreation areas, POL services, housing, medical clinics/hospitals, weather services, storage areas, maintenance facilities, and property disposal.)

Code: 11 Title: Inadequate/improper equipment design or equipment not provided.

Explanation: Improperly designed equipment and materiel or lack of equipment/materiel become root causes when the design or lack of equipment leads to accident-causing personnel errors or materiel failures/malfunctions.

Code: 12 Title: Insufficient number or type of personnel.

Explanation: Insufficient number or type of personnel becomes a root cause when people make accident-causing mistakes or materiel fails/malfunctions because the number or type of personnel provided was insufficient.

Code: 13 Title: Inadequate quality control, manufacture, packaging, or assembly.

Explanation: The inadequate manufacture, assembly, packaging, or quality control of materiel becomes a root cause when it leads to accident-causing personnel errors or materiel failures/malfunctions. (Note: Includes original manufacture and rebuild.)

Code: 14 Title: Inadequate maintenance.

Explanation: Inadequate maintenance (inspection, installation, troubleshooting, recordkeeping, etc.) becomes a root cause when it leads to accident-causing personnel errors or materiel failures/malfunctions.

E. INDIVIDUAL FAILURE

Code: 15 **Title:** Fear/Excitement/Anger (inadequate composure).

Explanation: Each person is a part of the system. Therefore, his state of mind is a system element. Inadequate composure is a temporary state of mind that becomes a root cause when a person makes an accident-causing error because of fear, excitement, or some related emotional factor made clear, rational thought impossible.

Code: 16 Title: Overconfidence/complacency in abilities.

Explanation: Overconfidence is a temporary state of mind that becomes a root cause when an accident is caused by a person's unwarranted reliance on: his own ability to perform a task, the ability of someone else to perform a task, the performance capabilities of equipment or other materiel.

Code: 17 Title: Lack of confidence.

Explanation: Lack of confidence is a temporary state of mind that becomes a root cause when an accident is caused by a person's unwarranted lack of reliance on; his own ability to perform the task, the ability of someone else to perform the task, the performance capabilities of equipment or other materiel.

Code: 18 Title: Haste/Attitude (poor motivation).

Explanation: Haste/attitude (poor motivation) is a temporary state of mind that becomes a root cause when a person makes an accident-causing mistake because he/she is in a hurry (haste), or has a poor/bad attitude.

Code: 19 Title: Fatigue (self-induced).

Explanation: Fatigue is a temporary physical and/or mental state that becomes a root cause when a person makes an accident-causing error because of reduced physical or mental capabilities resulting from previous activity and/or lack or rest.

Code: 20 Title: Effects of alcohol, drugs, illness.

Explanation: The temporary effects of alcohol, drugs, or illness become a root causes when a person makes an accident-causing error because of reduced physical or mental capabilities resulting from one or more of these effects.

Code: 21 Title: Environment conditions.

Explanation: Unknown or unavoidable conditions, which result in materiel failure or induce human error.

Code: 97 **Title:** Insufficient information to determine system inadequacy/cause.

Appendix B: Glossary of Safety Related Terms

Aborted takeoff

An unplanned event that occurs before intent for flight exists, with engine(s) running, that interrupts a planned flight (except for maintenance test flights and factory acceptance flights).

Accident

An unplanned event that causes personal injury or illness, or property damage.

Active Army personnel

Members of the Army on full-time duty in active military service, including cadets at the

U.S. Military Academy.

Aircraft

A manned weight carrying structure for navigation of the air that is supported by its own buoyancy or the dynamic action of the air against its surfaces.

Aircraft ground accident

Injury or property damage accidents involving Army aircraft in which no intent for flight exists, and the engine(s) is/are in operation.

Army accident

An accident that results in injury/illness to either Army or non-Army personnel, and/or damage to Army or non-Army property as a result of Army operations (caused by the Army).

Army civilian personnel

a. Senior Executive Service, General Management, General Schedule, and Federal Wage System employees. b. Corps of Engineer Civil Works employees.

c. Army National Guard and Army Reserve technicians.

d. Nonappropriated fund employees (excluding part-time military).

e. Youth/Student Assistance and Temporary Program employees; Peace Corps and Volunteers in Service to America (VISTA) volunteers; Job Corps, Neighborhood Youth Corps, and Youth Conservation Corps Volunteers; Family Support Program volunteers.

Commander

An individual that exercises authority and responsibility over subordinates by virtue of rank or position. The purpose of that authority and responsibility is to effectively use available resources and plan the employment of, organize, direct, coordinate and control the actions of an Army organization for the purpose of successful mission accomplishment. Examples of commanders are as follows:

a. Commander of a major Army command, CONUS and OCONUS.

b. The Chief of Engineers (civil and military works).

c. Commander, U.S. Army Space and Strategic Defense Command.

d. The Chief, Army National Guard Bureau.

e. Commander, U.S. Army Medical Research and Development Command.

f. Commanders of Army installations with a full-time safety professional. This includes posts, camps, stations, and military communities.

g. State adjutants general (ARNG).

h. Commanders of Army Reserve organizations with a full-time safety professional.

i. Commanders of medical treatment facilities.

j. Commanders in direct support of general support maintenance units.

k. Director of Facilities Engineering.

1. Provost Marshal/Law Enforcement Commander.

m. Director of Industrial Operations.

n. U.S. Army Plant Representative Office.

o. Commander of TOE, MTOE, or TDA organization.

Competent medical authority

Any duly qualified physician (Government or private), who is approved by the Office of Workman's Compensation to render treatment. "Competent medical authority" includes surgeons, podiatrists, dentists, clinical psychologists, optometrists, chiropractors, and osteopathic practitioners.

Contractor accident

An accident that occurs as a result of a Government contractor's operations in which there is damage to U.S. Government or Army property or equipment, injury or occupational illness to Army personnel, or other reportable event.

Destroyed aircraft

An aircraft is considered destroyed/total loss when the estimated cost to repair exceeds the current full-up replacement cost.

Drone aircraft

Those serial vehicles having a "Q" designator and which can be flown or operated by rated or non-rated personnel, or which can be flown or operated in the remote control configuration.

Emergency

An event for which an individual perceives that a response is essential to prevent or reduce injury or property damage.

Environmental factors

Environmental conditions that had, or could have had an adverse effect on the individual's actions or the performance of equipment.

Fair wear and tear

Damage to time-between-overhaul (TBO) items such as gearboxes, tires, and other items that deteriorate with use. (Hot starts, overspeeds, and overtorques are not considered fair wear and tear.)

Flight crew

Personnel on flight pay who are involved in operation of the aircraft.

Forced landing

A landing caused by failure or malfunction of engines, systems, or components that makes continued flight impossible.

Foreign object damage (FOD)

Damage to Army vehicle/equipment/property as a result of objects alien to the vehicle/equipment damaged. Excludes aircraft turbine engine(s) defined as a FOD incident.

Fratricide/Friendly Fire (FF)

A circumstance applicable to persons killed or wounded, or equipment damaged, in military action, mistakenly or accidentally, by friendly forces actively engaged with the enemy, who are directing fire at a hostile force or what is thought to be a hostile force. Fratricide/FF incidents will be primarily investigated and reported under DODI 6055.7.

Ground accident

Any accident exclusive of aviation (flight/flight related) (for example, AMV, ACV, POV, marine.)

Hospitalization

Admission to a hospital as an inpatient for medical treatment.

Human error

Human performance that deviated from that required by the operational standards or situation. Human error in accidents can be attributed to a system inadequacy/root cause in training, standard, leader, individual, or support failure indicated below:

Human factors

Human interactions (man, machine, and/or environment) in a sequence of events that were influenced by, or the lack of human activity, which resulted or could result in an Army accident.

Individual failure

Soldier knows and is trained to standard but elects not to follow standard (self-discipline--mistake due to own personal factors).

Initial Denial Authority

The official at HQDA-level with the authority to deny release of a document, in whole or in part, under the Freedom of Information Act.

Injury

A traumatic wound or other condition of the body caused by external force, including stress or strain. The injury is identifiable as to time and place of occurrence and member or function of the body affected, and is caused by a specific event or incident or series of events or incidents within a single day or work shift.

Installation-level safety manager

a. The senior full-time safety professional responsible for providing safety support to Army installations, including camps, stations, military communities, and USAR organizations.

b. State Safety Manager or Specialist (ARNG).

Intent for flight

Intent for flight begins when aircraft power is applied, or brakes released, to move the aircraft under its own power with an authorized crew. Intent for flight ends when the aircraft is at a full stop and power is completely reduced.

Investigation

A systematic study of an accident, incident, injury, or occupational illness circumstances.

Lost-time case

7

A nonfatal traumatic injury that causes any loss of time from work beyond the day or shift in which it occurred or a nonfatal non-traumatic illness/disease that causes disability at any time. This definition will be used when computing civilian lost-time frequencies for DOL reporting.

Lost-workday case involving days away from work

Cases in which an accident results in Army personnel missing one or more days of work. Days away from work are those workdays (consecutive or not) on which Army personnel would have worked but could not because of injury, occupational illness, or job-related physical deficiencies detected during medical surveillance examinations. Excluded are days that Army personnel would not have worked even though able to work (for example, weekends or holidays) and the day of the injury or onset of occupational illness.

Materiel factors

When materiel elements become inadequate or counter-productive to the operation of the vehicle/equipment/system.

Medical treatment

Any treatment (other than first aid) administered by a physician or by registered professional medical personnel under the orders of a physician.

Nonappropriated fund (NAF) employees

Employees paid from nonappropriated funds, including summer and winter hires and special NAF program employees. Military personnel working part-time in NAF employment are excluded.

Nonfatal case without lost workdays

Cases other than lost-workday cases where Army military or civilian personnel, because of an injury or occupational illness, experienced one or more of the following:

- a. Permanent transfer to another job or termination.
- b. Medical treatment greater than first aid.
- c. Loss of consciousness.

d. Restricted work activity or profile.

e. Diagnosis as having an occupational illness that did not result in a fatality or lostworkday case. This includes newly diagnosed occupational illnesses detected on routine physical examinations.

Occupational illness

Non-traumatic physiological harm or loss of capacity produced by systemic infection; continued or repeated stress or strain; for example, exposure to toxins, poisons, fumes; or other continued and repeated exposures to conditions of the work environment over a long period of time. Includes any abnormal physical or psychological condition or disorder resulting from an injury, caused by long- or short-term exposure to chemical, biological, or physical agents associated with the occupational environment. For practical purposes, an occupational illness is any reported condition that does not meet the definition of an injury.

Occupational injury

A wound or other condition of the body caused by external force, including stress or strain. The injury is identifiable as to time and place of the occurrence and a member or function of the body affected, and is caused by a specific event or incident or series of events or incidents within a single day or work shift.

Off-duty

Army personnel are off-duty when they:

a. Are not in an on-duty status, whether on or off Army installations.

b. Have departed official duty station, temporary duty station, or ship at termination of normal work schedule.

c. Are on leave and/or liberty.

d. Are traveling before and after official duties, such as driving to and from work.

e. Are participating in voluntary and/or installation team sports.

f. Are on permissive (no cost to Government other than pay) temporary duty.

g. Are on lunch or other rest break engaged in activities unrelated to eating or resting.

On-duty

Army personnel are on-duty when they are:

a. Physically present at any location where they are to perform their officially assigned work. (This includes those activities incident to normal work activities that occur on Army installations, such as lunch, coffee, or rest breaks, and all activities aboard vessels.
b. Being transported by DOD or commercial conveyance for the purpose of performing officially assigned work. (This includes reimbursable travel in POVs for performing TDY, but not routine travel to and from work.)

c. Participants in compulsory physical training activities (including compulsory sports).

Permanent total disability

Any nonfatal injury or occupational illness that, in the opinion of competent medical authority, permanently and totally incapacitates a person to the extent that he or she cannot follow any gainful employment. (The loss or loss of use of both hands, feet, eyes, or any combination thereof as a result of a single accident will be considered as permanent total disability.)

Permanent partial disability

Any injury or occupational illness that does not result in death or permanent total disability but, in the opinion of competent medical authority, results in the loss or

B-9

permanent impairment of any part of the body, with the following exceptions:

a. Loss of teeth.

b. Loss of fingernails or toenails.

c. Loss of tip of fingers or tip of toe without bone involvement.

d. Inguinal hernia, if it is repaired.

e. Disfigurement.

f. Sprains or strains that do not cause permanent limitation of motion.

Precautionary landing

A landing resulting from unplanned events that makes continued flight inadvisable.

Preexisting physical condition

A medical condition that existed prior to the occurrence of the accident.

Recommendations

Those actions recommended to the command to correct system inadequacies that caused, contributed, or could cause or contribute to an Army accident. Also referred to in this pamphlet as corrective action, remedial measures and/or countermeasures.

Recordable

Reportable accident that meets the minimum criteria stated in the regulation for Class A-

D accidents and Class E and FOD incidents.

Reportable

All occurrences that cause injury, illness, or property damage of any kind must be reported to the soldier's/employee's/unit's servicing/ supporting safety office.

Restricted work activity

Individual's injury is such that they are unable to perform their normal duties (for example, light-duty, profile).

ROTC personnel

a. Members of the ROTC during periods of basic or advanced training at premises owned or under the control of the Army whether on or off duty.

b. Cadets performing professional enrichment training while under Army supervision and directed by competent orders, regardless of the location of the training site. Regular

training on

campus is excluded; that is, weekly drill and classroom instruction.

c. Cadets involved in rifle and pistol marksmanship training under Army supervision on any firing range.

d. Cadets undergoing ROTC flight instruction.

Standards failure

Standards/procedures not clear or practical, or do not exist)

Support Failure

Inadequate equipment/facilities/services in type, design, availability, or condition, or insufficient number/type of personnel, which influenced human error, resulting in an army accident.

System inadequacy

A tangible or intangible element that did not operate to standards, resulting in human error or materiel failure. Also referred to in this pamphlet as causes, readiness shortcomings and/or root causes.

Training failure

Soldier/individual not trained to known standard (insufficient, incorrect or no training on task--insufficient in content or amount)

Appendix C: Army Accident Classification and Investigation

A.1 General

Army accident classes are used to determine the appropriate investigative and reporting procedures. The same classification system is used for accidents that occur in the air or on the ground. The classification definitions below are from Army Regulation 385-40 <u>Accident</u> <u>Reporting and Records</u>, which prescribes policy on accident reporting and recordkeeping procedures for the Army. This updated publication was extensively revised in early 1994 and released in November of 1994. Due to the extensive revisions, the changed portions of the regulation were not highlighted. This is the most current regulation update and is currently used in the U.S. Army Safety Center. AR 385-40 applies to the Active Army, the Army National Guard, the U.S. Army Reserve, and Army appropriated fund employees. This regulation is also applicable during full mobilization. This covers the spectrum of personnel involved in the accidents in this research and anything reported during the conflict in the Middle East.

A.2 Accident Classifications

Accident classes are as follows:

(1) **Class A accident:** An Army accident in which the resulting total cost of property damage is \$1,000,000 or more; an Army aircraft or missile is destroyed, missing, or abandoned; or an injury and/or occupational illness results in a fatality or permanent total disability.

(2) **Class B accident:** An Army accident in which the resulting total cost of property damage is \$200,000 or more, but less than \$1,000,000; an injury and/or occupational illness results in permanent partial disability, or when five or more personnel are hospitalized as inpatients as the result of a single occurrence.

(3) **Class C accident:** An Army accident in which the resulting total cost of property damage is \$10,000 or more, but less than \$200,000; a nonfatal injury that causes any loss of time from work beyond the day or shift on which it occurred; or a nonfatal occupational illness that causes loss of time from work (for example, 1 work day) or disability at any time (lost time case).

(4) Class D accident: An Army accident in which the resulting total cost of property damage is\$2,000 or more but less than \$10,000.

Note. Nonfatal injuries/illnesses (restricted work activity, light duty, or profile) will only be recorded in ASMIS in conjunction with recordable property damage accidents.

(5) **Class E aviation incident:** An Army incident in which the resulting damage cost and injury severity do not meet the criteria for a Class A-D accident (\$2,000 or more damage; lost time/restricted activity case). A Class E aviation incident is recordable when the mission (either operational or maintenance) is interrupted or not completed. Intent for flight may or may not exist. An example of a recordable Class E incident is: during maintenance operational check (MOC) the engine quits. Examples of nonrecordable Class E incidents are: chip detector light illumination and the component is not replaced; mission interrupted/aborted because of weather, unless mission is canceled; failure of Fair Wear and Tear (FWT) items found on pre- or post-flight inspection; radio failure where radio is replaced; closing a door found open in flight.

(6) Foreign Object Damage (FOD) aviation incident (Also known as Class F incident): Recordable incidents confined to aircraft turbine engine damage (does not include installed aircraft Auxiliary Power Units (APU)) as a result of internal or external FOD, where that is the only damage.

A.3 Accident Investigation Boards

There are different investigation procedures for different classes of accidents, these are described below.

(1) The following accidents will be investigated by a board consisting of a minimum of three members. The members should be officers, warrant officers, or DA safety and occupational health specialists/managers/engineers, GS-018/803-11/12/13/14.

(a) All Class A and B accidents, except those involving off-duty military fatalities/injuries not involving military operations.

(b) Any accident, regardless of class, that an appointing authority believes may involve a potential hazard serious enough to warrant investigation by a multimember board.

(2) Class C aircraft accidents (flight, flight related, or aircraft ground) will be investigated by a board of at least one officer, warrant officer, or DA safety and occupational health specialists/ managers/engineers, GS-018/803-9/11/12/13/14 (DA safety professional must directly manage an aviation safety program).

(3) When an accident involves Army property and another U.S. Military Service's property, a single joint board may be convened. Board members may be from the two Services involved. Appointment of the members and identification of a senior member as president will be made by mutual agreement between the commanders of the two Safety Centers. For uniform reporting within each service, the board's proceedings will be recorded in the format required by each service.

(4) The following accidents will be investigated by one or more officers, warrant officers, safety officers/NCOs, supervisors, or DA safety and occupational health specialist/manager/engineer, GS-018/ 803-9/11/12/13/14:

(a) Class C accidents.

(b) Class D accidents, Class E, and FOD incidents.

•

Appendix D: Accident Severity Data

General: This data set was generated in cooperation with the Operations Research & Systems Analysis / Statistics Division of the U. S. Army Safety Center. The data was extracted from the Aviation Safety Management Information System (ASMIS) database. The information in this spreadsheet depicts the Class a through C accidents that occurred from November 1987 to October 1998. The measures across the top of the spreadsheet represent the measures that were developed to determine accident severity. Each line represents a single accident. These are the raw scores that were input into the value functions described in Chapter Three.

c

COUNT	EVENT1	DAMCOST	INJCOST	TOTFTL	TOT DIS	PART DIS	DAYHOSP	DAYLOST	DAYREST	FLSPL	OLSPL	HYSPL
	5	0	340	0 0	0 0	0 0	0 0	2 1	0 0			
	5 5	11523	5 0	5 0	5 0	5 0	5 0	5 0				
	5 8	30200	5 0		- c				• c			
	5 2	1244U					• -	o c				
	01 Max	59688	340	0 0	0	0	0 0	0 01	. 0	0	0	0
	01 Min	0	0	0	0	0	0	0	0	0	0	0
	01 Total	113857	340	0	0	0	0	2	0	0	0	0
5	01 Average	22771.4	68	0	0	0	0	0	0			
01 Count	0.003687316											
	02	248086	0	0	0	0	0	0	0			
	02 Max	248086	0	0	0	0	0	0	0	0	0	0
	02 Min	248086	0	0	0	0	0	0	0	0	0	0
	02 Total	248086	0	0	0	0	0	0	0	0	0	0
1 02 Count	02 Average 0.000737463	248086	0	0	0	0	0	0	0			
	εÜ	125256	c	c	c	C	C	o	0			
	3 8	10000	o c	, c		• c						
	3 8	C00071	-	5 0	5 0	5 0	,	.				
	50 13	14640	5 0	5 0	5 0	5 0	5 0			c	c	c
		05251	5 0		5 0				o c	o c		
		14640		5 0	5 0		5 0	5 0			5 0	
	03 i otal	260381		5	5		5 1	5 0		5	5	2
3 03 Count	03 Average 0.002212389	86860.33333	0	o	o	0	o	Ð	D			
	04	10697	13885	0	0	0	10	25	30			
	04 Max	10697	13885	0	0	0	10	25	30	0	0	0
	04 Min	10697	13885	0	0	0	10	25	30	0	0	0
	04 Total	10697	13885	0	0	0	10	25	30	0	0	0
1 04 Count	04 Average 0.000737463	10697	13885	0	0	0	10	25	30			
	05	0	755	o	0	o	-	ю	0			
	05	24179	0	0	0	0	0	0	0			
	05	55563	0	0	0	0	0	0	0			
	05	121923	0	0	0	0	0	0	0			
	05	10000	0	0	0	0	0	0	0 0			
	05	54053	0	0 0	5 0	5 0	5,	5 (5 0	c	c	c
	U5 Max	121923	cc)	5 0	5 0	- 0	- c	° C		5 0		
	O5 Total	U 765718	0 755				- -	5 00		00	0	. 0
ų	US FURI	01/002	125 833333				- c	o) C	5	0	•
05 Count	0.004424779	000000000000000000000000000000000000000	0000000	þ	5	5	>	-	,			
	90	0	820	0	0	Q	0	-	14			
	90	15853	٥	0	0	0	0	0	0			
	90	190817	15844	0	0	0	34	0	0			
	06 Max	190817	15844	0	0	o	34	-	14	0	0	0
	06 Min	0	0	0	0	0	0	0	0	0	0	0
	06 Total	206670	16664	0	0	0	34	-	14	0	0	0
e	06 Average	68890	5554.666667	0	0	0	1	0	5			
06 Count	0.002212389	00000	c	c	c	c	c	c	c			
	ò	2380U	5	5	5	2	5	5	2			

HYSPL	0	0	0		n	4	-	-						¢	0 4	r		ю	ę		ı	N	Ŧ		-		Ŧ	. 4 -1									ব	-	28	2.153846154				0	0	0	
OLSPL	0	0	0		ы	4		-						- 0	04	r		ო	ę			с о	<i>с</i> о с	n			e.) M									4	÷	38	2.714285714				0	0	0	
FLSPL	0	0	0		ъ	5	2	-						•				e	e		ı	Ð					ç	1 01									5	-	28	3.11111111				0	0	0	
DAYREST		. 0	0	0	0	0	0	-	0	0	0	0	0	0 0	- 7	50	0	0	e	0	0 0	0 0	0 0	- 0			ی ای	0	0	0	0	0 0	0 602	1411	0	0	1411	0	1881	54	0	0	0	0	0	0 0	Þ
DAYLOST		0	0	o	0	0	0	4	9	o	0	0	0	0 0	- c	00	0	o	180	0	0 (0 (0 0	5 0	5 0		> %	0	0	0	0	0 0	0 750	155	0	0	180	0	583	11	o	o	0	0	0	0 0	c
DAYHOSP		0	0	o	0	0	0	с	~	0	0	0	0	180		0	0	5	60	0	0 1	0 1	0 0	0 0	5 0	5 0	5α	0	0	0	0	0 0	0 %	156	0	0	180	0	491	14	0	0	0	0	0	0 (c
PART DIS	50	0	0	o	0	0	0	0	0	0	0	0	0	- (-	00	0	0	-	0	0	0	0 0	0 0	5 0)	0	0	0	0 (- c	00	0	0	-	0	ę	0	0	0	0	0	0	0 (c
TOT DIS		00	0	0	o	0	0	0	0	o	o	0	0	0 0		0	0	0	0	0	0	0	0 0	0 (0 0			00	0	0	0	0 1	- c	- 0	0	0	-	0	•	o	0	0	0	0	0	0 (D
TOTFTL	5 0	00	o	0	17	0	0	0	0	0	0	0	. .	0 0		0	0	0	0	0	0	0	0 0	5 0	50		כ ג	00	0	0	0	ი ი	2 4	00	0	0	17	0	34	0.971428571	0	0	0	0	0	0 (þ
INJCOST		0	0	0	1092000	1232000	0	2065	2716	0	o	0	0	1310000	1680	000	0	2330	210360	0	0	2200000	0 0	0 0	0 0		1415778	2585000	0	0	0	2200000	110000	1650891	0	0	2585000	0	16023058	457801.6571	0	0	0	0	0	0 (c
DAMCOST	79850	15881	45731	22865.5	4618000	4628026	460000	460000	1256406	1198225	19388	177237	446373	92290	2/013/ 778627	13134	13134	10219706	10219706	370111	199458	3697532	1041616	922/04	922704	11430	1432	922704	8443	42221	38474	6459168	6453598 ee1e073	6600000	0	200000	10219706	0	78972527	2256357.914	14539	34966	23342	34966	14539	72847	24282.33335
EVENT1	07 Max	07 Min	07 Total	07 Average 0.001474926	08	80	80	80	08	80	80	08	80	88	80 80	88	88	08	80	88	88	08	88	88	88	9 8	88	88	80	80	08	88	80 8	88	80	80	08 Max	08 Min	08 Total	08 Average 0.025811209	11	11	1	11 Max	11 Min	11 Total	11 Average
COUNT				2 07 Count																																				35 08 Count							ю

-

HYSPL																																											
OLSPL																÷																											
FLSPL																6	I																										
DAYREST	00	00	0	0 (0 0	- c	0	0	0			0	0	0 0	0 0	- c	0	0	0 0		00	0	00	0		0 0	0 C	0	0	ə c	00	0	0 0	00	>0	00	0	0 0	5 0	00	0	00	5
DAYLOST	00	5 6	0	0	00	- c	00	0	0	00		0	0	0	00		00	0	0 0	5 C	00	0	00	00	0	0 (0 0	00	0	5 0	00	0	0 (00	> 0	00	0	0 0	50	00	0	00	2
DAYHOSP	00	00	0	0	00	5 C	00	0	0	00		00	0	0	00		00	0	0 0	5 c	00	0	00	00	0	0 (0 0	0	0	- -	0	0	0 (00	> 0	00	0	0 0	50	00	0	00	5
PART DIS	00	00	0	0	00	- c	0	0	0	0 0		00	0	0	0 0		00	0	0 0	- c	00	0	0 0	0	0	0 (0 0	0	0		0	0	0 0	0 0	- c	0	0	0 0	5 0	0 0	0	00	-
TOT DIS	00	00	0	0	00	- c	00	0	0	0 0		00	0	0	0 0		00	0	0 0	- -	00	. 0	00	0 0	0	0	0 0	00	0	5 0	00	0	0 1	00	5 C	00	0	0 0	00	0 0	o	00	þ
TOTFTL	0 0	0 0	0	0	0 0		00	0	0	00	- c	0	0	0	0 0	0 0	00	0	0 0	- -	00	0	00	00	0	0	0 0	0 0	0	0 0	00	0	0 0	00	. c	00	0	0 (0 •	- 0	0	00	þ
INJCOST	0 0	1200	0	0	0 0	- c	00	0	0	0 0		00	0	0	0 0		00	. 0	0 (00	00	0	0 0	0	0	0	0 0	• •	0	0 0	00	0	0 (0 0	5 C	00	0	0 0	1100000	000001	0	0 0	5
DAMCOST	63822	0	17526	40005	40385	49605 76089	141216	150864	92731	10193	02128	103424	123104	238012	87301	43138 710137	375360	10000	42169	24433 30226	30220 86175	57553	10931	54000	24433	297659	6266399 33800	33805	18427	339/9 127011	69545	36613	25302	114202	10000	51238	25390	24260	15805010 EEA	202 98947	77120	10000	76671
EVENT1 0.002212389	55	2 2	15	5	5 5	5 5	1 12	12	. 12	5 5	5 5	1 5	12	12	5 5	2 5	15	12	12	t 1	2 2	12	5 5	12	: 12	12	5 5	10	12	5 ¢	1 51	12	12	66	<u>5</u>	4 6	12	5	5 Q	<u>5</u> 5	12	12	<u>7</u>
COUNT 11 Count																																											

,

Jan San San San San San San San San San S	000	000	000
OLSPL 4	ی ت	000	000
FISPL	N N N N	000	000
D A YRES A C C C C C C C C C C C C C C C C C C C	000040	000000 00	0000
D AYLOS 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 7 0 7 0	000000 %0	ဝမ္ပဝမ္ပ
Ч КОН ХОН ХО С С С С С С С С С С С С С С С С С С			0 2 0 2
Free C C C C C C C C C C C C C C C C C C			
5 6 7 8 9 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9			0000
H H 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 2 3 3 0.033707865		0000
INJCOST 1878 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	u 1370000 2474278 27800.8764	0 0 23752 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 23752 0 23752
DAMACOST 35436 85846 85846 85846 85846 17004 75142 75142 75142 75140 75140 65603 45509 77540 65603 45509 178576 178576 178576 178240 178598 45409 89559 895598 10197 751137 128571 991301 10197 75245 2588575 10000 8000000 8000000 8000000 8000000 8000000	102000 15805010 0 38083734 427907.1236	23190 10000 23190 10000 33190 16595 60309 4720000	50773 4720000 50773 4831082
M NA C C C C C C C C C C C C C C C C C C C	12 12 Max 12 Min 12 Total 12 Average 0.065634218	15 15 15 Max 15 Min 15 Min 15 Average 0.001474926 20 20	20 20 Max 20 Min 20 Total
OOUNT	89 12 Count	2 15 Count	

HYSPL		0	0	0										-																									ო				-	•	-				
OLSPL		0	0	0									(n							-																				-		-	•	-				
FLSPL		0	0	0																																					-			•					
DAYREST 0	0	0	0	0	0		0	0	0	0	0	0	0	0 0	5 0		• -	00	0	0	0	0	0	0	0	0	20	5 0		- c	0 0	0	0	0	0 0		0	0	0	0	0	0	4	0 0	5 0		00	0	
DAYLOST 12	0	0	0	0	0		0	0	0	0	o	0	0	0 0	5 0			. 0	0	0	0	0	0	0	ò	0 1	0 0	5 0			0	0	0	0	0 0		00	0	0	0	0	D	0	0 0	50		>0	0	
DAYHOSP 7	0	0	0	0	0		0	0	0	0	0	0	0	0 0	5 0		o c	0	0	0	0	0	0	0	0	0 .	4 (5 0	,		0 0	0	0	0	0 0	- c	00	0	0	0	0	0	0	0 0		0 0	> 0	0	
PART DIS 0	0	0	0	0	0		0	0	0	0	0	0	0	0 0	- -			00	0	0	0	0	0	0	0	0	0 0	0 0			0 0	0	0	0	0 0		00	0	0	0	0	0	0	0 0	5 0	b c	0	0	
TOT DIS 0	0	0	0	0	0		0	0	0	0	0	0	0	0 0	5 0		o c	00	0	0	0	0	0	0	0	0	0 (- 0	5 0		0	0	0	0	0 0		0	0	0	0	0	0	0		5 0		00	0	
TOTFTL D	0	0	0	0	0		0	0	0	0	0	0	0	0 0	5 0		o c	00	0	0	0	0	0	0	0	0	0 0	0 0	5 0		, a	0	0	0	0 (0	0	0	0	0	0	0	0 0	5 0		00	0	
INJCOST 7917.333333	0	0	0	0	0		o	0	0	0	0	o	0	0 0	5 0		0 0	00	0	0	0	0	0	0	0	0	4264 ĵ	0 0	5 0	- c	0	0	o	0	0 0		00	0	0	0	0	0	480	0 0	50		> 0	0	
DAMCOST 1610360.667	289019	289019	289019	289019	289019		33499	152675	81957	287387	19810	18246	53215	246415	0006/	, 2100 , 2000	15/15	36490	190817	341141	117432	10000	55366	41669	167819	37085	92290	51909	60/101	00000	93472	25504	19132	362114	49318	133281	173333	76674	103285	106917	272407	19753	10680000	92110	38020	7 1001	43024 332608	277856	
EVENT1 20 Average 0.002212389	22	22 Max	22 Min	22 Total	22 Average	0.000/3/463	23	23	23	23	23	23	23	83	S 8	S 5	3 8	23	ន	23	5	23	23	23	23	53	53	82	5 6	3 8	3 8	18	23	23	53	5 5	23 23	ន	23	23	23	23	23	23	R 8	38	3 8	53	
COUNT 3 20 Count	1000 04																																																

HYSPL	б	ი – თ წ	000
OLSPL	N	1 3 1.428571429	000
FLSPL			000
DAYREST 0 0 0			
DAYLOST 0 0 0			
DAYHOSP 0 0 0		0000000004040 000	
PART DIS 0 0 0			
TOT DIS 0 0 0			
ТОТFTL 0 0 0	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~		
INJCOST 0 0 0		0 0 0 0 0 4744 63.2533333 65.25533333 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
DAMCOST 57222 29635 151269 396000	18305 18305 11890 271552 271552 296726 99620 99620 99620 78791 11878 111878 111878 11878 15154	4.6401 4.6603 108823 122198 133885 153885 16000 10000 10000 100000 10880000 1088000 1088000 100000 1086000 1086000 100000 83314 83314 100000 83314 83515 83515 83515 83515 83515 83515 831	207002 207002 40000 150000 263751 64351 101089 34569 34569 34569 5874808 5879892 20572 20572 7241733 517266.6429 517266.6429
EVENT1 23 23 23 23	***	23 23 23 23 23 23 23 23 23 23 23 23 23 2	24 24 24 24 24 24 24 24 24 24 24 24 24 2
COUNT		75 23 Count	14 24 Count

HYSPL 1	0 – 0 – 0 – 0 – 0 – 0 – 0 – 0 – 0 – 0 –	000	
Tasto Orsbr		000	0-
FLSPL 3 3 3	40°5° 000	000	യ തയന
DAYREST 90 00 00 00 00 00 00 00 00 00 00 00 00		0 00000	00000000000000000000000000000000000000
DAYLOST 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 00000	
D VHO V V V V V V V V V V V V V V V V V V		0 00000	
A A D D D D D D D D D D D D D D D D D D		0 00000	
TOT 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 00000	
		0 00000	
NJ 000000000000000000000000000000000000		0 00000	0 0 0 0 0 0 0 0 0 0 0 0 0 8
DAMCOST 96492 28527 56394 140418 27301 27301 277595 55292 55292 55292 55292 1128491 14232 55600 55550 55500 115304 115304 35665 55600 115304 115304 11555 55600 55755 55600 11566 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 1157555 55600 56600 56600 56600 56600 56600 56600 56600 56600 56600 56600 56600 56600 56600 56600 56600 56600 566000 56600 566000 566000 566000 566000 56600000000	110000 1140418 110000 1114711 58669 922704 922704 922704 922704	922704 811211 811211 811211 811211 811211	82.802 852.802 855.1892 855.1892 80.4688 187.237 1716.23 187.238 1716.23 1716.23 187.239 87.15847 87.15847 87.15847 87.15847
E V N N N N N N N N N N N N N N N N N N	25 Min 25 Min 25 Nura 25 Total 25 Average 0.014011799 26 26 Min 26 Min 26 Total	26 Average 0.000737463 28 Max 28 Min 28 Min 28 Total 28 Average 0.000737463	29 29 29 29 29 29 29 29 29 29 29 29 29 2
COUNT	19 25 Count	1 26 Count 28 Count	10 29 Count

HYSPL	-		Ŧ	-	-	-	-	c	ŋ	2				·	- •	-		n	-	c	ю (. 1	4 •	- 1	c	10	2			ο.	-				~		-											
OLSPL	۲		c	0	-	7	ო	c	n	•						-		ю	en L		4	ю ·	4 (20	σ -	- •	D	-	-			-					۴-											
FLSPL	4		•	-			-	L	n				5			c,		5	5			ლ I	<u>م</u>	۵ ч	n	-	- S			-		e.	,		o م	þ	ю											
DAYREST	14	0	0	<u>6</u> 0	06	0	06	0 4	0	189	-	ę	0	0 0	5 1	- 0	0	0	0	28 î	0 !	1 8	-	- 0	5 0	0 0	73	0	o	0	00	14	240	0	0 0	00	0	0	42	0	0 1		⊃ 7	- 	00	0	0	c
DAYLOST	21	0	0 8	7 0	50	0	4	0 0	- c	3 6	0	0	0	0 0	- Ç	2 0	0	17	0	0 (0	0 1	- 0	- 0		מי כ	101	0	0	0	09 0	o c	107	0	00	0	0	0	0	0	0 1	- c	- c		00	0	0	o
DAYHOSP	5	0	0 [à	12	0	¢	0 0	5 0	o vo	0	0	0	0 0	-	10	0	0	0	0 (0	0 1	5 0	5 0	ə c	0 m	. ი	0	0	0	ę c		, 198	0	00	00	0	0	44	ო	0	5.	4 C		00	0	0	n
PART DIS	۴	0	0 0		0	0	0	0 0	5 0	00	-	0	0	0 0	5 0	00	0	0	0	0 0	0	0	0 0	5 0		00	0	0	0	0	00		00	0	00	00	0	0	0	0	0	0 0	5 0			0	0	D
TOT DIS	0	0	0 0	- c	0	0	0	0 0	5 0	00	0	0	0	0 0		- 0	0	0	0	0	0	0	0 0	0 0	0 0	00	0	0	0	0	0 0	• c	00	0	0 0	00	0	0	0	0	0	0 0	- c	- c		0	0	D
TOTFTL	0	0	0 0	- <i>c</i>	10	0	•	0 0		0 0	0	0	-	0 0	5 0	9 0	0	9	e	0	0	0	4 (NG	ກດ	00) m	0	2	0	00	• c	00	0	ი ,	- 0	ę	0	-	Ð	0	0 0	- -	- c	- c	0	0	D
INJCOST	85595	0	432000	246050 480000	16990	0	341330	0	06/5	37385	210120	360	460000	0 0	0	2845000	0	5776375	2470000	3360	0	2160	2740000	1495000	3365000	4473	1650908	0	2200000	0	27160 0	1680	162593	0	2470000	220000	2470000	0	295544	1398	0	0	2407520	0 0		0	0	0
DAMCOST	922704	1156512	198221	922/04 1971318	948158	349382	244479	17026	932653 77660	581216	1112548	922704	5879892	392231	19531/	922704	16987	4778060	156037	5023882	922704	13005000	5308397	100026	989124 032061	750000	10907500	3756000	3777694	962367	190817 80067	3756000	4691994	11973	956192 060770	0	932835	10000	4691994	1000000	200000	91414 2007700	369/532	3440100	3442U02 150000	541140	15694	14700
EVENT1	30	30	8	5 6	8	30	30	8	20	90	30	30	30	8	5	6	30	30	30	30	30	30	8	30	00	00	30	30	30	30	00	5	9 8 9	30	30	30	30	30	90	30	8	30	000	0.05	6	88	30	30
L																																																

COUNT

HYSPL	0	0		0	בב בבטב ש
OLSPL	ო ო -	- m m - m r	- 0 - 0	10 FNN F 0F	0 TTCNT
FLSPL		a a a	، מימימ	· ບ ບ -	ഡ പ⊤ വവ
DAYREST 0 17	0000 ~ 0 % 0	00 <u>7</u> 00 <u>3</u> 00	* = = = 5, = = = = = = = = = = = = = = =	• • • • • • 8 2 • • • • • 2 • • •	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DAYLOST 0 17	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	4 º º º º º º 5 c	<u>%</u> %) 	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DAYHOSP 0 0 0 102	<u>,</u> 00800%0	N 0 4 0 0 0 N C		, ⁸	
PART DIS 0 0 0 0					000000000000000000000000000000000000000
TOT DIS 0 0 0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0) o o o o o o o o o o o o o o o o o o o	000000000000000000000000000000000000000
101FTL 7 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	- n o o o o n o c	000000000000000000000000000000000000000		м с о о о о о о о м <i>м</i> Ф
INJCOST 3405000 0 9265	247.352 13770000 13770000 2643380 840 0 15228 1370000	0 2200000 4384 0 0 2200000 8542	400 2870000 1225000 3628419 0 0 30074 1370000 22000000	480 0 1251010 1440 152000 113225 13225 0 460000 0 0 0	1495000 0 2200000 2200000 110000 2470000 5776375
DAMCOST 11500000 225014 850000 3756000	000000 190817 1021305 3756000 73056 922704 221402	10680000 3767666 3831672 91468 231746 10680000 922704 10680000	238224 238224 123800 4969211 200000 521025 521025 105306 10683045 295408 295408 358828	922704 922704 10680000 1932271 1853065 92290 165994 6600500 3577325 3357325 3357325 3357325 3357325 348976 8500000 25551	934704 990277 959227 153918 153918 15539000 87524 157773 21421 850000 95000000000000000000000000000000
EVENT1 30 30 30 30		8 8 8 8 8 8 8 8 8 8		3 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	30 30 30 30 30 30 30 30 30 30 30 30 30 3

COUNT

HYSPL 1	70 1.489361702	.	
0LSPL	89 1.893617021	-	
FLSPL 1	128 3.555555566		-
DAYREST 0	62G1 14		o w o o o o o o o o o o o o o o o o o o
DAYLOST	10		
DAYHOSP 0	69/		
PART DIS 0	4 0	• • • • • • • • • • • • • • • • • • •	
TOT DIS 0	4 0	••••••••••••••••••••••••••••••••••••••	
TOTFTL 0	108 0.990825688	• • • • • • • • • • • • • • • • • • • •	
INJCOST 0	/ 4288316 681544.1835	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	350 350 451800 0 0 451800 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DAMCOST 0	253884605 2329216.56	27833 27833 70350 59460 15686 15686 7993 33798 18122 33798 33798 33798 33798 33798 18122 23217 70600 11003 11252 91994 70000 71003 61198 15519 15519 11071 12277 51980 25160 517700 517700 51700 51700 51700 51700 51700 51700 51700 51700 51700 51700 51700 51700 51700 517000 5170000000000	149005 238244 43458 129126 83802 83802 83802 83802 1073030 15156 15156 3515 190817 75327 12599 30037 35461 13550000 1059955 10000 105955 10000 105055 20647
EVENT1 30 Min	30 I otal 30 Average 0.080383481	***************************************	***************************************
COUNT	109 30 Count		

НХЅРĹ	0 0-2		N N
OLSPL	0 0-0		r r
FLSPL	on orta	N 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	വ വ
DAYREST 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 - 0 0 0 0 0 2 0 2 7		o 7 o o o o o o 8 o 8
DAYLOST 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
AXHOSP 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			30%0000000%0%
			00000000000
101 D D D D D D D D D D D D D D D D D D			
101F1L 000000000000000000000000000000000			
NJCOST 0 19572	120 0 0 1100932 1592839	2222 2022 2000 2000 2000 2000 2000 200	0 19762 0 0 0 234780 234780 234780 234780
DAMCOST 30711 30711 37306 88812 24820 130499 14377 24851 10000 10000 10000 25064 777567 48288 22919 6026 6026	1437384 1437384 1437384 14078 10000 1350000 1350000 6026 31555395	444455.26/6 49053 49053 65970 55970 33768 107862 136222 28280 45156 168191 11495 67361 11495 67361	10247128 22902 98937 1500000 75240 64978 189315 100170 78057 10918530 127114 10918530
EVENT 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	32 32 32 32 32 32 32 32 32 Max 32 32 Max 32 7 Otal	32 Average 0.052359882 33 33 33 33 33 33 33 33 33 33 Max 33 33 Min 33 37 Otalal 33 Average	0.007374631 34 34 34 34 34 34 34 34 34 34 34 34 34
COUNT		32 Count	33 Count

нуярц	2	N	4		۴			4	÷ (5 22									ო											ę														
OLSPL	ო ო	n	ç		-			5	 1	6 6)								4											ю														
FLSPL	οιο	a	Ω		ы		5	5	ო (13 4 33333333									5											÷														
DAYREST	0 111	10	00	0 ¢	2 0	0	0	10	0	6 4		0	34	0	0	0 0	5 0	- c	. 0	0	0	0	0 0	0 0	0 0	00	0	0	0 1	o Q	0	0	0	0 0	5 6	2 c	00	0	0	0	0 0	5 0		5
DAYLOST	0 0	0	00	00	00	0	0	0	0	0 0	5	0	61	0	0	0 0	5 0		159	0	0	0	0 (0 0		. 0	0	0	0	0 26	0	0	0	0 (5 0	o c	00	0	0	0	0 0	50	- c	þ
DAYHOSP	0 67	Q	00	0 4	r O	0	0	4	0	4 -	-	0	14	0	0	0 0	5 0	- c	21	0	0	0	0 (00) C	00	0	0	0	0 0	0	0	0	0 (- c	• c	0	0	0	0	0 0	5 0	- c	þ
PART DIS	0 +	0	00	00	00	0	0	0	0	00	5	0	0	0	0	0 1	5 0		0 0	0	0	0	0 1	00	o c	00	0	0	0	0 0	0	0	0	0 (5 0		0 0	0	0	0	0 0	5 0	- c	þ
TOT DIS	0 0	0	00	00	00	0	0	0	0	00	5	0	0	0	0	0	0 0		0	0	0	0	0 1	0 0) C	0	0	0	0	0 0	0	0	0	0 0	5 0		0	0	0	0	0 (5 0		þ
TOTFTL	00	0	сı	00	00	0	0	5	0	5 0 711786711		0	0	0	0	o (5 0	- c	- -	0	0	0	0	0 0) C	0	0	0	0	0 0	0	0	0	0 1	5 0	o c	0	0	0	0	0 0	50		þ
INJCOST	0 254542	23140.18182	3010000 0	0 3064	0	0	0	3010000	0	3013064 430437 7143		0	18300	0	0	0 1	0 0		1177361	o	0	0	0	00	o c	00	0	0	0	0 32060	0	0	0	0 0	0	0.42		0	0	0	0 0	50		þ
DAMCOST	22902 23426381	2129671	10028403 1923133	59264 10102000	146869	167924	15539000	15539000	59264	38047793 6426300		36367	150000	166519	54901	27386	50526	37000	10112506	29978	41595	42376	13016	137697 33810	1000	62379	130037	11498	29644	64777 1055868	11493	151448	10966	51758	16851	95357	11825	14516	169494	18009	63559	13360	310/4	222
EVENT1	34 Min 34 Total	34 Average 0.008112094	35 35	35 25	35	35	35	35 Max	35 Min	35 Total	0.005162242	36	36	36	36	36	36	36 36	36	36	36	36	36	36 36	8 ¥	e Se Se Se Se Se Se Se Se Se Se Se Se Se	36	36	36	36 36	36	36	36	36	36	9 9 9	98	36	36	36	36	36	30 95	00
COUNT		11 34 Count								٢	35 Count																																	

Ъ	nn0n ←n44←00	2.96666666 0 1.1 2 2 2 5 5 66666 0 2.1 2 2 2 2 2 5 5 66666 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2
OLSPL	4 10 10 10 10 10 10 10 10 10 10 10 10 10	و. 2332 د میں میں میں 2332 د 2332 میں میں میں میں 2332 د 200 (200 میں 200 میں 200 میں 200 میں 200 میں 200 میں 200
FLSPL	ა⊷ია ≁იაიიაც	4 - τουοουτο 4 - τουουτο 4 - τουστο 4 - τουστ
D A A A R R S T S A A A B A B A B A B A A A A A B A	¥o&+ 600000605	۰۰۰ ۵۵%۵۵%۵۵%۵۵ N
DAYLOST 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	22 22 20 20 20 20 20 20 20 20 20 20 20 2	0 0 0 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
D Y HOX V O V O V O V O V O V O V O V O V O V O	20%- 000000000	0 00700707% NNN
PART DIS 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 00-00-0-0 000
F O Z S S		• • • • • • • • • • • • • • • • • • • •
101 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	20202000000000000000000000000000000000	1,833333 83333 0 0 5 6 5 7 8 0 0 5 5 8 0 0 5 5 5 5 5 5 5 5 5 5 5 5
NJ CO CO CO CO CO CO CO CO CO CO CO CO CO	1177361 1230121 1230121 20502.01667 3090 480000 395000 395000 395000 3010000 3888090	648015 0 27745000 277522 2020000 272522 200000 2740000 7232522 1446504.4 932 932 932 932
DAMCOST 22317 22317 19494 19494 14078 188739 48543 14148 45535 111255 45732 11255 45733 15733 45733 15726 95539 16226 95639 28349 28349 28349 28349 28349 28349 101652 281480 37024 101652 101652 281480 37024 101652 281480 37026 37056	10112506 101000 15989.438 266490.6333 266490.6333 266490.6333 140450 922704 11301843 15107563 15107563 140450 140450 28535714	4755952.333 104083 5911210 348844 3650593 17010160 17010160 17010160 17010160 5404900 5404980 5479892 5879892 5879892 5879892
E V E V B X X X X X X X X X X X X X X X X X X X	36 Max 36 Max 36 Average 36 Average 37 37 37 37 37 37 37 37 37 37 37 37 37	37 Average 0.004424779 38 38 38 38 Min 38 Min 38 Total 38 Average 0.005697316 39 Min 39 Min
COUNT	60 36 Count	6 37 Count 38 Count

•

0 HYSPL		
0 OLSPL		
D D	,	
DAYREST 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	000000 6 0700000087000r0088
DAYLOST 0 0		0 0 0 0 0 0 % % 0 0 0 0 0 0 % 0 % 7 0 0 0 % 0 0 0 ¢
DAYHOSP 2 2	0 0 0 0 0 0 0 0 0 0 7 0 ½ 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 ⁰ 0 ⁰ 0 0 0 0 0 0 0 0 0
PART DIS 0 0		° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °
TOT DIS 0 0		
TOTFTL 0 0	000000000000000000000000000000000000000	000-00000-000000000000000
INJCOST 932 932	0 125000 125000 125000 100000 1100000 1100000 110000 0 0 0 0 0 0 0 0 0 0 0 0	0 270000 270000 0 20892 13365 3918 3918 3918 0 12080 115000 115000 7203 0 115000 7203 0 0 10297 0 0 0 0 172156
DAMCOST 5879892 5879892	31523 21525 152855 152855 1504 17077 94955 25000 0 0 0 0 15354 173354 0 0 153354 173355 173354 173355 173354 173355 173356 173355 173356 173356 173355 173356 1735567 1735567 1735567 1735567 1735567 1735567 1735567 1735567 1735567 1735567 1735567 1735567	85000 86056 20000 0 7 47276 47276 47276 17134 17134 17138 1018 20232 1018 20232 1018 20232 26725 0 27239 67739 67739 0 358128 25459 0 0 358600 0 358600 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
EVENT1 39 Total 39 Average 0.000737463	6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6 6	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$
COUNT 1 39 Count		
AXMOSP DAYLOST DAYREST FLSPL OLSPL HYSPL 16 13 243 243 243 6 147 0 0 0 6 147 0 0 6 147 0 0 6 147 0 0 6 147 0 0 6 36 30 16 7 10 150 16 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 156 16 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
--	---	
OLSP OLSP	000 000 000	
FLSP	000 000 000	
DAYREST 243 243 243 243 243 240 240 240 240 240 240 240 240 240 240	0 8 8 8 8 7 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
Daytoo 24,00 25 26 26 26 26 26 26 26 26 26 26 26 26 26	04040 000000000 000000	
СА А С С С С С С С С С С С С С С С С С	060600000000000000000000000000000000000	
NA NA DI C C C C C C C C C C C C C C C C C C	0-0-0 00000000 000000	
0 T 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0-0-0 00000000 000000	
4 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
INJCOST 41491 41491 0 0 127330 127330 0 127330 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1100000 2762939 34973.91139 0 0 1680 1680 1680 1680 1680 1680 1680	
DAMCOST 0 240024 106791 106791 0 540024 1083337 4765 0 0 133737 4765 0 0 1236 68309 62576 619933 70000 19933 619933 70000 19933 619933 619933 70000 10000 10000 10000 0 0	10000 1973901 6914071 87519.88608 512578 542578 25145 14578 14578 145735 23790 11600000 11600000 11600000 11600000 11600000 11600000 11600000 11600000 11600000 11600000 1160000 11600000 1160000 11600000 1160000 1160000 11743 11743 12775 12755 127555 127555 127555 1275555 12755555 127555555 1275555555555	
м VЛ П 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	40 40 Max 40 To Min 40 To Min 40 Average 0.058259587 44 44 44 44 44 44 44 44 44 45 46 46 46 46 46 46 Min 46 Min 46 Min 46 Min 46 Average 0.005162242 0.005162242 0.005162242	
COUNT	79 40 Count 44 Count 45 Count	

•

HXSPL H	0000-	
OCSPL	1.333 4 - 1 0 0 0 0 1 - 1 0 0 0 0 0 1 - 1 0 0 0 0	
LLSPL FL	000	
A A R E S T S	000000 £0£0£20 <u>6</u> 57 000	000
	000000 0000000000000000000000000000000	
4 OH 2 O O O O O O O O O O O O O O O O O O O	000000 4000040000 000	000
с К С С С		000
S C F C C C C C C C C C C C C C C C C C		000
L L L L L L L L L L L L L L L L L L L		000
N N N N N N N N N N N N N N N N N N N	0 0 0 0 0 0 0 19188 19188 3837.6 3837.6 0 0 0 0 0	000
DAMCOST 18640 69042 70434 71459 71459 77455 16306 887365 887365 887365 887365 887365 887365 887365 887365 887365 887365 83365 83365 83365 83365 83365 83365 11214	15413 70302 568210 10000 2003730 60719.09091 922704 4651994 1146885 398053 398053 398053 398053 8083496 1616599.2 13592 13592	10499 25160 60000
Ш НА И Г Г Г Г Г Г Г Г Г Г Г Г Г Г Г Г Г Г	47 47 47 Min 47 Total 47 Total 47 Average 0.024356283 48 48 48 48 48 48 48 48 Min 48 Min 48 Total 48 Min 48 Total 49 Coordes 49 49	64 64 64 64 64 64 64 64 64 64 64 64 64 6
COUNT	33 47 Count 48 Count	

HYSPL	000		ς	-	- ത	- 0 -
JdSJO	000		m	0 0		0 N
FLSPL	000		Q	0	4	4 W
DAYREST 0 0					<u> 6</u>	4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DAYLOST 0 0		0200000			500000000000000000000000000000000000000	90000000000000000000000000000000000000
DAYHOSP 0 0					%	∞ o o ∧ o o o o o o o o o o o o o o o o
PART DIS 0 0		0-00000				
TOT DIS 0 0		0000000			• • • • • • • • • • • • • • •	
TOTFTL 0 0	00000	0 - 0 0 0 0 0				~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~
INJCOST 0 0		7 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9		24201 24201 0 0 0 0 0	75528 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	135844 210000 210000 0 0 0 0 0 0 0 0
DAMCOST 20000 10508	16399 60000 10508 181928 22741	30031 923874 43437 43437 43607 28576 10918 349309	23/09 1708759 231904 20550 28100 28100	2542.10 2542.10 103289 11276 59003 1550000 15510 23853 69759 69759 66149 66149	999275 49138 11086 32126 23641 34536 1000000 32211 71883 200000 220000 101462 253060	3756000 227400 28612 1500000 39988 35919 35919 266031 10546 16544 29396 29396 23470
EVENT1 49 49	49 49 Max 49 Min 49 Total 49 Average 0.005899705	ନ ନ ନ ନ ନ ନ ନ ନ ନ	ନ ର ର ର ନ ଜୁନ ନ ଜୁନ	ରି ଜି ଜି ଅ	ରି ଜି ଜି ଅ	ର ଜି କି କି କି କି କି କି କି କି କି କି କି କି କି କି କି କି କି କି କି
COUNT	8 49 Count					

HYSPL	-	-
OLSPL	~	۵
FLSPL		4
DAYREST 0 0 0 0 0 0 0 0 0 0 0 0		
DAYLOST 0 0 0 0 0 0 0 0 0 0 0 0		
DAYHOSP 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
PART PART 000000000000000000000000000000000000		
TOT DI 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
TOTFTL 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0)
	9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
DAMCOST 40148 31081 40036 40036 277348 127348 127348 127348 12250 53688 53688 53688 53688 53688	14259 14259 162344 162344 48659 12607 12607 12607 12607 13644 177525 97409 96409 96995 52495 520197 255295 520197 26974	44606 230864 32512 130265 169808 13292 26942 35104 13292 35104 14451 35104 14750 10080000 97614 17795 13200 1072000 1072000 1072000 1072005 135815 33965 256579 64258
even 88 89 89 89 89 89 89 89 89 89 89 89 89	2	ን ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ ଜ
F		

COUNT

OLSPL	
FLSPL	
D AYR 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
D VA 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
D VY HO N N N N N N N N N N N N N N N N N N N	
R F S S S S S S S S S S S S S S S S S S	
F O I I I I I I I I I I I I I I I I I I	
TOT 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
NUCOST 000000000000000000000000000000000000	
DAMCOST 32006 51422 51422 3487794 11277 98938 28139 28139 28139 58033 58037 17292 17220 17220 17220 17220 17858 118485 177858 118485 177858 118485 177858 1184855 177858 1184855 177858 1184855 177858 1184855 177858 118645 177850 119348 57140 57140 57140 57140 57212 55112 551140 119348 34714 117760 119348 34714 57212 551140 57212 55712 55714 55744 55744 55744 55744 55744 55744 55744 55744 55744 55744 55744 55771 55774 55774 55774 55777 55777 55777 55777 55777 55777 55777 55777 55777 55777 55777 55777 55777 55777 55778	430.08 49098 182993 1970824 113860 37010
M V B B F T	ର ର ର ର ର ନ ର

-

COUNT

HYSPL

ЧЗХН	- ∩	- ∞-0	1,428571425 1 2 2 2	- ~ ~	∞ - α
OLSPL	N M	- o-2		ე — თ	ო იი ათ
FLSPL	ம ம	ы м <mark>%</mark>	4. 7.75 7.75	0 1	م – م
DAY REST 0 0 0 0 0 0 0 0 0	<u> </u>	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	α ου ^χ οοοος	000003	000 <u>800800</u> 0
DAYLOST 0 0 0 0 0 0 0 0 0 0 0		0000000 600000 800000000000000000000000	n 00800000	- 4 o o o o o	00£00000
DAYHOSP 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			- 00200000	ο <u>τ</u> ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο ο	00200000
А К Т О О О О О О О О О О О О О О О О О О О		000000000000000000000000000000000000000	0 000000		
101 102 1000000000000000000000000000000			0 0000000		• • • • • • • • • •
TOTFT 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		• • • • • • • • • • • • • • • • • • •	0.065934086 0030 0030 0030 0030 0030 0030 0030 0	»←00000	0000000
NJCOST 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	4215000 0 0 0 0 0 0 0 0 0	0 0 0 0 4215000 4215000	39293.69231 0 762000 163855 2470000 2 0 0 0	2225000 0 0 0 0 0 0 0	2200000 0 1115749 0 3360 22200000 720 0
DAMCOST 28031 360643 40346 139725 1196722 1196722 119654 143457 15260 12561 17219	6498665 6874690 16422 160152 179000 10000 15539000 15539000	2924975 168544 10000 10000 10000 27000 93000 15539000 1192170708	655011.1429 124295 966744 141740 88390 984138 192260 37734	1937.78 4616589 430865 25878 43641 10004000 53764	10300000 19877 274755 92290 92290 92290 92290 948588 10035285 975173 975173
E VENT 50 50 50 50 50 50 50 50 50 50 50 50 50	ଜୁ କି କି କି କି କି କି କି କି କି	50 50 50 50 50 50 50 50 50 50 50 50 50 5	50 Average 0.134218289 51 51 51 51 51 51	<i>ភ ភ ភ ភ ភ ភ ភ</i> ភ	<u> </u>
COUNT			182 50 Count		

3 3 HYSPL	∞ ∞-00	
0LSPL 3	4 0 0 4-4	
FLSPL 5 4 5 5	റ്റ്റെ നമ്പ ന്നാം നമ്പം	
DAYREST 93 00 00 00 00 00 00 00 00 00 00 00 00 00	224 250 250 250 250 250 250 250 250 250 250	8 ~~~~ 00000000000000000000000000000000
DAYLOST 83 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		ດທານທີ່ ເວີດອອອອອອອອອອອອອອອອອອອອອອອອອອອອອອອອອອອອ
DAYHOSP 22 22 3 3 0 0 0 3 3 0 0 0 0 0 0 0 0 0 0		m 00000 000000000000000000000000000000
PART DIS 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 , 0 0 0 0 0 0	
TOT 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
101 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
INJCOST 52537 0 2200000 0 0 1112648 1112648 0 0 1518 0 0 0	2740000 2740000 1100000 0 211515 174984 0 2740000 2740000 19962861	415995 2965 2965 2965 2965 2965 2965 2965
DAMCOST 5751771 5751771 22675 159398 159398 29096 113438 113438 113438 133477 203293 102154 121432	1111665 10098317 10098317 137693 8619000 8619000 19633 2018973 2018973 2018973 380399 568094 568094 5680940 100000 1000000 82080535 82080535	1/10011.167 10683000 10683000 10683000 10683000 10683000 10683000 64530 64530 64533 67023 1100000 23044 14701 16818 14701 16818 25247 37453 37453 37453 37455 17512 367023 17512 367023 17512 367023 177512 374555 177512 374555 177512 377523 3775120
E V ም ማ ማ ማ ማ ማ ማ ማ ማ ማ ማ ማ T	51 전 월 51 전 월 51 Ma 51 Ma 21 Total 1 Total	51 Average 51 Average 52 Max 52 Max 52 Total 52 Average 54 54 54 54 54 54 54 54 54 54 54 54 54
COUNT		51 Count 52 Count 52 Count

J

R 1000 0	EVENT1 54	DAMCOST 17064	0 0	TOTFTL 0 2	TOT DIS 0	PART DIS 0	DAYHOSP 0	DAYLOST 0	DAYREST 0	FLSPL	OLSPL	HYSPL
Allen Titologo O </td <td>24</td> <td>16157 10000</td> <td>00</td> <td>00</td> <td>00</td> <td>0 0</td> <td>00</td> <td>00</td> <td>00</td> <td></td> <td></td> <td></td>	24	16157 10000	00	00	00	0 0	00	00	00			
Afrikin NEOCI 0 <th< td=""><td>54 Max</td><td>1100000</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0 (</td><td>0 0</td></th<>	54 Max	1100000	0	0	0	0	0	0	0	0	0 (0 0
G. A. Amana L. Martine S. Martin	54 Min 54 Total	10000 1675511	0 0	0 0	00	0 0	- -	o c	0 0	5 0	0 0	00
5 100/1 0 <td>54 Average 0.014749263</td> <td>83775.55</td> <td>0</td> <td>00</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	54 Average 0.014749263	83775.55	0	00	0	0	0	0	0			
6 4708 0	55	190817	0	0	0	0	o	o	0		-	
6 5115 0	55	407568	0	0	0	0	0	0	0		ę	-
5 5220 0	55	56115	0	0	0	0	0	0 1	0 1		ſ	
C FORM C <thc< th=""> <thc< th=""> C <thc< th=""></thc<></thc<></thc<>	22	92290	0 0	0 0	0 0	0 0	00	00	00		N	-
1 1	00 88	84192 150000	- c		5 6	5 c		. .	-	ĸ	V	÷
55 27200 0 <td>55</td> <td>1200000</td> <td>00</td> <td>00</td> <td>00</td> <td>00</td> <td>00</td> <td>00</td> <td>0</td> <td>)</td> <td>r</td> <td>•</td>	55	1200000	00	00	00	00	00	00	0)	r	•
65 22700 500 0 0 42 65 20700 500 0 0 0 0 42 65 20700 500 0 0 0 0 42 65 20700 500 0 0 0 0 42 65 20000 500 0 0 0 0 1 3 3 1 3 3 1 3 3 1 3	55	92290	0	0	0	0	0	0	0		-	-
5 2230 0 0 0 0 1 2 <th2< th=""> 2 <th2< th=""> <th2< th=""></th2<></th2<></th2<>	55	92790	5040	0	0	0	0	0	42		-	۰-
5 20735 00 0 0 1 5 1 3 54 20030 0 0 0 0 0 1 4 2 1 54 15000 500 0 0 0 0 1 4 1 4 2 1	55	92290	0	0	0	0	0	0	0			
5 23680 0 <td>55</td> <td>92735</td> <td>600</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>17</td> <td>S</td> <td></td> <td></td> <td></td>	55	92735	600	0	0	0	0	17	S			
SF 60/305 0 </td <td>55</td> <td>250820</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>~</td> <td>с</td> <td>. .</td>	55	250820	0	0	0	0	0	0	0	~	с	. .
Share Share <th< td=""><td>55</td><td>602309</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0 !</td><td>0 !</td><td>ı</td><td>N '</td><td>. .</td></th<>	55	602309	0	0	0	0	0	0 !	0 !	ı	N '	. .
SF Table SS Table ST Table Table <thtable< th=""> <thtable< th=""> Table<td>55 Max</td><td>1500000</td><td>5040</td><td>0</td><td>0</td><td>0</td><td>0</td><td>17</td><td>42</td><td>۰ n</td><td>4 ,</td><td></td></thtable<></thtable<>	55 Max	1500000	5040	0	0	0	0	17	42	۰ n	4 ,	
57 Total 56200 0 0 1 47 5 7 132656 433041539 0 0 1 47 5 7 132650 433041539 0 0 1 47 5 7 13265 433041539 0 0 0 1 47 5 7 13242 0 0 0 0 13 4 3 3 1 1 47 5 1 1 47 5 1 1 47 5 1 1 47 3 3 1 <td>55 Min</td> <td>56115</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>o !</td> <td>, - 1</td> <td>- !</td> <td>- 1</td>	55 Min	56115	0	0	0	0	0	0	o !	, - 1	- !	- 1
55 Arenage 56 Arenage 56 Arenage 56 Arenage 56 Arenage 57 2304 for 0 0 1 4 3 2125 7 157265 157305 0 0 0 0 0 0 0 0 0 3 2125 7 13742 0	55 Total	4759216	5640	0	0	0	0	17	47	9	11	` '
57 152268 430 0 13 57 23476 0 0 19 34 57 23056 0 0 0 0 34 57 23075 0 0 0 0 0 34 57 23076 0 0 0 0 0 0 0 7 1307 1307 0	55 Average 0.009587021	366093.5385	433.8461538	0	0	0	o	-	4	m	2.125	
	57	1532658	4930	0	0	0	0	19	34		-	~
	57	28476	0	0	0	0	o	0	0			
57 13742 0 <td>57</td> <td>22037</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	57	22037	0	0	0	0	0	0	0			
57 41327 0 <td>57</td> <td>13742</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	57	13742	0	0	0	0	0	0	0			
57 19077 0 <td>57</td> <td>41327</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	57	41327	0	0	0	0	0	0	0			
	57	19077	0	0	0	0	0	0	0			
	57 Max	1532658	4930	0	0	0	0	19	34	0	-	-
57 Total 1657317 4930 0 16 167317 4930 0 1 <th1< th=""> 1 1</th1<>	57 Min	13742	0	0	0	0	0	0	0	0	~	-
57 Average 276219.5 821.066667 0 0 3 6 1 58 28064 0 0 0 0 0 3 6 1 58 28064 0 0 0 0 0 0 3 6 1 1 58 53271 0 0 0 0 0 0 0 0 5<	57 Total	1657317	4930	0	0	0	0	19	34	0	•	-
58 28064 0 <td>57 Average 0.004424779</td> <td>276219.5</td> <td>821.6666667</td> <td>0</td> <td>0</td> <td>0</td> <td>o</td> <td>m</td> <td>Q</td> <td></td> <td>~</td> <td>-</td>	57 Average 0.004424779	276219.5	821.6666667	0	0	0	o	m	Q		~	-
58 42295 0 <td>58</td> <td>28064</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	58	28064	0	0	0	0	0	0	0			
58 91291 0 <td>58</td> <td>48295</td> <td>0</td> <td>D</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	58	48295	0	D	0	0	0	0	0			
58 53271 0 <td>58</td> <td>91291</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	58	91291	0	0	0	0	0	0	0			
58 20134 0 0 0 58 24670 0 0 0 0 58 25613 0 0 0 0 58 25613 0 0 0 0 58 27243 0 0 0 0 0 58 27243 0 0 0 0 0 58 57243 0 0 0 0 0 58 57243 0 0 0 0 0 58 51784 0 0 0 0 0 58 51733 0 0 0 0 0 58 51734 0 0 0 0 0 58 54250 0 0 0 0 0 58 54251 0 0 0 0 0 58 54177 0 0 0 0 0 58 541717 0 0 0 0 0 58 541717 0 0 0 0 0 59 603109 0 0 0 0 0	58	53271	0	0	0	0	0	0	0			
58 43670 0 0 0 0 0 58 236013 0 0 0 0 0 0 58 23958 0 0 0 0 0 0 0 58 23958 0 0 0 0 0 0 0 58 27243 0 0 0 0 0 0 0 58 27243 0 0 0 0 0 0 0 58 57733 0 0 0 0 0 0 0 58 57733 0 0 0 0 0 0 0 58 5303109 0 0 0 0 0 0 0 58 412117 0 0 0 0 0 0 0 58 41214 0 0 0 0 0 0 0 59 41214 0 0 0 0 0	58	20134	0	0	0	0	0	0	0			
58 25013 0 0 0 58 23958 0 0 0 0 58 23958 0 0 0 0 58 23543 0 0 0 0 58 27718 0 0 0 0 58 27243 0 0 0 0 58 27260 0 0 0 0 58 7783 0 0 0 0 58 7750 0 0 0 0 58 7750 0 0 0 0 58 7750 0 0 0 0 58 74217 0 0 0 0 58 44217 0 0 0 0	58	43670	0	0	0	0	0	0	o			
58 23958 0 0 0 0 58 97718 0 0 0 0 0 58 97718 0 0 0 0 0 0 58 7743 0 0 0 0 0 0 0 58 54260 0 0 0 0 0 0 0 58 17694 0 0 0 0 0 0 0 58 17694 0 0 0 0 0 0 0 58 17694 0 0 0 0 0 0 0 58 17694 0 0 0 0 0 0 0 58 14217 0 0 0 0 0 0 0	58	25013	0	0	0	0	0	0	0			
58 97718 0 0 0 58 53250 0 0 0 0 58 54250 0 0 0 0 0 58 7694 0 0 0 0 0 0 58 17694 0 0 0 0 0 0 0 58 603109 0 0 0 0 0 0 0 58 472177 0 0 0 0 0 0 0 58 472177 0 <td>58</td> <td>23958</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td></td> <td></td> <td></td>	58	23958	0	0	0	0	0	0	0			
58 2/243 0 0 0 58 7/250 0 0 0 0 58 17824 0 0 0 0 58 603109 0 0 0 0 58 42717 0 0 0 0 58 42717 0 0 0 0 58 42717 0 0 0 0	58	97718	0 (0 (0 0	0 0	0 0	0 0	0 0			
58 77634 0 0 0 0 58 603109 0 0 0 0 0 58 472177 0 0 0 0 0	202	21243 54350	5 0	5 0		, ,		- c				
	9.8	17604				o c			• c			
	80.3	1/094	5 0	5 0		5 0	5 0	, ,	5 0			
	on G	501000					o c	o c	o c			
	00 85	414115) c) C	, c	, 0) C	, o	0			

ЧХХН	000
OLSPL	000 -
FLSPL	000 -
DAYREST 000000000000000000000000000000000000	
AXH AXH C C C C C C C C C C C C C C C C C C C	
AA AA BA BA BA BA BA BA BA BA BA BA BA B	
FO FO SI SI SI SI SI SI SI SI SI SI SI SI SI	
TOTF 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
LS 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2855000 2855000 2855000 2855000 2855000 2855000 2855000 2855000
DAMCOST 121744 55392 49985 49985 49985 34160 10000 10000 85000 85334 45941 135417 53394 45541 135417 53320 55332 55332 135612 135612 135612 55320000000000	00000 603109 10000 3929042 89296.40909 52680 157072 673718 157072 167071 164490 1124782 1100256 1100556 10000000000
E V S S S S S S S S S S S S S S S S S S	55 54 58 Max 58 Min 58 Total 58 Average 60 60 60 60 60 60 60 60 60 60 60 60 60
COUNT	44 58 Count

HYSPL 1	0-		000	
1 OLSPL	~ ~ N ~	N NNNN	000	
FLSPL		പം നംനം നം പം	000	വ ന
DAYREST 70 00 00 00 00 00 00 00 00 00 00 00 00	348 0 10 10	ည္ ၀ ၀ ၀ ရွ ၀ ရွ <i>ဖ</i>		
DAYLOST 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 1 0 1 0 0 1 0	20002024		000000
AXHOSP 7 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	01010	20002024		
A A T T T T T T T T T T T T T T T T T T				0-00000
FOF 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		, 		0 ~ 0 0 0 0 0
	4 0 4 0.093023256			
INJCOST 11662 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	28555000 0 2879424 66963.34884	12270 0 0 12270 12270 3067.5		531120 0 0 0 0
DAMCOST 92290 92290 15597 158454 158454 38113 38113 38113 38113 38113 38113 38144 158726 18748 18724 18724 18724 187298 187298 187298 177081 17759 23384 17759 200000 200000	1000000 10000 15803817 367530.6279	998503 98718 468945 192735 98785 98718 1758901 439725.25	21000 263089 31409 263089 21000 315498 105166	72575 924766 67507 44518 18841 61536 61536 2857048
EVE NT1 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	60 Max 60 Min 60 Total 60 Average 0.031710914	61 61 61 61 61 81 Min 61 Total 61 Average 61 Average	63 63 63 63 63 Max 63 Min 63 Total 63 Average 63 Average	222222222
COUNT	43 60 Count	4 61 Count	3 63 Count	

í

۲Z	EVENT1	DAMCOST	INJCOST	TOTETL	TOT DIS	PART DIS	DAYHOSP	DAYLOST	DAYREST	FLSPL	OLSPL	HYSPL
:	5	32823	0	0	0	0	0	0	0			
	64	78858	0	0	0	0	0	0	0			
	64	98037	0	0	0	0	0	0	0			
	64	51124	0	0	0	0	0	0	0			
	64	15381	0	0	0	0	0 0	0 1	0 0			
	64	10000	0 0	5 0	5 0	ə c						
	40 Q	40110	- c			-			م د			
	5 4	4000B					- C	, c) c			
	5 73	52728	, o	00	00	0	. 0	0	0			
	64	67566	0	0	0	0	0	0	0			
	64	49034	0	0	0	0	0	0	0			
	64	27956	22440	0	0	0	0	0	187			
	64	136551	0	0	0	0	0	0	0			
	64	10680000	38190	0	0	0	15	48	06	2	2	
	64	38847	0	0	0	0	0	0	0			
	64	11631	0	0	0	0	0	0	0			
	64	54257	0	0	0	0	0	0	0			
	64	11126	0	0	0	0	0	0	0			
	64	13061	0	0	0	0	0	0	0			
	64	374451	17410	0	0	0	10	30	0			
	64	35233	0	0	0	0	0	0	0			
	64	1934172	420000	0	0	2	60	300	360	e		
	64	948158	120	0	0	0	0	0	-	5	2	-
	64	24830	0	0	0	0	0	0	0			
	64	17988	0	0	0	0	0	0	0			
	64	27431	0	0	0	0	0	0	0			
	64	43304	0	0	0	0	0	0	0			
	64	14044	0	0	0	0	0	0	0			
	64	100943	891	0	0	0	~		0	4		
	64	165067	0	,o	0	0	0	0	0			
	64	16919	0	0	0	0	0	0	0			
	64	10000	0	0	0	0	0	0	0			
	64	320611	3600	0	0	0	0	0	30	2		
	64	326426	0	0	0	0	0	0	0		-	
	64	454266	19341	0	0	0	26	68	0			
	64	53410	0	0	0	0	0 0	0 0	0 (
	64	64239		5 1		5 0	5 0		5 0			
	64	5/021	5 (5 0		5 0	5 0					
	64	14086	5 0			5 0			5 0			
	40 5	14000	5 0	5 0		5 0						
	49	14086	5 0	5 0	5 0	- c	- c				Ŧ	
	40 0	1101320	5 0						o c		-	
	40	610/4	5 0		5 0	-	- c					
	64	19564	5 0	5 0	5 0	. .		- c				
	64	14594	5	5	5 1		5 (5 0				
	64	14594	0 0	0 0	0 0	5 0			5 0			
	64	39746	0 0	0 0	0 0	5 0	5 0	5 0				
	64	26105	5 0	5 0		5 0	5 0	5 0				
	64	10000	0	5 1	5 1	. .	- ;	5 8	-			
	64	922704	28005	0	0	0	13	87	544			
	64	68357	0	0	0	0	0	0	0 0			
	64	72762	0	0	0 1	0 1	0 (0 (5 0			
	64	18086	0	0	0	0	0 (0 1	5 0			
	64	14500	0	0	0	0	0	D	o			

D-26

COUN

C EVENT1 DAMCOST INJCOST 64 1000000 20825 54 Max 10680000 531120 64 Max 10680000 531120 64 Max 100000 0 64 Max 64 Max 100000 0 0 0 64 Max 10000 0 0	64 Average 384584.7619 17491.14286 nt 0.046450177	66 13354 0 66 11424 0 66 222713 0 222713 0	66 11134 0 66 13093 0 66 Max 22713 0 66 Min 11154 0	65 Total 71738 0 66 Average 14347.6 0 nt 0.003687316	68 67171 0 68 20652 0 88 Max 67171 0 68 Min 206652 0 68 Total 87833 0 68 Average 43916.5 0 nt 0.001474926	69 73474 0 69 Max 73474 0 69 Min 73474 0 69 Total 73474 0 69 Total 73474 0 69 Average 73474 0 nt 0.000737463	70 5694953 2460040 70 169677 373000 70 169677 373000 70 370450 0 70 370450 0 70 370450 0 70 370450 0 70 370450 0 70 52519 14650 70 74351 0 70 81828 0 70 81828 0 70 81647 0 70 84160 0 70 74590 0 70 84564 0 70 84564 0 70 64561 0 70 64264 0 70 54284 0 70 708478 0 70 64264 0 70 708479 0 70 708479 0 70 708490 0 </th <th>70 76231 0</th>	70 76231 0
TOTFTL 2 0 2	0.031746032	0000	0000	00			2	0
TOT DIS 0 1 1	0	0000	0000	00			~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	0
PART DIS 0 3	o	0000		00			₩ ← □ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○ ○	0
DAYHOSP 0 60 131	0	0000	0000	00	00000	00000	50000%00000000000000	0
DAYLOST 51 300 598	თ	0000	0000	00		00000	0 0 0 0 0 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0
DAYREST 0 360 912	14	0000		0 0		0 0 0 0 0	ω ο ο ο ο Ο Ο Ο Ο Ο Ο Ο Ο Ο Ο Ο Ο Ο Ο Ο	0
FLSPL 5 27	3.857142857		00	0	000	000	ഗ	S
OLSPL 2 6	1.5		00	Ð	000	000	10 N N	
HYSP 1 1	*		000	5	000	000	4	

HYSPL	n	- N	o -
0LSPL 1	-	N	σ –
FLSPL		ن ب	ით ო
DAYREST 0 70 10 468 468 0 0 0 0	8000820200000	000012000072007 0000072007	
DAYLOST 0 5 141 141 0 0 0	, o o o + w o o o r & t o c		10000000000000000000000000000000000000
DAYHOSP 0 0 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3	<u>,</u> 		
PART DIS 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			
101 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			
TOTF1L 0 0 0 0 0 0 0 0 0			
INJCOST 0 239128 4723 151793 151793 0 0	22331 0 0 28255 3061 1680 1680 28255 7225 72255 72255 0 0	5380 5380 5380 5380 5380 5380 0 54774 0 6 6 1580	2200000 22200000 0 240 240 240 240 240
DAMCOST 237884 237884 64610 92290 92290 92290 12268 7216000 11216000 11236 85318 85318	2483 1115983 92290 59888 59888 100770 100770 1200000 12043 922704 922704 922704 922704 922704 922704	17,000 58084 58084 922704 10720425 46275 87225 16200 79049 79049 79049 79049 79049 79049 79049 712768 84756 84756 84756 84756	77097 77097 17097 147263 817056 81713739 817137375 817137375 817137375 817137375 817137375 8171375 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817137575 817157575 817157575 81715757575 817157575 817157575 817157575 817157575 8171575757575 81715757575 8171575757575757575757575757575757575757
EVENT1 70 70 70 70 70 70 70 70	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2		

COUNT

НУЗРL	-	1 ~~ ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	4 1 25 1.66666667
OLSPL	*	o ~ ~ ∩	5 1 1 1.88888889 1.88888889
FLSPL		סי טי אי אי	5 3 58 4.461538462
DAYREST 3 0 0 0 0			၁၀၀၀၀ဖို့ဝန္ကိလ ဝဝဝ
DAYLOST 30 0 0 0			20000 <u>5</u> 0880000
DAYHOSP 2 0 0 0 0 0			00000 <u>8</u> 084 000
PART DIS 0 0 0 0 0			000 01000000
TOT 0 0 0 0 0 0 0 0			000000000000000000000000000000000000000
TOTFTL 0 0 0 0 0 0 0		4 o o 0 o o o o o o o o o o o o o o o o	0 0 10 25 0,203252033 0,203252033
INJCOST 14042 0 0 0 0	2000 33340 33340 00000000000000000000000	2/4000 0 1370000 0 2530 0 2530 0 1213788 1213788 0 0 0	0 0 0 2740000 0 13585954 110454, 3106 0 0 0
DAMCOST 1525000 853588 72056 73675 119024	50085 50085 55662 773305 112791 123154 958561 303754 153154 95856 54220 110498 110498 110000 474277 90739 90739	12/25908 12/25908 858399 6755253 579129 100568 109568 109568 3016225 50522 50522 50522 50522 50522 50522 50522 952428 952428 952428 952428 952428 952428 952424 2009011	193000 66875 34448 75530 10000 10720425 3488 82542687 671078.7561 148520 207727 207727
EVENT1 70 70 70 70	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	70 70 70 70 70 70 70 70 70 70 70 71 71 71
COUNT			123 70 Count

HYSPL	000	000	Ν	N N N N	ω –		4 ~ ~ 4 ~
OLSPL	~ ~ ~ ~			000	ю	* **	4 4 -
FLSPL	000	000		000	۵	τ w	տտ տտ–
DAYREST			2000 0000 0000	8 519 0 65	770 90 0 0 0 0 23 0 0 0 0 0		28 0 0 0 0 0 28 0 28
DAYLOST			2000000	0 73 8 8	8 6 v c c c C c	0 0 0 0 0 0 0 0 0 0 <u>8</u>	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
DAYHOSP			8000-000	0 <u>6</u> 0 <u>6</u> 0	110 70 0 0 0 7 0	. 4	0000000000
PART DIS			0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0-00000		000070
TOT DIS					• • • • • • • • •		
TOTFTL							0 N O O O N O
INJCOST			48915 0 7666 0 0 0	960 48915 0 57541 6393.44444	94830 255620 4635 0 720 23341 0 0 0	0 0 0 44465 213078 213604 118604	5142 2200000 0 0 2200000 0 2200000
DAMCOST 58000	207727 58000 559860 139965	38106 22993 12865 38106 12865 73964 24654,66667 24654,66667	922704 68412 64245 689600 2131785 18411 56671 51410	125718 2131785 18411 4128956 458772.8889	4204192 190817 4601000 171863 251756 102242 187792 3756000	190817 29526 190817 190817 190817 145709 1306000 3771069 169174 190817	1220841 190817 6143967 169973 3769775 6143967 29526 29526
EVENT1	71 Max 71 Min 71 Total 71 Average 0.002949853	73 73 73 Max 73 Min 73 Min 73 Average 0.002212389	76 76 76 76 76 76 76	76 76 Max 76 Min 76 Total 76 Average 0.006637168	66666666	:	77 77 77 77 77 77 Max
COUNT	4 71 Count	3 73 Count		9 76 Count			

COUNT	EVENT1 77 Total	DAMCOST 31145781	INJCOST 2960435	TOTFTL 2	TOT DIS 0	PART DIS 2	DAYHOSP 207	DAYLOST 446	DAYREST 1275	FLSPL 26	01 0LSPL	HYSPL 13
22 77 Count	77 Average 0.016224189	1415717.318	134565.2273	0.090909091	0	0	თ	20	58	4.333333333	0	1.625
	78 78 78	41580 20104 81986	000	000	000	000	000	000	000			
	78 78 Max 78 Min 78 Total	163319 163319 20104 306989	0000	0000	0000	0000		0000	0000	000	000	000
4 78 Count	78 Average 0.002949853	76747.25	00	00	00	0	0	0	0			
	8888	43107 3442062 192316	0 1104130 0	0-0	000	000	000	-00	15 0	a	ი	←,
	8888	54249 969196 4600000 140686	0 89890 783358 0	00-0	0070	0000	0 23 53 0	0 140 17 0 12	0 29 5 0 0	υ	с , с,	2
	8 8 8 8 8	187616 1500000 34101 6298517 31615	0 1100000 0 1780315 0	0 ~ 0 % 0	000 - 0	00000	0 0 0 7 7 0 0 0		0 0 44 0 0	ىر س	2 +	+
	8 8 8 8 8	25209 981812 92290 948158 744544	0 2865000 480 0	040000	000000	000000			00400	، م	200	0 - 0 -
	8 8 8 8 8 8 8 8 8 8	10000000 967146 11212 25348 12552 12552 138471 23742 54067	8800 9800 9800 9900 9000 9000 9000 9000						2 r 8 a a a a a a a a	-	ო	- N
29 80 Count	80 80 80 Max 80 Max 80 Total 80 Average 80 Average	16632795 552538 552538 6604397 16632795 11212 11212 5555599 1915710.31	92886 1125 0 2865000 0 7826424 26976.6897	0 0 5 12 0.413793103	000-000	0000000	21 0 0 248 342 12 23	60 3 0 3 25 1 325	480 0 1273 44	5 5 1 4.33333333	5 5 7 3 5 5 5 4 7 3 5	1 1 7 7 7 7 12 1 7 7 7
81 Count	81 81 Max 81 Min 81 Total 81 Average 0.000737463	60000 60000 60000 60000			00000		00000	00000	00000	000	000	000

HYSPL		4			-	000 000
OLSPL		ო	ოო ო	000	, .	000
FLSPL		Q	ດດດວ	0 0 C	2	000 000
DAYREST	0000			337 0 0 0 0 0 0 3 37 3 37	38 000000000000000000000000000000000000	0000 0000000
DAYLOST	0000			800000808	20 000000000000000000000000000000000000	
DAYHOSP				800000808	20 000000000	
PART DIS	0000	0 0 0 0 0 C				
TOT DIS	0000	000000	00000			
TOTFTL	0000	0 1 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
INJCOST	0000	5 0 0 0 0 0 0 5 0 0 0 0 0 0	220000 220000 220000	80635 80635 80635 80635	13439.16667 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
DAMCOST	11410 16696 13534 20609	13807 968233 91868 64067 16314	968233 10000 1226538 122653.8	10202757 37500 128956 29817 106740 10000 10000	1752628 333 325600 325600 30522 21008 353670 69647 900752 10000 69900 65771	900752 10000 1920626 1920626 192062 6 192062 6 195000 99348 195000 99348 195000 99348 196000 94950 101155
EVENT1	82 82 82 82	8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	82 Max 82 Min 82 Min 82 Average 82 Average 0.007374631	84 84 84 84 84 84 84 Max	84 Avenual 0,004424779 85 85 85 85 85 85 85 85 85 85 85 85 85	85 Max 85 Min 85 Min 85 Total 85 Average 86 86 86 86 86 86 86 86 86 86 86 86 86
COUNT			10 82 Count		6 84 Count	10 85 Count 86 Count 86 Count

COUNT	EVENT1	DAMCOST	INJCOST	TOTFTL	TOT DIS	PART DIS	DAYHOSP	DAYLOST	DAYREST	FLSPL	OLSPL	HYSPL
	88	193731	0	0	0	0	0	0	0			
	88 Max	193731	0	0	0	0	0	0	0	0	0	0
	88 Min	193731	0	0	0	o	0	0	0	0	0	0
	88 Total	193731	0	0	0	0	0	0	0	0	0	0
-	88 Average	193731	0	0	0	0	0	0	0			
88 Count	0.000737463											
	68	594009	0	0	0	0	0	0	0			
	89 Max	594009	0	0	0	0	0	0	0	0	o	0
	89 Min	594009	0	0	0	0	0	0	0	0	0	0
	89 Total	594009	0	0	0	0	0	0	0	0	0	0
-	89 Average	594009	0	0	0	0	0	0	0			
89 Count	0.000737463											
	9	26478835	3840000	5	0	0	0	0	0	5	5	4
	91	10000	0	0	0	0	0	0	0			
	91 Max	26478835	3840000	5	0	0	0	0	0	5	5	4
	91 Min	10000	0	0	0	0	o	0	0	S	5	4
	91 Total	26488835	3840000	5	0	0	0	0	0	5	5	4
2 91 Count	91 Average 0.001474926	13244417.5	1920000	2.5	o	0	o	0	0	с,	с,	4
	8	16540	0	0	0	0	o	0	o			
	<u>8</u>	131025	0	0	0	0	0	0	0			
	93 Max	131025	0	0	0	0	0	0	0	0	0	0
	93 Min	16540	0	0	0	0	0	0	0	0	0	0
	93 Total	147565	0	0	0	0	0	0	0	0	0	0
2	93 Average	73782.5	0	0	0	0	0	0	0			
93 Count	0.001474926											
	94	52876	0	0	0	0	0	o	0			
	94	128686	0	0	0	0	0	0	0			
	94 Max	128686	0	0	0	0	0	0	0	0	0	0
	94 Min	52876	0	0	0	0	0	0	0	0	0	0
	94 Total	181562	0	0	0	0	0	0	0	0	0	0
2	94 Average	90781	0	0	0	0	0	0	0			
94 Count	0.001474926											
	Grand Average	810960.9425	128701.9336	0.204277286	0.008849558	0.022123894	2.277286136	4.544247788	9.718289086	3.693430657	2.098837209	1.616766467
	Grand Total	1099663038	174519822	277	12	30	3088	6162	13178	506	361	270
	Grand Min	0	0	0	o	0	0	0	0	←	-	-
	Grand Max	26478835	5776375	17	0	5	248	300	1411	5	S	5

Appendix E: Hazard Severity Calculations

General: This appendix displays the data used for the hazard severity calculations described in Chapter Three and Presented in Chapter Four. In order to establish the severity of a hazard_i, the severities of all the accidents that were a result of hazard_i were summed up in the following manner.

Severity of Hazard_i =
$$\sum_{j=1}^{94} P(\text{Accident}_j | \text{Hazard}_i) * \text{Severity} (\text{Accident}_j)$$

To determine the conditional probability that Accident_i occurred given that Hazard_j occurred, P(Accident_i | Hazard_j), the database had to be manipulated. A primary hazard was associated with each accident. Once the hazards and accidents were organized into a matrix format each conditional probability could be accessed. To verify that these probabilities were correct the law of total probability was. By viewing the sample space, S (Hazard_j), as a union of mutually exclusive subsets the law of total probability states: $S = B_1 \cup B_2 \cup ... \cup B_k$ where P(B_j) > 0, for i = 1,2,...,k and $B_i \cap B_j = \emptyset$ for i \neq j. Then for any event A

$$P(A) = \sum_{j=1}^{k} P(B_j)P(A|B_j).$$
 (Wackerly, 1995; p.61)
Where,
$$P(A|B_i) = P(Accident_i | Hazard_i)$$

 $P(Bi) = P(Hazard_i)$ (directly available from the database) $P(A) = P(Accident_i)$ (directly available from the database)

Once these probabilities were confirmed the severity of each hazard was computed as shown in this appendix.

The table below represents Praccident I occurred | hazard i was a contributing factor to an accident)≖ # times: hazard St contributed to avent xi # times hazard St contributed to any accident

•	

SI01 SI02 SI07 SI04 SI05 0 0.047619 0.023355 0 0	102 \$104 \$105 17619 0.023256 0 0	SIQ1 SIQ4 \$105 323256 0 0 0	5104 \$105 0 0	0 0 0	5106 2005	791 C 2013 1	1 0 027027	0 011834	•	0		13 515 0303	5115 d	•li* 0		0 007299	0 019608	•	0		0 006601	50016 0 035312
	0 0 0 0 0 0	0 0 0 0 0	0 0 0 0	0 0 0 0	0 0 0	0	°		. 0	0	0 0 0	5152	0	0	0	0	•	0	0	0	0	0 034894
						•••	00			00	0 0	0 0	• •		• •	0 0	0 0	600	5102 000434	9 C	0 0000	0 000359
				0 0 006319 0 0 005917	0 0 006319 0 0 0 006917	0 0 0 0 0	0 005917		• •		0		0 0	0 0 003597		0	. 0	0 00	5102 0 00434		0 006601	0 004891
					0 0 0 0 0 0 0	0 0 0 0			• •	0 007576	• •	0 0	0 0		••	0 0		0 0	0 0 00434	0 9	0 0033	9 2E-05
0 0 0 0 2 3 2 5 6 0 0 2 0 8 3 1 5 8 0 0 0 5 3 1 9 0 0 2 7 0 2 7 0 5 1 7 5 1	0 0 023256 0 020833 0 0 013158 0 005319 0 027027 0 017751	023256 0 020833 0 0 013158 0 005319 0 027027 0 017751	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 1 3 1 5 8 0 005 3 1 9 0 0 2 7 0 2 1 7 5 1	158 0 005319 0 027027 0 017751	9 0 027027 0 017751	0 017751	-	051282	022727	0	0	0	0 0 035971	0	0 029197	9096100	00 0	0408 0 00434	9	0 016502	0 157205
0 0 0 0 0 0006944 0 0 006579 0 005319 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 006944 0 0 006579 0 005319 0 0 7619 0 0 46512 0 0 0 13131 0 0 44515 0 0 13905 0 0 648511 0 0 54054 0 1 2435	0 0006944 0 0006579 0005319 0 0 146512 0.081333 0.064516 0.032805 0.058511 0.054054 0.12426	0.006944 0 0.000579 0.005319 0 0 0.081311 0.064516 0.012805 0.058511 0.054054 0.12425	0 0 006579 0 005319 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	579 0 005319 0 0 0 195 0 065511 0 054054 0 124255	9 0 0 0 0054054 012435	0 124/26		0	0 505500	0 000	0 5455 0116	0 413 0.02043	0 057554	• •	0 007299	0 058824 (00 0	5102 0 00434 0 0 09565	80	0 033003	0 016706
	0 0 0 0 0	0 0 0 0	0 0 0 0	0 0 0	0 0	0			۰	۰	0	0	0	0	0	•	Þ	0	0	0	0 0033	0 000242
			0 0 0 0010638 0 00118	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 1 18	8 0 00118	0 0118	3	<u> </u>	015152	0	0 0	0 0	0 00000	• •	•	0 019608	0 0	0 0 00434	90	0 0033	0.045518
u u u 0 9071429 0 0 046512 0 111111 0 1 29032 0 085526 0 1 2234 0 054054 0 0414	v v v 0 0046512 0111111 0129032 0085526 012234 0054054 00414	u u u uuuoosya u 000000 0 346512 0111111 0129032 0085526 012234 0054054 00414	0 0 0 0000079 0 0 000079 0 0054054 0 0414	9032 0.085526 0.12234 0.054054 0.0414	726 0 1 2 2 3 4 0 05 405 4 0 0 41 4	U U 054054 0 0414	0 0414	ں ی د	0 076923 (000000		0 0012	0 987 014583	1905300 E	0 181818	0 043796	0 058824	20	6122 0 05217	, o , e	0 049505	0.014764
0 0 0 0020833 0 0 006579 0 005319 0 0 01183	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 020833 0 0 006579 0 005319 0 0 01183	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 006579 0 005319 0 0 0 01183	61100 0 616300 678	0 0 01183	0 01 183	4	0	015152	000	0303 0 038	ī.	0 0014388	0	0	0 019608	0	0 0 00434	8	E07920 0	0 021328
0 0 0 0 0 0064516 0 0010638 0 001775 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 4 5 1 6 0 0 0 0 0 0 0 0 0 7 7 5 0 0 0 0 1 7 7 5	0 0 064516 0 0 010638 0 0 01775 0 0 0 0 0	4516 0 0010638 0 001775	0 0 0 0 0 0 0 0 0 0 0 7 7 5	B D D 01775	0 01775		- -	060606	6 0 0 0	5152 0051 0	eg c	0 0 000194	0.045455	0 0	0 C	000	0204 0 02173 0	6 0 6 0	0 006601	0 004565
									. 0		0 0 0 0 0 0	5152		. 0		0	• •	00	• •	, o	00	0.057159
0 0 0 0 4 7 6 1 9 0 0 2 3 2 5 6 0 0 2 7 7 7 8 0 0 0 2 6 3 1 6 0 0 6 3 1 9 0 0 5 4 0 5 4 0 0 1 8 3	(7619 0 023256 0 027778 0 0 0 0 2 6 3 1 6 0 0 0 5 4 0 5 5 4 0 5 5 4 0 5 4 0 5 5 4 0 5 4 0 5 5 4 0 5 5 5 4 0 5 5 5 5	023256 0 027778 0 0 0 026316 0 005319 0 054054 0 01183	0027778 0 0026316 0005319 0054054 001183	0 0 026316 0 005319 0 054054 0 01183	16 0 005319 0 054054 0 01183	9 0 054054 0 01183	0 01 183		0	015152	0	0	•	0 0017986	606060 0	0 007299	•	00 0	5306 0 00865	8	106600 0	0 043216
0071429 U 0033023 0059444 0032258 0059654 0079767 0081081 004733 3021429 0047619 0069767 011111 0129073 0059211 0074468 0054054 007100	0 0 093023 0 069444 0 032258 0 098664 0 079787 0 081081 0 0433 7619 0 069767 0 11111 1 1 129037 0 059211 0 0 14458 0 0554054 0 07100	093023 0.069444 0.032258 0.098684 0.079787 0.081081 0.04/33 269267 0.11111 0.120042 0.050931 0.074468 0.054054 0.07100	0.059444 0.032258 0.098584 0.079787 0.081081 0.04/33 0.111111 0.129037 0.059211 0.074458 0.054054 0.07100	2258 0098684 0079787 0081081 004/33 9032 00540511 0074468 0054054 007100	884 0079787 0081081 004/33 111 0074468 0054054 007100	7 0.061061 0.04/33 1 0.054054 0.07100	0.04/33		0.153846	0060606 01	22222	8000 ZGIG	92 0 0 2 9 2 9 2 9 2 9 2 9 2 9 2 9 2 9 2	5 D D68345	0.045455	2812/10	0 058039 0	10 1111110	2449 0 04347	5 0 5 0	0 072607	0 021594
			1013889 0032258 0032895 0021277 0 0 000591	2258 0032895 0021277 0 000591	162 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1 0 0 00291	0 00591		0	0	0	0 0 0 12	987 0.062	792E00 0 2	0.045455	0	0 019608	00 0	5102 0 00869	90 02	0 013201	0 000664
			0006944 0 0 0 0 0	•	0 0 0 0	•	00		00	007576	000	5455 0 025 5455 0 025	974	0 0 007194		0 007299	0 0	00	564100 0	=	0 006601	0.058256
00/1429 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0013889 0064516 0006579 0015957 0 0047337	013889 0064516 0006579 0015957 0 0047337	4516 0006579 0015957 0 0047337	79 0 015957 0 0 0047337	0 0 0 0 1337	0 047337		00	113636		5455 0 155	1	0 0 014388		0 007299	• •	0000	5102 0 05652	2 22	0 009901	0 017149
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 005917	0 0 0 0 0 002017	0 0 0 0 005917	0 0 002017	0 005917		0	972700 (0	0 0 012	987	0	•	•	0 019608	0	0 001304	13	0	0 314912
				2258 0 0 005319 0 0	0 0 0 0 0 0 0 0 0	00	00		• •			00	00	0 0010791	00	0 007299	0 019608	0 0	0 0 00865	~ ~	0 0033	0 383428
0 0 0093023 0027778 0064516 0096684 0047872 0135135 0106505	0 0 093023 0 027778 0 064516 0 098684 0 047872 0 135135 0 106505	293023 0027778 0064516 0098684 0047872 0135135 0106505	027778 0064516 0098684 0047872 0135135 0106505	4516 0 098684 0 047872 0 135135 0 106505	84 0 047872 0 135135 0 106505	0 135135 0 106505	0 106505		025641 0	0 128788	000 0	5758 0 051	348 0 04166	7 0 021583		0 043796	0 039216 (00 111111	0612 0 05217	, ,	0 039604	0 023543
0 0 0 0 0 2 3 2 5 0 0 0 6 9 4 0 0 3 2 2 5 8 0 0 0 1 0 6 3 8 0 0 2 7 0 2 7	0 0 023256 0 006944 0 032258 0 0 0 0 00 0638 0 027027	023256 0 006944 0 032258 0 0 0 0 0 0638 0 027027	3 006944 0 032258 0 0 0 10638 0 027027	2258 0 0 010638 0 027027	0 0 010638 0 027027	9 0 027027			•	•	000	5152	0	0 0010791	0	0	•	o	0	0	0 0033	0 02949
0 0 0 023256 0 0 0 0 0 0 0 0 0 0 01483 0 0.047610 0 0.070933 0 0.0769316 0.016647 0.077077 0.035503	0 0 023256 0 0 0 0 0 0 0 0 0 011834 7610 0 0 000833 0 0.055415 0.015057 0.035501	0232256 0 0 0 0 0 0 0 0 0 0 011634 A A A A A A A A A A A A A A A A A A A	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 11634	0 0 0 0 0 0 0 0 0 0 0 0 1834	0 0 0 0 0 0011834 145 0.015057 0.07277 0.015507	0 0011634	0.011634		00	0	• •	0	0 274 0.063	0 C	0.00000	0 791920 0	0 0		0204 0204 0.0391	٥ c	CODELU O	0 06029
0 0/1429 0 0 023256 0 020833 0 032258 0 006579 0 021277 0 0 0 0591	0 0 023256 0 020833 0 032258 0 006579 0 021277 0 0 0 00591	723256 0 020833 0 032258 0 006579 0 021277 0 0 0 00591	1020833 0.032258 0.006579 0.021277 0.000591	2258 0.006579 0.021277 0.00591	79 0.021277 0.0200591	0 0 00201	0 00591	2 ~		0		0	0	0 0 007194	0	0 007299	0 019608	200	0		0	0 1289
0 0 0 0 0 0 0	0 0 0 0 0 0	0 0 0 0 0	0 0 0 0	0 0 0	0 0 0	•		0	•	0	0	0	0	0	•	0	•	00 0	0612	0	0	0 00024
0285/14 0380952 0162791 0159722 0064516 0197368 0138298 0162162 009467 0 0143787 005877 0014722	09552 0162791 0159722 0064516 0197368 0138298 0162162 009467 2857 00562757 0014722 0064516 01072348 0017234	162791 0159722 0064516 0197368 0138298 0162162 009467 Ve0767 0014722 0064516 0197368 0138298 0162162 009467	0159722 0064516 0197368 0138298 0162162 009467 0014722 0064516 0197368 0138298 0162162 009467	4516 0197368 0138298 0162162 009467 0 0044053 0037234 0 0556	168 0138298 0162162 009467 63 0037234 0 052162 009467	3 0162162 009467	7.09467 1.03550 L		0 179487 (075758 03	03333	0 0012	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	5 0190640	0 181818	0 109489	0 213/220	0 0 0 0	12653 0 0608 0612 0 02606	20	0.039604	0 135081
				0 0006379 0005319 0	(79 0.005319 0	0		3 0	• •	• •				165E00 0 0		0	0	. 0	0		•	0 204771
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 1 3 8 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 2 7 0 0 2 9	0 0 013689 0 0 006579 0 010638 0 027027 0 029	0013689 0 0 006579 0 010636 0 027027 0 029	0 0 006579 0 010638 0 027027 0 029	179 0 010638 0 027027 0 029	\$ 0.027027 0.029	0 029	2985	051282 (0 9/5/00	11111 001	5152 0 051	345	0 0 003597	•	0 007299	0	00 0	5102 0 00865	9 9	0 019802	0.011873
0 071429 0 047619 0 0 013889 0 0 013158 0 026596 0 0 005 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2				0 0 0 0 1 3 1 5 8 0 0 2 6 5 9 6 0 0 0 5 7 7 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	58 0 026596 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0002	917	025641		0 0	0 0012	867	0 0 0 0 2 1 5 8 5	-	0 021898	0 0	8	5102 0 00434 0 0 0 00466	9 9	102510.0	0.068102
							100	634	> 0	007576	000	70		0 0 003597		0	00	0	2857	20	•	0.004018
0 0 0 0 0 46512 0 0 34722 0 0 32258 0 0 1 9737 0 0 58511 0 0 54054 0 0	0 0 0 0 46512 0 0 34722 0 0 32258 0 0 1 9737 0 0 58511 0 0 54054 0 0	046512 0 034722 0 032258 0 019737 0 058511 0 054054 0 0	0034722 0032258 0019737 0058511 0054054 00	2258 0.019/37 0.058511 0.054054 0.0-	37 0 058511 0 054054 0 0-	0 054054 0 0-	00	1142 (128205 0	030303	•	0 0 012	967 0.062	5 0.057554	•	0 021898	0 019608	00 0	0408 0 03043	5	0 046205	0 02135
0 0047619 0046512 0006944 0032258 0.006579 0015957 0	(7619 0 046512 0 006944 0 032258 0.006579 0 015957 0	046512 0 006944 0 032258 0.006579 0 015957 0	0 006944 0 032258 0.006579 0 015957 0	2258 0.006579 0.015957 0	79 0 015957 0	0		0	025641	0	0	0	0 0 0 0 2083	13 0.007194	-	0 (0 019608	00 111110	15102 0 00865	9 9 9 9	0.006601	0.05588
0 0 0 0 0 0 0.032258 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 032258 0 0 0 0 7619 0.069767 0.069444 0.037258 0.032105 0.079787 0.076	0 0 0 0032258 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 032258 0 0 0 0 0 0 0	2258 0 0 0 0 0 2254 0 000105 0 079787 0 077027 0 075	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0	960 0	- 5	0 CRC1201	0 (67,620)	- e	0 0012	U 247 0 02081	1965000 0	0 136364	0 10719	0 019608	000	U 5918 0.03043	5 0 5 0	0.059406	0.03014
		0 0 006944 0 0 0 0 0 0 0 0 0 27027	1006944 0 0 0 0 0 0 227027	0 0 0 0 0027027	0 0 0 0 027027	1 0 027027		0	•	0		0	0	16/0100 0	°	0 021898	0 019608	0000	5102 0 00434		106600 0	0 01016
0 0 0 0 0 0 0	0 0 0 0 0 0	0 0 0 0 0	0 0 0 0	0 0 0	0 0	•		0	0	007576	0	0	0	0	•	0 007299	•	00 0	5102	0	0 0033	5 38E-0
0 0 0 0 0000944 0 032258 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0006944 0.032258 0 0 0 0 0 0 00059	0 0 006944 0 037258 0 0 0 0 0 0 0059		2258 0 0 0 0 0 00059 A A A A A A A A A A A A A A A A A A A		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	65000	2 2	0	0	0 0	0 0142	U 157 0.04166	195500.0 0	0.045455		910000		5102 0 18606		0.023102	0.05836
						0 0 0 0118	81100	18	7871001	121210		0303	0	0 0 003597		0	0	。 。	0 0 00434	. 9	0	0 007064
0 0 0 0 0 0 3 8 8 0 0 0 0 0 0 0 2 7 0	0 0 0 0 0 3889 0 0 0 0 0 0 0 0 27027	0 0 013889 0 0 0 0 0010638 0 027027	1013889 0 0 0010638 0 027027	0 0 0010638 0 027027	0 0 0 0 0 0 0 0 0 0 0 2 7 0 2 7	3 0 027027		•	0	0	0	0	0	0 0 003597	•	o	0	D	0	•	0 0033	0 001872
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 013158 0 005319 0 027027 0 0059	0 0 013158 0 005319 0 027027 0 0059	58 0 005319 0 027027 0 0059	9 0 027027 0 0059	0 0059	2	0	•	0000	5152	0	0 0 010791	0	0 029197	0 019608	0	0 0 00434	9	0 0033	0 04891
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0046512 0013889 0064516 0013158 0021277 0 00414	046512 0 013889 0 064516 0 013158 0 021277 0 0 0 0414	0013889 0064516 0013158 0021277 0 00414	4516 0013158 0021277 0 00414	58 0 021277 0 0 0 414	7 0 0 0 4 1 4	0 0414	~ ~	025641	037879	000	0303 0 051	948 0.02083	19/010.0 51	0 0	0 014599		000	0408 0 03478	5	0 016502	0 116685
			0 0 0 0	0 0 0	0 0	0		_	025641	0	0	0	0			0 000000			0204	6 °	0,0000,0	97700.0
0.0/1429 0.095238 0 0.020833 0.032258 0.026316 0.031915 0 0.001//5	5238 0 D 020833 0 032258 U 026316 U 031915 0 0 001/75			2//10 D 2191915 0 001915	2//100 D 219100 911	S//100 D 0	\$//100			9/9/00	5	0606 0 036	-90700 (94	206/100 C	-		-		5/660 0 074/6	20		1100000
			0 0 000579 0 0	0 0006579 0 0							000	5152 0.012	26			eez700.0	• •		0 0 0 2506		0 0033	0.041793
			006944 0 0 0 0 0	0 0 0						015152	000	0303 0.012	967 0 02083	G 0 003597		0 014599		. 0	0	0	0	0 043294
				0 0 0	0 0	0000	ğ	5917		0	000	0303 0 025	374	0	0	D	0 019606	0	0 0 00965		0 0033	0 013255
0 0 0 0 0 0 0	0 0 0 0 0 0	0 0 0 0 0	0 0 0 0	0 0 0	0 0 0	0		0	0	0	0	0 0 0 025	374	0	•	0 007299	•	00 0	5102 0 00434	8	0	0 015566
				0 0 0	0			0	•	0	0			0 0 00	••	0	0 0	0 0	0	0	• •	0 011157
ם 1915-0000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0000000 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ם אובכשטעיט ט ט ט א-כרוכר אחתית ע ע ע א-כרוכר	0 6156000 0 0 0 0 700 0 0 0	0 8150000 0 0	0 8156000 0	200	8	5a17	- c	U 0.07576		2 0	5 c	1 0 003597	, 0	, o	» o	, 00 , 0	5102 V VV-V-	, o	» o	0 325594
							3	20	50	0	, 0	0 0012	и 187 0 02083		, o	, 0	, 0	;	0 0 000865	, o	0 0033	1 79E-05
			· · · · ·	· · · · ·	, o , o	, .		, 0	, 0	015157	, 0	, , ,	0	10	0	Ö	. 0	0	0	20	0	0 000436

This chart is P(accident I occurred I hazard) was a contributing factor to an accident)' Accident Severity Each column is added to give the total Hazard Severity in the bottom row.

2615	- 0	5903E-06	0	0	9836E-07	00683498	01597953	o	90626100	0	00//02/0	2372E-05	0	0	2675755	07842111	C/886500	01111110	002246484	000969304	00410755	03334156	•	00122833		37863E-05	•	0	0018/5466	0	000103241	000296097	00027077	0	000645562	0	000917422	41991E-05	0	0	07142F-05	0	000212687	004058606	•	004454916	0 00109024	Q	000115257	76766E-05	0	0	55693E-07	•	054876563
121	00	947E-06 1 5	0 2	7 CD-3/70	5C 0	3208257 00	0 0 0	0	00 0	0		80.7F-05	0	0	0661475 00	7607432 0.0	2644178 00			496E-05 0 (0	0	0929502	0720704 0	0	259E-05 5	0	101E-06	4087251 0		748E-05 0	0347461 0	•	0573965	0435834 0	0	11384157 0	3663E-05 4	1414E-10	3/32E-10		, 	0	02381325 0	14995-05	0 1973593 0		0	0	4164E-05 6	0 0	01661195	-	0	68752913 0
0	• •	0 1 829	2 0 0		•	0000	208 9 20	0	•	2		0 465		0	0000	445 0 02	347 0.00	8 2 2		0 87	0	0	000	000 698	0 0	9 -	0	0 7 50	019 000	3	0 6 05	000	0	0000	0.0 0.00	20	000	0 516	0 274		5		0	000	0 23	2682 000		0	0	6 7 9	0 0	000	0	•	9787 0.0
5120							0.001856									0 044535	0 002399							0 002615					0 006847						SUCANO D			_	_	_				_	_	00143				_	_			_	9 0 07878
6115	0 000692385	0	0 (0	0 003082443	0 000982699	•	0 000892516	0	0.000668451		00	Ō	0	0 019647991	0 002117071	CO-310ZOE 4		• •	0 006174748	0 007518196	•	0 000923255	00		0 002527972	0	0 00966638	0.002648648			c	•	0 000418743		0 000591056	0 000199329	0				0 000959175	0	0		5 0	, 0	0 000259892	J					0 06398649
1 115	0 000257749		0		0	0 004589915	0.000609704	0	•	0	0 000646584		00	0	0 000315448	0 035108234	0 000945728	0	977071-000.0	0 000125127	0	0 002798744	D	801601000	0 0	4 01326F-05	0.00094107	0	0 004048238	0 014769895	B 66618F-05	0 00149129	¢	0	0 000467647		0 003080394	0 000222609	3 92593E-10	0	ESOCCEZIO D		0 001428261	0 001703429	0	0 000941178		0 000632031	•	0 000113618	Ċ	00		0	0 079702329
5117	0 0		•	00	. 0	0		0	0	0	0 002684304	0		0	0 003928763	0 018219046	0 000981551	3 0183E-05		o c	• •	0	0	0	0 0	0 000124959	0	0	0 005602107				0	0			0 004110526	0	0	8 58598E-09	F1/759700.0		• •	0	•	0	00		0	c	o	0 0		D	0 038541656
Sile			0	1 75926E 05	• •	005654841	0 DOMORIARO	0	0	8 23497E-05	109962000	0 000306879	0-311507 0	• •	000777273	0 022347822	0 001475857	2 38656E-06	0.000419106	0 000246751	0	0 004137712	•	0 000508122	0 000318339	0 1 A6107E-05	0.000427529	•	0 005874152	0 00971806	4 27074E-06	0 001469833	•	1 44524E-05	0 001229115	570704000 0	0.002493916	0.000109703	•	6 79466E-10	0.000419854	20-3116-2 6 734055-06	0.000527891	0 00125919	0	0 00231909	0 0	0 000155736	0	0	0	0 000130467	0	0	0 067180833
\$115	0 0		•			0		0	0	•	002153035				0	0 050102376	0 001349633	4 15016E-05					•	0 000980958	•	0 B 500275-05			0 001925724	0 008442565		. 0	• •	0	0 001334743	0.001164196	0 000627997	0	0	•	0 002431654			0 002430936	•	0 002686279	0 0	1901060000	0	۰	0		3 73015E-07	0	0 01 7007882
\$114	0 0		0		0	0	0		0	0	9 000151736	0 000830965	971/52000.0	• -	0	0 007806163 (0 000280443 (8 62372E-06	0 001513142	60/062200 0	002012000000000000000000000000000000000	0	0	0 001223013	0	0		00	0 00040015	0	0	0 000684445	-	•	0 000277349		0 000391479	0	0	0	0 0008337099	5 0		0 006061554	0	0 00502369	0	0.000542761	0 000344273	0 000404302	0	00	U 2 32529E-07	0	0 046927051
tlis	0 00107005	0	0	0 0		c	0	0	0	Ð	0	0 000646306	6 91653E-05		0	0 003036508	0	0	0 002647999	0.00/828656	106//000.0		0	0 001783561	0 00044696	00			0	0 1	0.0000	0	0 000471796	•	0	0 0			0	0	0 021221708	0 000214069	0.000741181	0 003535907	•	0 007814629	0	27550000 0	0.000401652	•	0	0		. 0	0 056988794
\$112	00	0 0	0	• •		0	0	0	0	0	0	0 1			0	022267723	004798694	0					• •	0	D	0 0		0	0 010270529	•	0 001210100		. 0	0	0	• •		• •	ø	0	0			• •	0	0	0	,			0	•			0 042368548
5111	0	- o	0	0 0	0 5 96683E-07	003572831	0		0 000689672	•	0 000447384	0 000323153	0 000276661		000654794	0 012146031	0 000981551	•	0 000441333	0 002609552	C/ / 20/5100 0	DEBCOCZOD D	0	0 003032053	•	0	cu-ace21 (00	0 002334211	0	90,000,000		• •	3 04376E-05	0 000647148	0 0	U D D D D D D D D D D D D D D D D D D D	0	4 07464E-10	•	0 007073903	0 0001 60552		0.004419883	•	0 000976829	•	0	7/80000000		0	0	0 002466623	6 59943E-06	0 049974531
SI10	0 (• •	. 0			008061773 0	0	0 21/959000	00	0	001135667 (0	00		• •	015416116	003322173	0	0	0 0				000603667	0	0			005530285	0	0	001246212	0	0	EE273733	0 00143286	0	C C		0	002992805	•		1/10/100/10/1	81715E-05	D	•	0 0		0	0	0	0 0		049040987
5103	004179	0 0	. 0	394E 05		027906 0	0	0 65/0700	005387	0	006115 0	002524	1036-05		005114	0 9999600	0 555330	929E-06	0	0	2008118	0018034		0025075 0	o	961/000	1 88E-05	0	0 1716200	0047958	-		0005528	755E-05	0008846 0	•	0 0020300			118E-09	0024173 0	3 36E -05	0	0.4833000	0	0022889	o	0		0	0	•	0019266	0	0479799
80	09544 0(0 0	0	0	0 J	42488 0(•	0 0000	0	0	00296 00	õ	8 0		00336 01	62494 01	0 273110	0	0		00	5		0 21815	£7973	0	20-95 20-95	5	0 49965 0	ō	•	607500	08415 0	0	011544 0	0	0	0 141000	0	0	031546 0	•	06E-05	0 0	, , ,	0 0	•	•	•			•	•		436053 0
5	8783 0.00			E-05	0 0 747	3619 000	E-07	7.47 00	8424	0	0619 00	1345	۲. ۱		0 C 20	9012 001	0807 0.00	3E-05	0	•	17366	0 1905		2706 0.00	11382 0.00	•	1E-05 37	7154	20118 0.00	1967	18921	00 59210	1123 0.00	•	24955 0.00	39173	0			0	12417 00	•	BE-05 5	0.0 20020	0	11515	0					19292	0 0		41843 0.0
sk	231 0 0001	0 0	, ei	0 2 6015		345 0 0006	06 9 697	954 0 000	0 0 0 0	961	268 0 0016	1000 0 200	0 4 856		U 27 0 0000	725 00159	786 0 0016	05 14128	•	0	282 0 000	0		332 0 001	0 0 000	•	05 2 193	482 U 002	123 0.0043	084 0 0050	718 0 0010				215 0 001:	764 0 000				0	000 £/6	0	1991 0	000 000	0	319 0 004	0	495	. .			00000	00		238 0.05
SID6	0 00023		0 00015			0 00206	0 11994E	3 0 00054		0 00015	9 0 00126	0 00014		<u> </u>		779100 0	4 0.0012	5 2 1843E			4 0 00011		* -	9 0 00232	0		0 36172E	8000 2	4 0 00608	0 0 00622	0 0 00134	0 7811E	69000 D D		9 0 0004	3 0 00036	4	17000 8			0 0 00191	0	0			4 0 00339		0 0 00027				0			1 0 05605
SIOS	i			Ť				0 001077			0 001 9049	-	0 0002945			0.0064648	0 0027863	2 142E-0			0 001106		0006710.0	0 001518	0 0009515			0 0041589	0 0019878						0 000668	0 0018026	0 0002385	77/5000.0		6 0933E-0				780.3504.0	1070 100 0	0 004155									0 0500321
SIO4	°	• •	0 2E-04	•	•••	2 000	0 15-06	0.001			2 0.002	0 4E-04	•	• •			7 0 002	0 9E-06	0 4E-04	0	0 2E-04	0 100 0		6 7E-04	2 2E-04	6 0	0 3E-05	2 0003	4 0 005	5000 6	9	0 20 0		, o	17 TE-04	4 4E-04	0 000	20 C C C C C C C C C C C C C C C C C C C		0 1E-09	100 0 20	•	0 35 05	• • •		0 0 003	0	° .	0 35-04		。 。	0	80		32 0.051
5103	0 00082120			Ū		0 0036559;		0 00077701			0 00068668					50000000000000000000000000000000000000	0 00150656							0.00219004	0 00068603	0 00140221		0 00299829	0 0050158	0 00942425	0 00476211				0 00099329	0 0025991		0.0021030			0 00135720			0.005.477.0									0 0075719		0 0736258:
5102	001682	00		0			0	96/000 i	ə c		0	0	0	0	0		001028	•	0	0	0	¢ (,		. 0	0	6 55E-05	0 0	0011738	79297	0	0	0 003243		• •	0 002661	0	0 001 435	- c	0	0	0	0	0 0	5 0	0 01228	0	0	0 1			0	•	0 0	o 056283
5101		• •		0	6		•	0			550	•	۰	0	0 0	, se	1542 (0	•	0	001225	•			0	0	0	602600	008803	ø	o	•	004864	L 77700	• •	0	•	002153		, o	004169	0	•	•		0 00921	0	0	0	• •		•	0	• •	0 05877
											8					ŝ	5 8				÷							0					o e	>				•			•														-

Appendix F: Accident Severity Data

General: This data set represents the unweighted/weighted value scores and the unweighted/weighted utility scores for each measure. The raw scores were evaluated using the value and utility functions that were developed in Chapter Three and presented in Chapter Four.

100 million 100	Score	e	• •	• •	• •	0	•		,	•	0	• •	¢	• •		• •	0	• •	•	• •	• •	• •	Đ	• •	• •	0.0035	• •		• •	0.0035		• •	• •	• •	• •	0.0035	• •	• •	• •	• •		•		• •	•	• •	•	• •	• •	• •	• •	• =	
TAT DISABILITY	Score	0				0	e			0	Q I	• •	0	• •			0		•	• •		• •	0			0.25		• •		0.25		•		e e	• •	0.25		• •		• •		0			0	5 0	e (•		• •	,	-
LA TA	ISABILITY									_							_	_							_			_				_			_															_			_
	PARTIAL D	-				0											•		-																							-			-								
	Score	0				0	0	• •		•	¢,	0 0	•	• •			0	0 0	•	0 0			0	• •		• •		0 0		c (• •	•	• •			• •	• •	00	• •	0.0105		0			Ð		0 0		•		0 0		-
TOT IN DISTRIBUTION	Score	0			• •	0	o	00		0	0		0				0	• •	•	• •	0	0 0	¢	0 0	0	0 0		•	5 0			0	• •	0 0	• •		0 0	• •		0.75		0	0		0	0 0	0 0		50		0 0	2	c
	OTAL DISABILITY	0	0 0		0	0	0	0 0		0	0	• •	0	c c			9	0 0	5	• •			. 0			•		0 1		0	0 0	0	. 0	0 0	•		0 0	0 (0	• •	0	0	• •	0	0 0	0		0 0	00	0 0	>	c
	Score	0.176	~ ~		• •	•		• •		•	0	• •	•	• •			0	• •	•	0.352		• •	•	• •	0.176	¢				•	• •	0.187		• •		10		0	947-0	0.231		0	•		0	• •			• •			,	c
and the second	Score	0.5	• •			ø	ċ			•	0	• •	• •				0	• •	5		• •	• •	• •	• •	0.5	• •	• •	•	• •	Ð		0.53125			•	0		0	c70570	0.65625		0			0	• •	•	• •		• •	• •	•	4
	TOTFTL	-	• •	• •		0	c			c	6	e ¢	¢	0 C		• •	•	•	•	5		• •		• •		•		0 1	• •	•	• •	7			• •		• •		n 0	9 4		¢			0	• •	•	• •	• •		• •	,	e
	Score	6.039E-09	• •			0 0			ð	3.2813E-07 0	1.3523E-08	• •		0 ¢	• •	1.4707E-08 0	3.91E-07			0.02344028	0	3.7984E-08 5.0629E-08	0	•		0.02424273	0 3.0661E-08	0	0 4.3091E-08	0.00160867	• •	0.02776507			0	0.02941957	• •	0	0.02346933	0.02317387	0	• •				0 7 1688E-08	0		• •		0 0	`	¢
	Utility	1.2849E-07	• •	• •	• •	0 0	0 0		0	6.9814E-06 0	2.8772E-07	0 0		• •	• •	3.1291E-07 0	8.3192E-06		• •	0.4987294	0	8.0818E-07 1.07775-06	0	0 0		0.51580285	0 6.5236E-07	0	0 9.1684E-07	0.03422711	• •	0.5907462			0	0.62594828	, o o	0	0.49934744	0.49306107	0	• •		• •	• •	0 4 6145E-07	0		0 0				c
	Score	5.50566E-07	0 0			•	e			2.99137E-05	1.23287E-06	• •	. 0	0 0	,	1.34077E-06 0	3.56454E-05	•	•	0.749123721	0	3.46295E-06 4.61568E-06	0	0 0	• •	0.760740337	0 2.7953E-06	0	0 3.92854E-06	0.125387545	• •	0.808165972		• •	0	0.828681556	• •	0	0.749550019	0.745193148	0.1100001.0	0	•		0	0 1 97727F-06	0		0 0	• •	• •	~	c
	NJCOST	340	0 0			0	¢			13885	755					820 0	15844	00	ь	1092000	0	2065 2716	•	• •		1310000	0 1680	0 0	0 2330	210360	00	2200000		• •	0	2585000	• •	0	1100000	1018238	1/200/01	0	• •		¢	0	0	0 0	e (• •	• •	, ,	¢
	Score	-	3488E-10 4263E-00	4362E-U9 9318E-10	3625E-08	0 12125134	0 1169F_05	3958E-05 5461E-10	•	9638E-10 0		9853E-09 6752E-08	S625E-05	482E-10 1998E-08	0	0 5846E-10	0849822	319E-09	01-310	11133428	95842111	11124836	09544255	1356E-09	02764866	.05E-06	03450688 05286108	4065E-10	1392956	1392956	02472703 01917402	10697199	11801920	07610811 3324E-10	2936E-10	11801920	1354E-07 0734E-08	S616E-09	12022542	12102392	7/05-0071	02019437 0	4647E-10	1297E-09	0 .826E-08	7691E-10 0	219E-10	72856-09	1259E-08	0634E-05	00021837		UI assure
		•	272E-09 4.	601E-08 3.	225E-07 5.	0048053 0.0	0 00007 0.5	437E-05 1.	0	721E-09 3.	• •	11 00-318-00 11	206E-05 1	542E-09 3.		0 382E-09 7.1	ra 6016009:	638E-08 3	0 1201	1755404 0.	0 HSELGEL	1391354 0.	10 STST	204E-09 1.	1155327 0.	1 90-36681	16215598 0U	3 60-3160	091E-09 5. 0235598 0	0235598 0	0477557 0J	0 8211128	0 6/61677	0 616F-09 0	1932E-09 4	0 6/616123	7604E-06 5. 1816F-08 1:	106E-08 7.	19545728 U. 19429736 U.	12813219 0.	0 6/818171	15569363 0. 0	3926E-09 6.	2923E-09 1.	0 3161E-07 7	5825E-09 6. n	0637E-09 9	9852E-08 8. 3231E-08 9.	0791E-08 2	1598E-06 2. 00384043 9.	0 925299 0.		1 00 000
	1 1 1	0	(E-09 1.84	E-08 1.45	E-07 2.17	8255 0.09	20 0 3011	E-05 5.91		E-09 1.63	e	TE-08 7.95	1748 6.6	(F-09 1.47)	í.	0 E-09 3.21	H127 0.03	E-08 1.40	16-09 3.23	0107 0.47	14-10 2429 0.47	17.15 0.47	0110 6666	E-09 4.81	1110 0111 2069 0111	E-06 1.44	7811 0.14	NE-09 2.25	JE-09 2.25	12872 0.55	7726 0.08	16722 0.45	3435 0.32	2F.0 251-63 2F.0 251-63	E-09 1.81	19435 0.31 19435 0.31	SE-06 2.15 P-08 2.15	1E-08 3.20	1212 0.56	6628 0.51	0	0.125 0.06	TE-09 2.73	5E-08 2.30 1E-08 7.33	(E-07 3.5	(E-09 2.8)	ЭЕ-09 3.9(SE-08 3.65 TE-08 3.85	1E-07 9.06	5E-06 1.0. 90096 0.06	21277 0.00		
	Score		2.8320	-1162-2	3.49221	0.13126	1000.0	31680'6		2.51622		77722.1	0.00010	2.26758	1000017	1,93932	0.0541-	2.1614	4.9558	0.5710	0.5706	0.5706	0.50385	13956	0.16813	6.8376	0.2065	3.520	3.520	0.680	0.151-0.1189	0.5529	20211-10	0.4170	2.796	0/11/0	3,3443	1.9243	0.6071.	0.61020	0.6098		42099	3,5354	5.096	4.4082	6.0036	5.6842	1-3844	0.0005	3-100.0		
	DAMCOST	•	11523	30206	59688	248086	176766	11640		10697	0	24179 54563	121923	10000		0 15853	190817	29850	19961	1618000	0000091	000091	1198225	19385	146373	06226	576137 778527	13134	13134 10219706	10219706	370111	3697532	9191101 922704	922704	11432	10/226	84443	38474	6453598	6615873	0	200000	66311	23342	63822	14912	17526	40005	19605	/908/	150864	174	
L	EVENTI	E	5 1	8 8	5 5	10	e	888	l	2	S 0	20 X	3 8	50 X	3	88	8	6	01	8 8	8 8	89	88	8	5 8	80	\$ \$	8	8 8	8	88	80	8 8	80 80	80	8	80	80	8 8	80 93	80	80	= :	= =	11	22	12	2 2	121	2 2	2		1 :

Total	0.176	5.8E-05	9.7E-05	1000'0	0.0349	0	1.4E-05	0.00019	0.02368	0	0.00263	0'00116	0,00732	0,01681	0	0.00055	0.07889	0.00016	0	0.60727	0.18625	0.18619	0.10326	1.5E-05	0.20461	0.12431	0.15424	5.3E-05	0.23398	0.26627	0.01984	0.43899	0.15111	0.14136 1 9F-05	4.3E-10	0.43924 0.19053	1.5E-05	0,00031	0.40802	0.50238	0.28846	0.02019	0.00048	3.9E-05 2.4E-05	0.00016	0.00011	0.00279	0.00031	0.0039	0.00063	0.00046	4.8E-06	0.00029 1E-06
IRS Weighted lity Score	0	9375 5.8127E-05 0007 0.00021001	1001C0010 1002	6407 0.00010172	0 0 02126 0.01365318	0 0 6473 0.00084768	0 0	1251 0.00019376	3438 0.00014532	0	0 0 0	7506 0.00116254	8913 0.00730926	1561 0.01681168	•	1 0.00054736	• •	16563 0.00016469 13125 1.9376E-05	0 0	• •		0 0 19178 0.01289091	51243 0.00781971	12344 1.4532E-05	100096394	0 0	17877 0.06018884	18594 5.3283E.05	0 0		17035 0.00066362	0 0 secto numenato	0 0	B 0 B125 1.9376E-05	0 0	* *	02344 1.4532E-05	50783 0.00031485	• •			• •	78128 0.00048439	0625 3.8751E-05 13906 2.422E-05	0 0 25782 0.00015985	17969 0.00011141	49985 0.00278991	50783 0.00031485 62502 0.00038751	52502 0.00038751	40095 U.R0053767 86722 0.00053767	39064 0.0002422 67507 0.00038751	3E-05 4.8439E-06	46877 0.00029063 0 0
phed MI	0	09375 0.000	15625 0.001	640625 0.001	206733 0.220	100 32875	0	13125 0.00	24375 0.00	-	0 34375 0.02	1875 0.01	887785 0.11	149407 0.27		U 828125 0.001		265625 0.002 03125 0.000				0 912213 0.20	120581 0.12	234375 0.00	546875 0.01	0	1670 E0198U	MS9375 0.00	0	0	140625 0.010 703125 0.010	0	0	0 00125 0.00			1234375 0.00	6078125 0.00		0		0	018125 0.00	00625 0.00 1390625 0.00	578125 0.00	1796875 0.00	997062 0.04	5078125 0.00 10625 0.00	9625 0.00	0000 5/1875 0.00	390625 0.00 NK75 0.00	078125 7.81	146875 0.00 0
Unweig dHRS So	•	000 51 51	50 50 6'00	21 0.001	1649 0.220	175 0.013	0	-40 0.04	30 0.00	,	100	240 0.0	1150 0.117	1831 0.271		113 0.008	0	34 0.002	,			0 1600 0.207	1200 0.126	3 0.000	199 0.015	0 0.704	126/0 0.97t	11 0,000	-0 -0	e	740 0'01C	0	0	0 1			3 0.000	65 0.005				e	100 0.00	8 0.0 5 0.0HK	33 0.00	23 0.001	560 0.04	65 0,000 80 0,000	80 0.0	111 0.00	0070 05 05	1 0.000	09 0-0
Veighteit	0	• •			0	e	. 0	o	0		0 0		•	a o		, 0	0,062	0 0		0.062	0.062	0.062	0		, o	0.062		0 0	0.062	0.062	- 0	0.062	0.062	0.062	. 0	0.062	•	• •	0.062	0.062	790'n	0	0	0 0	0			0 0	0		0 0		• •
TOTLOSS V Score	0	• •			0	c		Ð	0		• •	00	0 0				-	0 0	2				•	0 0				0 0				- 9					-	• •			- 0	•	0	• •	0		• •	0 0	•				0 0
TOTLOSS	0	• •			0	e		0	0		• •	• •	0 0				1		•			- 0	0	• •	0		5 0	0 0				- 0	-	- 0			• •				- 0	Q	0	0 0	0			0 0	• • •		0 0		0 0
Weighted Score	0	• •		÷	0	-		¢	e		• •					• •	0	• •	5	0.00975	0.00325	5 0.00325 D			• •	0	2/50000 26100		0.04975	0.00975		7 0.0065 A	9 0.04325	3 0.04(325 0	. 6	3 0.00325 3 0.00325		• •	e c	•	• •	•	0		0	• • •		• •	• •		• •		
rw HYSPL SPL Score	0	• •		• •	0	-	• •	0					•	• •		• •	Đ	• •	•	5 625	0.08333	1 0.08333.			• •	- 2	67 0 67	-	3 0.25	3 0.25		2 0.16666	1 0.08333.	1 0.08333		1 0.08333	•	• •	• •			0	0		0	•	• •	• •	•	• •	• •	• •	
ighted Ra Score HY	0	0 0		. 0	0	-		0	0				•			:	0	• •	•	200975	0.00325	0.00325	. 0	•		0.00325	5610.0	0	00075	3.00975		200975	c/ 600.0	0 0		0.00975 0.00975	0	• •	• •			0	0		0			• •					0 0
OLSPL We Score	0				0	-		0	0		• •		0			• •	0	• •	5	0.25	0.083333	0.083333		• •		0.083333	0.2 0.5		0.25	0.25		0.25	3 23	00	• •	0.25	e (• •	0 0			o	0	00	Ċ			• •		• •	•		a o
a olstr																				6 7 6 7		25 1				- ,	n 4		75 3	75 3			n m			m m 10 10																	
.SPL Weighte core Scor	0	•	 		0 0			0 0	0 0				0 0				0 0	0	•	1 0.03	100.0	83333 0.003		•		e .		•	0.009	0.009	 	1 0.03		0 0		66667 0.000 66667 0.000	0 0		•••			0 0	0 0	8 8 8 8	6 0					••	0 0		
S 14S14 14																				w. w	2 0.1	1 0.0							ñ	m		¥0				2 0.1																	
Weighted Score	0	0 0			0	c	• •	ø	6.01009073		• •	• •	a (ə o		0.0021875	0	00	•	• •	• •	5 0.00015625 0		•		•	0.0021875	•	• •	7 0.00046875		0	• •	0 0		3 0.01769217 0	0	• •	• •	7 0.02194546	0.035	0	0		a					• •	0 0		
DAYREST	0				e	c		Ð	0.28830645		•		0			0.0625	G	0 0		• •	• •	0.00446428		0 0			0.0625			0.01339285	• •	0 0	• •	• •		0.50549048	•		• •	0.62701317	- 0	0	0	0 0	e					• •			
DAVREST	°	• •		• •	o	c	• •	o	30		0 0		0		:	10	e	0 0	2	0 0			, o	•	• •	•	0 1	0 0		9		0		• •	. 0	900 1000	0		0 0	1 392	5 I4II 0	0	0	• •	0		 -	• •			00		
T Weighted Score	05 3.913E-07	÷ ,		• •	•	4	• •	0	0.009		05 9.6031E-0	• •	•	• •		06 1.2481E-0 0	¢	•	•	• •	• •	46 2.1752E-0		0 0		•		•		36 0.0176683	• •	•	. •			21 0.0152644 0			* *	03 0.0174348	0 0	0	0		•		0.00010H			• •	• •	• •	• •
DAYLOS T Score	2.17391E-	e :			¢	c	• •	0	0.5		5.33504E	• •	0	• •	:	6.93396E	0	~ ~	•	• •	• •	0.0001205		¢ (• •	•		•		0.9815754		•				0.8480241	•			0.9686021	0.967908	6	0		•		0	• •		• •		• •	
DAYLO	1	•			•	a	• •	0	31 25		5 M		•			- •	03 0	• •	•			885 4		•		021 0		•	0 693	136 180			• •	• •		346 82 0	0		• •	128 156	128 155 0	0	0	• •	•	• • !	2 0	• •		• •	• •		• •
ISP Weighted Score	0	•		• •	0	¢		0	231 0.004442		5 0.0026		0				325 0.008391	• •	2	00		12000001	0	0		496 0.031320	0 0	0	0 769 0.003432	547 0.01247				• •	• •	1846 0.00403: 0	•		00	(239 0.01530)	2906 0.02755 0	0	0	0 0	c			• •			0		• •
DAVHO IP Score	•	0		• •	0			0	0.105769.		0.062:		0			0 0	0.199786	• •	•			0.072115	0	•		0.745726		0	0.081730	0.297008	0 0					0.096153	•		0 0	0.364316	0.655982	0	0	00	c			• •		0 0	0	• •	• •
DAYHOS	0	•	0 0		0	4	• •	Q	01				0	• •		• •	34	• •	•	• •		ю-	- 0	ð	• •	180	0 0	0	• •	. 8	• •		- o	• •		* C	0	• •	• •	28	156	0	Ð	00	-		• •	с с	• •	• •	0	• •	

.

Scare	• •	0		•	. =	• •	• •		•		0	-		• •		• •	. 0	• •		•					0	• •	¢	•				•		•	• •	•	• •	• •				• •	•	• •	0		•	• •		•	• •	• •		e
Score 0		•		• •	• •	• •	• •	• •	•				• •	• •		• •	•	• •	•	•		0			0	• •		• •	• •	0		0		0		•		• •		• •					0		• • •	• •	00	þ	• •	• •		0
PARTIAL DISABILITY 0	- 0	•					• •		e 1		0				••	• •				•		•	•		9	c c		-	• •	0		0	• •	•	• •	•	• •	0	. 0	0		0 0	. 0		0					•	0 0	• •	, e	0
Score I	8 0	ė	• •	0 0	. 0	0 0	0 0		e (c c	o	0 0		• •	• •	0 0	. 0	0 0		0		0	•		0	•	. 0	0 0		0	0 0	0	• •	0	• •			•	• •	0 0		• •	. 0	0 0	0	• •	. 0 .	• •		2	• •	•	• •	0
Score	0 0	0	5 0	00	e	0 0	• •	• •	0	0 0	. 0				• •					0			• •		0	•		•		0		0				0				00		• •	• •	• •	c	c 0	0	• •		•				0
FOTAL DISABILITY 0	0 0		o o	0 0	0	0 0	00	5 6	0	0 C	0	• •	0	• •		• •	• •	0	Þ	0		, Q	0 0		0	0 0		6	- 0	0		0	•	0	0 0	0		0 0	9 0	• •		0 0			0	0 0	. 6 .	• •		5	• •	0 0	0	0
Scare 1	• •			•			• •	• •	0	• •	•	• •		•	• •	•		•		0		•	0.176	• •	0	• •				0				•	• •	0				• •		• •	0.187		e			• •	•	•	* *	0 0		•
L Score	• •		• •	ф (• •	• •	• •	• •	•	• •		• •		•		• •	• •	•		•		•	50	• •	0	• •		•	- 0	0		•	• •	Ð	• •	•		¢	• •	• •	• •	• •	0.53125	• •	•			• •	-	•	• •	• •		•
TOTFT	00		• •	•	• •	• •	• •	• •	0	• •		• •	• •	•	• •	• •		• •	• •	0			33			• •		• •	• •	•		•	• •	. 0	• •	0	• •	¢ .	• •	• •	• •		61 2		•	8 8		• •		•		0 d 5		0
Score 0	0 0			• •		• •	с (0	• •	• •	• •		• •		• •		0 0		•		• •	0.023469		•	0	0	0	• •	0	0 0	0	0 0	0	0 0	0		• •		0 0		• •	3 0.024467	• •	0	7 2.1688E- 0	Ģ	0 C				5 7.018E-0		0
Utility				• •	• •	• •	• •		0	• •	•	• •	• •	0 0	• •	• •	• •	• •	• •	0			0.49934744		0	0	0	0 0	• •	•	c 0	0	• •	• •	a o	¢	• •	• •	- 0	• •		o c	0.5205874	• •	0	4.6145E-07	• •	• •		• •		1.4932E-0		0
Score 0	• •			0	• •	• •	0 0		0					0 0						0			0.749550019		0	0	0	•		0						0		• •	• •	•	• •	• •	0.763937571	• •	0	1.97727E-06 0	• •	• •		•	0 0	6.39764E-05		0
NJCOST 0	• •		• •				• •							• •		• •	• •	0	• •	0		•	1100000	• •	0	0	0	0		0						0		0 (•		• •	1370000	a c	¢	000	¢	• •		2	• •	23752		0
Score 8959E-06	1.74E-05 02101721	6636E-07	04526263	02490713	0682E-08	9349E-09 4429E-09	0137E-07	9975E-10	.482E-10	1843E-08 9349F-09	02251191	11930681	8605E-09	0208E-09	9660900'	3198E-07	1137E-09	73195-06	1482E-10	4707E-08	-1.526E-09	16975305	2302E-11	60-16276	482E-10	C307E-10	6836E-07	5.836E-07	.6821E-10 .1796E-09	7105E-09	.7853E-08 .1989E-07	S361E-08	C 115-09	6887E-08	.4564E-08 .7724E-09	.6212E-10	.7037E-08 .3857E-08	.4436E-08	.5875E-10	8788E-08	8087E-10	.4035E-08	12094572	.6609E-08	07380687	.10180004	02019437	5.482E-10	5436E-06	01-3784-0	.7027E-09 5.482E-10	5.676E-08	3672E-08	02227854
Utility 1.2271E-05 2	7.37276E-05	2.82357E-06 6	4.95312E-08 1 0.191790805 0	0.1055387 0	4.5262E-09	8.19889E-09 1 1.45885E-08 3	2.54817E-06 6	1.86884E-07 4 1.69387E-09 3	1.47542E-09	1.34928E-07 3	0 1916803600	0.505537348 0	2.05953E-08 4	4.32545E-09 1	0.025845776 (5.59248E-07	8.95617E-09 2	3.27623E-05	1.47542E-09	1.04689E-07 2	9.03625E-09 2	0.719292578 0	S21266E-11 1	8.16042E-06 1 1.11614E-06 2	1.47542E-09	2.24871E-09	2.47289E-06	2.47289E-06	3.67885E-09 8 9.23447F-09 2	1.99597E-08	1.18021E-07 2 9 31746F-07 2	3.19327E-07	1.13179E-08	1.56302E-07 3	6.17138E-08 1 7 \$101E-09 1	1.95814E-09 4	7.2192E-08 1 5.87143E-08 1	6.11678E-08 1	0.0112/905 0.0 3	1.64355E-07 3	1.61384E-09 3	5.94697E-08 1	0.512481879 0	1.12752E-07 2	0.312740975 0	0.431356088 0	0.085569363 0	1.47542E-09	1.07778E-05 2	0 0	7.21479E-09 1	2.4051E-07	0.1100305E-07	0.094400601
Score 1.88591E-05	0.000113307	4.33953E-06	7.61245E-08 0.264471912	0.152444684	2.20/282-09 6.95631E-08	1.26009E-08	3.91627E-06	2.6033E-09	2.26758E-09	2.07371E-07	0.138572022	0.603348505	3.16528E-08	6.64779E-09	3.21831-08	8.59508E-07	1.37647E-08	5,03513E-05	2.26758E-09 2.26758E-09	1.60896E-07	1.38878E-08	0.790535458	8.01132E-11	20-3711-221	2.26758E-09	3.45604E-09	3.80057E-06	3.80057E-06	5.65402E-09	3.0676E-08	1.81386E-07	4.90773E-07	1.73945E-08	2.4022E-07	9.48479E-08	3.00946E-09	1.10952E-07 9.02379E-08	9.40087E-08	0.01/214819 2.33626E-09	2.52596E-07	2.4803E-09	9.1399E-08	0.609895135	1.73288E-07	0.40626746	0.531174404	0.125	2.26758E-09	1.65642E-05	2,267586-09	1.10884E-08 2.26758E-09	3.69639E-07	1.54159E-07	0.137212045
DAMCOST 103424	123104	87301	43138 710137	375360	42169	24433 30226	86175	10931	1000	24000	297659	6266399	33805	18427	187244	51569	25302	11-12-02	10000	51238	25390	15805010	562	77120	10000	12992	85846	85846	17004	33477	52542	601-69	27640	55603	15509	19611	112US	15413	10197	56151	10599	15109	660000	52045	102116	2588575	200000	10000	102000	TOO T	23190	60309	50773	289019
EVENTI L	11	1 2	11 11	1	3 12	1 1	1 :	2 12	11	22	1 2	: :	3 3	1 1	3 3	1	1 11	1	2 12	1	2 2	12	2	2 1	12		1 1	11	2 2	12	1 1	12	3 5	12	2 2	12	8 8	# (3 13	¤ :	3 12	5	12	8	1 2	2 2	: 9	22	121	71	15 15	23	8 8	n

Score	0.00043 0.00114	0.02186	0.0002	0.11701	0.02544	0.00047	1.9E-09	16-05	4.8E-05	2.4E-05	3.2E-10 0.00035	1.9E-09	0.02715	0.00027	0.00024	6.8E-05 4 9E 06	0.002	0.00085	0.00012	0.00012	3.5E-10	4.8E-05	0.00024	0.23175	8.9E-05	9.7E-05	5.8E-05	0.00365	0.00015	4.8E-06	0.00021	0.00024	0.00019	61000.0	0.00029	0.00015	0.0002	0.00066	0.00121	0.00267	0.00034 4.8E-05	0.00012	0.00015	0.02175	9.7E-05	0.00059 0.07539	0.10915	0.02019	3.5E-10 3.5E-10	2.5E-06 7 sc-10	0	0.00048 3.5E-10	0.02466	0.11182 7.8E-05	0.02289
Utility Score	0.0180475 0.00111894	0.0136723 0.00084768	0.0078128 0.00048439	0.9999196 0.06199501	0.0086722 0.00053767	0.0075003 0.00046502	0 0	0.0001563 9.6878E-06	0.0007813 4.8439E-05	0.0003906 2.422E-05	0.0056252 0.00034876	0 0	0.0748628 0.00464149	0.0043751 0.00027126	0.0039064 0.0002422	0.0010938 6.78155.05	0.0500261 0.00310162	0.0136723 0.00084768	0.0019532 0.0001211	0.0017969 0.00011141	. e	0.0007813 4.8439E-05	0.0039064 0.0002422	0 0001175 1 9376F.05	0.0014063 8.719E-05	0.0015626 9.6878E-05	0.0009375 5.8127E-05	0	0.0023438 0.00014532 0.0023438 0.00014532	7.813E-05 4.8439E-06	0.0034376 0.00021313	0.0039064 0.0002422	0.0031251 0.00019376	200110000 00000000000000000000000000000	0.0046877 0.00029063	0.0023438 0.00014532	0.0032814 0.00020344	0.0106254 0.00065877	0.0195319 0.00121098	0.0001563 9.6878E-06	0.0054689 0.00033907 0.0007813 4.8439E-05	0.0019532 0.0001211	0.0023438 0.00014532	0.3508139 0.02175046	0.0015626 9.6878E-05	0.0254696 0.00157911	0.0933038 0.00578484		• •	• •		0.0078128 0.00048439 0 0 2	0.1178913 0.00730926	0 0 0.00125 7.7503E-05	0 0.009222 0.00061518
Score	0.018046875	0.013671875	0.0078125	0.999919572	0.008671875	0.0075	0	0.00015625	0.00078125	0.000390625	0.005625	0	0.07486044	0.004375	0.00390625	0.00109375	0.05002451	0.013671875	0.001953125	0.001796875		0.00078125	0.00390625	0 0003125	0.00140625	0.0015625	0.0009375	8	0.00234375	0.000078125	0.0034375	57906500'0	0.003125	0.003125	0,0046875	0.00234375	0.00328125	0.010625	0.01953125	0.00015625	0.000546875	0.001953125	0,00234375	0.350806161	0.0015625	0.005234375 0.02546875	0.093300913	0		• •	2	0.0078125 0	0.117887785	0.00125	0.009921875
MHRS	8 F	175	77 100	9006	Ξ.	× %	• ;	9 19	10	vn e	• ¢	0	55	* *	÷;	2 9	613	175	2	ล	• •	9 9	8	• •	8	8	2	•	R R	-	75	8 57	9 8	3 4	88	8 8	4	136	250	1	<u>ج</u> ع	នេះ	88	2065	2	326	982 •	• • •		• •	•	90 9	1150	0 %	127
Score	• •	0		o	•	• •	• •	• •	0	• •	• •		0.062	0	•	• •	, e	• •		00	00	00	• •	0.062	• •	• •		0	• •	0	0 0		• •	5 0	• •	• •	• •		0 0		• •	0 0	• •	0 0.062	0	• •	• •	• • •	0 0	00	•	0 0	0	0 0	0
Score	• •	0		•		• •	• •		0	•		0	0 -	• •	•		• •	0 0	• •	• •		0			. 6	0 0		0	• •	0	• •		•	• •	• •	• •	•	• •	• •			0 0		o -			• •		50	00	•		0	00	o
TOTLOSS		0		0			• •		•	• •		0	0 -	• •	• •			0 0		00		00				•		0	• •	0	00		•		• •	• •	00		• •	• •		0 0		o -	• • •	0 0	0 0		50	0 0	2		ø	• •	0
Score		0		0	•		•	• •	0	• •		0	• •	• •	• •	• •	• •	•	• •	• •	• •	• •		• •				¢		0	e 6	• •	•		• •		• •		• •		• •	•		• •		• •	• •	• • •		0 0	2	¢ 0	0	• •	0
sPL Score		0		0	-				8	e 0	-	æ (• •	• •	•	• •	• •	•	• •	• •	• •	•	• •	• •	•	ф (• •	0	• •	•	• •	• •	• •		• •		•		• =			•		• •		• •	• •			• •	•		o		Ð
Score HV		0		.00325	• •		0 0		0			0	• •		0			0		0 0					. 0	0 0		0	00	0	0 0		¢ 0		0 0					• •	0 0	0 0		0 10195	0		e 0			0 0	5		0	00	o
Score	• •	0	• •	0 22233 0	• •		•		0	• •		0 0	0 0	. 0				• •		• •		0 0		• •		0 0		0	0 0	0	0 0		0 0		0 0		0 0		• •		0 0	0 0		- č] = 1							0 0	D		0
1dS10				1 0																																								4											
Score		0		0.0065	•				•	0		Ð	0 0		•	• •	• •			•		•		• •		0		0	• •	• •	• •	• •			• •	• •	•	• •	• •		• •	•					• •	• •	• •	•	2	• •	Ð	0 0	¢
PL Score	• •	0		0.166667	•		•	• •	•	•	• •	•	• •	• •	•	•	• •	•		•	• •			• •		•	• •	÷	• •	•	e :	• •	• •			• •	• •	• •	• •		• •	•	• •	• •			• •	• •		• •	•	• •	0		•
m FLS				"	_										_													9625																			5625				_				_
Š		U				, ,	0.			ψ.		0.																143 0.00																			100.0				-		Ũ		U
Score	0 0	0		. 0	0 0				•	• •		0	• •	, o	0	0 0		0		• •	. 0	•		00	e e	• •		0.017857	0 0	• •	0 0	• •	0 0	• •	0 0	0 0	0 0	• •	00		• •	• •		0 0			0.044642			00	•		0	00	0
DAYREST		0		0	0 0	• •			0	0 0		0	• •	0	•	• •		0		0 0		•	00	• •	• •	• •	00	4	0 0	• •	• •	• •	• •	• •			0 0		0 0		• •	0 0				• •	9		• •	• •	•	00	ø	• •	o
Score		c	• •	ė	¢ :	• •	•	• •	0	0	• •	0	• •	• •	•	• •	• •	0		-	• •	• •		• •	• •	• •		0	• •		• •		• •		• •		• •	• •	• •	• •	• •	•	• •	0 00071161	0	• •	-			• •	•	• •	0.01084358	• •	٠
Score	• •	0			-		•	• •	. 0	•		•	• •	• •	•	• •	• •	•	• •	• •	• •	• •		• •	• •	•	• •	0	• •		• •		•		•		6 6		•	-	• •	0 0	• •	0 371700550	0	• •	•			• •	•		0.60242088	• •	•
DAVLOST		e	• •	•	-	• •	•			•	• •	•	• •		•	• •		•	• •	-		e :		• •	• •	•	• •	0	• •		•			• •	• •		•	• •	• •		• •	•		0 1		8 8	÷.		e e	0	5	0 0	36	• •	0
Score		0	00	. 0	•		•		• •	0		0	• •		0	• •		•	s 0	• •		•				0		00302885	a o	o o	0		•		0 0		0 0		• •		• •			0 0		0 0	0 0		• •	•	>	00	.00650641	0 0	•
Score	• •	0	0 0	• •	•	• •	•			•		0	• •	00	•	• •		0		9 9		0 0		0 0		0,		072115385 0	• •		0 0		•		0 0		0 0		0 0	. •	0 0	5		• •		• •	0	• •	• •	• •	2	00	.15491453 0		0
DAYHOSP		0			•		0		0	•		0	• •		0	0 0		0		0		0		0 0	• •	•		3	• •		0 0		0 0		0 0		0		0 0		0 0	0	0 0	0 0		• •	•		• •	• •	•	0 0	22 0	• •	0

_																																																										
core	0	ф	¢	• •	• •									• •				• •		• •		•		•	• •	• •	0			•	• •		•		0		÷	• •	•	• •	• •	• •	• •					•	• •		• •		e e	-		• •	• •	• •
80																																																										
Score	¢	¢	•	•	• •	• •	0	• •	¢	•			•	• •		• •		• •	• •	• •	• •	•		•	•		•		• •	•	• •	• •	•	• •	•				•	• •	• •		• •	•	• •	Đ,	• •	¢,	• •		• •	•		-		• •	• •	• •
MLITY																																																										
DISAE			¢	• •			0	• •	• •				•		• •	• •		• •				•		•	• •		•	• •	• •	•		• •	•		•	• •	•		•	• •	• •	• •	• •	• •	• •	•		ę,	• •		• •	• •		-		• •		
RTIAL																																																										
Z.																																																										
Score	0	0	0	• •	• •		0	• •	• •	• •	• •	, e	0 0	• •						•		•	• •	°	0 0		0	• •	• •	•	• •				•		•	• •	• •	• •	• •	, o	• •	•	• •	0	• •	•	• •	, .				e		0 0	0	00
Score	¢	¢	۰	• •	• •		•	• •		•	• •	• •	•		00	0 0	• •	• •	• •	e (• •	۰		•		• •	÷	• •	• •	o	• •		0	• •	•	• •	0		•		0 0	• •	• •	• •	• •	0	• •	0	• •	• •	• •	• •	• •	-	• •	• •		0 0
λĽ																																																										
SABIL	_	_	_	~ .				~ ~		~ .								~ ~																												0			• •									
IAL DI	Ĩ	Ĩ	-										-							-		-		-			-			-			-		-		-		-																			
5																																																										
core			•	•	• •		•	• •			• •		•			•				•		0	• •	0	• •		0	• •		•			•	• •	¢	• •	•	• •	•		• •	• •	• •		• •	•							•	4		• •	• •	
Ĩ																																																										
Score	•	¢	•	•	•		¢			•	• •	• •	•			• •		• •	• •	•		ø		Đ	• •	• •	•	-	• •	•	• •		•	• •	•		•	• •	•	• •	•	• •	• •		• •	•		•	• •		• •	•	• •	•	• •	• •		0 C
Ę																																																										
TOTH	•	0	Ð	•	•		÷	• •		•		••	•	• •	• •	•		• •				Ð	• •	¢	•		÷	••		•	•		•	••	0		-	• •	•				• •		• •	•	• •	•	00	• •	• •		••	-	• •	• •		• •
ž			0								• •						0 9E-08					•	• •							8E-09									0				•	, o								, .						00
×																	8.20													8.549																												
fellity		0	0		•				, o	0			0	• •		¢ i	0 159E-06	•	• •			0								191E-07			•	• •	o			a a	0	• •			• •		0 0	. 0		0				, o				• •	• •	• •
a																	1.74													18.1																												
SOTE		0		•	• •					0						0	0 12E-06			0		0	• •	•	ф с			• •		68E-07			0						0				• •		• •	• •												
υč.																	7.48]													7.79																												
JOST			0							0			0			0	164			•		0								. 2			•						•	• •			•															00
FNI [_									-						_	4	_		_		_								ч																-												
core	u 04E-09	025757	41E-07	223513	93E-09	20-30C	121202	71E-07	60-3E9	63E-10	24E-09	137829	78E-05	82E-10	96E-08	102564	149E-09	19E-08	1/8CU	11E-10	94E-00	HIE-09	145809	81E-05	38E-09	CC / CO 1	995E-06	812E-06	82E-09	417052	129E-06	12/15-US	93E-08	198283	87E-08	501E-09	564379	69E-09	163E-10	182556 146E-07	218642	576E-07	13E-06	194E-07	1064147 249E-07	722E-10	569E-07	507E-06	021E-05	185E-10	82E-10	82E-10	255E-08	0	10-37C	598E-07	141E-05	020183
s	0 8 4.72	0.00	6 4.09	3 0.02	81.1	5 F F 6	6 0.02	11		61.7 6	8 627	8 8 8 8	5 1.03	9 3 4 9 5 9 5	101 8	4 0.00	8 970 8 970	7 2.62	187 9	6.61	91 - 16 91 - 16 91 - 16	9 1.10	1 0.02 2.02	67	8 8 5	8 12 27 29 9	5 2.8	5 3.98	2010 S	12	8 1.63	5 5 1 8 1 8 1 8	8 1.45	0.02	177	8 3.2	8 0.02	91 1 92 12 1 92	6 151	1 0.02	16 0.02	9 7 7 7 9 7	8 2.03	181	19 0.00 M 5.82	6.9	8 2.4	19	8 1.6	4 <u>5</u> 8 8	8 9 7 7	17	1 3.8		8 F.	112	15	16 0.00 35 0.00
Alla	0-37.10	109140	348E-0	421667	0-3550-0	1-38189	241886	924E-0	318E-0	1928E-0	5782E-0	11-611094	973E-0	7542E-0	022E-0	1434591	0-34118 1800E-0	1351E-0	1248784	1131E-6	1096E-0	823E-0	1363595 1745E-0	69810	LISIE-C	9516120	1165E-0	8695E-4	N83E-0	CISH00	5766E-L	7022E-4	5278E-0	1199660	1302E-0	7717E.4	1866012	6667E-1 1353E-0	3912E-6	092481) 628E-0	9528144	7652E-0	5071E-4	5166E-0	121180	5431E-0	4531E-(7065E-0	8858E-(7053E-C	7542E-(7542E-C	2095E-1	-	1221E-0	5352E-1	50Z1E-U	008552) 9031310
ā	2.00	80.0	1.7	0.09	8	1	0.0	61.6	133	3.04	2.65	010	\$	5	15	0.00	11	Ξ	2.06	2.86	57	1.67	0.16	00'0	3.62		17	1.64		0970	4	325	5	970 S	81		.	1 5 1	1.93	7.81	0.05	1.29	38.	111	10.01	15	101	61	6.71		2:	15	1.6		4 86 8 4 1 86 8 4 1 86 8	52	2 S C C	5 0.04
Ę	80~290	676243	21E-06	9268356	27E-09	112E-05	958776	83E-06	97E-06	44E-09	179E-08	952625	03E-05	158E-05	14E-08	661324	67E-06	35E-07	57E-06	32E-09	S26E-06 87E-08	96E-09	8555718 1935-07	286382	51E-08	207420	18E-05	64E-05	1690148 26E-09	651966	538E-06	90-3198 79E-09	158E-08	511157	-13E-07	557E-08	680272	745E-05 21E-05	23E-05	(56801) 28E-06	123533	73E-06	182E-07	135E-06	1170413	148E-09	90-3655	366E-05	104325	19201	758E-05	758E-05	124E-07 0.125		729E-UK	413E-07	012E-08	1313684
Sco	770 E	0.001	2.666	0.136	7.745	6.514	0.130	EIFT	2.048	4.686	1.084	0.145	6.756	126	6639	0.006	4.320	171	3.173	1,305	1.404	7.185	0.145	0.000	5.570	191	1.862	1.592	0.13.	0.68	6.726	5009	9.763	0.14	2.786	2.116	0.156	6.55	2.98(0.13-	0.135	1.997	1.32	161.1	0.00.	157	1.60	3.02	0.001	5.026	2.26	2.26	2.49]	1	1.380	8.07.	1893	0.00)
ST	~	N.	~	5			, w				. ·		5	.	• •	6	~ ~	~ '	£ •		~ ~	~	z ~	Ē	• •	2 -	20	51	= m	ŝ		~ ~	14	92 .¥	5	wa A	. 2	v o va	¢	c -	36			0 m	£ *		- 2	12	8 ¥	2 o			e 2		, 2		9.0	8 5
AMCC	33495	15267	8195	28738	1981	1225	24641	7500	2930	1541	3649	34114	11743	1000	11665	16781	3708.	5190	83861	1472(2550-	1913.	36211	13326	3978	7667	10325	10691	1975	106800	9211	1597	1582	33260	5722	2963	3960	1830	1189	7321.	29671	7879	1161	7312	1626	1515	7640	10862	1221	1600	0001	001	5600		10001	6886	0001	1500
1																																																										
EVEN	52	ន	12	1	2	3 5	ង	ង ដ	ន	53	ន រ	រ ព	2	ព ព	า ส	ដ	ន ន	នេះ	ន ន	ដ	ន ន	2	71 F	ន	ສເ	ន ជ	ព	នា	ន ជ	ង	នេះ	ន ន	ន	នន	ព	នះ	ា ព	ង ង	52	ន ន	នេះ	ន ន	នេះ	ន	ສະ	រង	ង ង	ោ	ព រ	រ ព	នេះ	ន	ន ន	3	4 7	51	3 3	24

Score	0.0056	0.00541	0.01222	0.02289	0.00019	0.00839	0,09621	2.2E-07	0.00452	0.00246	0.0002	0.0705	0.02379	3.5E-10	2.96-05	0.00128	0.0052	0.07038	0.00239	0.00156 9.7F.05	0.00025	0.00392	0.00048	0.02464	0.00139	0.00155	0.00994	0.05178	0.00037	0.02957	0.00039	0.00155	0.01321	0.0012	0.08216	0.02509	0.00015	0.00035	0.02564	0.00031	0.00052	0.02858 0.00204	0.03978	0.00592	0.00219	4.8E-05 0.00139	0.0031	0.00266	0.00591	0.00124	0.00837	7.7E-10	9.7E-06 9.7E-06	3.5E-10	3.8E-08 0.02019	0	4.6E-07 2.1E-06	0.04776 0.00116	8.7E-09	0.02219
Utility Score	0.0378672 0.00234777	0.0830676 0.00515019	0.1971134 0.01222103	0.0106254 0.00065877	9/06100070 102100010	0.1352829 0.00838754	0	0	0.0728304 0.00451548	0.0396003 0.00245522	0.0148442 0.00092034	0 8	0.0001563 9.6878E-06	0 0 0 0	0.0004688 2.9963E-05	0.0206257 0.00127879	0.0672634 0.00417033	0 0	0.0385566 0.00239051	0.0156255 0.000196878 0.0015626 9.68788-05	0.0040626 0.00025188	0.0631399 0.00391468	0.0078128 0.00048439	0.0028907 0.00017922	0.0224226 0.0013902	0.0250008 0.00155005	0.1329942 0.00824564	0.8352127 0.05178319	0.0059377 0.00036814	0.0196882 0.00122067	0.0062502 0.00038751	0.0250008 0.00155005	0.0558219 0.00346096	0.0193756 0.00120129	0.9458367 0.05864187	0.0501234 0.00310765	0.002422 0.00015016 0.0093753 0.00058127	0.0019532 0.0001211	0 0	0.184512 0.01143974	0.0084378 0.00052314	0.1089644 0.00675579 0.0328589 0.00203725	0.1216306 0.0075411	0.0953853 0.00591389	0.0353279 0.00219033	0.0007813 4.8439E-05	0,0396003 0.00245522	0.0428932 0.00265938	0.0953853 0.00591389	0.0200007 0.00124004	0.1347523 0.00835464	0 0	0.0001563 9.6878E-06 0.0001563 9.6878E-06	0 0	0 0 0 0	•	••	0.0187506 0.00116254	•	0.014063 0.0008719
Score	0.037865999	0.083065001	0.197107985	0.910625	CZ160011 0	FF68225210	0	•	0.072828056	0,039598963	0.01484375	Ð	0.00015625	0	0.00046875	0.020625	0.067261288	0	0.03855529	0.0015625	0.0040625	0.063137921	0.0078125	0.002890625	0.022421875	0.025	0.132990275	0.835208055	0.0059375	0.0196875	0.00625	0.025	0.05582008	0.019375	6261-2821670	0.050121788	0.002421875 0.009375	0.001953125	•	0.184506824	0.0084375	0.108961095 0.032857795	0.121626992	0.095382324	0.0353267	0.020859375	0.039598963	0.04289184	0.095382324	0.02	0.134748312	0	0.00015625 0.00015625	0				0.770332752 0.01875	• •	0.0140625
MHRS	180	908	1555	136	₹ ;	1253	•	•	823	9	P 61	0	2 2	0		764	EL.	nc -	887	90 S	6	1504	001	37	787	320	1240	3256	2	252	8.	320	129	248	111	F19	31 120	22	• !	2 00 S	108	1093	1173	1000	957	10	8	537	1000	1 22	051 251	-	11 11	• •	• •			3000 240	-	180
Score	0	0	0	•	•	• •	0.062	0	•		•	0.062	0 0		. 0	0	• •	0.062	0		. 0	0 0	• •	0	• •	• •	0	• •		e	0 200	7007-0	0	• •		0	• •	0	•	• •	0	• •	Ģ	• •	• •	• •			• •	• •	• •	• •	• •		• •			• •	00	
Score	0	o	a	•			-	o	•	- 0	0	-	0 0		. 0	0	0 0		• •	o é	• •	• •		0			o	•		0	ο.	- 0	0	•		0		. 0	÷,		0	• •	0	0 0	• •				0 0		0 0	. 0	• •		0 0			0 0	0 0	
TOTLOSS	0	0	0	•				0	• •		÷	-	• •		• •	0	• •		• •			• •		0			0	• •		0			0	• •	. .	0	0 0	•	0	• •	0	0 0	0	0 0	. 0	• •			• •		0 0				0 0			• •	0	
Score	0	0	0	•	• •	• =	0.00325	•	•			9			• •	0	• •		. 0	• •		• •		•			0	0	0	Ð	0	0	0.00325	• •		0		. 0	0		0	• •	0.00975	• •		• •			• •	• •	• •		• •							
Score	0	0	•	•	-		0.083333	0	•		Ð	0	•			•			•			• •		•			0	-	9	0	0	0	0.083333	• •		•	• •	•	0				0.25	•		• •			•	• •	• •			• •	• •			• •	e a	
HYSFL							-																						•		•	-	-										9																	
Score	0	¢	0	0	•		0.00975	0	0		• •	0	0	0 0	• •	0	0	ə c	• •	• •	• •			Ð	• •		0	• •		0.00325	0	0 0	0.00325	0	.	0	0 0	• •	0	0.0065	0	• •	• •	• •		• •		• •	0	0	• •	• •	• •	• •	0 0		• •		•	
Score	0	•	0	0	• •		0.25	0	• •		• •	•	0	0.0833335	0 0	0	0 0	• •	• •	• •	• •	0 0		0	• •	• •	0	• •		0.083333	0	0.052530,0	0.083333	• •		0	• •		0	0.166667	Ð	• •	• •	• •		• •		• •	0	0	• •	• •			• •		• •	• •	•	
Jasio [m							-																-		-	1							н										-										
Score	0.00325		e	•	•			0	•	• •	• •	e	•			•	•	• •		• •	• •	• •		•	• •		Ð	-		0.00325	•	• •	0.00325	•	••	c			0	• •	•	• •		•		• •		• •	• •	• •	• •	• •	• •		• •		• •	• •	•	
Score	0.083333	•	•	•	•		• •	¢	•	• •		•	•			•	•	• •		• •	• •	• •		-	•	• •	•	•	• •	0.083333	÷ (• •	0.083333	•	• •	•	• •		0	• •	•	• •	• •	•		•	•	• •	•		• •	• •	• •	• •			• •	• •		• •
IdST4	-																	-	1											-		•	-																											
Score	0	0	•	•	•	-		e	•		• •	0	0		• •	0	• •	0 0 00151	0	• •	• •	• •		0	0 0	5 0	. 0	•	• •	. 0	•	0.00062	¢	•		0	0 C	• •	0	• •	•	• •	• •	• •	• •	• •	• •		• •		• •	• •	• •		0 0	2	0 0	• •		
Score	¢	0	•	•	•			•	•		• •	0	o i		• •	0	0	0 1477770	0	• •		0		0	• •	- c	0	0		• •	۰	0.01785714	0	•		0	0 ¢		•	• •	•	• •		0	• •	• •		• •	0	• •	• •		• •	• •	0 0	•	0 0			00
DAVREST	0	c	0	0	0		• •	0	e -	• •		0	0			0		0 ¢	0	• •		0		0	•	o c	• •	0		. 0	۰ °	4 0	0	• •		0	0 0		0		0	• •		0 0		• •		• •	0		0 0		0 C	• •	0 0	3	0 0	• •		o c
Score	0	0	0	0	•			0	0			0	6		• •	-	•		• •		• •	• •		•	-		•	•		•	•	• •	0	e :	••	6	• •	• •	•	• •	•	• •	• •	•	• •	•	• •				0 0	• •	•		• •		• •	• •		• •
Score	0	0	0	e	0	• •		0	-		• •	•	0	• •	• •	• •			÷	•		•		•	•	• •	•	•			e	• •	0	0	• •	0	•		e		•	• •	• •	•	• •	• •	• •		•		•	• •	• •		• •	,	• •	• •		• •
DAYLOST	0	0	0	0	•	•		0	¢	• •		0	•			-	-		• •	• •	• •	•	• •	•	•			•	• •		•		0	•			• •		•	• •	•	•		•		• •	• •		•				• •	• •	• •	•	• •	• •		
Score	0	Ð	•	•	•			0	0	•	, 0	• •	0	• •			÷	0	0	0 0		•	• •	• •	•		• •	0			0	• •	0	0	• •	0		• •	0	• •		• •		0 1		0 0		• •			•			0 0	0 0	5	• •	• •		
Score	0	0	0	0	c		• •	0	¢	• •	• •	• •	0				0	0 07007007	0	0		0	• •	• •	0	• •	• •	0	o c	, o	0	0 C	0	0	• •	•	• •		0	0 c		• •		•		• •					0 0		00		0 0	5		0 0	0	° °
DAVIIOSP	0	0	0	0	ð	•		0	0	• •	• •		0			ē	0	•	10	0	• •	0	• •		0	• •		0	00	0 0	¢	ę c		0		0	00		0	0 0		•		0	5 O	۰ ه	• •	0 0	0		• • •		0 0		0 C	•	• •	0 0		• •

Score	• •		• •		•	• •	• •			•		0 0		• •	• •	• •	۰	e	¢	0		• •	• •			0.0035	• •	• •			• •	0 0.0035		• •	• •	• •	• •	• •	0 0				• •			
Score			•		e	• •	¢ ¢		• •	-	• •			•		• •	e	0	e	e						0.25	• •	• •			ə a	0.25	e e		• •	• •	0 0		0 0							. 8. 6
PARTIAL DISABILITY	0		•		D	¢ 0	• •		• •	0		• •		•			0	0	0	¢ ;		c 0	• •			-					• •	e -	• •		• •				• •				• •			
Score	• •		•	e a	o	• •	• •					0 0		0			0	o	0	0		c 0	0 0			0	00	• •		00	• •	0 0	0 0	6 0	0.0105 0	0 0	0 0	0 0	00				00			
Score	0 0		0 1		0	0 0	• •		• •	0	0 0	0		0 0		0 0	• •	a	0	•		• •	• •		• •	0	• •					0 0		0 0	0.75 0	• •	c c	с с				• •	00	00		
FOTAL DISABILITY	0 0		0		0		0 0		0 0	0	• •	e (0 0		0 0	0	0	0	0		0 0				0	0 0	0 0	00	0 0	0 0	• •	0 0	• •	1	0 0	a D	0 0	00				0 0	0 0		
Score			•	• •	•	• •	• •	• •	• •			• •		•	• •	• •	•	0	0				• •			6	0 0.187	0 0.187		0.176 0	a a	= 0	0 0.176	• •	0.231	0 0.231	9.198 0		0.209	0.264	• • •	0	0.187 0			0.176
Score	0 e	• •	•	• •	ø	• •	• •	• •	• •	0		•		•		• •		8	8	e		• •	• •		• •	0	0 0.53125	0.53125	• • !	5 <u>0</u> 0			0.5	• •	0.65625	0.65625	0.5625 0		0.53375	0.75		679670	0.53125	• •		0.5625 0.5
TOTFTI	e e																																													
Score			6	• •	0	• •	• •	• •	• •	0	• •	•		•		• •	• •	0	0		° °		00			9		• •	• • •	- 0	••		o =	• •	e e	••	8 0	••	-7 P	4 6 1		n = 0	4 9	00		
	• •		e .	e e o o	e 0 0	e e o o	• •	•	••	0	0 0 0 0			0		e e	•	•	•		a a 0 0	0 0 0 0	0 C		1.5071E-08 0 0 0	0 1.3314E-05 0	0 0.0018414 2	7.4399E-07 0 0.00193687 2	4.3015E-07 0 0 0	0.00171438 I 0 0	7.1405E-08 0 0 0	1.5353E-06 0 0.00160854 0	6.3968E-09 0 0.00189418 1	•••	0.02421203 0 0.03058462 6	0.04684221 6	0.02891667 3 6.3467E-08 0	0 0 3.9809E-08 0	4 85010938 4 2210938	0.03303616 9	0 0 8.6455E-08 0 0.00454400	0.012554406	0.02776507 2 0	8.6941E-07 0 0 0	0.00031447 0	0.02891667 3
Utility	00		0 0	e a 0 0	0 0 0 0	e e o o o o		0 0	• • • •	0	• • • •	0	• • • •	• • •	0 0	• •		• • • •	• • • •		0 0 0	8 8 0 0	0 0 0 0	0	3.2067E-07 1.5071E-08 0 0 0 0	0 0 0.00028328 1.3314E-05 0	0 0 0 00.03917866 0.0018414 2	1.583E-05 7.4399E-07 0 0.04121009 0.00193687 2	9,1521E-06 4.3015E-07 0 0 0 0	0.0364761 0.00171438 1 0 0 0 0	1.5193E-06 7,1405E-08 0 0 0 0	3.2665E-05 1.5353E-06 0 0.0342242 0.00160854 0	1.361E-07 6.3968E-09 0 0.04030168 0.00189418 1	• • • •	0.51514963 0.02421203 0 0.65073668 0.03058462 6	0 0 0 0.99664275 0.04684221 6	0.61524838 0.02891667 3 1.3504E-06 6.3467E-08 0	0 0 0 0 8.47E-07 3.9809E-08 0	0.6406251 0.03010938 4	0.70289699 0.03303616 9	0 0 1 1.8395E-06 8.6455E-08 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.5907462 0.02776507 2 0 0 0	1.8498E-05 8.6941E-07 0 0 0 0	0.00669077 0.00031447 0 0.00669077 0.00031447 0 0 0	0.61524838 0.02891667 3 0.04030168 0.00189418 1
Score Utility	• •		0 0 0	0 0 0 0	e 0 0 0 0	0 0 0		•	• • • • • • •		• • • • •			0 0 0	•	• •	, e , o	•	• • •			8 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		L.37403E-06 3.2067E-07 1.5071E-08 0 0 0 0	0 0 0,001212048 0,00028328 1.3314E-05 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6.78229E-05 1.583E-05 7.4399E-07 0 0.146857575 0.04121009 0.00193687 2	3,9214E-05 9,1521E-06 4,3015E-07 0 0 0 0 0	0.132425767 0.0364761 0.00171438 1 0 0 0 0 0	6.50981E-06 1.5193E-06 7.1405E-08 0 0 0 0 0	0.000139943 3.2665E-05 1.5353E-06 0 0.125378343 0.0342242 0.00160854 0	5.83187E-07 1.361E-07 6.3968E-09 0 0.144127338 0.04030168 0.00189418 1		0.76030189 0.51514963 0.02421203 0 0.842536235 0.65073668 0.03058462 6	0 0 0 0 0 0.998741089 0.99664275 0.04684221 6	0.822553524 0.61524838 0.02891667 3 5.78612E-06 1.3504E-06 6.3467E-08 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.836941076 0.6406251 0.03010938 4 0.70508475 0.43067587 0.02494177 7	0.870245595 0.70289699 0.03303616 9	7.88185E-06 1.8395E-06 8.6455E-08 0	0.//890638 0.54349055 0.02554406 5 0 0 0 0 0 0	0.808165972 0.5907462 0.02776507 2 0 0 0 0	7.92551E-05 1.8498E-05 8.6941E-07 0 0 0 0 0	2.72352-00 0.22302-07 3.00012-08 0 0.027711885 0.00669077 0.00031447 0 0 0 0 0	0.822553524 0.61524838 0.02891667 3 0.142127338 0.04030168 0.00189418 1
INJCOST Score Utility	• •		e 0 0 0		a o o o o	• • • • • • • • • • • • • • • • • • •			• • • • • •		0 0 0 0 0 0 0 0						,	• • •				• • • • • •			840 I.37403E-06 3.2067E-07 I.5071E-08 0 0 0 0 0 0 0 0 0	0 0 85595 0.001212048 0.00028328 1.3314E-05 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	24655 6.782295-05 1.5835-05 7.43995-07 0 480000 0.146857575 0.04121009 0.00193687 2	1690 3.9214E-05 9.1521E-06 4.3015E-07 0 0 0 0 0 0 0 0	341330 0.132425767 0.0364761 0.00171438 1 0 0 0 0 0 0	3750 6.50981E-06 1.5193E-06 7.1405E-08 0 0 0 0 0 0	37385 0.000139943 3.2665E-05 1.5353E-06 0 210120 0.125378343 0.0342242 0.00160854 0	360 5.83187E-07 1.361E-07 6.3968E-09 0 460060 0.144127338 0.04030168 0.00189418 1		1301772 0.76030189 0.51514963 0.02421203 0 2845000 0.842536235 0.65073668 0.03058462 6	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2470000 0.82253324 0.61524838 0.02891667 3 3360 5.78612E-06 1.3504E-06 6.3467E-08 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2740000 0.836941076 0.6405251 0.03010938 4 1455000 0.770568475 0.43057587 0.03494177 7	142.000 0.870245595 0.7028659 0.03303616 9	0 0 0 1 4473 7.88185E-06 1.8395E-06 8.6455E-08 0	1600908 0.7/390638 0.3439909 0.00000 0 0 0 0 0 0 0	2200000 0.808165972 0.5907462 0.02776507 2 0 0 0 0	27160 7.92551E-05 1.8498E-05 8.6941E-07 0 0 0 0 0	1680 2./2356-00 6.2466-07 3.066114-08 0 162593 0.02711885 0.0669077 0.00031447 0 0 0 0 0 0 0 0	2470000 0.822553524 0.61524838 0.02891667 3 460000 0.144127338 0.04030168 0.00189418 1
Score INJCOST Score Utility	3.2136E-08 0 0 0 0 2.2136E-08 0 0 0	5.2277E-09 0 0 0 0 0	.02190248 0 0 0 0 0	1.11741877 0 0 0 0 0 0 (2917E-09 0 0 0 0 0	A.N.N.N.26651 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2.3561E-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	E827E-19 0 0 0 0 0	5.4355E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	27507E47 0 0 0 0 0	3.585E-08 0 0 0 0 0 0 0 0 1.1138E-08 0 0 0	3,4466E-10 0 0 0 0	2,8420,5-05 0 0 0 0 0 0 0 0 0		2,548(E-1)6 0 0 0 0 0 0 0	5,9112,5-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	3.482E-10 0 0 0 0 0	.47510811 0 - 0 0 0 0	\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	2751E-08 0 0 0 0 0 0	0.12070885 0 0 0 0 0 0 0 0	A02596834 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	5.4816E-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		LAZEZESES 840 1.37403E-06 3.2067E-07 1.5071E-08 0 1.02644691 0 0 0 0 0 0	0 0 0 h.07610811 85595 0,001212048 0,00028328 1.3314E-05 0	LU9525396 0 0 0 0 0 0 0 L01704178 432000 0.14072675 0.03917866 0.0018414 2	.007610811 24655 6.78229E-05 1.583E-05 7.4399E-07 0 .09896032 480000 0.146857575 0.04121009 0.00193687 2	1,0916(0441 [6990 3,9214E-05 9,1521E-06 4,3015E-07 0 1,02404312 0 0 0 0 0 0 0	0.0211667 341330 0.132225767 0.0364761 0.00171438 1 8.7042E-10 0 0 0 0 0 0 0 0	A07819549 3750 6.50981E-06 1.5193E-06 7.1405E-08 0 2.6763E-09 0 0 0 0	\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\	2,07610811 360 5.83187E-07 1.361E-07 6.3968E-09 0 2,11741877 460000 0.144127338 0.04030168 0.00189418 1	0.0255058 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1,05E-06 1301772 0.76030189 0.51514963 0.02421203 0 A07610811 2845000 0.842536235 0.65073668 0.03058462 6	8.66.5E.40 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.00034997 2470000 0.822553524 0.61524838 0.02891667 3 0.1132781 3360 5.78612E-06 1.3504E-06 6.3467E-08 0	A47610811 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.11464813 2740000 0.836941076 0.6406251 0.03010938 4 3 11464 1495000 0.836941076 0.6406251 0.03404177 2	2.12465-00 1493000 0.17039443 0.3000364 0.03036166 9	1,17847562 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	11.1.724719 16209908 0.74349053 0.74349053 0.002234406 3 3.10724719 0 0 0 0 0 0	3.10 734 9 37 2200000 0.808165972 0.5907462 0.02776507 2 3.0 8490491 0 0 0 0 0	0.00849822 27160 7.92551E-05 1.8498E-05 8.6941E-07 0 8.4847E-07 0 0 0 0 0 0	LIU/24/19 1060 2./2355-06 0.2436-07 3.06915-03 0 LIU/24/19 16:2593 0.02711885 0.00669077 0.0031447 0 A.A.168385-10 0 0 0 0	2000-2014) 2470000 0.822553524 0.61524838 0.02891667 3 3.08669559 460000 0.144127338 0.04030168 0.00189418 1
Utility Score I INJCOST Score Utility	3,48033E-07 8,2136E-08 0 0 0 0	2.215(4E-08 5.2277E-09 0 0 0 0 0 0 0	0.092807125 0.02190248 0 0 0 0	0.49753715 0.11741877 0 0 0 0 0 0 6 6 5.47318E-09 1.2917E-09 0 0 0	0.001112259 0.00102651 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6.52467E-06 1.5398E-06 0 0 0 0 0 1.33541E-08 1.5398E-06 0 0 0 0 0 0 1.33541E-08 2.9156E-09 0 0 0 0	1.68356E-07 3.9661E-08 0 0 0 0 0 0	0.000357.00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1.45573E-08 3.4355E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1.16555E-06 2.7507E-07 0 0 0 0 0	1.51907E-07 3.585E-08 0 0 0 0 0 0 4.71934E-08 1.1138E-08 0 0	3.57907E-09 8.4466E-10 0 0 0 0 0	0.000120448 2.84262-05 0 0 0 0 0 0 0 0 2.636348-49 6.22188-10 0 0 0 0	1.73709E-07 4.0995E-08 0 0 0 0	1.56259E-07 3.68//E-08 0 0 0 0 0 0 0 0 0 0 0 0	2.56473E.08 5.9112E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1.17542E-09 3.482E-10 0 0 0 0 0 0	0.332491979 0.07640811 0 · 0 0 0 0	0.24219722_0.65715855 0 0 0 0 0	1.17589E-07 2.7751E-08 0 0 0 0 0	0,122046622 0.028903 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.110435332 0.02596834 0 0 0 0 0 0 0 7.78639E-09 1.8376E-09 0 0 0 0	2.32271E-07 5.4816E-08 0 0 0 0 0 0	0,006152938 0,00145209 0 0 0 0 0 0 0	0,111293853 0.02626535 840 1.374036-06 3.20676-07 1.50716-08 0 0.1061340972 0.022614691 0 0 0 0	0 0 0 0 0 0 0.022491979 0.07610841 85595 0.001212048 0.00028328 1.3314E-05 0	0.403618486 0.09525396 0 0 0 0 0 0 0.172210931 0.01704178 432000 0.14072675 0.03917866 0.0018414 2	0.322491979 0.07610811 24635 6.78229E-05 1.583E-05 7.4399E-07 0 0.419323395 0.09896032 480000 0.146857575 0.04121009 0.00193687 2	0.345781395 0.081604441 [6590 3.9214E-05 9.1521E-06 4.3015E-07 0 0.101877508 0.02404312 0 0 0 0	0.089689409 0.0211667 341330 0.132425767 0.0364761 0.00171438 1 3.68822E-D9 8.7042E-10 0 0 0 0	0.331336837 0.07819549 3750 6.50981E-06 1.5193E-06 7.1405E-08 0 1.13404E-08 2.6763E-09 0 0 0	0.147610364 0.03483605 37385 0.000139943 3.2665E-05 1.5353E-06 0 0.402776845 0.09505554 210120 0.125378343 0.0342242 0.00160854 0	0.322491979 0.017610811 360 5.83187E-07 1.361E-07 6.3968E-09 0 0.49753715 0.11741877 460000 0.144127338 0.04030168 0.00189418 1	0.108075435 0.0255058 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	4,44899E.06 1.05E.06 1301772 0.76030189 0.51514963 0.02421203 0 0.312491979 0.07610811 2845000 0.842536235 0.65073668 0.03058462 6	3.67162E-09 8.665E-10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0.11209933 5776375 0.99864275 0.04684221 6	0.001482924 0.00034997 2470000 0.822553524 0.61524838 0.02891667 3 0.479991936 0.1132781 3360 5.78612E-06 1.3504E-06 6.3467E-08 0	0.332491979 0.07610811 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.485797172 0.11464813 2740000 0.836941076 0.6406251 0.03010938 4 0.4856797172 0.11464813 2740000 0.836941076 0.6406251 0.03010938 4	9.004551E-06 1.142-000 0.170294595 0.70289699 0.03303616 9 0.388211999 0.09161801 3365000 0.870245595 0.70289699 0.03303616 9	0.3332523809 0.07847562 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.605519657 0.142241264 1650908 0.7/1390658 0.5449053 0.122534406 5 0.454437262 0.10724719 0 0 0 0 0	0.454870219 0.10734937 2200000 0.808165972 0.5907462 0.02776507 2 0.359766582 0.08490491 0 0 0	0.036009409 0.00849822 27160 7.92551E-05 1.8498E-05 8.6941E-07 0 3.59523E-06 8.4847E-07 0 0 0 0 0	0.454437262 0.101/24/19 1680 2.7232502 0.22505-07 2.00615-08 0 0.473255021 0.11166771 162593 0.027711885 0.00669077 0.00013447 0 1.464774204 4.62588510 0 0 0 0	0.555557188 0.08669559 460000 0.82253534 0.61524838 0.02891667 3 0.555557188 0.08669559 460000 0.144127338 0.04030168 0.00189418 1
Score Utility Score INJCOST Score Utility	534891E-07 3,48033E-07 8,2136E-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1.52448E-US 9.21514E-08 5.2277E-09 0 0 0 0 0 0	0.135017414 0.092807125 0.02190248 0 0 0 0	0.595763641 0.49753715 0.11741877 0 0 0 0 0 8.41172E.09 5.47318E.09 1.2917E-09 0 0 0	0.001734545 0.001129259 0.00026651 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1.00277E-05 6.52467E-06 1.5398E-06 0 0 0 0 1.8987E-08 1.374541E-08 2.9155E-09 0 0 0	2.82858.07 (268568-07 3.96618-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1.68191E-08 1.09435E-08 2.5827E-19 0 0 0 0 0 0	2.23331E.08 1.45573E.08 3.4355E.09 0 0 0 0 0 3.40261E.07 7.1355E.07 6.224E.08 0 0 0	1.791338-06 1.16555E-06 2.7507E-07 0 0 0 0	2.33466E-07 1.51907E-07 3.585E-08 0 0 0 0 0 0 0 7.253456E-08 1.1138E-08 0 0 0	S.SEMGAFL-09 3.S7907E-09 8.4466E-10 0 0 0 0	0.000185103 0.000120448 2.84406-05 0 0 0 0 0 0 4.05178E-09 2.63634E-09 6.2218E-10 0 0 0	2.66973E-07 1.73709E-07 4.0995E-08 0 0 0 0	2.40154E-07 1.56259E-07 3.5847/E-08 0 0 0 0 0 0 0 5.56694E-05 3.5227E-05 8.5486E-06 0 0 0	3.84952E-08 2.50473E-08 5.9112E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2.26758E.09 1.47542E.09 3.482E-10 0 0 0 0 0 0	0.417023435 0.222491979 0.07610811 0 0 0 0	0.3255461 0.242197462 0.05715855 0 0 0 0	1.8072E-07 1.17588E-07 2.7751E-08 0 0 0 0	0,174682773 0,122046622 0,0289019 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.158541763 0.116035332 0.02596834 0 0 0 0 0 0 1.19669E-08 7.78639E-09 1.8376E-09 0 0 0 0	3.6676E-07 2.32271E-07 5.4816E-08 0 0 0 0 0	0.009420601 0.006152938 0.00145209 0 0 0 0 0 0	0.160242868 0.111293853 0.02626535 840 1.37403E-06 3.2067E-07 1.5071E-08 0 0.153249472 0.106130972 0.025614691 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.503071412 0.403618486 0.09525396 0 0 0 0 0 0 0 0 0 0 0 0.106296239 0.072210931 0.01704178 432000 0.14072675 0.03917866 0.0018414 2	0.417023435 0.322491979 0.07610811 24655 6.78229E-05 1.583E-05 7.4399E-07 0 0.519061272 0.419323395 0.09896032 480200 0.146857575 0.04121009 0.00193687 2	0.442342122 0.345781395 0.08160441 [6990 3.9214E-05 9.1521E-06 4.3015E-07 0 0.147458314 0.101877608 0.02404312 0 0 0 0	0.130712255 0.089689409 0.0211667 341330 0.132425767 0.03564761 0.00171438 1 5.6684215-09 3.6882212-09 8.704212-10 0 0 0	0.426699897 0.331336837 0.07819549 3750 6.50981E-06 1.5193E-06 7.1405E-08 0 1.7429E-08 1.13404E-08 2.6763E-09 0 0 0 0	0.208352556 0.14761u364 0.03483605 37385 0.00013943 3.2665E-05 1.5353E-06 0 0.502208657 0.402776845 0.09505534 210120 0.125378343 0.0342242 0.00160854 0	0.417023435 0.322491979 0.07610811 360 5.83187E.07 1.361E-07 6.3968E-09 0 0.595765641 0.49753715 0.11741877 460060 0.144127338 0.04030168 0.00189418 1	0.155887978 0.108075455 0.0255058 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6.83762E-06 4.44899E-06 1.05E-06 1301772 0.76030189 0.51514963 0.02421203 0 0.417023-445 0.322491979 0.07610811 2845000 0.842536235 0.65073688 0.03058462 6	5.64291E-09 3.67162E-09 8.665E-10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.002277015 0.0001482924 0.00034997 2470000 0.82253524 0.61524838 0.02891667 3 0.578965188 0.479991936 0.4132781 3360 5.78612E-06 1.3504E-06 6.3467E-08 0	0.417023435 0.322491979 0.07640811 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.885484.88 0.485797172 0.11464813 2740000 0.856941076 0.6406251 0.03010338 4 0.885484848 0.485797172 0.11464813 2740000 0.856941076 0.5406251 0.03010338 4 • ***********************************	1.4871474545 0.4886211949 0.09161801 3365000 0.870245595 0.70289699 0.03303616 9	0.427392748 0.332523809 0.07847562 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0.57556853 0.531400819 0.12541059 4473 7.881852-0 1.88195E-06 8.6455E-08 0 0 0.57755685435E-08 0 0.5775685455656565656565656565656565656565656	0.63442608 0.605519657 0.14290264 1650908 0.1/1890538 0.5449053 0.0454437262 0.10724719 0 0 0 0 0 0 0	0.554509831 0.454870219 0.10734937 2200000 0.808165972 0.5907462 0.02776507 2 0.457301333 0.359766582 0.08490491 0 0 0	0.054144427 0.036009409 0.000849822 27160 7.92551E-05 1.8498E-05 8.6941E-07 0 5.5255E-06 3.59523E-06 8.4847E-07 0 0 0 0	0.554084103 0.454437263 0.101/24/19 1050 2.1/2012-06 0.2450-07 3.06012473 0 0.572452175 0.473252012 0.11662713 165259 0.02771188 0.0665077 0.00031447 0 0.64772-06 0.464372-09 0 0 0 0 0 0	1 811566 0.00168 0.001715 0.00000 0.00000 0.000000 0.000000 0.000000
DAMCOST Score Utility Score INJCOST Score Utility	64351 5.34891E-07 3.48033E-07 8.2136E-08 0 0 0 0	101007 1.52448E-05 9.91945E-06 23192E-06 23192E-09 0 0 0 0 0 0 0 0 0 0 0	274608 0.135017414 0.092807125 0.02190248 0 0 0 0	5879892 0.595763641 0.49753715 0.11741877 0 0 0 0 0 20572 8.41172E.09 5.47318E.09 1.2917E-09 0 0	153049 0.001734345 0.001129259 0.00002651 0 0 0 0 0 0	96492 1.00277E-05 6.52467E-06 1.5398E-06 0 0 0 0 28527 1.8087FE-08 1.33441E-08 2.85562-09 0 0 0 0	56394 1.582865-07 1.680565-07 3.96615-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	27301 1.68191E-08 1.09435E-08 2.5827E-09 0 0 0 0 0 0	30204 2.227318-08 1.45573E-08 3.4355E-09 0 0 0 0 0 0 50402 3.40504E-07 5.224E-08 0 0 0 0	77595 1.79133E-06 1.16555E-06 2.7507E-07 0 0 0 0	55222 2.33466E-07 1.51907E-07 3.585E-08 0 0 0 0 0 0 42618 7.35315E-08 1.1138E-08 0 0	16767 5,50067E-09 3,57907E-09 8,4466E-10 0 0 0 0	128491 0.0001885103 0.000120448 2.84208-05 0 0 0 0 0 0 14232 4.051788-09 2.636348-49 6.22188-10 0 0 0	56755 2.66973E-07 1.73749E-07 4.0995E-08 0 0 0	25000 2,40154E-07 1,50259E-07 3,6874E-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	35863 3.84952E-08 2.50473E-08 5.9112E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	10000 2.26/58E.09 1.47542E.09 3.482E.10 0 0 0 0 0	922704 0.417023435 0.322491979 0.07610811 0 0 0 0	811211 0.3255461 0.242197262 0.05715855 0 0 0 0 0	52502 1.8072E-07 1.17589E-07 2.7751E-08 0 0 0	472200 0.174682773 0.122046622 0.02800 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	404698 0,158541763 0,110035332 0,02596834 0 0 0 0 0 239229 1,196695-08 7,78639E-09 1,8376E-09 0 0 0	59928 3,56976E-07 2,32271E-07 5,4816E-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	18/22/ 0.009420601 0.0065823938 0.00145289 0 0 0 0 0 0 0 0	412461 0.160242868 0.111293853 0.026285555 840 1.37403E-06 3.2067E-07 1.5071E-08 0 379377 0.155249472 0.106134072 0.02644691 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1154512 0.503071412 0.403618486 0.09525396 0 0 0 0 0 19 198221 0.106296239 0.072210951 0.01704178 432000 0.14072675 0.03917866 0.0018414 2	922704 0,417023435 0,322491979 0,07610811 24655 6,782295-05 1,5855-05 7,43995-07 0 1971318 0,51906(1272 0,419323395 0,09896032 480000 0,146857575 0,04121009 0,00193687 2	948158 0.442342122 0.345781395 0.08160441 (6990 3.92145-05 9.1521E-05 4.3015E-07 0 349382 0.147458314 0.101877608 0.02404312 0 0 0	244479 0.130712253 0.089689409 0.0211667 341330 0.132425767 0.0364761 0.000171438 1 17026 5.6684215-09 3.6882215-09 8.704215-10 0 0 0	912653 0.426699897 0.33135887 0.07819549 3750 6.50981E-06 1.5193E-06 7.1405E-08 0 27660 1.7429E-08 1.13404E-08 2.6763E-09 0 0 0 0	581216 0.2003525556 0.147610364 0.03485605 37385 0.000139443 3.26655.05 1.5353E.06 0 1112548 0.502200657 0.402776845 0.09505554 210120 0.125378343 0.0342242 0.00160854 0	922704 0.417022435 0.322491979 0.01610811 360 5.83187E.07 1.361E.07 6.3968E.09 0 5879892 0.595763641 0.49753715 0.11741877 460000 0.144127338 0.04030168 0.00189418 1	392231 0.155887978 0.108075435 0.0255058 0 0 0 0 0 0 0 0 0 195317 0.05587978 0.02550589 0 0 0 0 0 0 0	92290 6.827/62E-06 4.4489/E-06 1.05E-06 1.301/72 0.76030189 0.5151/963 0.02421203 0 922704 0.417/024445 0.322491979 0.07610811 2845000 0.84256255 0.65075668 0.03058462 6	16987 5.64291E-49 3.67162E-09 8.665E-10 0 0 0 0 0 0 0 1 4778060 0.57141145 0.47499313 0.11200933 5776375 0.298741089 0.99664275 0.04684221 6	156037 0.002277015 0.001482924 0.00034997 2470000 0.82253524 0.61524838 0.02291667 3 5023882 0.579505188 0.479591936 0.1132781 3360 5.78612E-06 1.3504E-06 6.3467E-08 0	912704 0.417023435 0.312491979 0.07610811 0 0 0 0 0 1 13005000 0.735587635 0.653165694 0.15414805 2160 3.62929E-06 8.47E-07 3.9809E-08 0	2006397 0.5845484.80 0.48579712 0.11464813 2740000 0.836941076 0.6406251 0.03010938 4 1000056 1.5845484.80 0.48579712 0.11464813 2740000 0.836941076 0.5406251 0.03010938 4	1000440 1.385/3124.0 9.0045512-00 1.444-05-00 1.472-000 0.870245555 0.70289699 0.03303616 9	933961 0.427992748 0.33252809 0.07847862 0 0 0 0 1 7 7801855-06 1.63955-06 8.64555-08 0 7500000 0.627556853 0.551400819 0.12541059 4477 7 770075 7 770055 7 7700575 7 770055 7 770055 7 770055 7	10907500 0.69412608 0.6015519657 0.14290264 1650908 0.77390638 0.5424025 0.022344005 0 3756000 0.55408766 0.454437262 0.10724719 0 0 0 0	3777694 0.554509631 0.454870219 0.10734937 2200000 0.808165972 0.5907462 0.02776507 2 962367 0.457301333 0.359766582 0.08490491 0 0 0	190817 0.0.54144427 0.036009409 0.000849822 27160 7.92551E-05 1.8498E-05 8.6941E-07 0 89952 5.5255E-06 3.59521E-06 8.4847E-07 0 0 0	255000 0.554004115 0.45447262 0.101/24/19 1560 2.19552-0.0576247 0.0669017-03 0 4691994 0.572452175 0.47325021 0.1168771 162593 0.027711888 0.00669077 0.00031447 0 11973 3.014424.04 9.64378482.00 0 0 0 0 0	56692 0.45571266 0.55587928 0.09669559 46000 0.82255524 0.6154828 0.02891667 3 56692 0.45512567 0.555387928 0.08669559 460000 0.144127138 0.04030168 0.00189418 1 56972 0.46514257 0.567354188 0.08669559 460000 0.144127138 0.04030168 0.00189418

Score	0.00145	0.00116	0.02327	0.00083	0.018222	0.02023	0.00038	0,00386	0.00045	0.00287	0.00137	0.00036	0.0004	0.02823	0.00029	0.00844	0.0028	100.0	3.5E-10	0	0.138110	0.05716	0.00055	0.03302	0.22821	0.0005	0.00028	0.01593	0.03327	0.03353	0.18153	0.10853 0.76788	0.19001	0.3499	0.09579	0.30415 8 7F-10	0.21658	0.00107	0.16232	0.13858 0.39631	0.02682	0.08145	0.43869	0.51719 0.51719	0.34127	0.14786	0.24631	0.32269	0.50265	0.14301	0.1725	957350 134736	1560.0	0,18443	0.24453 0.00046	0.33634
Utility Score	0.0234383 0.00145317	0.0187506 0.00116254	0.0221101 0.00137083	0.013438 0.00083315	0 0 c/occ/10m c706870	0.0642403 0.0039829 0.0336742 0.0020878	0.0061721 0.00038267	0.0085159 0.00052799 0.0378672 0.00234777	0.0071877 0.00044564	0.0463024 0.00287075	0/0221000 0/0212000	0.0058596 0.00036329	0.0064846 0.00040204	1.1403082 0.00869911 1.0773447 0.00169537	0.0046877 0.00029063	0.031251 0.00193756	10449985 0.00278991	1.0161724 0.00100269				e :	0 0 0	0.0156255 0.00096878	0 0 01770/2 0.05010/2	0.0081253 0.00050377	0.0045314 0.00028095	0.1585741 0.0098316 0.0235164 0.00145802	0.0953853 0.00591389	0.1368838 0.0084868 0 0 0	0 0	0.2141201 0.01327545 0 0	•	8 8 8 8		0 0 # 0	0	0.0171881 0.00106566	0 0	 	0.0212507 0.00131754	• •	0 0		0	, a , a	0 0 0 0		e e	•		e e e e	0	0 0 0	0 0 0.0074221 0.00046017	0 0 0
Score	0.0234375 0	0.01875 0	0.022109375 0	0.0134375	16/19/2710	0.064238284 6	0.006171875 0	0.037865999 (0.0071875	0.046300895 (0.0210001220.0	0.005859375 (0.006484375	0.140304314 (0.0046875	0.03125	0.044997062	0.016171875 (. 0		•	0	0.00890625	0.015625	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.008125	0.00453125 (0.15856959 (0.023515625 (0.095382324	0.13687982	¢	0.214114381 (• •	• •	• •	• •	•	0.0171875	0		0.02125		0	0	6 6	• •	• •	, e		•				0	0 0.007421875	
MHRS	300	240	283	- 11	1681	9 <u>5</u> 1	6	60 98 1	92	t s	187	75	8	1281	; 5	901	260	202	. 0	¢	•	0	111	200	0	101	8	301	1000	1262	•	1625	• •	• •	• •	• =	• •	220	9 0	• •	272		•	8 o	• •	• •	• •	, .						817 0	• %	0 8
Score	• •	• •	0	7907.D	•	• •	•		c	•		9	a .	• •	• •	0		• •			790.0	0	c	c	0.062		0	° °	• •	0	0.062	0 063	0.062	0.062	0.062	0.062	0.062	• •	0.062	0.062	۰	0.062	0.062	0.062	0.062	0.062	0.062	0.062	0.062	Ð	0.062	0.062	0.062	u 0.062	0.062 0	0.062
Score		• •	o -	- 0	•	0 0	0	0 0	0	• •	• •	0	0			0	•	• •	• •		-	0	c	0			0	e e	• •	0	-	e -				c		0 0			• •							•						o =	- 0	
LOTLOSS	• •	• •	۰ -	- 0 -	-		. 0	• •	•		• •	0	0	• •	• •	0	0	0 0			-	0	Ċ				0	• •		0	-	o -						0 0	•		• •			s											- 0	
Score 1	• •				8	0.0065 B		0.00325	0	• •		0	•	~ ~		•	0	• •			•	8	-	•	0.00325		0			•	0.00325		0.00325	0 00075	0.00325	0.00325	0.00975	0	8	• •	•	0.00325	0	0.00975	0.00325	0.00975	0.0065	0.00325	0.00325 0.0065	0.0065	0	0 0005	0.00325	2 0		0.00325 0
Score	• •	• •	•		•	0.166667		0.083333	•	-		•	-	• •	• •	•	•		• •		-	•		•	0.083333		8	• •		¢	0.083333	• •	0.083333	0 00111	0.083333	0.083333	0.25	0 166667	0	e e		0.083333	0	0.25	0.083333	0.25	0.166667	0.083333	0.083333	0.166667	- 0	0 166667	0.083333	• •		0,063333
HYSPL						1		-																	-						-			•	• •	-	ñ	•	4					£	-	£	4 4		- 6	. 14 1	n	۴				-
Score	• •	• •	• •		•	• •	o	0 C	• •	• •		0	0	• •	• •	•	0	• •			•	0	c	0.00325	0.00325	• •	۰	• •		•	0.00325	• •	0.00975	0	0.0065	0.00975	0.00975	0	0	• •		0.00325	0	0.00975	0.00975		2000-0	0.0065	0.00975	0.00325	0.0335	0.00325	0.00325	0.00325	c 0	0 0
Score	¢	• •	0 0		0	6 6		• •		•		0	0	0 0		• •	0	• •			-	0	c	0.083333	0.083333	• •	•	• •		•	0.083333	0 0	0.25	0	0.166667	0.25	0.25	0	0	0 0		0.083333	6	0.25	0.25		0.25	0.166667	0.25	0.083333	0.083333	0.083333	0.083333	U 0.083333	• •	• •
14S10																								-	-						-		3		- 14	m 5		-	-					'n	£		۰. ۲	. 0	m -	- ·	n	-	-			ĸ
Score	• •	• •	0		•	0.00975	ø	0.007	¢	-	- 0	e	•	0.0195	• •	7 0.0065	0			,	-	o	e	•	0.039		•	• •		•	0.0195		3 0.0032		• •	3 0.0032	0,039	• •	• •	0	•	• •	0.039	0.039	0.039	• •	1600.0	0,039	0.039	3 0.0032	0.039	0 0.000	0	0.0097		0.039
L Score	•		0		•	0.25		0	•			0	0	0.5	• •	0.16666	•	• •	• •		•	e	-	• •		• •	•	• •		¢	0.5	• •	0.08333	• •	• •	0.08333	-	• •	• •	• -	• •		-				0.25			0.08333	- 0	0	0	U 0.25	• •	1 0.25
FLST						'n			•					4		7									¥î				375		175 4		867 1		3	1 29	433 5	97	625	875 6	,	375	ŝ	ŝ	5	787	331 3	3 W 3	ŝ	1 22	871 5	-	•	875 3	817	wi en
Scurr	•	• •	•	• •	•			00		• •		•	•	• •	• •	0	0	•	00		e	0	c		0 0		0		0.00109	0	0.00211	00	3 0.01826	0 0	0	19 0.0180	56 0.01788	0 0000	stoooro 98	57 0.00046	• •	0.00100	0	• •	0	0	13 0.00416	, o	0 0	57 0.0015	0 0.01785	• •		0.0021	81 0.01999 0	00
Score	•				0	06		0 0	• •	•		0	0	• •	• •		0	• •			•	Ð	e	• •	6 (• •	0	0 0	0.03125	0	0.0625	0 0	0.5219619	0	0	0.5164714	0.5109809	0	0.0044642	0.0133928		0.03125	c	• •	0	0.2600806	0.1189516	» o	00	0.0446428	0.5102489	• •	• •	0.0625	0.5713762	
DAYREST	•	• •	c 1	• •	o					• •		o	•	• •		. 0	0	0			0	0	c		0 0		0	0 0	•	0	4	• •	105	0 8	2 0	8	5	0	<u>-</u>	n c		0 1	0		• ;	87 0	8	, .	0 0	9 2 1	F 0	• •		• 1	0 0	• •
Score	•		•	• •	Ð	00		• •		•			•	• •	• •		0	• •			-	0	e		•		¢	• •			0.00828637	• •	0.0085191	0	0	0,0045	• •	0	0	• •		0 00.04461	•	0.00674461	•	• •	• •	••	• •	4.7693E-06	3.913E-07 0	• •	0.01366352	• •	0.01638415	
Score				• •	•	•		e 6		• •	• •	0	0	• •			0	•	• •		•	•	e		•		÷	• •		0	9685560911	0 0	0.473283303	0	0	0.25		00132220	0	• •		0 00000000000 000000000000000000000000	Ð	0	•		• •	, e	• •	000264959	0	• •	0.75908419		1010230101	
AYLOST	•		•		e	•		• •		• •		•	•				e				•	e	đ		•					0	12	• •	• ¤	• (- 7 =	⊒ •		¢ ;	, 2 =	•	• •	- 2	•		•		•	••	• •	-	~ =	• •	9		107	8 8
Score I			0	• •	0	0 0		• •		¢ i		0	0		• •		•				e	0	c	• •			. 0			0	00343269		00886218	0	0	00444231		0	0	• •		0	0	0 0	0		•		00	00302885	0.00302885	• •	0.00444231	- -).03414744 0	
Score	0		0	00	0	0 0	• •	•		•	• •	0	0	•	• •		Q	• •			Ð	0	c		0			00		0	0.081730769 0		0,211004274 0	0	0 010490011/0	0,105769231 0		0	0 0 00/06/180/0	•••		0 005707700	0		0		0	, .	• •	0.072115385 6	0.072115385 (• •	0.105769231 6		0.813034188 (0 0
JAYHOSP	e i	• •	0	• •	0	0	. 0	0 0		0		. 0	0	0 0			0	•			0	0	e		0		. 0	0 0		0	s	0 0	- E	•	2 0	10		0	^ 0	0 0		• •	. 0		0	0 0	0 0		0 0	o m	r o	0	9 8	c c	861 0	00

					•						• •			•												• •				96125							• •	• •				• •	• •					
×																														0.0																		
p																														75																		
Scor 5				• •			• •				0 0			0		• •			* *	•		•	• •	• •					••	57-0			••				• •			•	50				• •	••		, o
ABILITY																																																
0 UIVT DI		• •		• •	• •		• •			0 0	¢ ¢			0 0		• •			• •	•	• •	•			• •	• •		• •	• •		• •	• •			• •	e e				0	• •		•		* *	• •		; 0
PAR																•																																
Score 0 0		• •	• •	0.0105	0 0		• •	• • •	• •	• •	• •	• •	• •	0 0		• •		• •	• •	0	0.014	•		• •	• •	• •	• •	• •	• •	• •		• •	• •			• •	• •	• •		0	ə o	60	• •		0 0	00		> 0
Score 0 0				0.75	•		• •			• •	• •	• •	• •	0					• •	•	o	•	. .	• •	9 9	00	• •		0 0			00	• •			• •	° °	• •	• •	0	- o	• •	0	- o	00	• •		
ABILJTY																																																
TAL DIS, 0 0		0 0	00	- 0	0 (• •			0 0			0 0	0		00			0 0	0	¢ 14	0		0 0		0 0			0 0	5 0		00	0 0		00	00	• •	• •		0	- o	• •	0 0			00		50
<u>٩</u>							_																											_														
Score 0.187 0.198	0.176	• •	• •	0.176	•	• •	0.242	•	• •	0.231	0.187	6		0.187	28170	• •		0.18		0.25	0.18	•		•	0.18	• •		0.170	•	0.176		0.17	-	0		0.187		- 2	610 617	•		• •	00		• •	• •		
Score 0.53125 0.5625	0 S.O		• •	50 1	•	• •	0.6875		• •	0.65625	0.53125	6795'N	• •	0.53125	0.53125	• •		0.53125		0.71875	0.53125 0.59375	•		0	0.53125	• •	• •	0 S O	•	0.5	• •	0 9 8 9	0	0 0		0.53125	0 0	0	0.5625	0		• •	•		• •	• •	• • •	• •
1FTL 3	• -						0 r			o ve		n 0		. 61	- 7			• •			~ ~			. ,	4 14	• •		• •		• •		• -		, e		0 7		۰,	4 F	•	• •	0 0	•		• •	• •		
6507 TG	£E65	8-08	8-08	2506			6918		E-07	E-06 0938	1945	F-08	202	8761	6507	E-08		6507	E-07	9874	2715 4268			E-06	6507	E-09		5552	E-08	3752	E-07	9418				6507			1667									E-07
Scot 0.02776 8 0.0289	0.0016	0 0	0 4.2298	7 0.0286	0		0 01322		0 1.9808	2.5309	0.0244	1.5071	0 3.707	0.0244	0.0277	5 8.4579		0.0277	5 1.7983	0.0306	0.0239		• •	1.0321	0.0277	7 8.5498	••	0 0.0240	7 2.6153	0.0250	5 2.8149	0 0.0018	0	0	• •	0.0277	0 0	0	8 0.0289	00	20	00			00	00		5 2.1864
Utility 0.5907462 0.61524838	0.03551757	5.3976E-07 0	0 8.9995E-07	0.60904387	•	• •	0 70705714	0	0 4.2145E-06	5.3849E-0	0.5205874	0.6314888 3.2067E-01	0 7.8871E-00	0.5205874	0 0.5907462	1.7995E-06		0.5907462	3.8262E-00	0.65316455	0.73069533	0	• •	2.1961E-0	0.5907462	1.8191E-0		0.5111350	5.5645E-0	0.5327132	5.9891E-00	0.0403016	0	80/90/0'n		0.5907462	00	0	0.6152483		• •	o c	0	00	• •	00		0 4.6519E-0
ore 165972 553524	0 440779	79E-06 0	0 18E-06	957702 0			0 377084	0	0 82E-05	230672	172759	819108 03E-06	0 42E-05	172769	0 165972	79E-06		165972	44E-05 68E-07	868416	210923 282464			81E-05	7/6291	68E-07		0 596924	31E-06	U 930656	18E-05 0	0 127338	0	0		0 165972		0	553524	0		• •			0 0	0 0		0 25E-05
0.808	0.129	2.312	3.856	0.818			0.877		1.805	0.000	0.763	1.374	57E.E	0.763	808.0	7.710		0.808	1.635	0.843	0.756			9.408	0,808	7.794 7		0.757	2.384	0.771	2.566	0.144	ł	2.0		0.808		200	0.822									1.99
INJCOST 2200000 2470000	0 295544	0	0 2289	2402520	•	• •	0.3405000	0	9265	47532 2740000	0000261	2643880	0 15228	1370000	0 2200000	4384		2200000	8542 480	2870000	1225000 3628419	0	• •	30074	2200000	480	• •	0 1251010	1440	1520000	0	0 460000	0	0	• •	0 2200000	0 0	0	2470000	0	• •	• •	• •	• •	• •	• •		0 10055
Scare 0 17823439	482E-10 11168771	13815163 32019437	694E-07 10697199	14409180	0020183	419E-10	5948E-10	2072532	10724719	81/12/111	0849822	10724719	8182E-07 07610811	02064619	1417052	10760376	60578020	11801920	1417052	10057273	8692E-05 11301554	02019437	1377E-06	117212	946464770	11801920	151/16-05	19878161 1941945	.05E-06	61856021	10640697	02403014	1895E-09	11335982	4494E-06 00028848	0302336	1383E-07 00041001	4144E-09	12094572	0 7226E-09	4204E-07 0759E-08	5146E-07 2518E-08	4347E-10	4021E-07 8641E-10	7915E-09 8572E-09	7047E-10 7075F-09	9396E-08	0604E-06 02289154
y 0 1656 0.1	2E-09 3. 53021 0.	59363 04	3E-06 9.	81395 0.	55216 0.	10 12 10 10 10 10 10 10 10 10 10 10 10 10 10	4E-09 60	10 11161	37135 01 37262 0.	10 90617	10 60F60	31968 U	2E-07 1.	10 15858	45777 0. 70067 0.	48128 0.	23752 0.	91979 0.	122177 0	55629 0.	JE-05 10	69363 0	30196 0J 4E-05 3,	13568 0	n 22622	0 61919	1 1124	66162 UJ 31567 UJ	9E-06	68601 U	77064 0.	22642 0.	8E-09 2	00601 U	9E-06 1. 22373 0.	08476 0. 84093 0.	7E-07 1.	4E-09 1.	37135 U. 81879 U.	0 3E-08 2.	2E-07 1. 6E-08 1.	6E-07 1.	1E-09 7.	7E-06 J. 2E-09 9.	SE-08 7.	1E-09 7.	2E-08 1.	6E-05 3. 98039 0.
Unite 0.3315(0.47325	0.58534	4.1076	0.34578	0.008	3.1533	2.794	0.08781	0.45443	0.40963	0.03600	0.4010	71044	0.08748	0.6804	0.4559	0.0884	0.32240	1,6004.0	0.4261	7.9201	0.0855	0.1324	0.6005	0.0951	0.3224	0.0003	0.4185	4,4489	0.0036	6660 0	0.1018	9.2775	0.1837	6.1415	0.1281	4.8232	1666.3	0.5124	1.1536	6.0188	5 2253	3.1503	11797	3.3014	3.2647	8.2186	1.2967
Score 0 1.426879499	2.26758E-09 1.572452175	0.67661718 0.125	6.31299E-06	0.442342122	1.001313686	0.194735564 4.84635E-09	4.29471E-09	1.128122476	0.354110265	2.0 2.1111221	1054144427	1.554084105	1.18409E-06	1.127657624	0.68996159	0.555569103	0.129001831	0.68996159	0.68996159	0.525953679	0.606121719	0.125	0.188469229 1.23869F.05	0,690021345	0.138214448	0.417023435	0.000505102	0.518295008	6.83762E-06	0.005625501	0.14488816	0.147383298	1.42587E-08	0.428729307 0.254417853	9.43897E-06 0.001877243	0.182749798	7.41288E-07 0.002667207	9.21085E-09	0.354110265	1.77301E-08	9.25032E-07 7.00636E-08	9.8632E-07	4.84171E-09	2.21554E-06 6.42381E-09	5.07404E-08 3.16317E-08	5.01753E-09	1.26312E-07	1.99297E-05 0.140781094
ST 6		ž -	+ 2	9 5	2	 	. 5	1 =	28	8 8	10	88	9 7	R	- 	22.	. 2	- 1 -	87	6	8 =	2	5 X	5 I		2	200	5.5	ç	, ÷	13 X	22.00		32	e 2	8 Ø	- 	-	38	5	9.9	g =		8 6	2 *	19 1		-
DAMCY 0 93283	1001	100001 20000	9141	1846	1500	126951	1470	22501	37560	10000	19081	37560	7305	2214	37676	38316	1162	92270	106800	23225	1238:	2000	5210	106830	14562 19885E	17229	13950	19322	9229	1659	35773	3489	2565	2069	16531	5015	6792	21-12	00099	2783	7035	1105	1568	1812	3879 3379	1598	1984	J04U 3112:
EVENTI 30 30	R R	9£	0 0	90	R	R 8	R 7	8 8	R R	£ 5	8	8 8	g g	8	R R	R 8	88.	R R	£ 5	ŝ	er er	36	<u>e</u> e	R i	R R	8	R R	R R	96	R 8	8 8	8 A	R	R	R R	R R	R 8		R	32	88	8 8	8	88	5 5	88	: R	88

Score	0.27677	3 5E-10	0.37734	0.20318	0.00081	0.1722	0.16777	0.00338	0.0035	0.48327	0.02145	0.17966	0.11362	0.28197	0.43357	0.00543	0.17795	0.40722	0.23048	0.00012	0.22905	0.40812	0.0869	0.48527 0.3217	0,50481	E1280.0	0.00773	0.29592	0.31411	1/081.0	0.01357	0.45018	0.0048	0.39986	0.04442	0.0406	0.00039	0.12714	6E-05 0.0095	0.04037	0.02818	0.00665	9.7E-U0 0.3354	0.40986	0.00082	0.01048	0.00048	4.9E-06	0.0006	0.00095	0.0001	6.8E-05	0.0012	2.7E-05 0.10088
20					180409			591389	19926	17755	172174	0				15433		•		01211	170070				•	922479	123035				024502 210815	• •	393508	0	179187	0	038751	152823	27E-05 271579	038751	168005	624008	0	• •	082347	139829	20-392	39E-06	989342	1161-60	010172 938301	15E-05	319611	22E-05 027372
š					692 0.000			995 0.00 853 0.00	396 0.000	1000 N'NH	101 0.00					529 0.0				532 0.00	7070 B/0					385 0.05	1111 0.00				423 8.00 9024 0.00		469 0.00	0.01	243 0.00	1121 0.00	502 0.00	101 0.04	1375 5.81 803 0.00	1502 0.00	5782 0.02	9101 1915	an's cost		817 0.00	103 0.0	125 1.93	6-05 4.84	5878 0.00 5713 0.00	313 0.00	5407 0.00 3389 0.00	938 6.78	00.0 2761	3906 2.4 1567 0.06
HUN	•	5 9	¢	• •	0.0129	• •	•	1120.0	0.0564	0.000	0.0116	0	• •		• •	0.0876	• •	•		0.0019	0				•	0.9552	0.0198		• •		0.0340	• •	0.063	0 0.2474	0.1261	0	0.0062	96997	0.0000	0.0062	0.3496	0.100	0	•	0.0132	1 0.0642	2000.0	5 7.8131	0.0096	0.015	5 0.001(8 0.1513	0.0016	5 0-0192	9 0.9721
Score	0		. 0	• •	0.01296875	• •		0.051197887	0.05643779	0	0.01164062	0		• •	• •	0.08762623	• •	•		0.00195312	0	• •			•	0.95523707	0.01984375	-		• •	0.03400131	• •	0.06346694	0.24740145	0.12612058	0.07141280	0.00625 0	0.66980255	0.0009375	0.00625	0.34967045	0.10064332	0	0	0.01328125	0.06423828	0.0003125	0.00007812	0.0096875	0.0153125	0.00164062	10010375	0.01929687	0.00039062
MHRS	0	• •		0 4	166	• •		1000	677	: •	£	ξ¢	• •		• •	55	• •	•	• •	25	0				-	4261	52			• •	불로	• •	ę.	0	1200	6 e	8 4	2720	1 F	8 -	2062	1037	4 0	Ð	170	150	- ș	-	124	196	21	18	57	4627
Score	0.062	0.062	0.062	0.062	0	0.062	0.062		0	0.062	0	0.062	0	0.062	0.062	0	0.062	0.062	0.062	e (0.062	0.062	0.062	0.062	0.062	70//n	0	0.062	0.062	0	0.062	0.062	700.0	0.062	00	0.062	0 00	0		0	0	• •	0.062	0.062	0		• •	• •	0 0	0	00	0 0	0 0	0 0
Score													۰ -			- 0				0 0						- 0	с.			- 0	o -		- 0	- 0	• •	• -	0 -	• 0		o -		•		-	0		• •		0 0	• •	• •	00		00
OTLOSS	-							• •	0		0		e -			- 0		_		0						- o	0,			- 0	• -		- 0	- c	• •	0 -	0 -	• •	0 0	0 -	. 0	0		-	0			. 0	0 0	• •		• •		0 0
Score 1	•	0.00325			• •	• •		• •	•	• •			9	1	• •		0.00325	0.00325	0.00325		0.00325	0.0065	•	0 0.00325	0.00325	0.00325	0.00325	• •	0.00325 0.00325	0	0 0.0065	• •		0.00325	0.00325	0.00325	0	0.00325	0 0.00325	0.00325	0.00325	•		0	÷	0.00325	<i>o</i> •		• •		• •	•		• •
Score	0	0				• •		• •	•				0	00001			083333	083333	083333		083333	166667		0 083333	083333	u 083333	083333		083333		0 166667	• •		083333	083333	083333 0	0	083333	0 083333	083333	CT-0	•			•	083333	• •		• •					
HYSPL		-											,	4				1	1		1 0.	2 9.		1 0.		1	1			-	2					-	•		-		d n					1								
Score	•	0.00325	• •	• •	• •	• •		e e	•		0	50	0	0	0 0	• •	0.00975	0.00325	0.00975	0	0.00325	0.00975	0.00325	0.0065	0		0.00325		0.0065	0	0.00325 0.0065	0.0065		0.00325	0.00975	0.00325	0	0	0 0.00325	0.00325	0.00325	0	• •	0	c	0.00325					• •	0 1		• •
Score	0	0.083333	• •	0 0		0 0			0	0 0	0		0	0	• •	0 0	0.25	0.083333	0.25	0	0.083333	0.25	0.083333	0.166667	0		0.083333	• •	0.166667	0	0.083333	0.166667		0.083333	0.25	0.083333	0	0	0.083333	0.083333	0.083333	0		0	e	0.083333	• •		• •		• •	0		• •
OLSPI		-											,	'n			~ -	-	6		n –		-	2			-		n .	n	- 1	7		-	ñ	-		-	-	- '	- -					-								
Score	e	0.00975		•		•		• •	•	• •	•		0	0	0	0	0 0.0175	0.0065	0.039	-		0.039	-	0.039	0.039		-		0.00325	0.039	• •	0.039		0.00325	• •		0	6:0.9	• •	0.00325	0	•	• •	0	e		• •	• •	0 0		• •	•		• •
Score	•	0.25						• •		• •			0	0	0	0	0	66667		0									083333	• -	• •			083333		• •			• •	083333		•		-			•		•			•		
Tast		Ē											•	-	•	-	-	. 5	-			•0		un un	ŝ				1	ŝ		¥0		-				n wa		- ·	n													
ore F]	~ ~	01815			0	01040					0 66935			0	0	67936		0 64516		0 84517	0	0625		07863		0	80765	0	c700		69217	c/81			64516				•		•			-									0 76153
s	Ī		29 0.016			500.0 50	-					71 0.003				100.00	51 0.017		23 0.005		82 0.018	- 000	43 0.00		58 0.011			149 0.01	000			83 0.017	30.0			123 0.002																		29 0.01
Score	0	• •	0.4576612	•	• •	0	0	• •				0.1048387	0		0	0	0.5051244	• •	0.1612903	0	0.5384333	0 308009C 0	0.0178571	• •	0.3165322	0 0	e	0.5164714	0	0		0.5054904	0.03	00		0.1612903		00	• •	•	• •	0	• •	•	-	• •	• •		0 0		• •			0 0.503294
AYREST	o	0 0	6	0 (• ;	17 0	• •	• •	• •	• •	0 1	0		• •	- 0	\$		o 17	•	120	0 8	4	• •	33.		•	8 -		• •		9 S	0 12	• •		51			0 ¢	0 0		0		• •	-		• •	e e	0	• •	• •	• • •		o \$
- au]_	• •				10-311		• •				0	0		192589 C		13503	125891				0			151-808		-	506363 N				018		0		0 0				•		0			4		•		• •		• •			0 01013
ŭ						1.248						90.00			10.017		900.0	110 0.011				7004			10'0 10			88 0.00				đ		27 N D1	130 UNI	-02 9,60																		651 0.00
Score	1					Ş.						- 24			×.							- F	2		¥2			Ξ.			• •		* •	0 20	£ -	쁖_									-				-		~ ~			0.005627
	•	• •	• •	•		6,93396E-06	• •	• •		• •	-	0.37470055	-	• •	0.99588280		0.340834	0.649178		¢	• •	0.0000	0	• •	0.72696	• •	•	0.54946						0.570	62°0	555														• •				•
VLOST	0 0	•		•		1 6,93396E-06	0 0 0	e c		• •		0 0 0	0		240 0.99588280		16 0.340834 n	42 0.649178		0		0 0	0 0		54 0.72690	• •	•	30 0.54946 0 0.54946	•		• •	300		0	60C-10 62 0	3 5,3351						0			-		e .			••	• •			0 0 6
· DAVLOST	0	•	154 0 0	885 0 0		077 1 6,93396E-06		•		•		0 0 17 0.37470055	0 0 150		051 240 0.9588280		2885 16 0.340834 n n	(692 42 0.649178	0 0 440. 0 0 440.	0	••	0 0 0	0 0 7407	• •	1526 54 0.72690	••	•	25 30 0.54946 0 0	•		• •	1808 300		0	905.0 92 0	3 5.1354			• •			0			-		e .		0 1		• •		• •	•
Score DAVLOST	0 0	•	0.00996154 0 0	0.00302885 0 0		0.00323077 1 6.93396E-06		•	•	•		0 0 0 0 0 17 0.3747065	0 0 15020610'0	• • • •	0,03132051 240 0,99588290	• •	0.00302885 16 0.340834	0.00282692 42 0.649178	0 0 0 0 0 0 0 0 0 0	0 0	• • • •	0 0 0 0		•	0.00603526 54 0.72690	• •	•	0.00525 30 0.54946	•	•••	• •	0,00854808 300	••	0 0	900-00 0 0 0	0 3 5.335 0 0			• •			0 0	••	•	-		e .	• •	e :		• •		0	e •
Score Score DAVLOST	0 0 0	•		0.072115385 0.00302885 0		0.076923077 0.00323077 1 6.93396E-06		•	• •	• • •	a a a a	0 0 0 0 0 0 17 0.3747065	0.454059829 0.01907051 0 0	• • • •	0.745726496 0.03132051 240 0.958828		0.072115385 0.00302885 16 0.340834	0.067307692 0.00282692 42 0.649178	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0	• • • • •			• • • •	0.143696581 0.00603526 54 0.72696			0.125 0.00525 30 0.54946 0 0 0 0 0 0 0	•		• •	0.203525641 0.00854808 300	• • • •		905.0 92 0 0 0 0	0 0 3 5.335			• • •		•••	0 0 0	• •		-		• • •		e .				a a 0 0	e e 0 0

Source -			
ATTIMAKIN DISAMULTI VITAN			•••••
a			•••••
¥ • • • • • • • • • • • • • • • • • • •			
AL DISABILITY 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	
101 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 1 0 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2		0 2 2 2 2 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0
			5550 2550
Secore (1011	0.00187772 0 0.002447772 0 0.001847772 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	(1000) (1
Utility 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.49941949 0.49941949 0.40040 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	4.5149E.05	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Score 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0.749595883 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 1 1,13837E.05 0 0 1,93458E.07 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
INACOST 0 0 3 6 0 3 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 11100332 0 6 461800 6 6 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 19572 19572 1920 1900 1900 1900 1900 1900 1900 190	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Seore 10222-06 1.0222-06 2.44455-09 2.44455-09 2.44455-09 7.73778-10 7.73778-10 7.73778-10 7.73178-10 7.7316-10 4.07995-10 4.07995-10 4.073115-10 4.075115-10 4.075115-10 4.075115-10 4.075115-10 4.075115-10 4.07	5.0119E-55 5.0119E-55 7.0119E-56 7.01192561 7.011925-08 7.00176-08 7.011725-08 7.011725-08 7.011725-08 7.011725-08 7.011725-08 7.011725-08 7.011725-09 7.011725-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00 7.011755-00	7,6475E.07 20128E.09 2,0128E.09 2,1312E.09 2,1312E.09 2,1342E.10 2,146121E.09 2,04621E.09 2,04721E.09	4,3118E.10 2,0218E.00 4,44218E.00 4,44218E.00 4,44218E.00 4,44218E.00 4,44218E.00 4,44218E.00 0,0001010 1,24428E.00 0,00043105 1,15428E.00 1,0043105 1,15428E.00 1,0043105 1,15428E.00 1,0043105 1,15428E.00 0,0043105 1,15428E.00 0,0043105 1,1558E.00 0,00431000000000000000000000000000000000
Utility 8.477.288.07 8.477.288.07 8.477.288.07 8.477.288.07 8.477.288.07 8.477.68.09 8.437.465.09 8.437.465.09 8.437.455.09 8.447.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.09 8.457.455.00 8.457.455.00 8.457.455.0000000000000000000000000000000	0.0060177623 2.062548-06 0.26635694 8.560938-03 8.560938-03 2.353192-08 0.05609340 0.015609340 0.015609340 0.015609340 0.015609340 0.01560354756 0.01560354756 0.01560354756 0.01560354756 0.0155035556 0.0155035556 0.015505556 0.01550556 0.01550556 0.0155055556 0.0155055556 0.0155055556 0.01550555556 0.0155055555555555555555555555555555555	2.24452-06 8.237452-09 8.237452-09 1.1562822-08 1.1156282-08 1.1156282-08 1.1156282-08 1.1156282-08 1.1156282-08 1.115528-08 1.115528-08 1	1, 201555-0 1, 201557-0 2, 201
Seore Seore 6.655659-06 6.655659-06 6.655659-06 6.655659-06 7.12306382-05 7.1230632-07 7.1230632-07 7.1230632-07 7.1230632-03 7.1461002-03 7.1461002-03 7.14610000000000000000000000000000	0.000136128 3155545-66 0.299647035 1.2997428-07 1.2997428-07 1.2997428-07 1.241428-07 1.241428-07 1.241428-07 1.241428-07 1.241428-07 1.241428-07 2.241528-08 2.241528-09 2.241528-00 2.24	2003.02.000 2003.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.02.000 2003.0000 2003.00000 2003.00000 2003.00000 2003.00	2.82109422_09 2.8210142_012 2.821128_012 2.824128_012 2.824128_02 2.824128_02 2.824128_02 2.824128_02 2.81218_02 2.812128_02 2.81218
DAMCOST 91994 74000 74000 76010 16021 16021 1508 1519 15731 11071 12588 15787 15777 15797 15797 15797 15797 15797 15797 15797 157977 15797 1579	120126 120126 832327 832327 832327 832126 107304 15156 35115 35115 35115 35115 35115 12299 10000 1329060 13290000 13290000 13290960 13290960 13290960 13290960 13290960 13290000 132900000 132900000 132900000 132900000 132900000 132900000 132900000 132900000 132900000 132900000 132900000 132900000 1329000000 1329000000 1329000000 132900000000 132900000000 13290000000000000 1329000000000000000000000000000000000000	88812 2.1859 1.4377 1.4377 1.4377 1.4377 1.4376 1.4376 1.4376 1.4368 2.53685 2.25965 2.25965 2.25956 2.25919 2.6125 2.25919 2.6125 2.25919 2.6125 2.25919 2.6125 2.14928 2.149	8441 8787 8787 8787 8787 8787 8788 8788

Score 0.00070	0.02027	0.00024	0.0003	3.75-08	0.00025	4.8E-05	2.4E-05	6100'0	0.00197	0.00453	0.00011	0.00076	0.07322	0.41277 9.75-05	6.3E-05	0.02083	0.00839	0.03788 0.00029	0.00016	4.85-05 0.40269	0.0001	3.5E-10	4.8E-05	6.3E-05	0.00056	4.4E-05	0.0008	3.55-10	0.00048	57000'0	0.02811 7.8E-05	0.00966	0,1397	0.00268 3.5E-10	9.2E-09 3 9E.05	3.5E-10	4.86-05	0.00053 0.00012	0.00224	0,00026	0,00014	0.00136 2.9E-05	0	0.01497	26100.0	0.00023	0.00843	0.01816 0.0006	0.23592	0 0.54914	0.16074 0.00058	0.2059
Utility Score	0.3269076 0.02026827	0.0039064 0.0002422 0.023907 0.00148224	0.0047658 0.00029548	0 0017199 0.00010067	0.0039845 0.00024704	0.0007813 4.8439E-05	0.0003906 2.422E-05	0.0307041 0.00190366	0.0318104 0.00197224	0.0698997 0.00433378	0.0017188 0.00010657	0.0117191 0.00072659 0.0078178 0.00048439	0,0782817 0.00485347	0 0 0.0015626 9.6878F-05	0.0010157 6.2971E-05	0.3359304 0.02082769	0.1352829 0.00838754	0.0025001 0.00015501 0.0046095 0.00028579	0.0025782 0.00015985	0 0 0	0.0015626 9.6878E-05 0.0071096 0.0004408	0 0	0.0007813 4.8439E-05 0.00174 77403E-05	0.0010157 6.2971E-05	0.0089847 0.00055705	0.0001563 9.6878E-06	0.0128911 0.00079925	0 0	0.0078128 0.00048439	0.0039845 0.00024704	0.1025386 0.0063574 0.00125 7.7503E-05	0.0002344 1.4532E-05 0.0153911 0.0005435	0.3269076 0.02026827	0,0431653 0,00267625 0 0	0 0.000625 3.87515-05		0.0007813 4.84395-05	0.0018751 0.00011625	0.0361664 0.00224232 0.0246102 0.00152583	0.0032032 0.0001986	0.0022657 0.00014047	0.004688 0.00030032 0.0004688 2.9063E-05	•	0.0074221 0.00046017	0.0314903 0.0019524 0 0	0.0037501 0.00023251	FTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTTT	0.2928578 0.01815718 0.0096878 0.00060065	0 0 0.0108597 0.0006733	0 0	0.9999893 0.06199934 0.0093753 0.00058127	0
Score 0.0046875	0.32690012	0.00390625 (0.004765625	0 0011026	0.003984375	0.00078125	0.000390625	0.030703125	0.031809304	0.069897463	0.00171875	0.01171875	0.07827929	0 0.0015625	S79510100'0	0.335922838	0.135278944	0.004609375	0.002578125	0	0.0015625	0	0.00078125	0.001015625	0.008984375	0.00015625	0.012890625	6	0.0078125	91013984375	0.102535512 0.00125	0.000234375	0.32699012	0.043163868	0.000625	Ð	0.00078125	C156C300.0	0.036165213 0.024609375	0.003203125	0.002265625	0.00484375 0.00046875	-	0.007421875	0.031489252 0	0.00375	918561501m	0.292850692 0.0096875	0.010859375	9	0.999989335	•
MHRS	2000	30 S0	3	• ;	15	≘ ș	ŝ	66E	201	800	• X	<u>s</u> 5	870	0 02	4 11	2025	1253	R 8	R 1	2 e	50		9 2	8 12	115 115	10	165 1	7 O	901	5	1020 21	3	2000	7 -	- .	•	8	5 II	315	4	8 A	ų 19 29 4	, ,	56 26	403	48	215	1900	0 66	0	9999 120	•
Score	0	• •		• •	• •	0 0		• •	• •	0	0	• •	• •	0.062	• •	0	. U	• •	0	0.062	0 0		• •		0 0		0		0 0		• •	0 0		0 O	• •	÷	0	• •	0 C		• •	• •	5000	0	0 0	0 0	• •	• •	0.062	0.062	- 0	0.062
Score	0	• •	, o	• •		0 0		•		۰ ،	- 0	0 0	• •			0 -	- 0	e 0	•		• •		• •		• •				0 0			0 0		c c	• •	• •	0	• •	0 0			• •		- 0	c 0	0 0	•	0 0	- 0	-	0 0	-
TLOSS	0	0 0	0	• •						ъ.	- 0	• •		- 0		o -	- 0	0 0	6	÷ -	0 0		• •		•		0		• •			00		• •	• •	. 0	0	0 0	00		• •	0 0		- 0	c 0	0 0		• •	- 0	-	0 0	-
12								~ .			575			1325 J		9410	6761			1325			• •										965																	195		
2 S											0.06			1 0.00						933 0.04							-						667 0.0				-													5 0.0		_
SPL See				¢.							1000			1 0.08		0.000				10.08									Ψ.		•••		0.16				-									÷.				- -		-
۳.											2			325			C76 -		_				_		_	_	_						965			_	_													66		-
955 2		• •		с (• •	0 0		•			333 0.00			333 0.00		0 000	0	00				, .					0				00		667 0.01			, 0	0													0.0		
PL Scol		• •		• •		0		0		•	0.083	• •		0,083		0	0 0	00	0		00		• •		•	• •	0		00		• •		0.166	• •	00	0	o		00						00	0 0		00		-	••	e
* 018										•	-			-			- g																115 2																	5		
e Scor	. 0	• •		0				0		•		0 0	••			0 000	0000 CC	• •	•	• •	• •		-		Ċ		6	**		••	• •		9000 9				¢		• •							•		••		0.0	••	Ð
PL Scon		•		¢ (• •	•	• •	0		0	• •	•		• •	• •	0	ecenin O	* *	•	• •	• •		• •	• •	•	• •	•	• •	•		• •	•	1	• •	• •		0	• •	• •		••		, ,	- 0			• •	• •			• •	0
e FLS											875					•	-			8145													5625 3											t516					765	47		625
Scar	• •	• •	0	• •		0 0		0		0	7 0.00046	0 0	• •	00		00	• •	• •	0	0 3 0.00268	00		0 0		0	00	0		•		0 C	• • •	0.00015	0 0	00		0	00	00			00		3 0.0056	• •	¢ í	- o	0 0	0.0180	0	00	7 0.0015
Score	• •	•	• •	0 0				0		0	0.01339285	• •	• •	0 0		0		• •	0	0 0.07661290	• •	• •	• •		0		0	• •	• •		00	00	0.00446428	• •	• •	• •	0	0 0	0 0		• •	• •		0.16129032	• •	0 0		e, e	0.51647144	0	0 0	0.0446428
YREST		• •		• •		0		0		0	~ •	0 0		0 0		•	• •	• •	0	15					0 (0	••			• •			• •	00		0		0 0			• •		51 0	0 0				, g o	0	• •	9
M m		_		_				_		_		_		_		_			-										_																	_				_		_
See		φ.		•				-												••	••																													0		Ū
Score	• •	• •		•		•		e		0	• •	•	• •	• •			• •	• •	-	• •	• •	• •	•	• •	•		•	• •	• •	• •	• •						8	• •	• •		• •	• •				0		• •		0		8
DAYLOST		• •		•	• •	•	• •	•			• •	• •	• •					• •	¢	• •	• •		•	• •	•	• •	•	• •	•	• •	• •				-		0		•		• •	• •			• •			• •				
Seare		• •		e i	• •	0		•	• •	0	00	0 0		00282692	• •			0 0	•	• •	0 0		•	• •	¢.	0 0					• •	00964744	• •	• •			0	• •	• •			0 0		0 00886218		0		• •	00776282 0	0	• •	00323077
.		¢ 0				. 0		0	0 0	0	• •			307692 0.1				0 0					0		0				0			700855 0.					0	0 0	0		0 0	00	5.	0.004274 0.	0 C	0		00	482906 0. 0	0	0 0	923077 0.
З,	Ĵ					-							5	0.067.	-		-	-		-	-											0.229												0.211					0.18			0.076
WHO:	• •	• •	. .	÷	0 0		• •	0	• •	0	• •	•		61 0	• •	0	0 0	• •	¢	0 0	0 0		•	• •	0	0 0	• •	• •	÷	• •	• •	6 4	• •	• •	• • •	• •	0	0 0	00	• •	• •	• •	-	37	0 0	• •	• •	• •	9 g o	o	• •	; -1

Score						0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Score 0 0				* * * * * * * * * * * *		
PARTIAL DISABILITY 0 0				e e e e e e e e e e e e		
Score 0 0 0						
Sort 0 0	• • • • • • • •					
TOTAL DISABILITY 0 0						
Score 0 0		0,176 0 0 0 0 0 0	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~			0 0 0.187 0.
Score 0 0		o <u>%</u> = = = = = = = =			* * * * * * * * * * * * * * * * * * * *	9 9 0.57125 0.55125 0.55125 0.55125 0.55125 0.55125 0.55125 0.55125 0.55125 0.55125 0.0522 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.05225 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.0525 0.052125 0.05225 0.052125 0.05225 0.052125 0.05225 0.05225 0.05225 0.05225 0.052125 0.05225 0.05225 0.052125 0.05225 0.052125 0.05200000000000000000000000000000000
TOTFTI 0 0						
Score 0 0	0 0 4.771/TE-07 0 0 0	0 0.02375179 0 0 0 0 0 0	0 0 0 1.11545E-06 0 0 0 0	4,4 4,5 6 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	• • • • • • • • • • • • • • • • • • • •	0 5.804E.08 0.00178238 0.00178238 0.03134466 0.03134466 0.031134466 0.032701393 0.022701393 1.6753E.09 1.6753E.09 0.00 0.00 0.00 0.00 0.00 0.00 0.00
Utility 0 0	0 0 1,0152E-05 0 0 0	0.50535723 0 0 0 0 0 0 0	0 0 0 14564E-05 0 0 0	9.4574E-07 0 0 0 0 0 0 0 0 0 0		0 1,12355-06 0.04121009 0 0.017923 0.056690756 0.666690756 0.05665907462 0.05465271233 0.5907462 0.5907462 0.5907462 0.5907462 0.545E-07 0.5907462 0.545E-07 0.5907462 0.545E-07 0.5907462 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.59076 0.5007770 0.5007770 0.5007770 0.5007770 0.5007770 0.5007770 0.5007770 0.5007770 0.5007770 0.5007770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.50076 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.500770 0.50070 0.50070 0.500770 0.500770 0.50000000000
2 000	0 4,35001E-05 0 0 0	0.753672372 0 0 0 0 0 0 0	0 0 0.000105241	4.05336E-06 0 0 0 0 0 0 0 0 0 0 0 0	• • • • • • • • • • • • • • • • • • • •	0 5.329/1718-06 0.1458/57575 0.011868/57575 0.011818/57575 0.0118178/182 0.02819/41076 0.02819/41271 0.02818174127 0.02818174127 0.02818174127 0.02818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174282 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174271 0.00818174771 0.00818174771 0.00818174771 0.00818174771 0.00818174771 0.00818174771 0.0081817477770 0.0081817477777777777777777777777777777777
INJCOST 0 0	, 18300 0 0 0 0 0	0 0 0 0 0 0 0 0 0	32000 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 3000 450000 395000 3910000 2774000 2774000 277532 932 932 932 932 932 932 932 932 932 9
Score 0.00415172 0.00103552 0.16824065	0 6.1999E-09 0.09681059 0.0009108 3.4587E-08 2.6046E-09 2.314E-08 2.314E-08	7.295E-10 0.13873693 3.3607E-09 1.0126E-08 1.089E-08 5.3237E-10 6.5769E-05 4.867E-09	3.422E-10 6.859E-08 3.2726E-05 4.3336E-10 3.253E-09 8.5396E-09 6.09479946 0.09479946 4.0378E-10 4.0378E-10 5.5417E-08	8.2295E-10 1.4258E-09 4.5355E-10 6.4462E-10 0.00119526 9.7391E-10 7.6401E-08 7.7391E-10 7.5401E-08 5.5577E-10 5.5547E-10	1.25465.09 1.27565.06 1.27565.06 1.27565.06 1.49925.07 1.49925.07 1.299285.07 1.212952.05 1.212952.05 1.212952.05 1.212952.05 1.212192.05 1.2121921.05 1.212192.05 1.	4.11E-09 0 0 8.4523E-05 8.4523E-05 8.4523E-05 0.0761091 0.0761091 0.0761091 0.0761091 0.076109 0.0175692 0.027503 0.027503 0.01766851 0.00766851 0.01766851 0.000766851 0.00076851 0.000056552 0.000056555 0.000055555 0.00005555555555
Utility 0.000642881 0.004387802 0.712884093	0 2.62709E-08 0.410214356 0.0013859331 1.46556E-07 1.10363E-08 9.80508E-08 2.78895E-08	3.0911E-09 0.587868351 1.42404E-08 4.29052E-08 4.61426E-08 2.2558E-09 0.000278683 2.06228E-09	1,17525,0 2,906555,0 0,000138668 1,836276,09 1,836276,09 1,33486,09 0,401692634 1,00045896 0,100097896 0,1702456,09	1.031020 6.01115E-09 9.00615E-09 1.92181E-09 1.92181E-09 1.92181E-09 1.92181E-09 1.2555E-09 3.123735E-07 1.25521E-09 1.255521E-09 1.255521E-09 1.255521E-09 1.2555520 1.255521E-09 1.2555220 1.2555220 1.2555220 1.2555220 1.2555220 1.2555220 1.2555520 1.2555520 1.2555520 1.2555520 1.2555520 1.2555520 1.25555520 1.25555520 1.25555520 1.255555520 1.2555555555555555555555555555555555555	6.58772.09 6.9802.65.09 2.9802.65.09 2.9802.65.09 2.9802.65.09 2.9802.65.09 2.9802.65.09 8.1006.71-9 6.109.112.99.112.99 2.11.112.64 2.021.82.09 2.9519.640 2.03519.6400 2.03519.6400 2.03519.6400 2.03519.6400 2.03519.6400 2.03519.6400 2.03519.6400 2.03519.640	1,741546.08 10,000281.49 10,000281.49 10,000281.49 10,000281.45 10,000281.45 10,000281.45 10,000281.45 10,000381.45 10,
Score 0.000987649 0.006725355 0.785315243	4,03757E-08 0.509812066 0.005917274 2.25242E-07 1.69617E-08 1.50694E-07 4,28653E-08	4.75072E-09 0.678825013 2.1886E-08 6.5941E-08 7.09165E-08 3.46594E-09 0.000428234 3.16952E-08	2.26598E-09 4.46676E-07 0.000213101 2.82217E-09 2.18425E-08 2.561249E-09 0.5011986561 2.66499E-09 2.66499E-09	5.554(FL.0) 9.28494E-08 1.38415E-09 4.19796E-09 4.19796E-09 5.32587E-09 1.97547E-07 2.4344E-09 2.4344E-09 2.4344E-09 2.4344E-09	1.01246.08 8.2106778.09 7.400778.09 4.947728.50 4.947728.50 1.255546.97 0.002559559 2.738995.09 2.738955.09 2.713955.08 1.2255468.54 1.2255468.54 1.2255468.54 1.2255468.54 1.2255468.54 1.2255468.54 1.2255468.54 1.2255468.54 1.2355468.54 1.235468.54 1.2355468.54 1.2	2.67657E.08 0.0006580377 0.0006580377 0.117023455 0.1317547346718 0.1317345745 0.1317345438667 0.131734543867 0.131734543 0.131734548 0.131734548 0.31712-08 0.017173-08 0.017173-08 0.017173-08
DAMCOST 146869 167924 15539000	36367 1500040 166519 24901 27386 2573 25730 37000	15528 10112506 2978 41595 41595 13016 137697 33819	10000 62379 130037 11498 29644 64774 1055868 11493 151448 11493 151448	22251 22252 22255 22555 2255 22555 22555 2255 25555 25555 25555 25555 25555 25555 25555 25555 25555 25555 25555 25555 25555 25555 25555 25555 2	22317 44078 1494 14148 14148 14148 14153 14155 112555 112555 112555 15255 15255 15255 196895 2649 27304 196895 2649 27649 27649 196895 11655 2649 27649 196895 264	32056 144450 144450 14450 14251 1510756 1510756 14203 1510756 14203 1510110 12101010 1210112 15235 121215 1514 1514
VENT1 35 35 35	98 98 98 98 98 98 98 98	******	**********	*******	***************************************	* ****

Score 0.01902 0.01902 0.26924	0	0.18988	16000'0	7.85-05	0.0002	7.3E-10	0.49256	1.9E-05	0.00721	0.00089	0.01091	3.5E-10	0.0001	5.7E-05 6 8E-05	1.9E-05	0.00042	3.4E-05	0.00072	0.00048	2.9E-05	0.00534	0.00023	4.8E-06	0.00128	0.00024	5.6E-10	0.00017	0.00064	5.3E-05	0.00078	5.3E-05	4.9E-05	0.0009	3.4E-05	0.00029	3.4E-05	4.9E-06	6.3E-05	4.9E-06	0.01505	1.5E-05 4 8F-06	0.00016	0.00049	4.85-06	1.95-05	0	0.27052	0.177110	0.45249	0.57665	0.312	0.16601	0.44902 0.50545	0 0.18225	0	3.9E-09 1.5E-09	0.00026 3.9E-09														
Utility Score 0.0422615 0.00262021 0.0031251 0.00019376 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0	0 0 0	0.00125 7.7503E-05	0.0032032 0.0001986	0 0	0	0.0003125 1.9376E-05 0.0003125 1.9376E-05	0.1162905 0.00721001	0.0142974 0.00088644	0.0164068 0.00101722	0 0	0.0016407 0.00010172	0.00039906 2.422E-05 0.0010938 6.7815E-05	0.0003125 1.9376E-05	0.0067971 0.00042142	0.0005469 3.3907E-05	0.007969 0.00049408	0.0078128 0.000184439	0.0004688 2.9063E-05	0.003047 0.00018891	0.0003125 1.9376E-05	7.813E-05 4.8439E-06	0.0013282 8.2347E-05	0.0001563 9.68785-06	0 0	0.0027345 0.00016954	0.010391 0.00064424	0.0008594 5.3283E-05	0.0125004 0.00077503	0.0008594 5.3283E-05	0.0007813 4.8439E-05	0.0081253 0.00050377	0.0005469 3.3907E-05	0.0046025 7.7503F.05	0.0005469 3.3907E-05	7,813E-05 4,8439E-06	0.0010157 6.2971E-05 0.0019532 0.0001211	7.813E-05 4.8439E-06	0.0003125 1.9376E-05	7.813F-05 4.8439F-06	0.0025001 0.00015501	0.0078128 0.00048439	0.9889627 0.06131568	0.0003125 1.9376E-05	 		• •	0	• • •			• •	e e e e	0		• •														
Score 0.042260075 0.003125 0	0.00265625	Ð	8	0.00125	0.003203125	0	0	0.0003125	0.116287012	0.014296875	0.01640625	0	0.001640625	0.00109375	0.0003125	0.006796875	0.000546875	0.00796875	0.0078125	0.00046875	0.003046875	0.0003125	0.000078125	0.001328125	0.00015625	0	0.002734375	0.010390625	0.000859375	0.0125	0.000859375	0.00078125	0.008125	0.000546875	0.00125	0.000546875	0.000078125	0.001015625	0.000078125	0.0003125	0.000078125	0.0025	0.0078125	0.98896228	0.0003125	•	•		•	0	• •			Ð		• •	• •														
530 530 0	PL	, o	0 0	1 91	4,	n e	0	4 4	1140	183	210	0	21	~ =	ī 4	87		102	001	¢ ک	39	\$ 4		11	8 -	• •	35	EE :	\$ =	160	=	₽ '	104	-	\$ ¥	-	-	n ;	- i	4	m -	. 55	100	1 1	4	c	•	ф с		0	•	• •	• •	•		00	0 0														
Score 0 0.062	c	0.062	•		• •		0.062		• •	ð (0 0	• •	•	ə c	• •	0	700.0	0	•	0	•		0	•			0	• •	• •		0	•		0			0	• •		0			• •		0	0.067	0.062	0.062	0.062	0.062	0,062	0.062	0.062	0.062		• •	• •														
Score 0	c		• •	• •	• •		-	0 0	0 0	0	• •	• •	0 0		• •	÷.	- 0	0	• •		•		•	•			0	0		• •	0	•		0			0			0	• •		0 0			-				-				-																	
TOTLOSS 0 1	e		0		0	• •	-	• •		•			•		• •	0	- 0	0	• •			• •					0	0		• •	¢	0 0		0	• •	. 0	0			0			• •		• •	-				-				-			0 0														
Score 0.00325 0	-				• •	• •	0.00975	• •	• •	0	• •		•	• •		•	c: 400.0	8	• •		•	• •		•	• •	• •	0	•			8	•		0	• •		0	• •		0			•	• •	•	-	•	0	0.00975	0.0195	0.039	C00070	0.0065			• •															
Score 0.083333 0	e		•		•	• •	0.25	• •	• •	-			-		• •	•	c7 0	0	• •	• •	-		•	•	• •	• •	0	¢	• •	. 0	0	•		0	• •	• •	0			0			• •			-	-	0	0.25	0.5	-	0.166667	0.166667 0.083333	•		• •	• •														
I I							•										n																															-	• ••	7	1 0 1	7	- 1																		
Scote 0.00325 0 0	e		•		•		0.0195	0 0	• •	0	0 C	• •	•	• •		0	c/ 60070	0	0 0	• •	•	• •	• •	• •			¢	•	• •		0	•		0	• •		0	• •		0	• •	0	0 0		0	c	• •	0.065	0.00975	0.039	0.00975	0.00325	0.00975	0		00	0 0														
Score 0.083333 0 0	-	• •	•	• •	•		0.5	• •	• •	•	• •		•	• •	• •	•	9	0	• •		0			•	• •	• •	0	¢ '	• •	• •	0	0		0	• •	• •	0	• •	• •	0	• •	• •	¢ (c	• •	0 166667	0.25	-	0.25	0.083333	0.25	0			0 0														
] olspi							4										'n																															ſ	• •	ŝ	с,	m	6 10																		
Score 0.00975 0.039	-		•		•		0,039	• •	• •	¢	¢ ¢	• •	•		• •	•	0.00125	•		• •	•	• •		÷ 1			÷	•	• •	• •	•	•		e		• •	•	• •	• •	¢	• •		•	• •	• •	-	0.0195	0.039 a me7£	0.039	0.039	0.00325	0.039	0.039	-		• •															
Score 0.25 1	đ		•		0		-	• •	• •	•	• •	• •	•	•	.	e	0.0000	0			•		• •	•			•	•			•	6		•	• •	• •	•	• •	• •	•	• •		•			c	0.5	1] -	-	0.083333			e		• •	00														
FLSPL 3 5							¥î										-				-																										4	40 P	n va	va	1	чо чо 	un un																		
Score 0 0	-	0.0120665	0		0	• •	0	00		0	• •	, o	•	• •		0	060010.0	0	• •		0.0051512		• •	•		• •	0		• •	• •	0	0	• •	Q	• •	• •	e	• •		•	• •	• •	•		• •	6731000	0	00		0	0	0.010090	e e	e			00														
Score 0 0	c	0.344758065	0		0	• •	0	• •	• •	0	• •	00	0	• •	• •	0	0.288306452 n	•	0 0	0 0	0.147177419	• •		•	• •		0	0			0	•	• •	0			a	0 0		0	•		0	0 0	00	F39C636600	0	• •		0	0	0.288306452	0 0	c		0 0	0 0														
AYREST 0 0	4	- ¥	0		0	5 0	•			0	• •		¢,	a 6	• •	0	R -	. 0	• •		2	• •		•	• •		0	0	0 0	, o	0	0	• •	0			0	• •		0	c c		0	• •	• •	5	0			0	0	0 <u>8</u>	0 0	c		00	0 0														
0 0 0 0 0 0 0 0 0		0 375313	0				747077			0	• •		•			0	328545	. 0							• •							•														10.000	0			0	0	0 677343		-			• •														
2		8229 0.01					9836 0.01										10369 0.01																													10.0						56993 0.01																			
T Scal	d	0 0.76406	•	••		••	0.9705			0	•		6	-		•	URE2'0	. 0	•		•				•		5	•			-	•	• •	•	-	• •	÷	•	• •	¢	•		•	• •		00000	0	•		•	0	0.9318	* *	c			••														
DAYLOS 0 0 0	đ	- 2	•				159	• •	• •	e	• •		0	• •	• •	¢	8 -	• •	•	• •	•	• •	• •	•	•	• •	•	•	• •	• •	¢	¢	• •	•		• •	•	• •	• •	•		• •	•	• •		5	2 -	• •	• •	•	0	0 12 0		e		• •	-														
0 0 0	c	0.00525	0		0	• •	0.00634936	0 0		0	•		0			0	0.00444231	. 0			0	•	• •	0	• •		•	0	• •		. 0	c	• •	• •	0 0	• •	0	•		•	• •		0				0	0 0	• •	0	0	0.00572115	0 0	209080000		00	00														
Score 0 0	¢	0.125	0		6		151175214	•	• •	0			0	0 (0	0.105769231	• •	0		0			0	• •		• •	0	• •			0	• •		•			•		. 0	•	• •	0			A CONTRACTOR	0	0 0	, 0	o	0	0).136217949	0 0	CONTRACTOR		00	ð 0														
0 0 0 0		0 1	0		0	0 0	21 0	0		. 0	•		0	•		0	_ _		0		0	•		0				0	• •			0	0 0	. 0			. 0	0		• •	0		0	• •			• 0	•		0	0	0 1	0 C		•		• •														
Score	•	•		•	0	0	0	0	•	-						ç					•	•	-	-	• •	•	÷	•		• •	0	•	•	0.0035	• •	• •		e	•	•	•			0	•	• •	• •		0	0	• •		0		Ð	e (•	• •	0	0	• •	• •	0	0	e		0	0	• •	, .	
----------------	---------------	-------------	-------------	---------------	---------------	-----------------------------	-------------	-----	-------------	-------------	-------------	-------------	-------------	-------------	-------------	-------------	----------------	---------------	-------------	-------------	-------------	-------------	-------------	-------------	-------------	-------------	-------------	-------------	-------	---------------------------	-------------	-------------	-------------	-------------	--------------	----------------	-------------	------------------	--------------	-------------	-------------	-------------	---------------	-------------	--------	-------------	--------------	------------------	---	-------------	-------------	---------------------------	------------	----------------	------------------	-------------	--------------	-------------	---------------	-------------	-------------	------------------	-------------	-------------	-----	-------------	-------------	-------------	-------------	-----------------	---
Score	8			• •		0	•	-	• •												. 0	0	0				0	•			•	0	•	0.25				0	0	0	•			e	÷.	• •			•	•	• •		•	• •		¢ (0	0	• •		0	0	-			0		, 0	
λĽ,																																																																							
ARTIAL DISABII	•	•			•	÷	0	•	•	•	• •			•	•	•					•	e	0				0	•			Ð	Ð	e		• •	•		•	0	0	0			0	¢	• •	• •		0	0		• •	0	• •		•	•	• •	8	۰	• •	• •	•	•	e	• •	•	•	•	, .	
Score P	•	0 0		. 0	÷	ð	Ð	0	0	Ð I	P 0		0,0105	0000							. 0	0	o	• •			0				0	0	•	• •				0	0	0	0			0	Ð	• •			0	0	• •		• •			0	0 0		0	0	• •	• •	0	0	c		• •	0		, o	
Score	0	•	• •		. 0	0	0	0	0	5 1				5	•							0	0			, .	0			. 0	0	0	0	0 (0	0	0	0 0		0	0	0 0		• •	0	0			. 0		, .	0	0		• •	0	• •	• •	0	o	c			0		, .	
BILITY																																																																							
TOTAL DISA	。 1	Ċ			. 0	•	0	Ģ	0	0	•											0	•	0	2 0		0				0	0	0	•	•				0	0	0	• •		0	0	• •				0	•					0	0		Ģ	0	0 0		0	0	c		0	0	0	, 0	
Score	•	•	• •			0.176	0	o	0.176	•	•			0.1.0			-		-			0.176	•	• •	-	• •	0	•	9/170	• •	•	0	0	• •	• •	-	• •	• •	•	¢	0	0.176		0	÷			• •	•	0	• •	• •		• •	• •	•	¢ 1	• •	• •	÷	• •	• •	•	0	4		• •	•	-		
OTFTI. Scare	0	0	• •	•		1 0.5	0 0	0 0	1 0.5	0				-			•					1 0.5	0 0	•	•		8	0 ·	1		0 0	0 0	0 0	0	•	•			•	•	0 8	1 0.5		0	0 0	0 °				0 0				• •		0 0	0		•	e 0	•		0	0 0	4	• •		0 0	•	, o	
Score T	•	0	0 0			6.7114E-05	1.9656E-08	0	0.02346933	1.5339E-07	• •		0,000,000,0	61676700'0			5 0		a (0.00164835	0	0	10 20011 2	3.3029E-07	7.4863E-08	0	÷ (0	2.7442E-07	0	4,4622E-05	1.474E-07	0 0		2.2775E-07	0	• •	1.646E-07	0.00046978	0	. 0	0	0	0-360867	0		5.379E-07	1.329E-07	1.4464E-06	, o	0	0	0		- c	. 0	0	3.5785E-08	U 1.3432E-08	2.4565E-07	0	0 0		0	o	0 0	v 3.0661E-08	0
Tindley	0	0				0.00142795	4.1822E-07	0	0.49934744	3.2637E-06	•	.	00000000	62429100/0	• •	-	⇒ •	- c				0.03507138	0	•	0	7.0275E-06	1.5928E-06	õ	•		¢	5.8388E-06	0	0.00094941	3.1362E-06			4.8456E-06	•	•	3.502E-06	0.00999534	4.02126-02	0	0	0	0.0015704	0		1.1445E-05	2.8276E-06	3.07/4E-US		0	0	0	•		• •	0	7.6138E-07	0 2 8579E-07	5.2265E-06	0	0 0	• •		0		u 6.5236E-07	0
Score	¢	ð	• •		, o	0.006073457	1.79202E-06	0	0.749550019	1.39844E-05	0 (.	0	8866401/1.0		• •						0.128043845	0	0	0	3.01108E-05	6.82507E-06	0	•		0	2.50179E-05	0	0.004048081	1.34379E-05			2.07625E-05	0	•	1.50056E-05	0.040737775	0077/1000/0	. 0	0	0	0.0006674408	0-302/66-40		4.90366E-05	1.21157E-05	0.000131841		0	0	0	0		• •	0	3.26244E-06	U 1 22458E-06	2.23945E-05	0	¢			0	0	u 2.7953E-06	
INICOST	o	0	• •		• •	125000	1090	0	1100000	7456	0		0	0000279								270000	o	•	0	74907	3918	0	•		0	12080	0	115000	7203	• •		10397	0	• •	7922	172156	1491		0	0	065/21	667		96861	6580	36240	• •	0	79077	0	o			0	1950	- 52	11056	0	•	5 0	. 0	0	0	u 1680	
Scote	3.7556E-10	13374E-06	L.7512E-07	10-370771	0.1998.09	•	0	0	0	•	8.1091E-09	50-302tc./	5710/1001	U 000000	100001020	60-377091	33176-07	200020115	C17/62607	10-36704-06	00-3/2///	•	0.0268351	L71-49E-08	L-4067E-U9	• •	-	S.5225E-10	•	5242F-08	L2451E-09	2.2757E-11	2.9729E-09	•	0	E.0693E-07	0.02432646	0.000	0.02492929	8.6109E-09	0		0 011063.01	3.9358E-06	0	4.5847E-05	1.27296-10	0 3.2887E-09	0	0	0	2.788/5-11	4.7266E-08	6.9836E-08	2.3456E-07	0,01025898	0.03752731	3.4878-10	0.02063211	0.00119975	0	3.482E-1U	• •	3.482E-10	0	0.03080259	5.199E-07	0.00012954	1.8115E-09	0.1465742	0
Tettes	3.71002E-09 8	5.66678E-06	1.16576E-06	- 20-30/06-09	5 68617E-00-2			0	0	•	3.43606E-08	3.19853E-07	0.007208706		0.000855216	7.00078E-09	5.64273E-07	1 191601100.0	0.419373503	2.28937E-06	00-3/04757/	0	0.113708071	7.26669E-08	5.96076E-09			2.34003E-09	•	3.7345E-09	5.27597E-09	9.64274E-11	1.2597E-08	•	•	4.53085E-07	0.103078202	0.05060.0	0 10563358 1	3.64868F-08	Ð		0 011131000 0	1.66769E-05	Ð	0.000194266	5.39382E-10	0	0	•	•	1.18164E-10	2.0028E-07	2.95916E-07	0 9.93895E-07	0.043470274	0.15901404	5.82961E-07	0.087424175 (0.005083677	0	1.47542E-09	• •	1.47542E-09	•	0.130519451	2.20298E-06	0.0005489	7.67592E-09	0.109492166	•
een s	5.70193E-09	8.70923E-06	1.79166E-06	7.98567E-09	00-366811.2	0 Division of the second	. 0	0	8	•	5.28088E-08	4.91581E-07	0.011029897	0	0.001313686	1.07595E-08	8.6723E-07	0.00170396	0.519111961	3.51852E-06	1.1503/E-U5	0	0.16349958	1.11682E-07	9.16108E-09	• •		3.59639E-09	•	5.73956E-09 9 9757E 09	8.10863F-09	1.48199E-10	1.93603E-08	•	•	6.96346E-07	0.14909572	0 01-3/0/70/1	1800250510	80765F-08	•	0	8	2.56305E-05	•	0.000298531	8.28975E-10	0 2.14173F-08	0		•	1.81606E-10 2 0704E 00	3.0781E-07	4.54793E-07	0 1.52752E-06	0.065073282	0.22308258	8.95952E-07	0.127574865	0.007788608	0	2.26758E-09		2.26758E-09		0.18594377	3.38575E-06	0.000843318	1.17971E-08	0.70801579	
i naaroost	17077	94945	17597	16007	15401	•	. 0	0	0	÷	39219	12153	173354	0	ISINN	00677	69643	152855	1065761	82000	00002	0	426857	17276	21370		• •	13297	•	1/134	20232	1018	28725	Ð	•	62739	358129	10177	376000	39860	•	•	1 1009 1	106791		133737	4765	0 29756	0	0	0	1235	58309	62576	75848	192835	££6619	70000	120755	169535	0	10000		1000		512578	84578	145135	23790	0000011	
LINSAL	Ę	ę	ę :	ş 9	9 9	4	9	9	ŧ	Ŗ	ę i	q :	ŧ.	ş :	ş :	;	ç :	ş :	; :	\$ S	ş ş	9	Ŧ	Ŧ	ş :	;	4	9	Ę I	2 9	9	9	ŧ	ŧ	1	9 :	ş :	1	1	9	9	9 9	\$ \$	9	ŧ	ę :	q 9	7	4	Ŧ	Ş :	q q	¥ #	9 1	q q	ŧ	q	q 9	7	ŧ	ş :	7 9	9	9	:	# 7	; 7	\$	4	7 7	

885-10 885-10 885-10 1.35-66 1.35-66 1.255-69 0.00012 2.55-69 0.01767 0.01767 0.01767 0.017577 0.017577 0.017577 0.017577 0.0175777 0.017577777 0.01757777777777777777777777777777777777	0,1889 0,0002 1,35-07 1,35-07 1,35-07 0,00026 0,009877 5,45-05 1,75-08 0,17765 0,02565 1,775-08 1,775-08 0,02565 0,0256 0,00026 0,00000 0,0000000000	0.01349 0.00867 5.55-10 0.176 0.176 1.55-08 1.55-08 1.25-09 1.255-09 0.02674 0.02974	0.118-07 1.68-09 1.68-09 0.02453 0.02751 0.02771 0.02751 0.02751 0.02705 0.022065 0.022065 0.022065 0.022065	0.02457 0.02457 0.02457 0.03565 0.03656 0.030565 0.03414 4.7E-08 7.E-08 0.02189 0.02185 0.00185 0.02185 0.0018	0.02162 0.0012 0.0012 0.0015 3.55-10 1.45-07 0.0156 3.55-10 0.01676 3.55-10 0.02245 0.00023 5.8E-05 5.8E-05 0.0022 5.8E-05 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0022 0.0012 0.00000000
Hilly Score 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		• • • • • • • • • • • • • • • • • • •		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0
Sears U 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2				0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 00015525 0.0 0 0001552 0.0 0 0001550 0.0000000000000000000000000000
AMHKS				• • • • • • • • • • • • • • • • • • •	• • • • • • • • • • • • • • • • • • •
S S S S S S S S S S S S S S S S S S S					
TOTLOSS					
ore Skore					
se TatsAH			_		
λ. 					
naru L					
Score Score 9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0					
s s	34	51 57	56 16 16 173 16	22 21	23 ²³ 23
Score Score 28571 0.000312 28571 0.000312 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		90323 0.005645 90323 0.005645 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	11125 0.001093 11125 0.001093 11137 0.02003(0 11137 0.02003(0 11137 0.02003(0 11137 0.02003(0 11137 0.02003(0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	42857 0.001566 42857 0.001566 006452 0.010990 00 00 00 00 00 00 00 00 00 00 00 00
XR55T Sec		21 21 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 2280 0.5885 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	30 0 0 288. 0 0 0 288. 1 1 1 5 0 0 2 338. 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Score 02 0 0 0 0 0 0 0 0 0 0 0 0 0	EX3550000	0.0100554 0 0 0 0 0.00918553 0.0120683 0.0120683	0 0 0.00871106 0 0 0 0 0.0013503 0.001751 0.001731258 0 0.01731258 0 0.01731258 0 0.01731258	0 0.01084358 0.01084358 0.00021641 0.01089903 0 0 0 0 0 0 0 0 0 0 0	3.9136.07 3.9136.07 1.24815-07 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
C Score	6520620 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.558752243 0 0 0 0 0 0 0.510307188 0.670461295 0.02567769	0 0 10 11 11 11 11 11 11 11 11 11 11 11	0.0120 0.649.24087 0.649.24087 0.012022871 0 0 0.012022871 0 0 0 0	2.173918-015 2.173918-015 6.93396E-016 0 0 0 0 0 0 0 0 0 0 0 0
T D D VI D O VI	4515 28 28 28 28	3269 31 2885 0 0 0 1 269 0 26 26 26 26 26 28 285 11	2692 23 2692 23 115 0 5641 15 5641 13 3462 147 3269 0 1	1462 146 3269 10 7705 36 7705 30 7705 30 0 0 2692 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	264 200000000000000000000000000000000000
Seave Seave Seave Seave Seave 2000 0000 0000 0000 0000 0000 0000 00		SI 730769 0.0034 72115385 0.0030 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1730769 0.0040 6153846 0.00402 0 0 0 0 117307582 0.00302	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 86538462 0.008 86500.0 358462 0.008 8600.0 3170076 0	000 000 000 000 000 000 000 000
DAVIIOSP 0 0 0 0 14 1 14 1 14 1 14 113 113 113 113 113 1		ч м о о о о о о о о о о о о о о о о о о	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	00% v 2000200000000000000000000000000000	но области области области области области

Score 0			0.01 2000 2000 2000 2000 2000 2000 2000
Score 0			
PARTIAL DISABILITY 0 8			
Score 1			••••••
Score 0		•• •••••	
TOTAL DISABILITY 0 0			
Score 0			6 7 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8
Score 0			
10TFTL			• • • • • • • • • • • • • • • • • • •
Score 0 0 \.3407E-08		0 0 0 16 16 16 16 16 16 16 10 10 10 10 10 10 10 10 10 10 10 10 10	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Utility 0 1.9874E-06 9		0 0 1.481E-06 6 3.4917E-06 1 3.481E-06 1 0 0 0 0 0 0 0 0 0 0 0	4.314E-05 2 4.314E-05 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Score 0 8.51563E-06		0 0 6.34925E-06 6.34925E-06 1.49614E-05 1.4346E-05 0 0 0 0 0 0	0 0.000033534 0 0 0 0.006438951 0 0.006438951 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
INJCOST 0 4800		0 3664 7902 0 0 0 0 0 0 0	4000 41000 0 0 0 0 0 24201 24201 0 0 24201 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Score 1354E-10 09643009	0 20065.00 20165.00 2	19465-19 41425-07 41425-07 41425-07 4168771 11168771 11168771 11168771 11168771 11168771 11168771 11168771 11168771 117652-19 175525-19 183555-19 183555-19 183555-19 183555-19 183555-19 183555-19 183555-19 183555-19 18355555-19 1835555-19 1835555	0 010195-08 010195-08 010195-08 010195-08 010195-08 010195-08 010195-08 010195-08 010195-09 010155-09 010155-09 011355-09 011355-09 011355-01 000000000000000000000000
Utility 17601E-09 5 408602063 0.	143665.00 1 141655.01 1 141152.01 2 141152.0	0.4855E_09 7 9.92532E_07 1 0.1321491979 0 1713254312 0 1005981445 0 1005981445 0 1005981445 0 100598145 0 100598249 3 016498249 9 1.14487E_09 1 1.14487E_09 1 1.14487E_09 1 1.14487E_09 1 1.14487E_09 2 2.233186E_07 5 2.233186E_07 5 2.233186_05 2.23318_05 2.23318_05 2.233186_05 2.23318_0505050505050505050505050505	1111112-01 11111111111111111111111111111111111
Score 3.34431E-09 2. 0.50816776 0.	6.8019416.09 4 6.8019416.09 5 9.311518.07 5 9.311518.07 5 9.311518.07 5 9.311518.07 5 9.311518.07 5 9.311718.09 5 5.311718.09 5 9.311718.09 5 9.311748.06 5 9.3197448.06 5 9.3197448.06 6 9.3197448.06 6 9.3197448.06 6 9.317428.06 6 9.317428.06 6 1.320558.08 6 9.317428.06 6 1.320558.08 6	4.685318.40 2.09898.47 2.09898.47 2.09898.47 2.00898.47 2.003088.49 0.451715.40 0.451715.40 0.451715.40 0.451715.40 0.451715.40 1.35376.40 2.37359.40 2.37359.40 2.37359.40 2.37359.40 2.35376.40	1.1982(16.06) 1.1881(170) 2.1982(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.06) 1.941(170) 2.941(16.07) 1.941(170) 2.941(16.07) 1.941(170) 2.941(16.07) 1.941(170) 2.941(16.07) 1.941(170) 2.941(16.07) 1.941(170) 2.941(16.07) 1.941(170) 2.941(16.07) 1.941(170) 2.941(16.07) 1.941(170) 3.941(16.07) 1.941(170) 3.942(170) 1.941(170) 3.942(170) 1.941(170) 3.942(170) 1.941(170) 3.942(170) 1.941(170) 3.942(170) 1.944(170) 3.942(170) 1.944(170) 3.944(170) 1.
DAMCOST 12743 1416218	193101 1920000000000	11411 114111 114111 1141111 114111 114111 114111 114111 114111111	10000 1000000
EVENTI 46 46		77 # # # # # # # # # # # # # # # # # #	S S S S S S S S S S S S S S S S S S S

Score	9./E-U6 0.12056 0	0.00023	0.00107	0.00058	0.00015	0.00035	3.9E-05	5.8E-05 2E-07	0.003	9.2E-05	0.00087	0.00018	0.00012	0,00048	0,00097	0,00017	0.00027	0.00016	4.2E-10	0.00038	0.03548	3.5E-10	3.5E-10	2.4E-09	3.2E-05	0.00152	0.00121	0 0 14737	0.17369	0.13242	0.02693	0.16448	9.7E-05	0.00085	9.7E-06 9.TE-06	5.5E-08	1.25-09	3.9E-05 0.00087	0	0.26264	0.00208	3.4E-05	0,00433	0.08604	1.8E-09	190100	0.00012	0.062	0.0933	0.062	2.9E-05	0.13436 0.00116	7.3E-05	0.00051	0.00746	0.14109	77000.0	0.00136 0.062	3.9E-05	0.148 4.2E-09	0.00059
Utility Score	0.041992 0.0026035 0.041992 0.0026035	0.0037501 0.00023251	0.0171881 0.00106566 0.0043751 0.00027126	0.0093753 0.00058127	0.0023438 0.00014532	0.0056252 0.00034876 0.0025001 0.00015501	0,000625 3.8751E-05	0.0009375 5.8127E-05	0.0047658 0.00029548	0.0014844 9.2034E-05	0.014063 0.0008719	0,0029689 0,0001840/0	0.0018751 0.00011625	0.0078128 0.00048439	0.0156255 0.00096878	0.0027345 0.00016954	0.0043751 0.00027126	0.0025001 0.00015501	0	0.006094 0.00037785	0.0237508 0.00147255	0 0	0	0 0	0.0004688 2.9063E-05	0.0245321 0.00152099	0.0195319 0.00121098			0.1137602 0.00705313	0.0195319 0.00121098	• •	0.0015626 9.68788-05	0.0136723 0.00084768	0.0001563 9.6878E-06 0.00115636 9.6878E-06	0	0 0	0.014063 0.0008719	0 0	0.000625 3.8751E-05	0.0335105 0.00207765	0.0005469 3.3907E-05	0.0698997 0.00433378 0.0210163 0.00130301	0 0	0 0	0.0011719 7.2659E-05	0.0019532 0.0001211		0.0015626 9.6878E-05 0.0015424 0.00053344	0 0	0.0004688 2.9063E-05	0.0186725 0.00115769	0.0011719 7.2659E-05	0.0082034 0.00050861	0.1203209 0.00745989	0.0014844 9.2034E-05	0.0123442 0.00076534	0.0218757 0.0013563 0 0	0.000625 3.8751E-05	0.0016407 0.00010172 0 0	0.0095316 0.00059096
Score	520510000	0.00375	0.0171875	0.009375	0.00234375	0.005625	0.000625	0.0009375	0.004765625	0.001484375	0.0140625	0.00296875	0.001875	0.0078125	0.015625	0.002734375	0.004375	0.0025	0	0.00000575	0.02375	•	-	0 000	0.00046875	0.02453125	0.01953125	-		0.113756774	0.01953125		0.0015625	0.013671875	0.00015625	0	0	0.0140625	and courses	0.000625	0.033509398	0.000546875	0.069897463	÷	0 0 000000	0.001171875	0.001953125	. 0	0.0015625	0	0.00046875	0.018671875	0.001171875	0.008203125	0.120317247	0.001484375	0.01234375	0.021875 0	0,000625	0.001640625	0.00953125
MHRS	2 527	48	220	R 021	30	r 1	¦∞	ច	• ;;	6	180	88	9 P	0	200	8	- %	32	¢	82.62	304	0	0	• ;	2 °	314	250	-		1124	250	0	50	175	н 5	3 0	0	8 081	•	~ ~	。 428	-	800	•	•	186	Я ¢		20	0	9	180	15	105	1165	6]	158	280 0	, ec	21	, 122
Score		0	• •	, o	0	• •	0	• •	• •	0	0		• •	Ģ	o	•	• •	0	•			0	0	• •		o	0	0.067	0.062	0	0	0.062	Û	0		• •	e	• •	¢			0	0 C	0.062	-		0 00	0.062	0 0	0.062	0		0	• •	, ¢	• •	• •	0.062	0	0 0	, o
Score		0	• •	• •	0	• •	•	• •	• •	0	0 0		• •	0	0	•	• •	0	Ð			0	0	• •		0	0	-		0	o	-	0	0	• •	. 0	0	e o	e			•	• •	-	•	• •	e -		•	, _	0 0		0	•	• •	• •	• •	0 -	0	0 0	, o
TOTLOSS		0	0 0		0		0	• •	- c	0	•	• •		0	0			o	ð			0	0	•	• •	0	0	-		. 0	0	-	0	0	00		0	0 0		• •	•	¢	• •	·	0		0.		•	, .	0		G	9	0	0		0 -	. 0	0 0) O
Score	0,0065	8		• •	0	• •		• •		0		•	• •	•	•	•	• •	•	•		• •	•	÷			•	•	e		0.00325	•	0.00325	0	0	• •	• •	•	• •				•	• •		0	0			0.00325	, .	•	• •	0	• •		0.00325		• •	•	0.00975 0	, o
Score	u 0.166667	0	• •	• •	•	• •			• •	•	•	• •		6	8	•	• •	•	•			0	•	•		0	e	•		0,083333	Ð	0.083333	Û	0	• •	• •	•	• •		• •		•	•	• •	•	9	• •	• •	0.083333	, o	0	• •	0	• •	• •	0.083333	• •	00	. 0	0.25 0	, œ
IdSAH	7																												,	1		•													,	•			5 1			•				-				ñ	
re Scoli		0	• •		0	o c	. 0	• •		0	0	•	• •	e	0	•	• •	•	•	0 0	• •	0	0	•		0	c	2000 666	0	333 0.0033	0	667 0.006	0	0	00	• •	0					ð		• •	0	6000 c	0 0	• •	667 0.006	, 0	0	667 0.006 0	0	0 0		0 0		00	0		. 0
LSPL Scal		0	• •		0	00		•		0	0	••	• •		0	0		0	0	00		ð	¢	•		0	0	2300		1 0.083	¢	2 0.166	0	0	00		0			5 C		0			•		00		2 0.166	, 0	0.	2 0.166	0						0		, 0
Score O	0 0	Ċ	•		0	• •		•		0			į	. 0	0	•		0	0	• •		0	0	•		0	0	•		0.00325	8	0	0	•	• •		0					•	• =		•	6 0			0.0065		•	• •	0	• •				• •			
Score		0	• •		0		• •	•		¢	0	•			8	-	• •	•	•					-			8	-		0.083333	0	Ð		e	• •	• •	•	• •					• •	•	ф,	- •	• •	• •	0.166667		•		•	• •		• •		• •		• •	, .
IdSII	z																												2	1 L		6													,	n			4	2		86				61					
Score	0 4 0.0150302	0	0 0	• •	0	• •				•	0	• •			0	•		0	¢	• •		0	0	•	- 0	o	0	FIASCOUL E	0	3 0.0175768	¢	4 0.0135483	¢	0	0 0		0	c 0				0	•	, o	0	• •	• •		0	0 0	0	0.0182558	•		• •	2 0.0182046		• •	, o	• •	
Score	0.42943548	0 ,	0 0		0	• •		0 0		0	G	0 0		0	0	•	- 0	c	0			0	0			0	0	007133700	0	0.50219619	0	0.38709677	0	0	• •		0	e e				•	• •		0		е с	• •	0	0 0	6	0.5215959	0	• •	• •	0.52013177	• •	• •	, 0	0 0	• •
DAYREST	⇒ ₽	0			0		• •	• •		•	0	• •		• •	0	0		0	0	• •		0	0	•				ž	2 0	- 15	0	37	0	0	• •		0	0 0		• •	• •	0	• •		0		• •		• (30	0	<u>5</u> 0	0	00	00	100	• •	• •	. 0	• •	> o
Score	-	•		• •	•	• •		•		•	•	• •			•	•		¢	0	¢ (• •	-	0	-	• •	0	0	-		3.913E-07	0	1.0308E-05	0	0	¢	• •	0		1	0 00674461	0 0	0	• •		0		• •		0	0	0	0.00828637	Ð	¢ ¢	• •	0.01648352	• •	• •	, .	• •	
Score		0		• •	0	• •		•		-	•	-				-		0	e	•	• •	e	8	•			•	•	• •	2.17391E-05	•	0.000572663	•	•	e e		0			0 27.4700559	0 0	0			0		• •		0	B	0	0.460353836	0	00	,	0.91575116	• •	• •	, o	• •	• •
DAVLOST	• •	0	•		0	• •	• •	•		• •	0	•			0	•	• •	•	0	• •		. •	0	•	• •	•	•	•	• •	. 4	•	9	¢	¢	¢ (0	• •		• •	-	•		• •	0	• •	• :		- 1	<u> </u>	•	5 e	•	•		911	• •	• •		• •	, .
Scare		0			0	•		0			0	0 0			0	•		•	0	• •		. 0	0	0		•	0	1000000000	1105750000	0.00282692	0	0.00282692	c	0	• •		0	0 0				0	00	0	0	8 0	0 0	• •	0	0.001/10204	0	0.00363462	0	• •		0.00870513		o c	, o	• •	
Score	• •	0	• •		0	00		0			0	• •			0	0			Ð	0 0		• •	0	0	• •	• •	0	TTOPPOSTO	1/05769/07	0.067307692	0	1.067307692	e	0	•		0	• •				0	• •	• •	0		•		0	0.184829UD	0	0,086538462	. 0			0.207264957		• •	, 0	0 0	ə 0
DAYHOSP	• •	0	0	• •	0	•		0		. 0	0	0 0		• •	0	0			0	0 0			0	0	• •	Ģ	0		• •		Ð	2	c	. 0	•		0	• •				0	•		o	• •	0			R 0	0	- vo e		0 0		36	0 Q	ę (, o	00	o` 0

PARTIAL DISABILITY 0 COL TOTAL DISABILITY TOTFTL 0 5.6321E-07 0.02346933 1.3432E-08 3.7707E-01 Score 0 0 1.1983E-05 0.49934744 2.8579E-07 8.0228E-06 UHBHY 5.13439E-05 0.749550019 1.22458E-06 3.43753E-05 Scare 009418 0 INJCOST 0 0 0 20532 1100000
 Utility
 Sare

 ULINEXPLA
 Sare

 <t Revert
 Altername
 3.24877E-05 2.49284E-07 2.41192E-08 0.001196667 0.101196667 0.23886E-09 7.35095E-08 7.35095E-08 0.178877773 2.73551E-09 1.25314E-05 1.49763E-08 EVENTI So So So ****

Score	0.0397	0.0312	0.18443	2.45-05	0,16788	0.00591	0.0255	3.8E-10	0.00036 7.3E-05	0.00207	3.9E-05	3.9E-05	0,00017	0.02203	0.00031	4.8E-05	0.02444	3.9E-05	1.5E-05 6 00133	0.00055	9.2E-05	0.0029	0.062	0.15691	0.00149	C0-26-1	0.00044	0.00918	0.00259	0.00044	0.02847	0.00043	0.03131	0.00024 0.00146	0,0014	0.00066	0.02139	4.5E-05	0.00132	0.00011	0.00052	7.8E-05	0.062	0.20371	0.062	0.18412	0.00036	0.00022	0.00011	0.00031	0.00013	0.02176	0.00027	0.00316	5.8E-05	4.2E-10	66-05	0.00026	0.00015	2.9E-05 6 RF-05	100'0	0.00031	0.00246
Utility Score	0.0001563 9.6878E-06 0 0	0.1585741 0.0098316	0.0065627 0.00040689	0.0003906 2.422E-05	0.0107035 0.00066362	0.0953853 0.00591389	0.00000228 0.00056189	0	0.0011719 7.2659E-05	0.0333471 0.00206752	0.000625 3.8751E-05	0.000625 3.8751E-05	0.0028126 0.00017438	0.0009375 5.8127E-05	0.0050002 0.00031001	0.0007813 4.8439E-05	0.00125 7.7503E-05	0.000625 3.8751E-05	0.0002344 1.4532E-05	0.0057033 0.00035361	0.0014844 9.2034E-05	0,0367581 0,002279	0.0019532 0.0001211	0.0114066 0.00070721	0.0240633 0.00149192	0.0728304 0.00451548	0.0070315 0.00043595	0.003047 0.00016891	0.0418127 0.00259239	0.004297 0.00026642	0.0028126 0.00017438	0.0068752 0.00042626	0.0015626 9.6878E-05	775700070 556552000	0.0225007 0.00139505	0.0107035 0.00066362	0.008594 0.00053283	0.0001563 9.6878E-06	0.0014063 8.719E-05	0.0017188 0.00010657	910510000 8/518000 910510000 2212000	0.00125 7.7503E-05		0 0 0	0 0	0.0336742 0.0020878	0.0058596 0.00036329	0.0035157 0.00021798	0.0017158 0.00010657	0.004922 0.00030517	0.00125 7.7503E-05 0.0023438 0.00014532	0.0011719 7.2659E-05	0.0043751 0.00027126	0.050905 0.00315611	0.0009375 5.81276-05	0 0	0.0009375 5.8127E-05	0.0042189 0.00026157	0.0022657 0.00014047	0.0004688 2.9063E-05 0.0010938 6.7815E-05	0.0131254 0.00081378	0.0050002 0.00031001	0.0396003 0.00245522
Score	0.00015625	0.15856959	0.0065625	0.000390625	0.010703125	0.095382324	0.0090625	0	0.001171875	0.033346039	0.000625	0.000625	0.0028125	0.0009375	0.005	5718/000.0	0.00125	0.000625	0.000234375	0.005703125	0.001484375	0.036756866	0.001953125	0.01140625	0.0240625	9508282200	0.00703125	0.003046875	0.041811345	0.004296875	0.0028125	0.006875	0.0015625	0.02359375	0.0225	0.010703125	0.00859375	0.00015625	0.00140625	0.00171875	0.002421875	0.00125	0	0	6	0.033673063	0.005859375	0.003515625	0.00171875	0.004921875	0.00125	0.001171875	0.004375	0020000000	0.0009375	0	0.0009375	0.00421875	0.001265625	0.00046875	0.013125	0.005	0.039598963
MHRS	64 C	1377	4 8 c	. v	137	1000	116 116	0	5 5	426	œ	œ	36	12	5	<u>e</u> 2	2 21	80	۳ ș	7 <u>7</u>	6	467	⊳ ¥	5	308	4 ¥	8	39	22	4	98	88	8	8 5	288	137	110		18	52	108	91	•	89 0		430	5	5 ,	° 12	63	16 30	5	8	96 622	1	× 0	20	5 F	29	9 1	168	5	200 200
Score	0.062	0	0	0	0	•		¢	c 0	o	¢	0	•	•	• •	• •		0	• •		o	•	790.0	• •	• •		• •	0	0 0	• •	•	0	0 0		•	0	• •	• •	ð	•	• •	. 0	0.062	0.062	0.062	0 C	•	0 0		0	• •	0	0		•		•		0	• •	0	9	30
Score	• -	0	0 -	- 0	0	0 0	ə o	0	• •		0	0	0	e 1	• •		0	0	•	• •	¢	0		• •				0			. 0	0	•		0	0	• •		0	•					-	• •	•	• •	• •	0	• •	0	0	• •	0		0 0		0	0 0		0 0	
SSOTLO	• -	• •	e -	- 0	• •	•		0	• •		• •	0	¢		•	ə e	• •	0	•			0			0			0	•		• •	0	•	• •	• •	0	0 1	- e	0	•	• •	• •	~ 1	o -		• •	• •	0	• •	• •	• •		0		0		0		0		0	0 0	0 0
Score T	• •		0.00325		0	•	0.00325	0	• •	•	0	•	0	•	• •		• •	0	• •						•			0.00325				0	•		. 0	0	• •		0	•					0	0.00325	•	• •		0	• •	0	•		0		•				•		
Score			0.083333 (100001		•	0.083333 (• •	-	0	0	•	•		• •		•	• •						•	-	. 6	0.083333	•		•	đ	•		. 0	0	• •		0	¢ i	• •	Þ	6 6	• •	•	0.083333	•	• •		0		•	• •		•		• •	• •	•	• •	•	• •	
HYSPL				4			-																					-																		-																	
Score	0 0	• •	0.00975	0	0	•	• •	¢	ф с	, ¢	Ð	Ð	Ð	ð	• •			0	• •		0	0			0 1			0.00325	• •	• •	•	0	• •	• •	• •	0			0	0	0 0	• •	0	• •		0.0065	0	0		0	• •	0	0		0 0		0 0		0	• •	• •	• •	
Score	o e	. 0	0.25	0.100001	0	0		o	• •	• •	o	o	0	•	• •			0	• •			•			0 1		• •	0,083333	•	, c		0	0 0			0	•		0	•	• •		• •	• •	¢	0.166667		• •	• •	0	• •	. 0	0		0		0	• •	•	• •	• •	0 0	
OLSPI	v;		50 F	7	•																																									7	,																
re Scor	0.019 0	. 0	61019 °	• •	0.039	ė	• •	0	• •	•	0	0	0	0	•		•	0	Ð	• •		•			•			0	•	-		0	•			0	•		0		• •		÷	• •		0 0	-	•	• •	•	• •			••	•	••	0		6	• •			
LSPL Sco	1		7) ° 7	* *	5	•	• •	0				e	0	•	•		ē	0	0	• •		0	* *		•			•	•			0	•			•	•		•	•						• 3		•		•	••		•						•	••		•	
Score F	e e	• •	12217606 2		91767936	0		0	~ ~			0	0	0	• •			0	•			0	0 0		0			0	0 (0	0 0			0	0 0		0	0	• •	, o	0	• •		01776903	. 0	¢ (¢	• •		0	0 0			0		0		, .	0	0 0
Score	• •	, c	3601757 0.0		5124451 0.0	0	0 0	0	0 0		. 0	0	0	0	0 0			0	•			0			0			•	0 0		, o	0	•		• •	0			0	0						7686676 0.0	, .	•		0			0	0 0	0		0		0				
REST			10 0.63		90													0										0	0		, .	0	0			0										56 0.50							0		0	0 0	0		0	• •		0	
re DAY			5956 4		8637			-															202	8178																						1615																	
e Sco			1955 0.0177		3836 0.0082	÷	¢ e	•	¢ 6	. ¢	•	0	Ð	Đ	•		••	•	•			•	0 2 01 2 01 2	E01070 66ES	•			•	•			¢	•	••		•	•		-	-			•			3303 0.008		•			••								•		, .	•	
T Scor	• •		0.98664		SE094'0	0		0	~ ~	¢	•	8	8	0	•		• •	0	•	• •		0	10271 (0.57676		•		8					•	• •		0	•	• •		•			•	• •		0.47328		• •		•	66	•	0		•		0		•	• •		0	• • _.
DAYLOS			8 194	• •	4 21	0	• •	•	• •			0	•	0	• •	• •	• •	•	•	• •	• •	¢	¢ •	3.	e	•	• •	0			• •	e	•			•	•		¢	0	• •	• •	•	• •		4	• •	0	• •	• •		ē		• •	0	• •	0	• •	•	• •		0	
Score	• •	• •	0.0042403	• •	1 0.0038365-	0	• •	0	• •	, o	. 0	0	0	0	0 (, o	0	•			0	• •	0.0030288	0	• •		0	0			0	0	0 0		0	0			0	• •		0	• •		4 0.0038365	• •	0		• •	0 0		0		0		0		0	• •		0	0 0
Score	• •		0.100961538		0.091346154	¢		0	•		• •	9	0	o	•		• •	0	•	• •		0	• •	0.072115385	0		• •	0	0	-		0		• •		0	•	5 C	•	Ŷ	• •		0	0 0		0.09134615		0			• •		0		0		0		• •	00	• •	0	
DAYHOSP	00	. 0	6			0	0 0	0	•			0	0	o	•			0	0			0		- m	0	• •		0	0	• •		0	0	•		0	0		. 0	0	0 0		o	0 0	0	1		0			00	• •	0	• •	• •	0 0	0		0		• •	0	0 0

Score									00		• •	• •		• • •			• •	0 0			e o		• •					• •		• •	• •			0.006125					• •		÷
Score o o o			0 0	• • •									• •								• =	••	• •					• •	• •		• •	-		0	0 0	0 0					Ð
PARTIAL DISABILITY 0 0 0			• •						• •		• •						• •	• •		• •			• •			. e (. •	• •	0 0			e	00	7	0 0	8 0	8 8	0	• •		0
Score [• •	• •	00	0 0				00		00		0 0		0 0						• •	0 0	• •			00		• •	• •	• •	0 0		00	0
Score Score			0 0						• • •	o 0		0 0					00	0 4	000	0 0	0 0	0 0	0 0				00	00	0 0	0 C				0 0	0 0	0 0	0 0	00	0 0	00	0
TOTAL DISABILITY 0 0			• •							• •	0 0	• •					0 0			• •	0 0	00	0 0					0 0	00	0 0		, o	00	• •	0 0	0 0	00	0 0	00	00	0
Score										e e		• •				• •	• •	• •		• •	• •	• •	• •	0.176	8 .								0 0.198	• •	0.253	0 0.198	0.176 0	• •		0.187	0.176
Score D D			• •	• • •	• • •					• •		• •				• •		•		5 ¢	• •	c 0	• •	9.5	0		• •	• •		•			0 0.5625	• •	0.71875 0	0.5625	0.5			0.53125	5.0
1TATOT 0 0 0					• • •					• •	• •	• •					• •	-		• •		• •	• •	• - •	• = •			±	• •	• • •			e m		* •	0 10	- 0	• •	• •	н с	-
Score o					1.4962E-06 0 2			0	1.3771E-06	• •	00	0 0					00	0		0 0	c 0	0 0	• •	0.0234841	0		••	• •	0 0	• • •			0.0034577	0.00033146	0.02891667 0	0 0.0282932	0.02357479	00		0.02776507	0.0235266
Anno 0			• •	• •	3.1833E-05 0			0 3.4069F.06	2.93E-05	• •		• •				• •	0 0			• •		0 0	0 0	0.49966181	0 0	50	• •	0 0		•			0 0.07356799	0.0070524 0	0.61524838 0	0.60198303	0.50159129 0	• •		0.5907462 0	0.50056601
Score			• •	0 0	0.00013638			0	0.00012553	0 0		• •				• •	• •	• • •					0 0	0.749766685	0		• •	• •	• •	• • •		• •	0 0.233871373	0.029157674	0.822553524 0	0 0.814826876	0.751094016 0	с с	0 0	0.808165972 0	0.75038924
INJCOST 0 0 0			• •	0 O	36888 0			0 0	35310	00						• •	• •				• •	00	00	1104066	0			• •	00				0 762000	163855 0	2470000 0	0 2325000	0	00	• •	2200000 0	1115749
Score 0.00234501 1.1423E-05 5.7367E-10	2.43882-08 8.633E-10 0.00314473	1.0559E-07 2.3138E-09	2.6961E-08 1.1321E-07	9.4681E-10 5.5533E-10	0.05461495 5.3439E-09	3.482E-10 1.2357E-05	2.9165E-07 0.02420515 2.67 FC 08	4.2748E-08	0.13610879	7.7231E-09	5.5142E-09 5.7926E-07	1.0175E-09 1.30777E-08	2.3095E-07	0.03446186	2.1796E-08 6.2694E-07	2.0288E-08 0.00411993	0.09895806 7,4947E-06	6.5881E-09	0.02440933	9.0113E-09 7.9119E-05	0.09543575 1.6357E-08	0.00019208	5.0124E-10	0.12044697	8.1128E-10	0.00285412	3.482E-10 6.1503E-10	0.16824065 3.482E-10	0.00109587	3.482E-10 7.487E-10	3.482E-10 3.5065E-10	1.1201E-06	1.9394E-05 0.08595735	9.5067E-05 7.3589E-07	0.09031508 0.00972224	7.0533E-09 0.01120594	0.02699824	2.2402E-09 1.2249E-08	0.13817242 3.1161E-08	0.13971467	0.02190624
Utility 0.009936466 4.8401E-05 L43081E-09	1.0334E-07 3.65804E-09 9.013325146	3.1616E-08 4.47.412E-07 2.80433E-09	1.14243E-07 4.79692E-07	4.01192E-09 2.35311E-09	8.231419277 2.26437E-08	1.47542E-09 5.23604E-05	1.23582E-06 0.102564215	1.07634E-07 1.81136E-07	511267376720	3.27248E-08	2.33651E-08 2.45448E-06	4.31142E-09	9.78605E-07	1181209110	9.23557E-08 2.65655E-06	8.59679E-08 0.017457339	0.419313812	2.79158E-08	1.17644E-08	3.81836E-08 0.000335251	0.404388774 6.93098E-08	0.000813915 0.00108199	2.12391E-09	0.510368526	0.51822763 3.43762E-09	0.002158362	1.47542E-09 2.60605E-09	0.712884093	9112362418	1.47542E-09	1.47542E-09	1.06207E-08 4.74636E-06 0	8.21786E-05 0.364226058	0.000402828 3.11819E-06	0.382691031 0.04119595	2.9887E-08 0.047482775	0.471726862 0.114399328	9.49219E-09 5.19005E-08	0.585476348 1.32037E-07	0.592011322 5.07648E-09	0.092823035
Score 0.015178155 7.43852E-05 3.7359E-09	1.58823E-07 5.62204E-09 0.020312277	4.85907E-08 6.87626E-07 1.50683E-08	1.7558E-07 7.37238E-07	6.16592E-09 3.61649E-09	0.31273736 3.48011E-08	2.26758E-09 8.04701E-05	0.148394993	2.78388E-07	0.668880445	5.02947E-08	3.59098E-08	6,62622E-09	1.50402E-06	0.206290641	1.41941E-07 4.08284E-06	1.32124E-07 0.026544277	0.519051578	4.29038E-08	0.149574234	5.86843E-08	0.503860498	0.001250276	3.26424E-09	5.79603E-09	0.615285687 5.28327E-09	0.003312759	2.26758E-09 4.00523E-09	0.785315243 2.26758F-09	0.537775962	2.26758E-09	2.26758E-09 2.26758E-09	1.29465E-06	0.000126294	0.000618951 4.79234E-06	0.061751982	4.59332E-08 0.070911337	0.570972417 0.164430495	1.45885E-08 7.97658E-08	0.676695677 2.02927E-07	0.68250442 7.80203E-09	0.135039346
DAMCOST 176858 118485 13592	51097 16955 180056	57101 26200	52188 67864	17760 13340	792308 34800	10000	354412	21212 21212	F12FE	38699	35130	18398	75678	911513	49876 86632	49098 182993	1970824	37010	360643	40346 139725	1196722	149457	12561	61212	16422	150152	14140	10000	2924975	10000	10000	63000	124295 966744	141740	984138 192260	37734 193778	4616589 430865	15878	10004000 53764	10300000	274755
	888		57 57	57 5 7	និតិ	S S	888	8 8 8				8	6 8 8	8 58 1	F 75	\$ \$	8 8	8	7 6	F 7	7 , 7	8 8		7. 78	R .R	हे है	R R	7 , 7,	89	8 8 8	888	88	51 51	55	55	5 5	ភភ	5	5	5	15

Score	0.00238	0.00171	0.0033	0.00011	0.0033	1.9E-05	0.00019	1 9F-05	1.55-05	0.01465	0.00024	0.09532	0.00061	0.00038	CU-31.4	21000-0	0.062	0.01534	0.13918	0.17044	0.00011	0.00024	4 95 AS	0.00013	0.00023	0.00591	0.00422	0.0209	0.03456	2.9E-05	0.002	0.00709	0.00417	0.09941	6.6E-05	0,00155	0.00052	60970'0	900000	0.09547	0,00058	0.00025	0.0007	0.062	0.062	0.38148	8.1F-10	0.00051	0.00305	790'0 0 00077	0.23024	9.7E-05	0.10986	0.00113	3.5E-10	3.5E-10	2.56-09	0	77900.0	0.4023	0.00975	0.42748	0.07225	0.06217	0.2895	0.03704	0.062	0.00013	0.23267 3.15-08	0.47554	0.062	0.25517
Utility Score	0.000625 3.875tE-05	81741000 0 9618600 0 1700/1000 0 9618600 0	0.053285 0.00330367	0.0017188 0.00010657	0.0025782 0.00015985	0.0003125 1.9376E-05	0.003047 0.00018891	0.0003125 1.9376E.05	0.0002344 1.4532E-05	0.2363286 0.01465238	0.0039064 0.0002422	0	0.0098441 0.00061033	0.0061721 0.00038267	56711000 U 1568100 U	20-32622F 12020000	0 0	0.2474078 0.01533928	0.0018751 0.00011625	0.0081253 0.00050377	0.0017969 0.00011141	771700010 1000000	1070100176 F 11820000	0.0021094 0.00013079	0.003672 0.00022766	0.0953853 0.06591389	0.0156255 0.00096878	0.0016407 0.00010172	0.0015626 9.6878E-05	0.0004688 2.9063E-05	0.0322121 0.00199715	0.1143892 0.00709213	0.0007813 4.8439E-05	0.007344 0.00045533	0.0003/2 2.812/E-05	0.02SUNUB 0.0415SUNS	FICTONN'D 8/ CHRIST	0.0109479 0.00067815	0.0011719 7.2659E-05	0.0005469 3.3907E-05	0.0093753 0.00058127	0.0009375 5.8127E-05	0.0071877 0.00044564	0	0 0	0.0014063 8./196-05 0.0019532 0.0001211		0	0.0031251 0.00019376	0.001297 0.00026642	1 0.062	0.0015626 9.6878E-05	0 0	0.0011719 7.2659E-05	0 0		0.0181256 0.00112379	0 B	0 0 0	0.0014533 0.0002761	0 0	0 0	0.0085159 0.00052799	0.0028126 0.00017438		0.0046877 0.00029063	0 0	0.0020313 0.00012594	 	0.0089847 0.00055705	•	•
Score	0.000625	C/8176/70/0	0.053283317	0.00171875	0.002578125	0.0003125	0.003046875	0.0003125	0.000234375	0.236322482	0.00390625	•	0.00984375	0.006171875	2/2010/0	0.000703125	0	0747401459	0.001875	0.008125	0.001796875	5790X579110	C/C+0+0000 0	0.002109375	0.003671875	0.095382324	0.015625	0.001640625	0.0015625	0.00046875	0.032211058	0.114385781	0.00078125	0.00734375	C1.66000.0	0.025	C/SH20010	0.0100245000	0.001171875	0.000546875	375600.0	0.0009375	0.0071875	•	0	0.0011953125	0	0	0.003125	578392E00.0	1	0.0015625	0	0.001171875	•	• •	0.018125		0	0.004453125		0	0.008515625	0.0028125	• •	0.0046875	0	0.00203125	• •	0.008984375	0	e
MHRS	~	351	646	8	33	4	39	ю ч		1710	50	•	126	62 '	0 7	ť 0	n C	1750	24	104	53	8 I	6	27	4	1000	200	21	50	9	412	1128	01	94	12	320	108		¥ 1		120	12	92	0	•	18	9 e		4	- :	10400	20	0	8 51	•	0	727	°	•	ñ 5		0	109	36		99	30	26	00	, 115	0	0
Score	Ð		, .	Ð	ð	o	•		0	c	o	c	c	00		• c	0.062	0	0	0	¢	•		• •	0	0	0	0	0	0	0	0	0	0		0	-			• •	•	0	0	0.062	0.062			• •	0	70//n	• •	¢	•	• •	o	• •	• •		0	0.062	0	0	0.062	0.062			0.062	0	0.062	0.062	0.062	0
Score	0	ə c	• •	0	0	0				0	0	ð	•	•	5 0			0	0	0	•	•		• •	o	0	0	0	0	0	0	0	0	0	-	•	• •				0	0	0	-	,							ð	•	• •	•	• •			ο.		. 0	0	-				-	0	- 0	» –	-	0
SSOTLO	0			0	0	0			. 0	0	0	0	0	•			- ,	. 0	0	0	0	o 1	• •		¢	¢	¢	0	0	0	•	•	0	¢	•	0	•	-			0	0	e	-		• •		• •	0		• •	0	•	0 0	0	•			o ·		- 0	Ģ	-	-	• •	• •		0		ə	-	0
Score T	•			¢	•	0	•		. 0	0		00325		• •					0		-						.00325		•	•	•	•	e	•	• •						0	8	0	•	•	-011525 0.01165	(III)					0	.00325		0	•			.00325	67600°	.00975	3900.0	0		.00325	.00325	•	0	.00325	.00975	0	.00325
score	0			0	0	6				0	0	83333 0	•	• •			• •	0	0	0	•				0	0	83333 0	0			•	•	0	•	•	•					Ð	e	•	8		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	,	. 6	÷.				983333 0		•	• •			083333 0	0 0000	0.25 0	166667 (0	•	0 083333 0	83333 0	-	0	183333 0	0.25 0		083333 0
IdS/H												1															1																		•		4						-							-	ŕ	2 0.			-	1	,		-	3		1
Score	•	• •	• •	o	0	0	• •			0		0						0	0	•	0				•	0	•	0	0	0	0	0	0	0		0				. 0	6	0	•	0	•	C000.0	0		0			0	0.00325		0				0.00325	0		\$7,000,0	0	•	0.00975	0.00325	0	Ð	0,00975	0.00975	0	0.00975
Score	0				0	0				0	0	0	0	• •				•	0	0	0			, .	0	0	0	0	0	0	0	0	0	0	•	0	•			, o	0	0	o	0	0	0.75	- -		0			0	0.083333		0				0.083333	(7) U		0.25	0	•	0.25	0.083333	0	0	0.25	0.25	0	0.25
JISLO																																															n						-						- ,	•		3			•	-			•	۴		ñ
Score	-			•	0	0	•			0	0	0	ę	• •	• •			6	c	0	0	•	• •		Đ		•	•	•	•	•	•	•	•	-	•					•	•	-	-	-	0.039	0	-	•		• •	•	•		•				0.00325	0.00355	0	0.039	•	-	0.039	0.00325	•	•	0.0195	0.039	•	•
Score	•		• •	0	0	0	•	• •		•	e	e	-	• •			• •	•	•	0	0	•			0	0	Ð	0	0	0	0	Ð	¢	¢	•	•	•	• •	- 4	• •	0	e	0	Ð	•				•			0	•		0	•			0.083333	1 0.02111	0	1	•	-		0.083333	0	0	0.5		•	0
LISPL																			\$	4																										n v	n												- •	n -	•	5			n r	-	•		-	¥0		
Score	0	• •	• •	• •	0	0	•	• •		¢	0	0.0175384	¢	0 0	-				0.0170060	0.0184480	Þ				o	0	•	•	•	•	•	0	•	•	•	-					e	•	o	•	•	0.0100070			0			•	0	• •	¢	•	• •		0 0	0 0200750	0	0	0	0	0 001078	0	¢	ø	00	> o	0	0.0100907
Score	0	• •	• •		e	0	0		¢	¢	0	501098097	e	•				. 0	485887097	527086384	0	•			•	0	0	•	•	•	•	•	•	-		-	- •			• •	0	•	0	•	•	.288306452			0			0	0	• •	¢	• •			0 0	0 \$73\$77474	0	0	0	0	0	0	, o	0	0 0	, 0	0	.288306452
YREST	0				0	0	0			ę	o	48	۰.	• •	• •				44 0	119 0	Ð				. 0	0	0	0	0	0	0	•	0	0	•	0	÷ •				0	0	0	0	•	g -			0			0	0	0 0	0	•			•	0 746	ç o	0	0	0	0 % 2	9 G		0	0 0		0	30 0
M DA]_						_				_	58521	_	_					2E-06	52016	_				_	_	_	_	_	-	_	_	-	_	_	_							_	•	_				-			-	_		_					65535		_	_	_	1870			_			_	39599
Sci	Ĭ			. U	Ū	-					-	117 0.011	-						846 2.175	662 0.011	-				-		Ū	Ū	Ū	J	J	-	Ū	Ĵ		Ū	÷.				ų	Ū	Ű	Ű	-								-			-				10 D 011					1012			•				7209 0.005
Score	÷		• •	•		0	•			•	e	0.649178	0	-	-			•	0,000120	0.663513	¢	•	• •		•	•	•	•	÷	•	•	•	-	•	•	•	•		• •	• •	•	•	•	•	•				Ð	-	• •	Ð	Ċ		Đ	•	• •	,	•	0.75908	0	0	•	•	0 670161	0 0	• •	•	• •	••	•	LT7662.0
DAYLOST	•	• •		• •	0	e	÷	• •		. 0	0	4	•	•		• •		•	4	7	e	•	-		•	¢	¢	•	•	-	-	•	•	-	-	-	• •			• •	•	•	•	•	-				•	• •		•	÷	• •	•	•	• •		•	- 5	3 -	0	0	0	• *	9 -	• •	0			Ð	2
Score	0				0	Ð	0		, c	• •	0	0.00823397	0	0				• •	0.00282692	0.00343269	0				0	0	0	0	0	•	0	•	0	•	0	0	•				0	0	0	0		0.002625			0	• •		•	0	• •	• •	0	• •		•	0 0110600	0	0	0	0	0.00540705	cu/0+cu0.u		0	00	50	6	0.00525
Score	0	• •			0	0	0				0	96047009	0	•					67307692	69/06/18	•					0	0	0	0	0	0	0	0	0	0	0	0	• •			0	0	0	0	•	0.0625		• •	0	• •	, o	• •	0			0	• •	,	0	0	10/4/200	c	0	0	0	01666/87	• •	0	00		0	0.125
HOSP	0	• •									0	13 0.1	0	0	0				2 0.0	5 0.0	0	0	÷ •			0	0	0	0	0	0	0	¢	•	0	0	0					0	0	0	0				0			. 0		0 0				,	0	0 5	. 0		0	0	0		, ,	. 0			. 0	14
ΥN																																																																								

Score	0		•		•		•				• •			-		•		0	0.0035	0	0	e	•	0	• •	•	-		•		•				• •		0	• •	¢	¢ ¢	0	• •	• •		0	0		•	• •		• •	, c	• •	0	00
Score	0		•		•	• •	•		•	• •	• •	• •	• =		• •	•		•	0.75	0	0	0	0	•		•					• •	• •	• •	• •	• •	• •	•	• •	• •	•	•	• •	• •		0	0	• •	0	• •	,		, 6	00	0	0 0
ARTIAL DISABILITY	•		•	• •	6	• •	•				• •		• •		• •	0	• •	0			0	e	0	0	*	•	•		-								Ð	• •					• •		•	Ð		0			• •	, o	0 0	0	• •
Score	•		0		0	• •			• •	• •			e e		• •	c	0 0	0	0 6	0	0	0	0	0	• •	. 6	•		0 1		0	0 0	0 0		• •		a	• •		0 0		0 0	0 0		0	0	0 0	0			• •	, e	• •	0	• •
Score	0		•	0 0		• •	• • •	• •	0 0				0 0	. 0	0 0	0	• •	0	• •	•	•	0		0	. c	. 6	Ð,	• •	• •		0	0 0	0	• •	0 0		0	0 0		0 0			e (0	0	0 0	0		,	0 0	, 0	00	0	0 0
FOTAL DISABILITY	0		. 0			• •		• •			• •					•	• •	0	0 6	• •	0	0	0	0			0		0		0	0 0	0	• •	• •		0	0 0		0 0		0 0	0		0	0	• •	0	0 0	•	0	, 0	0 0	0	o o
Score	0	• •	0.187	¢ ¢		0	•		0	0			• •	0.209	• •	0.187	• •	•			0	0		•			ę		•		•	• •	• •	• •			6	• •		• •		• •	•	• •	0	0	• •	•	• •		• •	• •	e e	0	6 9
Score	9		0.53125	• •		0.53125	•	• •	¢) e	• •	• •	• •	0.59375		0.53125	• •		• •		¢	÷	¢	¢	• •	•	-	• •	•		•		-		• •	• •	•	• •	• •	• •		• •	•	• •	0	0	• •	•	• •	•	• •		0 8	0	• •
OTFL	0	• •	. 4	• •		¢ •	•		۰.	- 0	•		• •	• •	• •	• •	• •	¢	•		٥		•	•	• •		0	• •		• •	•	• •	•	• •	•		0	• •	• •	• •	• •	• •			0	0	• •		• •		• •		0 0	•	• •
Score	•	0 6 3467E-08	0.02776507	1.2887E-08 0	3.1937E-06	0 0776407	o		0	750105700	0 - 3117L C	0	0 0	0.03010938	0 0	0.02346933	• •	• •	0 00160024	0.00052968		5.5553E-08		0	• •		0		0	0 0	0	• •	0		0 0			00		00		0 9.8568E-08	0	0	• •	0 9.6196E-08	o c	• •	• •		•	, a		. 0	6 6
Unline	•	0 13504F.06	0.5907462	2.7419E-07 0	6.7951E-05	0 5907467	0	• •	0	0	0 0	0	0 0	0.6406251	• •	0.49934744	• •	0	0	0.01126977	• •	1.182E-06	- c	• •	• •		0		•		•	• •	0	• •	•		• •	• •				0 2.0972E-06	÷	2.2794E-07 0	•	0 2.0467E-06	• •	• •	• •	• •	0 0	, o	0 0		
Score	c	0 5 786175-06	0.808165972	1.17488E-06 0	0.000291059	0 0 \$0\$164977	0		0	0	0	0	0 0	0.836941076	o c	0.749550019			0 176421070	0.045652829	o	5.06463E-06	c	0	0 0	0 0	0	0 0	0		0		0		0 0		ø	0,	• •	• •	• •	0 8.98612E-06	0	9.76699E-07 0	0	8.76991E-06	•			3	0 0		00	. 0	• •
NJCOST	0	0	2200000	720	52537	0 770000	0		0	0	0	0	• •	2740000		1100000		• •	0	174984	0	2965	c	• •	• •		•	• •	0		0		0	- 0	0		o	0 0	0 0	0 0	• •	0 5040	0	0 0	0	4930				,	0 0		• •		c o
Scare	05E-06	1463E-09	3833508	18803341	1679545	5139E-09 M0.17441	12243956	13906425	212E-06	C/70060	2417485	5795E-06	1941E-05	13866306	5745E-05 13043549	63308559	1666E-09 30017857	2587E-09	1339963	2122647	482E-10	14172057	0 7235F-08	1212E-08	4202E-08	19866160	6876E-09	5956E-10 4968E-10	1019E-09	9923E-09 8691E-09	2043E-10	8768E-09 2143E-10	3459E-08	9927E-10 1631E-09	7425E-10	482E-10	0 00849822	02607733	05E-06	0193E-07	19545058	.05E-06 0989E-06	.05E-06	0213161	03661315	u 09695903	901E-09	8479E-10	8757E-09 0074E-09	8	7855E-09	5859E-07	9782E-08 232F-09	2282E-08	0526E-09 8431E-09
Linkin	L44899E-06	.33318E-08 3.	10 1095919851	1373022911 0/	0 286568761	6.8386E-09 1.	005082866 01	LI65526475 01 L59521E-08 24	0.05592E-05 7.	0 13535556501	102435794 0	00301E-05 2	C33094E-05 1	0 22232323 07	0.000278581 6.	118116732	1.94306E-09 1.	07571E-08 7	1144052101 0.	1.513671483 0.	L47542E-09 3	0.600512566 0.	0	4.75105E-08 1.	3.5679E-07 8.	1402536735 0.	7.15073E-09 L	2.79474E-09 6. 3.60036E-09 8.	8.90645E-09 2.	2.53912E-08 5. 9 91054P-08 6.	3.90012E-09 9.	2.06642E-08 4.	5.70304E-08 1.	3.81047E-09 8. 4.92847E-09 1.	3.70445E-09 8.	3.33194E-09 //	0.036009409 0.	0.110497145 0.	1,0-36136-01 1,44899E-06	2.12682E-06 5.	9 16515t10t10	4.44899E-06 1	4.44899E-06	4.63311E-06 1. 0.090322477 0	0.155140482 0.	0 0.410843342 0.	1.22922E-08 2	2.47791E-09 5.	4.18463E-08 9.	0	1.18028E-08 2.	7.9833E-06 9. 4.06184E-06 9.	1.26194E-07 2.	5.20404E-08 1.	8.69764E-09 2. 7.80961E-09 1.
Score	6.83762E-06	2.04897E-08	0.677309618	0.471316383	0.593249377	1.05102E-08	0.138150516	0.231416866 1.321E-07	4.69656E-05	0.059336232	0.148219853	1.67983E-05	9.72963E-05	0.678546566	0.000428078	0.354889545	7.59698E-09	4.72705E-08	0.203720364	0.611013121	2.26758E-09	0.690020462	LU ALLEL	7.30189E-08	S.4835E-07	0.501962413	1.099E-08	4.29524E-09 5.53339E-09	1.36883E-08	3.90237E-08	5.9941E-09	3.17588E-08 7 74145E-09	8.76499E-08	5.85631E-09 7.57457E-09	5.69337E-09	5.12086E-09 2.26758E-09	0.054144427	0.159166253	6.83762E-06	3.2687E-06	0.503924826	6.83762E-06 7 15638E-06	6.83762E-06	7.12061E-06 0.131587647	0.218098721	0.51045295	1.88919E-08	3.80829E-09	6.43136E-08	60-T+0+T*/	1.81397E-05	1.22695E-07 6.24262E-06	1.93947E-07 8 073E-09	7,99809E-08	1.33674E-08 1.20026E-08
DANCOST	92294	29303	10035285	975173 11102	5751771	12675	295004	19096	113438	131736	353477	102154	121452	1009317	137693 e.terrenz	851000	19633	38039	568094	6656970	10000	10683000	UUTUS	42690	64623	0000011	23104	14701	25247	37453	17512	33840	14657	17309	17064	16157	190817	107568	06226	24192	120000	92290 92790	92290	92735 250820	602309	1532658	28476	13742	41327		28064	16216	53271	43670	25013 23958
PVENT	5	5	តត	57 Q	5 5 5	55 5	55	55	5	7 7	5	7 5	ត	r 15	5	5	5	5 15	5	5	51	25	7	5	33	5 3 5	3	33	z		5	z 7	. J.	ಸಸ	75	र र र	55	55	8 8	18 1	8 18	8 ¥	18	8 8	52	22	5	5	5	ò	8	88	8 ¥	8	88

Score 0.08231 0.06293	0.1038 0.40859	0.15097 0.0619	0.00015	0.02244	0,06848	0.22572	0.02437	0.00713	0.06933 0.00146	0.44618	0.16954	0.29337 1.2E-09	0.10243	1960.0	0.11406	0.062	0 0,20482	0.00019	0.01144	0.00029	0.095	0.00016	100000	0.00288 0.00023	0.00011	0.00056	4,4E-U3 0,00039	2.4E-05 4.4E-05	7.9E-10 0.062	0	0.04101	3.9E-08 0.07175	0.062	0.15745 0.6685	0.08452	16310.0	0.03757	0 0.18541	9.2E-05 0.00012	0.00035	0.00048	0.00012 1.9E-08	9.6E-07 0.0002	0.00036 0.00061	0.00058
Utility Score 0.1178913 0.00730926 0.0150005 0.00093003	0 0 0.0039064 0.0002422	0 0 0 3137 0.06189545	0.0396003 0.00245522 0.002422 0.00015016 0.0007025 0.002089		0.1044541 0.00647615 0.1044541 0.00647615	0.0637989 0.00395555	10/7/2000 10110000	0.0101566 0.00062971	0.1180522 0.00731924 6 0	0.0025001 0.00015501	0.0017188 0.00010657	0.0051564 0.0003197 0 0	0 0 0	0.0017188 0.00010657	0.0051564 0.0003197			0 0 0.0031251 0.00019376	0.184512 0.01143974	0 0 8006700000 //8040000	6 6 0.0078128 0.00048439	0.0025001 0.00015501	CD150000 0.0009000	0.0463962 0.00287656 0.0037501 0.00023251	0.0017188 0.00010657 0.00178178 0.00010657	0.0090628 0.00056189	0.0062502 0.08038751	0.0003906 2.422E-05 0.0007031 4.3595E-05	• •	0 0	0.031251 0.00193756	• •	• •	• •	0 0 0.021300 0.0001520 0	0.1415853 0.00877829	0.0039064 0.0002422	0 0 00035939 0.00022282	0.0014844 9.2034E-05 0.0019532 0.0001211	0.0062502 0.00038751 0.00034876	0.0078128 0.00048439 0 0	0,0018751 0,00011625 0 0	0 0 0.0032032 0.0001986	0.0037814 0.00035845 0.009766 0.00060549	0.0092972 0.00057643 0.0144536 0.00089612
Scure 0.117887785 0.015	0.00390625	0.998313665	0.002421875	0	0.104450896	0.063796901	0,003203125	0,01015625	0,118048688	0.0025	0.00171875	0.00515625 0	0	C/ 00+C00070	0.00515625	• •	¢	0.003125	0.184506824	0	0.0078125	0.0025	0.015	0.046394682 0.00375	0.00171875	0.0090625	0.00625	0.000390625 0.000703125	0 0	1010200	0.03125		• •		0	St1185111.0	0 0.00390625	0.00359375	0.001953125	0.00625	0.0078125	0.001875	0.003203125	0.00578125 0.009765625	0.009296875
MHRS 1150 192	200	0 6783	300	070	1063	746	41	130	1151	33	22	99		2 12	99 99	•	0	9	1500	g o	0 1	8:	6 1 2 6	55 8 4	2 2	911	6 G	v. 6	• •		356	0 0			• • •	740	0	¢	88	6I 52	8	72 100	5 7 8	00	41
Score 0.062 0.062	0 0	0.062	0.062		0,062	•	0.00	• •	0.062	0	0	• •	0	0.062		0.062	0.062	Ċ	•	0.062	• •	• •	• •	• •	• •				0.062	-	790'O	0.062	0.062	0.062	0.062	790'n		0,062	00	• •	0	• •	00	0 0	9 0
Score 1	c o .	- 0	- • •		c		- 0 (• •	- 0	o -	- 0	• •	۰.				1	Ð	0 0		0 0	0 0	• •	0 0	0 0				0 -		- 0 1	o -				- 0 -		-	• •	• •	•	• •	00	• •	
1 I SSOLIOS	c 0	- 0	- 0 1				- 0 1	• •	- 0	•	- 0	• •			•	-	-	0	0	o -	• •			0 0			0 0				- 0	o -			· ·	- 0	° 0	1	• •	00		• •	0 0	0 0	0 0
Score 1 0.00325 0	0.00325	• •	0 0						e e	0.00975	• •	0.00325	0.00325				0	0	0						• •				• •		0.00325	0.00325	0 0.00355	0	0.00325		0.00325	0.00325	• •		•		• •	• •	
Score 0.083333	0.083333		0.25					• •	• •	0.25		0.083333	0.083333		• •		0	0	0		• •	•		• •							0.083333	0.083333	0	0.083333	0.083333		0,083333	0.083333	• •	• •	•	• •	° e		
1 Tasyn	- 7		•							£		-	1																		-	1	-	-	-			-							
Score 0.0065 0	0.00975	c 0	0.00975	6900'n				• •	00	0.0195	• •	0.0065	•		0	0	0	0	Q ·	• •	• •	• • •		0 0	• • •			0 0	00		22500.0	0.0065	0	0 0125	0.00325	> 0	0.00975 0.0065	0.00325	• •	• •	•	00	0 0	• •	0 0
Score 0.166667 0	0.25 0.25		0.25	0		• • •	000	• •	0 0	0.5		0.166667	• • •		0	0	0	Ð	0		• •	• • •		• •	00						0.25	0.166667	0 2	0	0.083333		0.25	0.083333	• •	0 0	0	0 0			
	е е		m 1	4						4		3			•	4																14	4	-	-		m n	-							
Score 3 0.00322	0 0.039	• •	0.03	0				0.0065		0.039	0.039	0.0097	•	• •	0	0	0	•	•	• •	• •	• •	• •	• •	•			• •	• •			• •	0		•		9 0.0032	0	• •		•	• •	00	• •	• •
L Score 0.08333	• -			60				0.16666	• •			0.25			0	6710	¢	¢	•		• •	• •		• •					• •				÷ -		•	• •	0.0833	•	• •				• •		
• [<u>rts</u>	282 S	375	493 5	+				1625 2		NG	W5	Ð			125	r 100	375																	•	815	8125	-	5653							
Scar 0 0	45 0.00910 0	14 0.0009	46 0.01811 0			0	• •	86 0.00015 0	• •						00003	0	0100.0	0	0	00	• •			• •				0 0		, ,			00		10910.0	0 37000.0		65 0.01206	00	00	•	00		00	
Contraction Contraction	0.2600806	0.0267857	0.5175695				• •	0.0044642	00	• • •	• •	0 0			0.0089285	0	0.03125	ð	0	• •	0 0			00			• •	• •				• •	• •		0.457661	0 0.0223214	0 0	0.3447580	• •	e e	. 0	00	0 0	00	
DAVREST 0 0	28 0	90	6 0 1			. 0		- 0	00	• • •		• •			7 5	9 0	٢	0	0		• •		• •				• •	0 0			• •				4	.		34		0 0		• •	00	. 0	0 0
Scure B	a e	• •	0.01532125			0.0098903	• • •	• •	e 0			• •		• •	9.6031E-07	0	4.7693E-06	0	0	• •			• •					. .			• •	• •	-			0.00674461	• •	0.00766214		0 0		• •		• •	• •
Score 0		• •	0.851180798			0.549461288	• •	• •	e e						5.33504E-05	0	0.000264959	¢	¢	•	0 0						• •			,	• •	• •	•			0.374700559	• •	0.425674214	• •	• •		• •	• •		• •
AVLOST 0	• •	• •	8 -			- R		• •				• •	•	• •	e .	90 0	5		•		• •						c 0	• •		•			-			9 5	* •	61		• •		• •	• •		• •
Score [1 0 0	00	• •	0.00650641			00302885	0 0	0.00302885 0	0 0			0 0		• •	0	0	0	0	•	0 0	0 0			0 0				• •		,		0 O					0 0	0	00	00	Ģ	00	0 0	• •	• •
9 0 0	• •	• •	0.15491453 (0			072115385 0	• •	0.072115385 (0 0			0 0	• •	• •	0	0.162393162 (0				• •		• •	• •			00	• •			• •	0 0				0 0		0	• •			00	• •	• •	• •
DAYHOSP 0 0	e 0	• •	- 7	• •		0	0 0	- n 0				00	. 0		•	5 7 7	0	•					• •	0 0			• •	• •		•	• •	• •				c c	• •	0			. 0	00	00	0 0	c c

Scure 0 0						• •	• •	e 0	• •			• •			0		• •		•							0.0035	* *	• •	• •		• •			• •		• •	¢ ¢	¢	• •	•					•	• •
Cort Scort					• •		• •				• •	• •			0		• •		•	• •	•					0 ⁰			• •				• •	• •		• •	• •					• •			0	• •
VILLIA																																														
PARTIAL DIS			• •	• •	• •	• •	• •		-		• •				•	* *	•		•							e -		8 8	•				• •	• •			• •	• •	• •	•	••			00	0	0 0
Score 0 0			• •	0 0	0 0			0 0			0 0			• •	0	• •	•		-					• •		• •	00	c 0	• •	> a (• •	• •	0 0						0		0 0		00	0	0 0
care 0				0 0	0 0	00	00	0 0			0 0	0		0 0	0	0 0			5	0 0				0 0			0 0	0 0				0 0	• •	0 0				. 0	• •	0		0				0 0
۶ ۲																																														
TOTAL DISABILIT 0 0				0	0 0		0 0			0 0	0 0	•		00	0	00	00		0	0 0			• •	0 0				00	0			00	0 0	0 0		• •	0 0			0		0 0		0 0	¢	00
Score 0 0				• •	• •		0 0	• •				e d	• •	• •	0	• •	•		•	• •				• •	-	0.209	• •	• •	• •		• •			• •			• •	• •	• •	•	• •	• •	• •			
Score 0							e e	• •			• •	-			•	• •	•		•	• •				• •		0.59375		• •	• •			• •	• •	• •		• •	66		• •	¢	• •	• •			c	
TOTFTL 0 0						• •	0 0	• •	• • •	• •		0		• •	0			• • •	•	¢ ¢				• •		• •	• •		• •		• •	• •	• •	• •			• •		• •	•		•				
Score 0 0	• • •				a o		0 0					0		• •	•	• •	•			c c	0	3.0699E-08	0 O	0 0		0.03063022	• •	0 0	• •		0 2.6253E-07			• •			• •		• •	c	• •	• •		• •	0 2.7989E-07	00
Utility 0 0						00	• •	• •	• • •		00	0		00	0	00	•		ə o	• •	0	6.5317E-07		0 0	0	0.6517069	0 0	0 0	•		0 5.5858E-06	00	• •	• •		• •	•		• •	e	• •	• •		• •	0 5 9552F-06	00
Score 0				0 0	° °	a a	a a				0 0	0		0 0	. 0		• •		Ð	e e	0	2.79874E-06	c 6	00	• •	0 0.843069108	0 0	• •	•		0 2.3934E-05	0 0	0 0		• •	0 0	0 0	• •	• •	0		• •	50	• •	2.55167E-05	• •
0 0 0				0 0	0 0		0 0	0 0		5 0	00	0	• •	• •	• •		0		0	• •	9	1682	0 0	•		0 2855000	• •	0 0	0 0		0 11662	• •	• •	• •		• •	0 0		• •	0	• •	•		e 9	12270	0 0
Score 1.7218E-06 2.5678E-09	3.2582E-08 9.3973E-10	0.03631707 0.02880193 9.02880193	1.5372E-05 3.618F-08	2.2016E-08 5.028E-09	3.482E-10 3.482E-10	5.4029E-07 3.482E-10	1.6639E-08 2.0687E-06	1.5072E-07 0.00516319	5.3431E-05	3.012E-08 1.2158E-08	0.00066599 2.9916E-08	1.568E-09	4.3226E-06	2.3245E-08 5.4886E-08	0.02019437	1.3449E-07 3.482E-10	0.00216609	1270107010	3.482E-10 0	2.8208E-08 0.00538131	0.00038461	1.6969E-05	9.8931E-10 0.02144977	5.5552E-09	0.00471284	5.2849E-06 0.13815163	1.4242E-07 0.02168723	6.4254E-10 2.0274E-05	2.8318E-06	20-31269.6	1.41/2E-08 1.05E-06	5.7404E-10 3.482E-10	0.00048991	8.0289E-09	3.979E-07	0.0403108 6.9131E-08	1.7488E-09 3.497E-10	3.482E-10	0.02019437 0.00620898	1.7372E-09	0.00045019	11505200.0	2.2716E-06	0.02019437	0.09413699	1.8861E-06 0.02865242
Utility 7.29586E-06 1.08806E-08	1.3806E-07 3.98192E-09	0.153885904 0.122042091	6.51349E-05	9.32873E-08 2.13049E-08	1.47542E-09 1.47542E-09	2.28937E-06 1.47542E-09	7.05048E-08 8.76552E-06	6.38664E-07 0.028877942	0.000226402	1.27627E-07 5.15164E-08	0.002821981	6.64414E-09	2.90555E-07 1.83162E-05	9.84939E-08	0.085569363	5.69862E-07 1.47542E-09	0.009178334	0.085569363	1.47542E-09 0	1.19526E-07	0.001629715	0.177394273 7.19032E-05	4.19199E-09 0.090888843	2.35388E-08	0.019969678	2.23938E-05 0.585388247	6.03477E-07 0.091895059	2.72264E-09 8.59073F-05	1.1999E-05	0.09120623	6.00509E-08 4.44899E-06	2.43237E-09 1.47542E-09	0.0020759	3.40209E-08	1.686E-06	0.170808454 2.92929E-07	60-36660F'L	1.47542E-09	0.085569363	7.36114E-09	0.004720531 0.001907574	0.010141689	7.03159E-05 9.6254E-06	0.085569363	() 1000024444	7.9919E-06
16-05 16-05				58	88	¥ 8	6 8	5	806	53	2954 F.417	803	8E-05	5E-07 1E-07	0.125	21E-07 58E-09	026612	0.125	158E-09	137E-07	502184	110503	370284	68E-08	110816	165E-05 7661718	84E-07 759514	42E-09	11E-05	\$18651 \$31252	22E-08 52E-06	83E-09 58E-09	186355	66E-08	12E-06	135611 02E-07	84E-08	58E-09	0.125	3E-08	18291	6968	E-05	0.125 0.125	1011	E-05 0576
Scon 1.121 1.6722-	2.12184E-07 6.11981E-09	0.1746767236	0.00010010	1.43373E-	2.26758E- 2.26758E-	3.51852E. 2.26758E.	1.08359E 1.34716E	9.81561E	0.00034	1.9615	0.0043	1.02114	2.8149	15137		8.758	0.014	1977	2.26	15	0.00.0	9-2-10 9-000	6.412	3.61	2.0	3.44)	9.274	1.181.1	1.844	0.000	9.229 6.837	3.73	0.003	5.228	1.59	0.238	1.138	2.267	0.010	1.1313	0,00292	0.0154	0.00010		10.007	1.22827
DAMCOST Score 97718 1.1215 27243 1.6722	54250 2.12184E-07 17694 6.11981E-09	603109 0.216480236 472177 0.174676723 412415 5 0070479 0.6	121744 0.00010010 55197 5 35517F.0	49985 1,43373E- 34160 3.27435E-	10000 2.26758E- 10000 2.26758E-	85000 3.51852E. 10000 2.26758E.	46949 1.08359E 99732 1.34716E	71000 9.81561E 185441 0.022176	135417 0.000347	53394 1.9615 43561 7.91755	163090 0.0043 53320 1.94822	22399 1.02114	101820 2.81491 107820 2.81491	20575 1.5137 51972 1.5137	20000	69751 8,7587 10000 2,2679	175992 0.014	20000	10000 2.26	52680 1.1 185889 0.07	157072 0.003	6/3/18 0.246 122829 8.000	18148 6.442 256393 0.132	35208 3.61	184452 0.030	110026 3.441	70379 9.274 266071 0.133	14190 4.184	103178 1.844	111956 0.0000	92290 6.837	13597 3.73 10000 2.267	159725 0.003 158454 0.003	39113 5.228	81644 2.59	655500 0.238 62465 4.502	23448 1.138	10000 2.267	200000 200000	23384 1.1313	158724 0.00723 158798 0.00292	177081 0.0154	122584 0.00010 100759 1.47932	200000 200000	100501	98718 1.22827

Score	0.00064	0.00121	9.4E-10	0.02885	9.2E-06	1.5E-05 2 20007	2.2E-08	5.8E-05	0.00073 3.5E-10	0.04704	0.00068	0.00024	2.1E-06	0.00048	0.00015	0.00035	4.8E-05	0.00016	1.6E-09	0.00069	0.0003	0.00039	0.02019	3.5E-10	0.00275	0.00024	0.00185	0	100.0	0.0038	0.10375	0.00284	0.02155	0.06365	0.00369	0.06216	0.38133	0.00019	0.025/1 6.4E-10	9.3E-05	2.8E-06 0.02539	9.7E-05	0.06217	0.00037	0.00246	0.00376	8E-09	5.2E-09 4E-07	0.04147	6.9E-08	0.00365	0.00052	0.00669	1.75-09	0,00093	0.00239	8.9E-05	0.02136 0.02019	0	0.17564	0,03937
Utility Score	0.0103128 0.0006394	0.0195319 0.00121098	0 0	0.0041408 0.00025673 0.0007813 4.8439E-05	0 0	0 0		0.0009375 5.8127E-05	0.00117191 0.00072659	0.7587392 0.04704183	0.0109379 0.00067815	0.0039064 0.0002422	0 0	0.0015626 9.68785-05	0.0014844 9.2034E-05	0.0056252 0.00034876	0.0007813 4.8439E-05	0.0025001 0.000155010	0	0.0111723 0.00069268	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.0062502 0.00038751	•		0.0093753 0.00058127	0.0039064 0.0002422	0,0297666 0,00184553	0	0.0161724 0.00100269	0 0	0.3269076 0.02026827	• •	0.0016407 0.00010172	0.0265634 0.00164693	0.0070315 0.00043595	0.0025001 0.00015501	0.0007813 4.8439E-05	0.0031251 0.00019376	0 0	0.0011719 7.2659E-05	0 0 00000000000000000000000000000000000	0 0	0.0027345 0.00016954 0.00016954	0.0059377 0.00036814	0.0396003 0.00245522	0.052685 0.00326647 0.0003125 1.9376E-05	0		0.0187506 0.00116254	••	0.05884 0.00364808	0.0084378 0.00052314 0.0002344 1.4532E-05	0.0078128 0.00048439	• •	0.0078128 0.00048439	0 00076049	0.0014063 8.719E-05	0.0187506 0.00116254 0 0	0 0	10055000.0 10054500.0	0.0156255 0.00096878
Score	0.0103125	0.01953125	0	0.00078125	e	0	0	0.0009375	0.01171875	0.758732975	0.0109375	0.00390625	0	0.0015625	0.001484375	0.005625	0.00078125	0.0025	0	0.011171875	0.00484375	0.00625	-		0.009375	0.00390625	0.029765625		0.016171875	0	0.32690012		0.001640625	0.0265625	0.00703125	0.0025	0.00078125	0.003125	0	0.001171875	0.062265026	0	0.002734375	0.0059375	0.039598963	0.052683286	0		0.01875		0.058838154	0.000234375	0.0078125		0,0078125	0 0.012265625	0.00140625	0.01875	0.01010	0.00546875	0.015625
MHRS	125	181	132	250	Ð	8 :	<u>e</u> o	0	580	, ដ	150	•	2962	₹ \$	q •	100	8	2 F	0	8	32	14	• (79 8	3 0	•	• •		120	009	381	¥,	6	. 0	2000		31	340	8 8	32	0 9	417	•		732	0 ¥	8	200 200	640	40		0	•	0	108	۳ ۱۹	0	0		157 18	240
Score	• •		e i		•	• •		0	0 0	• •	•	•	•	. .	• •	ø	• •		0	•	• •	0	• •		• •	.			0		0	• •	• •	0.062	• •	0.062	0	• •		0	0 0	0	0.062		ò		0		0 0		Ģ	0 0	0		0		, o	• •	4	20	0
Score	0	0 0	0	• •	0	•	• •	0	• •	•	0	0	• •			0			Ð	•		0	0 0	• •	• •	0			• •			• •		-	• •			• •		•	• •		~ 0	• •	0	• •	•		0		c	e e	. 0	0 c	•	• •	, 0		¢		0
TOTLOSS	•	• •	0		0	•		0	• •		0	•	0 0			0	•	• •	•	•	•	• •	¢ i	ə c		0					, 0				<i>ф</i> .					, 0					Ū	•	. 0		. 0			• •		• •		• •	, o	00		a o	0
Score	•		0	• •	0	•	• •	0	• •		0	•	• •			•	-		•	•	• •	0	-		•	•			•		0,00325				0.00325		Ð	e :	-	•	• •	• •	0 0176	0	0		•	• •	¢	• •	0	• •		• •	•	• •		• •		0.00325	0
PL Score	•		•		0	•	• •	0			•	•	• •	• •		0	•	• •	0	•		0	• •	• =	• •	•	• •		•		0.083333	•		• •	0.083333			•		•			0	0		• •	•		0		•	• •	•	• •	•	• •	• •	• •		0.05580.0	8
ore HYS			0	• •	0			•			0	0	0			0	0					. 0	0		• •	0					0325 1						, o	0			0 0		0	0	0		. 0		0	0 0	. 0	0 0	, ,			• •		0 0		0065 1	0
Score S	0		0	0 9	0	0		0			0	•				0	0		0	0		0	•		, o	0			0	• •	083333 0.0	0 0			• •			0			• •	, o	0	0	0	0 0	0	• •	0	0 0		• •	0 0	• •			5 0			0.166667 0.	0
OLSPI.	_																														1												-	-																2	
Score	•	• •	0	• •	•	•		•	• •		c	¢	•	-	• •	0	0	• •	•	•	• •	•	• •	• e		0	• •		0	• •	0.00325	•	• •		• •			•		•	• •	• •	• :	• •	•	• •	•	• •		0 0		• •		• •		• •	• •			0.039 8	0.00975
PL. Scare	÷ .		0		•	•		0	• •	• •	0	e			• •	0	0		-	•		•	•	• •		¢			•	• •	0.083333	•	• •		• •	• •		•	• •	• •		• •	• •	• •	0	• •		• •	Ċ	• •	•	•		• •		• •					0.25
ore FLSI]		0					0									•		. 0	0			0			0	c 0		0		38177 1	0			0			0					0	0	0	0 0			. 0			0.0		0 0	, .			00		77306 5	0 0
ano S			0					0				•	e 1										0			•			0		90776 0.02				6			0	• •		0 0		0 120005 001	Into coenci	0			0 0	. 0	0 0	. 0	0 0		0 0				0 0		558858 0.01 0	0
2FST Sc					_	_		_			_	_		_		_	_						_	_							18 0.610				_			_					(60C70 0												_				3 0.50	
TAVI -]		0	00		0											Ū						-						Ŭ		97 37	E-07 (_			-																						19112	-
Ser Ser			-									c	Ģ .										Ļ			ç					74214 0.0070	IE-05 3.913			-										_	_										_				00559 0.006	-
Sen Sen	•		0	• •		•			-			e	•	-		Đ	e			•			-			÷	••		-		0.4256	2.1739																												1476.0	÷
- I DAVIA	-		6	• •		0			0			0	0	•		0	0				• •		0		• •	0	• •	•	0	• •	692 19	2692 2	* *		•	• •		•					•	0 0 0		•				• •		• •		•		•		0 0	•	1 25	-
Second	0		0	00		0			0		0	0	e	•	• •	ð	¢	, ,		0	00	• •	0	ф с		0	• •	•	0	0 0	\$692 0.0028	7692 0.0028			0			0		0	00	0	0	6134 U.OU.58		0 0		00	• •	00		0 0		0 0		0 0		• •		5 0.00.	Ð
Contraction of Contraction	0	• •	0	o c	• •	o	• •		•		. 0	0	0	0 0		•	0	• •		¢	• •		0	• •		e	0 0	•	•	• •	0.067307	0.067307			0			0	• •		0 0		0	0.091340	0	•		0 0			• •	• •		• •		•		0 0	,	0,12.	0
1308710	0	• •	•	• •		0	e e	• •	•		• •	0	0	0	• •	• •	0	0 0	• •	0	00		0	• •		0	• •	•	•	• •		10	• •	• •	0	• •		o	• •	• •	•	• •	•			•	• •	00		• •	• •	0 0	• •	0	0 0	0			5	<u>7</u> 0	0

Score	0		0	0		0					0			• •			-	• •			• •	0	0	0	• •		•			•		0		•		0	• •	0					• •	0 0	• •	a			-		•	0.014	0.0035 0
																							-																														
Score	8		•	0	• •	•	• •		•		8			• •	•	• •	•	• •	• •	• •		0	0	6	. .	• •	•			• •		•		•		0		¢	• •	•		•	• •	• •	• •	0	• •		-		0	1	0.25 0
RUTY															•																																						
PARTIAL DISA	0	• •	0			•			•			• •		• •		* *	-		• •	• •			7 0	0	• •		•	• •	. ¢	•		•		•		•		•	• •	•	• •	•		•	• •	۰	• •		•	• •	0	s	- 0
Store	o	• •	0	0	0 c	•	• •		0		0			• •	. 0					• •	• •	0		0	0 0		0	0 0	, o	0 0		0	5 0	0		0		0	0 0	0	• •	0 (0 0		0	0 C		-		o	0.014	00
ore 0	0			¢. o	0 0	0			0	• •	0	0 0					. 0	0 0	. 0	0 0		•		0			0							0		0	0 0	0		0		0		0		0	• •		0	00	0	-	0 0
ž				Ð																																																	
L DISABILITY 0	0	0 0	0	- 0	0 0	0			0	0 0	•	• •	00	00	. 0	0 C		• •	. 0	0 0		0 1		0			0	• •		0		¢	• •	0 (0				0		0	• •	•		o	0 0		o	0 0	0	7	
TOTA			_								_							_		_		_		_	-		-					-		_		_		_		_		_			.8				_			75	76
Sco	0								0															0												Ū									2 5	-			-		-	5	19
Score	0	• •	•	• •	• •	-	-	• •	•	• •	0	•		-		• •		• •	• •	•	• •	•	• •	•	• •		•	• •		•				0	• •	0		•		0	• •	•	• •	•	0.5312	e	• •	• • •	-		6	0.7812	0.5
TOTFTL	•	• •	÷	• •	• •		•		¢	• •	•	• •		• •	• •	• •		• •		• •		•	• •	•	• •	• •	•			•	• •	•	• •	•	• •	e		•		0		•	• •	•	9 M	0	• •		¢		c	2	- 0
Score		00	0 0	0.00206877	0 0	0	• •	• •	0	• •	0	0 0		0 0	• •	6.4315E-07	1.6003E-06	0 0		0,	0 4.4495E-07	0	2.122E.09	•	0 0		ø	1.6003E-08 0	• •	0	6.8337E-08 0	5.1634E-07	ə o	0		0	0 0	. 0		0		9.1465E-07	• •	c (0 5.7509E-07	• •	• •	÷	• •	0 Q (• • •	0 0.02887347	0.00175224 0
Utility 0		• •	0 ¢	0.04401634 (• •		•		0		. 0	0 0		0 0		1.3684E-05	3.4048E-05	0 0		6	9.4671E-06	0	0.03874497 1 4.5149E.08	0			0	3,4049E-07 n		0	1.454E-06	1.0986E-05	• •	0		0		0	• •	0	• •	1.9461E-05		0 0	1.2236E-05	0 0	0 0		• •			0 0.61432918	0.03728176
Score 0	0		Ð	0.155178917			• •		o	• •	. 0	0		•		5.86309E-05	0.000145866	0 0	5 0	6 0	4.05637E-05	0	0.139405908 1 93458F-07	•	0 0		0	1.45896E-06 0	o o	0	6.23012E-06 0	4.70708E-05		0		0	0 0	. 0	• •	0	0 0	8.33788E-05	9 9	9	0 5.24269E-05	0	• •		0	• •	0	0.822022783	0.134917985 0
INJCOST 0	0	00	Ð	0 0	•		• •		c	0 0	. 0					22440	38190	• •		•	17410	0	420000	0	• •		0	168		0	9960	19341		0		0		. 0		0		28005		0 0	0 20825	0	¢ ¢	, e	o	0 0	0	2460040	373000 0
Score 0.0101635	0 .3523E-09	<pre>.02161333 .8613E-09</pre>	0 .7402E-07	.0958E-07	.3287E-08	3502E-08	U10304383	10-35-55-02	.7726E-06	11449E-08	3.482E-10	1073E-05	11594E-08	1017F-07	0169E-08	17559E-09	0.1417052	(8311E-09	12603E-08	L112E-10	02-187574	5.5684E-09	108160441	0148E-09	9.716E-10	1871E-08	0763E-10	2.31E-06	8.597E-10	3.482E-10	02316207	02799176	L1299E-08	L2007E-08	6.1086E-10 6.1086E-10	6.1086E-10	0.09536885	4393E-08	6.5089E-10	15195E-09	5.4315E-09 3.482E-10	07610811	L7701E-07	0.8242E-10	00454748	0 5.5634E-10	1.2888E-10	11276E-10	63777E-10 0	L0627E-07 L3042E-09	0 L.8889E-07	0.11651963	0.00764469
Utility 0.043065687	0 5.73028E-09 1	0.091581916 (1.63616E-08 3	0 7.37365E-07 1	0.324298828 (4.64309E-07 1	5.63002E-08	2.69075E-07	0.43662641	1.30781E-06 3	7.51105E-06	1.03597E-07	1.47542E-09	0.000347476 2	2.18617E-07 5	1.20055E-07	8.54623E-08 2	1.16775E-08	0.600445777	3.31825E-08	1.38148E-07 3	1.74201E-09	0.105405674 6	2.35948E-08	0.418603017	8.53742E-09	4.11696E-09	5.03012F-08 1	2.57471E-09 4	9.78814E-06	3.64282E-09	1.47542E-09	175441890.0 1 FLAFT8890.0	91609811'0	1.27815E-07 -	1.77994E-07	2.58839E-09	2.58839E-09	1 1102010010	6.09868E-08	2.75801E-09	3.60995E-08	1.45403E-08	0.322491979	5.01775E-07 7.50053E-07	4.16278E-09	2.72602E-09 0.400624906 (0 2.35738E-09 2	1.81727E-09	1.749E-(19	2.27867E-09	4.50281E-07 5.52641E-09	0 8.00388E-07	0	0.032392745
Score 0.064483095	B.80686E-09	0.133327341 2.51462E-08	1.13326E-06	0.419006307 7.13596E-07	8.65277E-08	4.135418-07	0.536442954	2.88133E-08 2.00997E-06	1.15437E-05	1.59218E-07	2.26758E-09	0.00053392	7.21076E-08 3.35993E-07	1.84512E-07	1.31347E-07	1.79471E-08	0.68996159	5.09982E-08	2.87516E-09 2.1232E-07	2.6773E-09	3.48745E-09	3.62628E-08	0.518332314	1.31212E-08	6.32735E-09	1.7057/E~08	3.9570TE-09	1.50433E-05	0.00518404 5.59864E-09	2.26758E-09	0.14235297	0.170084854	1.96438E-07 5 79445E-07	2.73559E-07	3.97809E-09 3.97809E-09	3.97809E-09	0.503570179	9.373065-08	4.23878E-09	5.54813E-08	2.2347E-08	0.417023435	7.71178E-07 1.15275F-06	6.39778E-09	4.18963E-09 0.5	3.62306E-09	2.79296E-09	2.68803E-09	3.50209E-09	6.92036E-07 8.49354E-09	1.23011E-06	0.592135158	0.048811816
DAMCOST 192735	21000	263089	72575	924766 67507	44518	61536	2857048	32858	26037	51124	10000	140118	29266	52728	15061	27956	10680000	38847	1001	11126	13001 374451	35233	1934172	24830	17988	10267	11011	100943	16919	10000	326426	154266	53410 64239	57021	14086	14086	1181928	18051	765F1	39746	30192	10/226	72762	18086	100000	13354	1424	11154	13093	67171 20662	73474	2694993	189679 370000
VENTI	3	88	3	33	33	1 2	33	1 1	J	33	3	33	13	33	13	33	1 2	33	13	3	1 1	64	3 3	3	33	13	3	33	1 3	3	33	33	33	3	33	3	33	5 3	33	3	33	3	33	3	1 1	38	38 %	8 8	\$	3 3	69	70	70

Score 0.01016	0 1.4E-09 0.02167	0	0.00012 0.10982	0.00097 0.00043	1.1E-09 0.06279	0.14236	0.00102	0.00066	0.0002	0.00078	8.2E-05	c000:0	0.00039	0.00013	0.09728	0.19062	7.8E-09	0.00097	7.8E-05 0.00043	0.10138	0.07648 0.16939	0.13076	2E-09 0.00477	0.00232	0.04063	0.02409	0.00276	5.35-05	0.07231	0.0498	0.00011	5.3E-05 5.3E-05	0.00017	5.8E-05 6.1E-10	0.09862	\$1000.0	5.3E-05 5.3E-05	8.5E-09	0.062	0.1176	0.00109	0.00034	0.2947	0	0.00047	0.00019	4.1E-10 0.0502	u 1.1E-07 1.3E-09	0	0	0.55271 0.26123 0.08697
Utility Score	0 0 0 0 0.0009375 5.8127E-05 0.0082834 0.00051345	0 0 0++++++++++++++++++++++++++++++++++	0.0019532 0.0001211 0.0035939 0.00022282	0.0156255 0.00096878 0.0068752 0.00042626	0 0.00.46877 0.00029063	0.0050783 0.00031485	0.0164068 0.00101722 0.0060158 0.000177298	0.0106254 0.00065877	0.0032032 0.0001986 0.1109586 0.00687943	0.0125004 0.00077503	0 0	0.0021094 0.00013079	0.0062502 0.00038751 7 9175 06 4 94105 06	0.0021094 0.00013079	0.2574487 0.01596182	0 0	0 0 0 0000 0 0000 0	0.0156255 0.00096878	0.00125 7.7503E-05 0.0068752 0.00042626	0.0027345 0.00016954	0.2336231 0.01448463 0.0143755 0.00089128	0.0040626 0.00025188	0 0	0.0373536 0.00231593	0.6553749 0.04063324	0.0316502 0.00196231	0.0316502 0.00196231	0.0008594 5.3283E-05	0.0008594 5.3283E-05	2/05054689 0.000339020	0.0017188 0.00010657	0.0008594 5.3283E.05	0.0028126 0.00017438	6 0 0	0 0 0	0.0023438 0.00014532	0.0008594 5.3283E-05 0.0008594 5.3283E-05	0 0	0.0082815 0.00051345	0.0023438 0.00014532	0.0175006 0.00108504	0.0055471 0.00034392	192820000 0.0002875000	0 0	0.0076565 0.0004747	0.0031251 0.00019376	0.8097027 0.05020157		e e	0	0 0 0.0093753 0.00058127 0.0039064 0.0002422
Score	0 0.0009375 0.003375	C7T272NI'N	0.001953125 0.00359375	0.015625 0.006875	0.0046875	0.005078125	0.01640625	0.010625	0.003203125	0.0125	0 0000 12976	0.002109375	0.00625	5/16012000	0.257442152	0	0 0034375	0.015625	0.006875	0.002734375	0.233617021 0.014375	0.0040625	0 076197871	0.037352412	0.655367209	0.031649128	0.031649128	0.000859375	0.000859375	0.00546875	0.00171875	0.000859375	0.0028125	C/ 66000.0	0.00076176	0.00234375	0.000859375 0.000859375	0	0 0.00828125	0.00234375	0.0175	0.005546875	0.00625	4	0.00765625	0.003125	0.809697423	• •	-		0 0.009375 0.00390625
MHRS	212 70	200	112 0	106	0 X	3	200	; c	33	210	1	<u>8</u> 4	1106	0	5 I	8	- ;	1785	140		30 102	16	88 %	1700	184	70	853	474 2687	117	405	= :	= =	600	P 7	= :	38	2 C		30	Ξ	10	o j	90 Q	5	11	121	8 8	96 96	111		
Score 0	000	•	• •	• •	0 0.062	0	• •		0 C	• •	• •		00	• •	0.062	• •	0 0	• •		0.062	0.062	0	• •	. 0	• •		0		•	. .	•		0		• •	• •	0 0	0	0.062	•				e		0		00	c		0 0.062 0.062
Score 0	• • •	Þ	00	0 0	e -	• •	• •	. 0	0 0	• •	• •	• •	• •	a q			• •		• •	-	- 0	0	00		0 0	• •	0	• •	0	• •	0	• •	0		• •		0 C	0	- 0			• •		e	• •	0		00			o - -
TOTLOSS	• • •	¢	0 0	0 0		- 0	•		0 0	• •	0		0				• •						• •	• •			0	• •	0				•		•		• •	0	~ 0	0	• •	• •	• •			0		00	, c	•	
Scure		•	• •		• •	• •			• •		• •	• •	• •	• •	• •	• •	• •		• •	•		0.00325	• •	. 6	•		•	• •	0		•	• •	•	• •	•		• •	0	• •	0		0 0		4	• •	•		0 0			0.00325 0.00325
0 gene		¢	e e	e e	• •	• •	• •	• •	• •		• •	• •	•		•		• •		• •	•	• •	0.083333	• •		-		0		•		•	• •	•		• •		• •	÷	* *	•		• •	• •	c		•	• •		-		0.5 0.083333 0
ASVH .																55						55 1							;	8					25																 -
are Scar		•	••	• •	• •					, e	••				•	6667 0.00						6667 0.00					0	• •	0 0	0 0 0000 0	÷ •	 	0		3333 0.003		00		0 0 0 0	0						0		00			1 0.00 6667 0.00 0 0
- Ispl																2 0.16						2 0.16								1 0.08					1 0.08																5 2 0.16
Score 0		•	0.00975	• •		0.039	• •		• •		• •		0 0		0	0.0065	•		• •		0 0 DN975	0.039	• •		•	0.0195	•		0.039	• •	0	• •	•	• •	•		• •			0		•				0		•		,	0.039 0 0
Score		•	0 0.25	• •	• •	-	• •		• •	• •	•	• •	• •			0.166667	•		• •		0 520	-	0		•	0.5	•		-		•		-	• •	• •		\$		• •	0	• •	•		e	• •	0				•	
FLSPL			e			5										4						n no				4			ŝ																						v;
Score ()	000	0	0 0	• •	• •	• •	• •	• •	• •			• •	•		0.01931918	0.0180765	•		• •	• •	0 07153551	0.00015625	• •	• •		• •			0,01009073		ð	c 0	•	0 0	• •		• •	• •	• •	0.02004941	• •		0 0	c		0		00	, e	•	0.0007812
Score 0	000	0	c 0	• •	00	• •	0 0	• •	• •	00	0 0		0 0	ə o	0.551976574	0.516471449	c (00		0 0.615300146	0.004464286	0 0	• •	0 0		0	0 0	0.288306452	• •	0				0 0		0 0	. 0	0 0	0.57284041		6	0 O	¢		0	• •		, c	5	0.022321429 0 0
DAYREST 0		¢	00	• •	• •		• •	• •					•		187	- 8	• •			• •	0	-	• •		0 0		0		30		0	0 0	• •	0 e	o d	, 0	• •	Ģ	• •	244		0 (•	• •	• •	, ,		~ o o
Score		•	0 1.06724741	• •	• •		• •		• •		¢	• •	•	• •	•	0.01242827	-		0 0	0.0098903	0	0	-		• •	0 1.2481E-07	¢	• •	8	0 0,01433167	0	• •			•		• •	•		0.01526443	• •	÷ (0.0127665		• •		• •			,	
Score 0			0.402633935	÷ •	• •	• •	•		• •		•		0	• •	\$	0 690459388	•			549461288			• •		• •	0. 93396E-06	•	• •	0	0	•	• •			•	• •	• •		• •	1211-208+8		•	0.70924999	¢		0	• •			2	
AVLOST 0		•	0 18 0	• •	• •					. 0	0	• •	•	• •	0	• ¥	•		• •	30	0	6	• •	• •	•		-		-	- 8	e	• •		• •	•		• •	• •	• •	82	• •	0	o 15		• •	0	• •		, .	9	
Score 1		0	• •	0 0	0 0				• •		0	• •	0	0 0	0	u 1.00540705	0		• •	00444231	0	0	• •		0 0	0.002625	0	• •	0	0 0.00713462	۰	• •					00		• •	0.00603526	0 0	0				0				2	0.00603526 0 0
Score 0	• • •	0	0 0	0 C	0 0		• •	• •	e d		0		0	00	0	0 0.128739316 C	Q (, 0	•	0.105769231 0	0 107005617	0	0		0	0.0625	0	0 0	-	0.169871795 0	ō	• •	. 0	0 0	0					0.143696581 (0	a a			o	• •	0		5	0.143696581 0 0
DAYHOSP 0		•	0 0	•	• • •		0		•		0	• •			0	o 2	0		0 0	- 2	0 (80	0		0	0 ~	• •	a c	0	- 20	0	• •	• •		0	• •	0			61		0	• •		• •	0		0	•	•	6] 0 0

														10																																					
Score	•	• •	•	• •	•	• •	• •					• •		۶ĝ	е Ф	•	• •	• •	• •	-				• •	• •		• •	• •	• •	• •	• •	• •	• •	• •	• •			• •	•	• •	•	• •	•	• •	• •			• •		0 0	
S.														6																																					
,																																																			
Scor	•		•	• •	• •	• •		• •	• •			• •		- 73		• •	• •	• •	• •			• •		• •	•		* *	-	• •	• •	• •	• •	• •		• •	• •	• •	• •			0	• •	•		• •		• •		• •	00	
È																																																			
ABIL																																																			
TDIS	•		•	• •	•	•	• •	• •	• •	-	• •	• •			• •	• •	• •		• •		• •	• •		• •	• •	• •		• •	• •	• •	• •	• •	• •	• •	• •		• •	• •	•	• •	•		•	• •	• •	• •	• •	••		00	
RTIA																																																			
- A																																																			
z															~ ~	~ ~									<u> </u>																										
ŝ																		•••																					-		-		-								
Score	e i	• •	•	• •	• •	• •	• •	• •	• •	• •	• •	0 0		• •	0 0	• •	• •		0 0		0 0	• •		• •	0		° °	• • •	• •		0 0	0 0	00	• •		• •	• •	00	• •	• •	•	• •	• •	• •	0 0		00	• •	00	00	
E11																																																			
ISAB	•				。。										• •	• •		• •	•													0 C	• •	• •						• •	•	• •		• •			• •	• •	0 C		
LALE																																																			
5	_																																																		
E												8								_			_		_									_				42	_				_								
Sce												6.1				-																						6.1					, Ç								
	Ĺ											n																										22													
Scor	-	• •		•	• •	•	• •	• •	• •	• •	• •	95.0		• •	• •	•					• •			• •	•		• •			• •	e 9		• •				• •	0.531			•	• •					¢ ¢		• •		
2																																																			
OTF	•	• •		•	• •	•	• •	• •	• •			n e	• • •	• •	0 0	•	• •	• •	• •	• • •	• •	• •		• •	•		• •			• •	• •	• •	• •		• •	0 0	• •	~ ~	•	• •	•	• •	•	• •	••	• •		• •	• •		
-	1	5	5									5		34	80	60		9			88	80		8 8	01			6	\$			01	yu,		ş			6					6		ş	ş				80	
Score	ę,	0 1200E-		• •		•	• •	• •	• •	• •	• •	129077		0 0	0 1762E	00201	• •	0 172E-	0 0	• •	2778E	0 MAGLE	•	1778E 136E	1792E		• •	396E	-	• •	• •	4015E	0	0	0	• •	• •	12776	• •	• •	ę,	• •	2543E	• •	70875	0		• •	0 0	0 2887E	
		÷	;									8.0.0		6 0.0	6 9.1	0.0		5			9 9 9 9	1		9.0	-i 9			213	r. P			5	÷ 2		- -			30.0					88 -4-		8.5	ń R				1 1	
diffy		0						• •	• •		• •	86703		0 46030	0 524E-0	04277		0 36E-0			229E-0	0	0	229E-0	172E-0		• •	546E-0	0	• •		748E-0	0	•	10		• •	90746	• •	• •	0	• •	SIGE-C	• •	0185	0			• •	0 19E-(
5		7.40	2									0.61		0,03	1.9	0.0		2.45			33	5		26	3.14			2.2	2			1.5	000		6			0.5					9.6		0.0	2				2.7	
۴		5.05	3									2334		2515	E-06	8776		6834			88	5.06		E-06	E-05			E-06	9043			E-05	1003		5			5972					E-07		3189	2				3E-06	
Seat	-	0 20984	0	0 0	• •	0		¢ ¢	• •	• •	• •	8245		12657	36564	01792		00010	• •		81165 23892	0	•	81165	34852		0 0	66048	40056	0 0	00	7472	0	•	0	00	0 0	80816	•	0 0	0	0 0	81852	0 0	54000	0	00	00	00	0	
		~	5									0		e	80	ø		ø			4 10		•	40	-			6	6			9	c		4			e					m		0,	n				-	
COST	ç	0							• •			06966		0	0	1793		0			1825	0		1825 4425	1225			080	<u>.</u>	• •	• •	4574	0	-	0		• •	00000		• •	•	• •	340	• •	5474	0				120	
N		-	-									22		23	•	2		~						" m					•			ч						52								-					
ž	73858	10-31	2E-07	6E-07	2E-09	7E-07	2E-07	04078 3£-08	3E-07	6E-07	1E-06	15785	01428	3E-08 E-06	82867	181-66	8E-07	6E-08 8E-11	1E-06	85-08	9E-06 15163	8E-10	19822	10811	67467	6E-119 36617	9E-05	10811	41,00	6E-07 31839	7E-07	50761 7F_09	2E-05	7E-09	/E-01	3E-09	5E-05 6E-07	04825	6E-05	30766 8E-07	E-07	80-36	1787	1E-06 9E-08	8-B	50026	7E-07	9E-07	80-31)	7E-07	
ŝ	0.024	2 2	1	627	8.826	5.004	3.633	0.003	1.075	1.482	1,119	0.078	0.021	1.05	0.026	0.123	5.561	8.744	F60"6	5.511	0.138	5.203	0.008	0.076	0.053	0.002	2.45	0.076	1.563	6.617	3.140	6.118	2.63	8.88	102	2.491	8023 1205	0.107	1.76	9.000	8.8	2.22	00	3.16	1.0	0.063	1.705	1.92	5.075	2.93	
	-	75	N B	~ ~			2 2	36	20-3	603	5 8 8 8	9819	3556	E-02	1186	971	5 % 2 %	E-07	E-05	2 G	E-06	E-09	6076	1979	2963	E-09	F61+	1979	E-08	E-16 4912	80-3	1605	9421	80	E-86	E-09	2098 F-06	1265	83	3649 F-06	88	E-08	8732	E-08	Egg	5889	E-07	E-07	E-07	E-06	
	5449	E-07 1.	E-01 2	E-06 3	88	2	낖낖	* #	_		P1 1	23	ŝ.	58	89.99	5	1 5	F697	Ħ	13549	63506	20501	03600	32249	22743	26955	00010	6777E	52464	80408	33079	5021	11000	67696	50812 60812	11417	53287	6525	387.17	00130	ğ	38	18	88 F	8	88	2 2	56 5	ž,	96	
Udility	10482449	L7668E-07 1.	000010070 2018E-01 20	53713E-06 3	71452E-06	12062E-06	94848E-(53951E-(0128840 56074E	55616	28202	7420	6	Ŧŝ	8Ŧ	= 2	- R -	23	¥ 7																		~ ~					6	98	2 5	8 Ľ	1 - 1	72	87	19 8	9,	1 2 8	
Utility	0.10482449	7 4.7668E-07 1.	8.67048E-07 2	5 1.53713E-06 3	5 1.71452E-06 3 3.73992E-08	5 2.12062E-06	5 8.94848E-(1 0.0128840 1 3.56074F	1 4.55616	6.28202	6 1.0292 5 4.7420	7 0.33		3.563	11.0 7	9 0.52	22	28.6	5	4 ii 0 m	ගේ ස් ගෙස		 		, w	~ .	e - 9 -	- 6 -	8 6 6 6	20	11 (d 14 (d 14 (d)			i ini i poci	- H 9 9	37	8 M 8 M		i ri e vo	89	3.7	0.43	8 0.17	9 9 1 9		6 G	6 7.03 6 7.24	8.15	1215	2 17 2 17 2 17	
re Utility	173509 0.10482449	09E-07 4.7668E-07 1.	STE-06 8.67048E-07 2	41E-06 1.53713E-06 3	04E-06 1.71452E-06 38E-08 3.73992E-08	18E-06 2.12062E-06	29E-06 E.94848E-0 77E-06 1.53951E-0	646111 0.0128840 19F-07 3.56074F.	35E-07 4.55616	4/E-0/ 1.36491 92E-07 6.28202	BIE-06 1.0292 31E-06 4.7420	509067 0.33	80'0 \$15818	99E-07 3.563 S2E-06 4.448	346287 0.113	983599 0.52	39E-06 2.12	72E-07 2.8 88E-10 3.70	21E-05 3.8	07E-00 4-	08E-05 9. 661718 0.	88E-09 2	0 124441	0 557520	968786 0	26E-08 7 314284 0	160125 0.	023435 0	754895 0. 14E-07 6.	58E-06 2J	28E-06 1.	0 163611	171731 0.0	84E-08 3.	86E-06 3. 18E-06 1	24E-08 1.(06E-09 4.1	054102 0. 65E-06 3.	925477 0.4	345-00 2.	001962 0.	43E-06 3.7	51-60 0-143 57-664 0-143		01E-06 4.89	62E-06 4.44	916741 0.2	85E-06 7.03 37E-06 7.24	38E-06 8.15	03E-07 2.15	14E-06 1.2 77E-05 2.8	
Score Utility	0.151473509 0.10482449	7.32609E-07 4.7668E-07 1.	1.332S7E-06 8.67048E-07 2	2.36241E-06 1.53713E-06 3	2.63504E-06 1.71452E-06 5.74788E-08 3.73992E-08	3.25918E-06 2.12062E-06	1.37529E-06 8.94848E-0 2.36607E-06 1.53951E-0	0.019646111 0.0128840 5.472498-07 3.560748-	7.002352-07 4.55616	2.65482E-07 6.28202	1.58181E-06 1.0292 7.28801E-06 4.7420	0.428509067 0.33	0.129818544 0.089	5.47699E-07 3.563 6.83762E-06 4.448	0.16346287 0.11	0.621983599 0.52	4.80039E-05 5.12 3.622E-06 2.3	4.34472E-07 2.85 5.69488E-10 3.70	5.92221E-05 3.8	0.83/02E-00 4- 3.58942E-07 2-	1.4808E-05 9. 0.67661718 0.	3.3888E-09 2	0 0.05414427 0	0.417023435 0	0.307968786 0	1.11726E-08 7 0.015314284 0	0.000160125 0.	0.417023435 0	0.690754895 0. 1.01814E-07 6.	4.30958E-06 2. 0.002071732 0	2.04528E-06 1- 1.05961E-08 6-	0 16367100970	0.000171731 0.0	5.784E-08 3.	1.7118E-06 3.	1.6224E-08 1.0 6.32306E-09 4.1	0.00054102 0. 5.42965E-06 3.	0 111212020	0.000114845 7.	0.002001962 0J	5.78943E-06 3.7	1.45019E-07 9.43	0.237425358 0.17	7.52901E-06 4.89 2.06175E-07 1.3	6.83762E-06 4.44	0.356916741 0.2	1.08085E-06 7.03 1.11337E-06 7.24	1.25288E-06 8.15	3.30503E-07 2.15	1.91114E-06 1.2 4.4277E-05 2.88	
. Score Utility	0.151473509 0.10482449	7.32609E-07 4.7668E-07 1.	1.33257E-06 8.67048E-07 2.	2.36241E-06 1.53713E-06 3	2.63504E-06 1.71452E-06 5.74788E-08 3.73992E-08	3.25918E-06 2.12062E-06	1.37529E-06 8.94848E-0 2.36607E-06 1.53951E-0	0.019646111 0.0128840 5.47249F-07 3.56074F	7.00235E-07 4.55616	2.1284/E-07 1.38491 9.65482E-07 6.28202	1.58181E-06 1.0292 7.28801E-06 4.7420	0.428509067 0.33	0.129818544 0.08	5.4769E-07 3.563 6.83762E-06 4.448	0.16346287 0.11	0.621983599 0.52	3.622E-06 2.12	4.34472E-07 2.85 5.69488E-10 3.70	5.92221E-05 3.8	0.63/02E-00 4- 3.58942E-07 2.5	1.4808E-05 9. 0.67661718 0.	3.3888E-09 2	0.05414427 0	0.417023435 0	0.307968786 0	1.11726E-08 7 0.015314284 0	0.000160125 0.	0.417023435 0	0.690754895 0. 1.01814E-07 6.	4.30958E-06 2.	2.04528E-06 1. 1.05961E-08 6.	0 1252100910	0.000171731 0.0	5.784E-08 3.	1.7118E-06 3.	1.6224E-08 1.0 6.32306E-09 4.1	0.00054102 0. 5.42965E-06 3.	0.55325477 0.4	0.000114845 7.	0.002001962 0J	5.78943E-06 3.7	1.45019E-07 9.43	0.237425358 0.17	7.52901E-06 4.89 2 06175E-07 1 3	6.83762E-06 4.4	0.356916741 0.2	1.08085E-06 7.03 1.11337E-06 7.24	1.25288E-06 8.15	3.30503E-07 2.15	1.91114E-06 1.2 1.4277E-05 2.88	
COST Score Utility	1450 0.151473509 0.10482449	795 7,32609E-07 4,7668E-07 1. 2010 0 0000000 0 27535000 0 0	351 1.33257E.06 8.67048E.07 2.	630 2.3624JE-06 1.537J3E-06 3	525 2.63504E-06 1.71452E-06 124 5.74788E-08 3.73992E-08	160 3.25918E-06 2.12062E-06	697 1.37529E-06 8.94848E-0 647 2.36607E-06 1.53951E-0	9690 0.019646111 0.0128840 601 5.47249F-07 3.56074F	300 7.00235E-07 4.55616	2014 2.1284/E-U/ 1.38491 819 9.65482E-07 6.28202	231 1.58181E-06 1.0292 990 7.28801E-06 4.7420	1182 0.428509067 0.33	7884 0.129818544 0.08	1610 5.47699E-07 3.560 290 6.83762E-06 4.448	6698 0.16346287 0.113 5236 0.114612915 0.093	6000 0.621983599 0.52	JB/8 4,800,99E-05 3.12 318 3.622E-06 2.1	076 4.34472E-07 2.85 183 5.69488E-10 3.70	5983 5.92221E-05 3.8	-7.0 -2.58942E-00 -7-	0770 1.4808E-05 9. 20000 0.67661718 0.	1843 3.38888E-09 2	0 027777777 0.054144427 0	2704 0.417023435 0 1704 0.417023435 0	5000 0.307968786 0	1263 1.11726E-08 7 6956 0.015314284 0	6900 0.000160125 0.	2704 0.417023435 0	20425 0.690754895 0. 275 1.01814E-07 6.	7225 4.30958E-06 2. 5000 0.002071732 0	019 2.04528E-06 1. 753 1.05961E.08 6.	0 1555100970 1500	7668 0.000171731 0.0	1191 5.784E-08 3.	1097 1.7118E-06 3.	5939 1.6224E-08 1.0 1982 6.32306E-09 4.1	0263 0.00054102 0. 760 5.42965E.06 3.	13739 0.55325477 0.4	3252 0.000114845 7.	H624 0.002001962 0/ 7784 5.44154P-06 7	1464 5.78943E-06 3.74	0109 1.45019E-07 9.43 0676 0.019E76691 0.07	3901 0.237425358 0.17	3347 7.52901E-06 4.89 1937 7.66175E-07 13	1290 6.83762E-06 4.4	1288 0.356916741 0.2	2056 1.08085E-06 7.03 1381 1.11337E-06 7.24	3675 1.25288E-06 8.15 9024 7 912012 05 5.09	9086 3.30503E-07 2.15	2791 4.4277E-05 2.88	
DAMCOST Score Utility	370450 0.151473509 0.10482449	67795 7,32609E-07 4,7668E-07 1. 976910 0.43047049 0.37634E004 0.0	74351 1.33257E-06 8.67048E-07 2.	80630 2.36241E-06 1.53713E-06 3	81828 2.63504£-06 1.71452£-06 40124 5.74788£-08 3.739927E-08	84160 3.25918E-06 2.12062E-06	74697 1.37529E-06 8.94848E-0 80647 2.36607E-06 1.53951E-0	179690 0.019646111 0.0128840 64601 5.47249F-07 3.56074F	67300 7.00235E-07 4.55616	70819 9.65482E-07 6.28202	76231 1.58181E-06 1.0292 92990 7.28801E-06 4.7420	934482 0.428509067 0.33	237884 0.129818544 0.08	64610 5.47699E-07 3.563 92290 6.83762E-06 4.448	426698 0.16346287 0.113 316236 0.141612915 0.093	7216000 0.621983599 0.52	85318 4,80039E-05 5.12 85318 3,622E-06 2.1	62076 4.34472E-07 2.8 3483 5.69488E-10 3.7	115983 5.92221E-05 3.5	59988 3.58941E-07 2.5	100770 1.4808E-05 9. 10000000 0.67661718 0.	12843 3.38888E-09 2	0 190817 0.05414427 0	922704 0.417023435 0 922704 0.417023435 0	785000 0.307968786 0	23263 1.11726E-08 7 176956 0.015314284 0	126900 0.000160125 0. 59091 7.015747 0.	922704 0.417023435 0	10/20425 0.690754895 0. 46275 1.01814E-07 6.	87225 4.30958E-06 2. 155000 0.002071732 0	79049 2.04528E-06 1. 22753 1.05961E-08 6.	0 102081 0.600143531 0	127668 0.000171731 U.C	40191 5.784E-08 3.	77097 1.7118E-06 3.	26939 1.6224E-08 1.0 17982 6.32306E-09 4.1	140263 0.00054102 0. 89760 5.42965F.06 3.	3713739 0.55325477 0.4	123252 0.000114845 7.	154624 0.002001962 0. 89784 5.44154F.06 1.	90464 5.78943E-06 3.7	50109 1.45019E-07 9.43	0.17 0.237425358 0.17	93347 7.52901E-06 4.89 53937 7.06175E-07 1.3	92290 6.83762E-06 4.4	11-0 69920501570 autoret 023	72056 1.08085E-06 7.03 72381 1.11337E-06 7.24	73675 1.25288E-06 8.15 119074 7 917017 05 5.09	59086 3.30503E-07 2.15	78305 1.91114E-06 1.2 112791 4.4277E-05 2.88	
TTI DAMCOST Score Utility	370450 0.151473509 0.10482449	67795 7.32609E-07 4.7668E-07 1. 075010 0.1000000 0.27221E006 0.0	24251 1,33257E-06 8.67048E-07 2.	80630 2.3624JE-06 1.537J3E-06 3	81528 2.63504E-06 1.71452E-06 40124 5.74788E-08 3.73992E-08	84160 3.25918E-06 2.12062E-06	74697 1.37529E-06 8.94848E-0 80647 2.36607E-06 1.53951E-0	179690 0.019646111 0.0128846 64601 5.47249E-07 3.56074E	67300 7.00235E-07 4.55616	70819 9.65482E-07 6.28202	76231 1.58181E-06 1.0292 92990 7.28801E-06 4.7420	934482 0.428509067 0.33	237884 0.129818544 0.08	64610 5.47699E-07 3.563 92290 6.83762E-06 4.448	426698 0.16346287 0.113 316236 0.114612915 0.09	7216009 0.621983599 0.52	85318 3.622E-06 2.1	62076 4.34472E-07 2.87 3483 5.69488E-10 3.71	115983 5.92221E-05 3.5	277201 0223/0212-02 72 20088 3.58941E-07 2.5	1000000 1.4808E-05 9. 10000000 0.67661718 0.	12843 3.38888E-09 2	0 190817 0.054144427 0	922704 0.417023435 0 922704 0.417023435 0	785000 0.307968786 0	176956 0.015314284 0	126900 0.000160125 0.	922704 0.417023435 0	10/20425 0.690754895 0. 16275 1.01814E-07 6.	1 87225 4.30958E-06 2. 155000 0.002071732 0	1 79049 2.04528E-06 1. 22753 1.05961E-08 6.	0 112001 1200113531 0	127668 0.000171731 0.0	40191 5.784E-08 3.	1 77097 1.7118E-06 3.	1 26939 1.6224E-08 1.0 17982 6.32306E-09 4.1	0 140263 0.00054102 0. 0 89760 5.42965E-06 3.	3713739 0.55325477 0.4	1 123252 0.000114845 7.	154624 0.002001962 0. 89784 5.44154F_06 7.	90464 5.78943E-06 3.7	0 50109 1.45019E-07 9.43 0 180676 0.040E7669 0.03	0.21 102212122358 0.17) 93347 7.52901E-06 4.89) 53937 7.64775E-07 1.3	92290 6.83762E-06 4.44	11-0 6927021570 602751 0 270 112261570 62258 0) 72056 1.08085E-06 7.03) 72381 1.11337E-06 7.24) 73675 1.25288E-06 8.15 1 119071 7 017012 AK E AG	59086 3.30503E-07 2.15	0 78305 1.91114E-06 1.2 1 112791 4.4277E-05 2.88	

Score	0.0872	0.15406	0.06398	0.00048	0.00491	0.00039	0,00246	0.00304	26100'0	0,00073	1.5E-07	0.068	0.39546	0.00076	0.02494	0.04379	0.02908	0.08963	0.1812	5.6E-07	6.8E-05	0.02878	0.00325	5.5E-08	0.07859	2720.0	0.14239	0.07213	0.13813	0.09154	1110000	0.00828	0.00058	0.06667	0.140/4	0.00121	6.6E-07	0.0014	0.06274	0.19077	0.00063	0.04293	8.9E-09	0.00146	0.00105	0.00047	0.00024	0.43164	0.06195	0.00358	0.00098 8.4E-07	8.9E-07	0.00212	0.00777	0.00021	0.0631	0.09144	0.06381	0.00061	0.00047	0.00305	18600.0	0.00509	0.00129
Utility Score	0.0075003 0.00046502 0.0089066 0.00055221	0.0093753 0.00058127	0.0318906 0.00197721	0.0078128 0.00048439	0.0791506 0.00490734	10/00/00 200200/0	0.0396003 0.00245522	0 0 0.006719 0.00041658	0.0318104 0.00197224	0.0117191 0.00072659	0 0	66000900 0 26829600 97622210 0 000000	0 0	0.0121879 0.00075565	0.0109379 0.00067815 0.0003753 0.00058377	0.0028126 0.00017438	0.0362507 0.00224754	•	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0	0.0010938 6.7815E-05			0	0.184512 0.01143974	0.2423708 0.01502699	0.0158599 0.00098331	0.0262509 0.00162755	e :	0 0 A AVTENUT A AVA 12503	2000E0000 26686900	0.0953853 0.00591389	0.0089847 0.00055705	0.0753457 0.00467144	0.018754 0.00073627	0.0195319 0.00121098	0 0	0.0032032 0.0001986	0.0119535 0.00074112	0.0305479 0.00189397	0.0101566 0.00062971 0.0019532 0.0001211	0.0062502 0.00038751	0 0	0,0058596 0,00036329	0.0625936 0.0038808	0.0076565 0.0004747	0.0025001 0.00015501	0.8277851 0.05132268	0.9991529 0.06194748	0.0574769 0.00356357	0.0108597 0.0006733	0 0	0.034167 0.00211836	0.0025782 0.00015985 0.0028320 0.00015985	0.0034376 0.00021313	0.017735 0.00109957	0.0511013 0.00316828	0.0050002 0.00031001	0.009766 0.00060549	10015784 0.00464686	0.0490583 0.00304162	0.1581766 0.00980695	0.0820458 0.00508684	0.0056252 0.00034876
Score	0.0075	0.09375	0.031889504	0.0078125	0.079148111	0.018671875	0.039598963	0.00671875	0.031809304	0.01171875	0	4CF10F/F70	0	0.0121875	0.0109375	0.0028125	0.036249498	•	0 016485370	0	0.00109375		• •	0	0.184506824	9611361196	0.015859375	0.02625	•	0 0026	270000	0.095382324	0.008984375	0.075343372	0.011875	0.01953125	0	27120220000	0.011953125	0,030546875	0.01015625	0.00625	•	0.005859375	0.062591595	0.00765625	0.0025	0.827780283	1222166674	0.057475077	0,0102937575	Ð	0.034165904	0.002578125	0.0034375	0.017734375	0.0510996	0.005	0.009765625	0.007578125	0.049056725	0.158172058 0.000675	0.082043272	0.005625
MHRS	0 F	0	120	8	114	408	•	00I	8	239	\$ 00	98	407	150	0	0101	0	156	140	36	461	-	211	0	4		• •	0	1500	•	26/1	336	•	• ;	96 B	1000	115	982	152	250	° ;	4	153	165	<u>8</u> x	38	0	213	555	86	32	3222	7307	687	0	0	436	3 6	4	624	905	49 <u>5</u>	01E	26
Score	0.062	0	0.062		0	• •	e ·	0 0	• •	0	• •	0.062	0	0	0 00	700/n	0	0.062		• •	0	0	0	0	0.062	0.067	0.062	0.062	0.062	• •	• =	0	•	0.062	790.0	0	•		0.062	0	0 063	0	0	• •		• •	•	790'n	0	•	• •	0	0	• •	• •	0.062	0.062		•	••	0	• •		0
Score		0			0		0	• •	Þ	0	•	ə -	, ø	0	o -	- 0	0	1	÷ •		0	÷ -	- 0	0	- •		-	-	-	• •		0	0			0	•	• •		•		- 0	0			• •	. .		•	•		0	•			-			0	, 0	0	0 0	• •	0
TOTLOSS		-	• •	00	0	0 0	0	• •	• •	0	0		0	0			-	Ō		00	0	- •		-	0.			-	0	•			-		• •	•	0		• •	0		• •	0	•			-			0	0 0	0	0	0 0	-	-	• •	•	0		0	• •		0
Score		0.00325	•		•	- 0	•	• •		0	•		0.0065	0	• •			•			0			0	0	0	• •	•	Ð	• •	• •	¢	•		0 00325	•	•		• •	0.0065		• •	•	• •			•	0.04975	•	•	• •	0	e	0 0.00176	8	6	• •	. e	•	• •	0	• •		0
Score	0	0.083333	• •		¢		8	8 6	• •	Ð	•		0.166667	0		• •	0	•		• •	0			0	- 2	9 -		0	0		• •	0	0	•	0 A83333	6	0 0	• •	• •	0.166667	• •	• •	0	• •	• •		0	0.25	0	-		0	0	0	0	0	•	. .	6		Đ			0
HASPL		s 1											5		~	ŗ.							\$,	•									-					2		2						5 3						-	-		2						7	
r Score		0.007	0 0		Ċ		0	• •		0	•		5 0.007	0	833 0.0032	433 0.0032	0	0	•	• •	0	0 0	933 0.0032	0	0 0			0	0	0 0		0	0	0	2500,0 553 2500,0 553	0	•			567 0.006:		0.0032	0	• •		• •	0	0.0097	•	•		• •	0	0 0	• •	•	333 0.0032	• •			0	0 255	0 0	0
SPL Scot	0 0	3 0.2	0 0		0		0	00	0	0	0		3 0.2	0	1 0.083:	1 0.083	0	0		. 0	0	0	1 0.083	0	0		. 0	G	0	• •		•	0	0 .	1 0.087	•	0 1		, 0	2 0.166		1 0.083:	Ċ	• •			0	3 0.2	•	0			0	•		0	1 0.083	0	0	» o	0	1 0.05	0	0
Score 01		0.039			6	• •	6	• •		0	•	0.039	0.039	0	• •		0	0			0	• •		0	•			0	0	• •			0	•			•			0.039			0	¢ 0			0	650.0	0	•	• •		•	0	0	•	0	0				• •		•
Score	• =	-	• •		ċ		÷	• •		0	•			0	• •		•	0	• •		0	•		0	•			0		• •		•	0	e	• •	•	•	• •		1	•		0	-						•	• •	•	0	- 2	- 	0	- 2	3 -			. 0	• •	• •	0
FLSPL		5										'n	5									-																		ŝ				~				¥7							'n		-	•						
Score	• •	0.01756406	• •	• •	o		0			•	•	• •	0.01453625	0		0.01782028	•	0.0015625	0.0229191		o	0.01771772		ð	0.00515121	. v.uut/16/.	0.0021875	0	0	• •		0	0	0	C/ 56000'0 1	ø	0	• •		0.0021875	• •	0.0184608	0	0.0010937	• •	• •	0		. 0				0	0	0	0	0 0	0	0	. 0	0	0 0		\$756000.0
Score	• •	0.501830161	0 0	• •	0	. .	•	o c	• •	0	0	ə c	0.415322581	0	• •	0.509150805	0	0.044642857	0.654831623	• •	0	0.506222548	• •	0	0.147177415	0	0.0625	0	0			0	0	0	EU621992010	0	0	• •		0.0625	• •	0.527452416	¢	0.03125			0		• •	0	• •	¢	0	0	0	0	0	0	0	. o	0	0 0	• •	0.026785714
DAYREST	• •	50	• •	• •	0		0			0	0		, e	0		2	Ð	9	468		0	62		0	8:	= <	1	0	0	• •		0	0	۰ م	• <u>*</u>	•	•	ə c	, a	14	• •	120	0		• •		0		. 0	•	• •	0	0	• •	• •	0	• •	n 0	0	00	0	00		9
Score	• •	2.1752E-06		• •	•		•		• •	0	•		0.01366352	0	-	0.01338293	•	4.7693E-06	0.01721742		0	9.6031E-07		0	1.2481E-07	/II-TICHO'S		¢	2.2134E-05	0.01543142	10++/orm/n	¢	•	•	0 7 7174E-05	•	•			0.01194325	• •	0.01638415	0	3.913E-07			0	• •	•	•	• •	•	¢	ð	• •	0	0.01775956	0	0		• •	•		•
Scote		000120846	• •		0	• •					•	• •	,75968419	0	• •	0	Û	000264959	956523593		0	33504E-05		0	93396E-06	0		0	.001229661	857300915	Account of		0	•	0	0	•			663513662		010230401	0	.17391E-05			0	• •		•	• •		0	• •	• •	•	.986641955 5 40 42 1955	60710±//+C+C*	•		. 0	•		0
AYLOST	• •	0 +	•	• •		• •	0	• •		0	0	• •	. 09	9	• •	0 15		5	- -		0	•0 •n		•	9 1	n n =			, 0	52 F	2 4			•			0	• •		- ∓	• •	107	0	n 1		• •	0		Ð	•	• •	• •	0	• •	. 0	Ð	194 0	, e	0		. 0	0 0		. 0
Score D	•	00713462			0	• •	o	• •	• •	0	0		0634936	0	• •	U 00403846	0	00302885	01687179		0	0110609		0	0	C797001		0	0				0	0	00444231		0			00424038		00444231	0	0 0	• •		0		, 0	0	• •		0	• •		0	0.00525	76079700	0	• •	, o	• •		, 0
kore	0 0	YO \$641486	0		0		¢	• •	• •	0	0	• •	1175214 0.0	0	-	0 6153846 0.0	0	72115385 0.0	01709402 0.		0	53354701 0		0	0	1 679670		0	0	•		, .	0	0	05769231 0.		0	• •		0 85319600	0 0	U 15769231 0.1	0	0			0	0 0	0	0			o	•		•	0.125	n 760/06/0	0	• •				. 0
HOSP		16 0.16			0	0 0				0	0	•	11 0.15	0	0	0 8 0.05	. 0	3 0.0:	38 0.4		0	51 0.2(0	0	0 1		, 0	0	0	10 0.1		0	0 0		9 0.1(0 0	0 10 0.10	0	0			0			0	• •		0	0		. 0	14	70 70						. 0
AVC																			~			•••																																										

Score	0	0	• •	• •	• •	0	0		• •	0	•			•	• •		0	• •	• •	• •	0	•	•		• •		•	0	• •	,	0	•	•	•	4	. 6		•	0	0	0	0	•		• •			¢	0.0035	0	• •	•	• •	•			0	• •	0 0035		•	0	•	0	0	•		•	• •	» Ф	
Score	0 0	0	•			6	0	e 6		0	•		. 0	0	• •		Ð	÷ •	•		0	•	•								0	0	¢.	•	e	• =			0	0	0	0	•					a	0.25	•	•	• =		•	• •		0	• •	0	} =	• •	0	0	0	0	0		0	•	• •	I
RTIAL DISABILITY		•				0	6	•		0	•			0			0		• •	. 0	0	6 1	c 1							;		•	-	•	-				0	0	0	0	•	•				0	1	0				0	•		0						•	•	0	0		•	•	-	
Score PAI	0 0	0			. 0	0	0	0 0		0	0	00	. 0	0	• •		0	0 1		0	0	0	0	0	COTO'O		. 0		. 0		0	0	0	•	c		5 6		Ð		0	0	0	0				0	0	0	0			0			0	0 0			• •	0	• •	0	0	0		0	0 0		,
Score	• •	0	• •			0	0	о (• =	0	0			0	00	• •	0	0	е с	• •	o	0	0	0 ¥	6					•	0	ð	0	ð	e			2	0	0	0	0	0	0	•			0	0	0	0			0	0 0	• •	0	• •	•		• •	•		•	0	0		•	• •	. c	,
TOTAL DISABILITY	00	• •	0	• •		. 0	0	0		0	0	0 0		0	0 0		0	0	0 0	• •	. 0	0	0					, e		,	0	0	0	0	c			>	0	0	0	0	0	0				0	0	0	0	5 0	0	0	0	• •	0	• •	• •			0	. 0	. 0	0	0		0	0	5 C	2
Score	• •	•	•	• •	• •	•	0	0 1			0	0	0	0	0.187	•	÷	ċ	• •	• •	•	0.176	0	0 10	/ern		-	•		•	0	0	0	0				•	٥	•	•	•	•	•			•	•	•	•	•		• •	•	• •	• •	¢	•				•	0.187	0	0	0		•	•		;
															-									2	G																																						23								
TL Score	• =			• •		•	0	•	• •	•	0	0 0 59375	0	0	0.5312	• •	•	0	• •		ē	9.6	¢	•		•	• -	• =		•		0	•	•	4			•	•		•	•	•	¢	•		•	•	0	•	•	-	• •	0	-			•	•			•	0.531	•	0	0		¢	•		
Score TOTFTL Score	0	1.5071E-08 0 0	1.2373E-06 0 0	• • • •		0 0	0 0 0	e .			0 0	0 0 0 0 0 0 0 0 0 0		0 0	0.02446761 2 0.5312		0 0 0	0 0	4.6982E-08 0 0			0.02388578 1 0.5	9 0								0 0	0 0 0	0 0 0	0					2.7009E-06 0 0	e 0	0 0 0	0 0	1.5842E-07 0 0	0 0 0	0 0 0	0 0 0 1.774/5-708 0	0	1.9518E-05 0 0	0.00163768 0 0	8.9888E-08 0 0	0	1.288/E-08 0 0	6.8309E-07 0 0	0 0 0	• •	•••	0 0	2.186E-06 0 0			5.1696E-05 0 0	1.0078E-07 0 0	0.02776507 2 0.531	0 0	0 0 0	0 0 0	0	0 0 0	-		,
Utility Score TOTFTL Score		3.2067E-07 1.5071E-08 0 0	2.6326E-05 1.2373E-06 0 0			• • • •	0 0 0	• • •		0 0 0	0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0	0.52058743 0.02446761 2 0.5312		0 0 0	0 0 0	9.9962E-07 4.6982E-08 0 0			0.50820803 0.02388578 1 0.5	6 8 0 0								• • •	0 0 0 0	0 0 0	0 0 0					5.7466E-05 2.7009E-06 0 0	0 0 0	0 0 0	0 0 0	3.3706E-06 1.5842E-07 0 0	0 0 0	• • • • •	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0	0.00041528 1.9518E-05 0 0	0.03484423 0.00163768 0 0	1.9125E-06 8.9888E-08 0 0	0 0 0	2.7419E-07 1.2887E-08 0 0	1.4534E-05 6.8309E-07 0 0	0 0 0 0	•		0 0 0 0	4.6511E-05 2.186E-06 0 0		0,03426037 0,00151024 0 0 0	0 0 91 816961 8 166601000	2.1442E-06 1.0078E-07 0 0	0.5907462 0.02776507 2 0.531	0 0 0	0 0 0	0 0 0	0 0	0 0 0	0 0 0		, 0
Score Utility Score TOTFTL Score		1.37403E-06 3.2067E-07 1.5071E-08 0 0	0,000112788 2.6326E-05 1.2373E-06 8 0			• • • • •	0 0 0 0			0 0 0 0	0 0 0 0	0 0 0 0 0 0 0 0 0 0 0		0 0 0 0	0.763937571 0.52058743 0.02446761 2 0.5312		0 0 0 0	0 0 0 0	4.28324E-06 9.9962E-07 4.6982E-08 0 0		e e o o	0.755613466 0.50820803 0.02388578 1 0.5	6 8 0 0 0								• • • •	0 0 0 0 0	0 0 0 0 0	• • • •					0.000246161 5.7466E-05 2.7009E-06 0 0	0 0 0 0	0 0 0 0	0 0 0 0	1.44422E-05 3.3706E-06 1.5842E-07 0 0	0 0 0 0		U U U U U U U U U U U U U U U U U U U	0 0	0.001775582 0.00041528 1.9518E-05 0 0	0.127330854 0.03484423 0.00163768 0 0	8.19479E-06 1.9125E-06 8.9888E-08 0 0	0 0 0	1.17488E-U6 2.7419E-U7 1.2887E-U8 U U	6.22714E-05 1.4534E-05 6.8309E-07 0 0	0 0 0 0	• • • •		0 0 0 0	0.000199249 4.6511E-05 2.186E-06 0 0		n n 6701010/0//02450/0 1076671/0	0.004686129 0.00109991 5.16968-05 0 0	9.18746E-06 2.1442E-06 1.0078E-07 0 0	0,808165972 0,5907462 0,02776507 2 0,531	0 0 0 0 0	0 0 0 0	0 0 0 0 0	0 0	0 0 0 0		• • • • •	, 0
INJCOST Score Utility Score TOTFTL Score		840 1.37403E-06 3.2067E-07 1.5071E-08 0 0	33319 0.000112788 2.6326E-05 1.2373E-06 0 0				0 0 0 0 0 0				0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		0 0 0 0 0 0	1370000 0.763937571 0.52058743 0.02446761 2 0.5312			0 0 0 0 0	2530 4.28324E-06 9.9962E-07 4.6982E-08 0 0			1213788 0.755613466 0.50820803 0.02388578 1 0.5	6 8 0 0 0 0								• • • • •	0 0 0 0 0	0 0 0 0 0	0 0 0 0 0					48915 0.000246161 5.7466E-05 2.7009E-06 0 0		0 0 0 0 0	0 0 0 0 0	7666 1.44422E-05 3.3706E-06 1.5842E-07 0 0	0 0 0 0 0		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		94830 0.001775582 0.00041528 1.9518E-05 0 0	255620 0,127330854 0.03484423 0.00163768 0 0	4635 8.19479E-06 1.9125E-06 8.9888E-08 0 0		720 1.17488E-U6 2.7419E-U7 1.2887E-U8 U U	23341 6.22714E-05 1.4534E-05 6.8309E-07 0 0	0 0 0 0 0			0 0 0 0 0	44465 0.000199249 4.6511E-05 2.186E-06 0 0		n n 67010100/n /¢n97460/n 1076671/n 8/n617	118604 0.004686129 0.00109991 5.1696E-05 0 0	5142 9.18746E-06 2.1442E-06 1.0078E-07 0 0	2200000 0.808165972 0.5907462 0.02776507 2 0.531		0 0 0 0	0 0 0 0 0	0 0	0 0 0 0 0			, o o
Score INJCOST Score Utility Score TOTFTL Score	2.855E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.02519857 840 1.37403E-06 3.2067E-07 1.5071E-08 0 0	0.02268022 33319 0.000112788 2.6326E-05 1.2373E-06 0 0	0.00026907 0 0 0 0 0 0 0 0 0	1 12/11E-02 U U U U U U U U U U U U U U U U U U U	2.855F.09 0 0 0 0 0 0	0.02265551 0 0 0 0 0	3.4648E-08 0 0 0 0 0 0		0 0 0 0 0 0 0 0	9.1156E-07 0 0 0 0 0 0	0.03459528 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		5.8643E-07 0 0 0 0 0 0	0.1217114 1370000 0.763937571 0.52058743 0.02446761 2 0.5312		481688-06 0 0 0 0 0	1.1027E-07 0 0 0 0 0	0.02262108 2530 4.28324E-06 9.9962E-07 4.6982E-08 0 0	0.00375212 0 0 0 0 0 0 0 0	7.12678-09 0 0 0 0 0 0	0.082578 1213788 0.755613466 0.50820803 0.02388578 1 0.5	0.07610811 0 0 0 0 8 0								0,00013531 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0	0.02035419 0 0 0 0 0 0	4.5948E-08 0 0 0 0 0 0					0.07610811 48915 0.000246161 5.7466E-05 2.7009E-06 0	11902E-07 0 0 0 0 0	8.1344E-08 0 0 0 0 0 0	0.04329712 0 0 0 0 0 0	0.09969589 7666 1.44422E-05 3.3706E-06 1.5842E-07 0 0	1.019E-09 0 0 0 0 0 0		2.512-08 U U U U U U U U U U U U U U U U U U U		0.1093653 94830 0.001775582 0.00041528 1.9518E-05 0 0	0.00849822 255620 0.127330854 0.03484423 0.00163768 0 0	0.11125313 4635 8.19479E-06 1.9125E-06 8.988E-08 0	0.00148432 0 0 0 0	0.0213384 720 1.174885-06 2.74195-07 1.28876-08 0 0	0.00641708 23341 6.22714E-05 1.4534E-05 6.8309E-07 0 0	0.10724719 0 0 0 0 0 0		0.10849822 0 0 0 0 0 0 0	0.00849822 0 0 0 0 0 0	0,0001365 44465 0.000199249 4.6511E-05 2.186E-06 0 0		n n 67n191n0'n /6n97660'n 1676671'n 2/n617 91816/n1'n n v v v v issuestinder	0.001849822 118604 0.004686129 0.00109991 5.1696E-05 0 0	0.09554485 5142 9.18746E-06 2.1442E-06 1.0078E-07 0 0	0.00849822 2200000 0.808165972 0.5907462 0.02776507 2 0.531	0.1187075 0 0 0 0 0 0	0.00124875 0 0 0 0 0 0	0.10731207 0 0 0 0 0 0	0 0	1.0111E-08 0 0 0 0 0 0	1.228E-09 0 0 0 0 0 0		
Utility Score I INJCOST Score Utility Score TOTFTL Score	1.20976E-08 2.855E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.106773619 0.02519857 840 1.37403E-06 3.2067E-07 1.5071E-08 0	0.096102628 0.02268022 33319 0.000112788 2.6326E-05 1.2373E-06 0	0.001140122 0.000140126907 0 0 0 0 0 0 0 0 0 0 0			0,0959597912,0.02265551 0 0 0 0 0 0	1.46812E-07 3.4648E-08 0 0 0 0 0		0.122457132 0.62889988 0 0 0 0 0 0 0	3.86256E.06 9.1156E-07 0 0 0 0 0	0,146590166 0.03459528 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6.87191E-05 1.6213E-05 0 0 0 0 0 0 0 0 0	2.48486E-06 5.8643E-07 0 0 0 0 0	0.515726274 0.1217114 1370000 0.763937571 0.52058743 0.02446761 2 0.5312		2.04101E-05 4.8168E-06 0 0 0 0 0 0	4.67243E-07 1.1027E-07 0 0 0 0 0	0.095852035 0.02262108 2530 4.28324E-06 9.9962E-07 4.6982E-08 0 0	9.3014/F_US 2.31312-US 0 0 0 0 0 0 0 0 0 0 0 0	3.0198E-08 7.1267E-09 0 0 0 0 0 0 0 0	0.349906783 0.082578 1213788 0.755613466 0.50820803 0.02388578 1 0.5	0.322491979 0.07610811 0 0 0 0 0	2.21604E-05 5.2298E-06 0 0 0 0 0 0							0.000135312 0 0 0 0 0 0	0,000747284 0.00017636 0 0 0 0 0 0	0.086246575 0.02035419 0 0 0 0 0	1.94693E-07 4.5948E-08 0 0 0 0					и 1734 1444 0.0764 0.0811 48915 0.000246161 5.7466E-05 2.7009E-06 0 0	surveyerury 1,1902E-07 0 0 0 0 0 0	3,44678E-07 8,1344E-08 0 0 0 0 0	0.182462354 0.04329712 0 0 0 0 0 0	0.422440221 0.09969589 7666 1.44422E-05 3.3706E-06 1.5842E-07 0	4.31771E-09 1.019E-09 0 0 0 0 0 0		1.06358E47 2.51E-08 U U U U U U U U U U U U U U U U U U U		0.463412271 0.1093653 94830 0.001775582 0.00041528 1.9518E-05 0 0	0.03484423 0.00849822 255620 0.127330854 0.03484423 0.00163768 0	0.471411576 0.11125313 4635 8.19479E-06 1.9125E-06 8.9888E-08 0	0.00628949 0.00148432 0 0 0 0 0	0.090416942 0.0213384 720 1.174885-06 2.74195-07 1.28872-08 0 0	1.10181E-05 2.000641708 23341 6.22714E-05 1.4534E-05 6.8309E-07 0 0	0.454437262 0.10724719 0 0 0 0 0		1.56258E-08 3.412/E-07 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.036009409 0.00849822 0 0 0 0 0 0	0.000157838 0.0001365 44465 0.000199249 4.6511E-05 2.186E-06 0 0		0 0 670131382 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 21/00/00/00/00/00/00/00/00/00/00/00/00/00	0 101851017 0.09554485 5142 9.18746E-06 2.1442E-06 1.0078E-07 0 0	102300 0.00843822 2200000 0.808165972 0.5907462 0.02776507 2 0.551		0.005291325 0.00124875 0 0 0 0 0 0	0.454712159 0.10731207 0 0 0 0 0	0 0 0	4.28452E-08 1.0111E-08 0 0 0 0 0	5.2033E.09 1.228E.09 0 0 0 0 0 0	1,73939E-06 4,102E-07 0 0 0 0 0 0 0 0	
Score Utility Score INJCOST Score Utility Score 707FTL Score	185977E-08 120976E-08 2855E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.154121122 0.106773619 0.02519857 840 1.37403E-06 3.2067E-07 1.5071E-08 0 0	0,139551899 0,096102628 0,02268022 33319 0,000112788 2,6326E-05 1,2373E-06 0 0	0,001751017 0,001140122 0.00026907 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			0,139408069 0,095997912 0.02265551 0 0 0 0 0	2.25635E-07 1.46812E-07 3.4648E-08 0 0 0 0 0 0	3,59289E-05 2,33779E-05 5,5172E-06 0 0 0 0 0 0 0 0	0.175230733 0.122457132 0.02889988 0 9 0 0 0 0 0 0	5,93656E-06 3,86256E-06 9.1156E-07 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.000105582 6.8701E-05 1.6213E-05 0 0 0 0 0 0 0 0 0	3.81899E-06 2.48486E-06 5.8643E-07 0 0 0 0 0	0.61294184 0.515726274 0.1217114 1370000 0.763937571 0.52058743 0.02446761 2 0.5512		3.13678E-05 2.04101E-05 4.8168E-06 0 0 0 0 0 0	7.18106E-07 4.67243E-07 1.1027E-07 0 0 0 0 0	0.139207675 0.095882035 0.02262108 2530 4.28324E-06 9.9962E-07 4.6982E-08 0 0		4,64113E-08 3.0198E-08 7.1267E-09 0 0 0 0 0 0 0	0.446773609 0.349906783 0.082578 1213788 0.755613466 0.50820803 0.02388578 1 0.5	0.417023435 0.322491979 0.07610811 0 0 0 0	3.40578E-05 2.21604E-05 5.2298E-06 0 0 0 0 0 0							0.14468808556 0.004573342 0.00013531 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.125940731 0.086246575 0.02035419 0 0 0 0	2.99223E.07 1.94693E-07 4.5948E-08 0 0 0 0 0					0 11707444 0 12701079 0.076(0811 48915 0.000246161 5.7466E-05 2.7009E-06 0		5.29735E-07 3.44678E-07 8.1344E-08 0 0 0 0 0 0	0.254085126 0.183462354 0.04329712 0 0 0 0 0 0	0.522210297 0.422440211 0.09969589 7666 1.44422E-05 3.3706E-06 1.5842E-07 0 0	6.63588E-09 4.31771E-09 1.019E-09 0 0 0 0 0 0	2.64926E-07 1.72377E-07 4.0681E-08 0 0 0 0 0 0 0 0	1.63461E-07 1.063381E-07 2.51E-08 U U U U U U U U U U U U U U U U U U U		0 20112582 0 00041528 0 000177582 0 00041528 0 0 00041528 0 0 0	0.05414427 0.05609409 0.00849822 255620 0.127330854 0.03484423 0.00163768 0 0	0.570666496 0.471411576 0.11125313 4635 8.19479E-06 1.9125E-06 8.9888E-08 0	0.009628863 0.00628949 0.00148432 0 0 0 0 0	0.131718219 0.090416942 0.0213384 720 1.174885-06 2.74195-07 1.28876-08 0 0	1.095455E-05 1.10184E-05 4.000641708 23341 6.22714E-05 1.4534E-05 6.8309E-07 0 0	0.554084105 0.454437262 0.10724719 0 0 0 0		2.09415E-08 1.56258E-08 3.413/E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.054144427 0.036009409 0.00849822 0 0 0 0 0	0.000888594 0.00057838 0.00001365 44465 0.000199249 4.6511E-05 2.186E-06 0 0		0.554379822 0.454737985 0.1074816 2130/8 0.12694231 0.026460/9 0.001121024 0 0	UXU/2200014 UXU/2200014 UXU/20012 V V V V V V V V V V V V V V V V V V V	0.601333313 0.0018551485 5142 9.187466-06 2.1442E-06 1.0078E-07 0 0	0.2042020010 0.444401040 0.00849822 2200000 0.808165972 0.5907462 0.02776507 2 0.531		0.008105705 0.002291325 0.00124875 0 0 0 0 0 0 0 0	0.554354428 0.454712159 0.10731207 0 0 0 0 0	0 0 0 0	6.58488E-08 4.28452E-08 1.0111E-08 0 0 0 0 0	7.99694E-09 5.2033E-09 1.228E-09 0 0 0 0 0 0	2.67326E.06 1.73939E-06 4.105E-07 0 0 0 0 0 0 0	
DAMCOST Score Utility Score INJCOST Score Utility Score 707FTL Score	28314 1.8927E-08 1.20976E-08 2.855E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	383681 0,154122122 0,106773619 0,02519857 840 1,37403E-06 3,2067E-07 1,5071E-08 0 0	303754 0.139551899 0.096102628 0.02268022 33319 0.0000112788 2.6326E-05 1.2373E-06 0 0	153154 0,001751017 0,001140122 0,000140126907 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2001 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2831.1 1 SECTION 1 2007KE-08 2.855F-09 0 0 0 0 0 0 0	302866 0.133408069 0.0355997912 0.02265551 0 0 0 0 0 0 0 0	54920 2.25635E-07 1.46812E-07 3.4648E-08 0 0 0 0 0		474277 0.175230733 0.122457132 0.02889988 0 0 0 0 0 0 0 0	90739 5,93656E-06 3,86256E-06 9,1156E-07 0 0 0 0 0	577509 0.207026216 0.146590166 0.03459528 0 0 0 0 0 0 0 0 0 0	0/22309 0.000105582 6.87018-05 1.6213E-05 0 0 0 0 0 0 0 0 0	85899 3,81898E-tK 2,45486E-06 5,8643E-07 0 0 0 0 0	6755253 0.61294184 0.515726274 0.1217114 1370000 0.763937571 0.52058743 0.02446761 2 0.5312	21/12/04/11/13/11/14/14/24/2011/14/14/11/2011/14/14/11/14/14/14/14/14/14/14/14/14/1	109008 3.1378E-16 2.44101E-05 4.8168E-06 0 0 0 0 0 0 0 0	67576 7.18106E-07 4.67243E-07 1.1027E-07 0 0 0 0	301625 0.139207675 0.095852035 0.02262108 2530 4.28324E-06 9.9962E-07 4.6982E-08 0 0 6 6653 4.55925-07 0.00 0	20224 1.200395-07 9.80147-68 2.20275-08 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	37844 4,64113E-08 3.0198E-08 7.1267E-09 0 0 0 0 0 0 0	952428 0.446773609 0.349906783 0.082578 1213788 0.755613466 0.50820803 0.02388578 1 0.5	912704 0.417023435 0.322491979 0.07610811 0 0 0 0 0 0	109911 3.40578E-05 2.2164E-05 5.2239E-16 0 0 0 0 0 0 0							145613 D. MHORSONRSS D. MONST3342 D. MONST33442 D. MONST3344444444444444444444444444444444444	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	207727 0.125940731 0.086246575 0.02035419 0 0 0 0 0 0 0	58000 2.99223E-07 1.94633E-07 4.5948E-08 0 0 0 0 0					9327h.4 0.117057446 0.17241059 0.07610811 48015 0.000246161 5.7466E-05 2.70095-06 0		64245 5.29755E-07 3.44678E-07 8.1344E-08 0 0 0 0 0 0	689600 0.254085126 0.183462354 0.04329712 0 0 0 0 0 0 0	2131785 0.522210297 0.422440221 0.09969589 7666 1.44422E-05 3.3706E-06 1.5842E-07 0	18411 6.63588E-09 4,31771E-09 1.019E-09 0 0 0 0 0 0	56671 2.649266-07 1.72377E-07 4.061E-08 0 0 0 0 0 0 0 0	21410 1.0340[E-47] 1.0628E-97 2.21E-08 U U U U U U U U U U U U U U U U U U U		4204192 0.563874484 0.463413271 0.1095653 94830 0.00177582 0.00041528 1.9518E-05 0	190817 0.051144427 0.0364949822 25520 0.127330824 0.03484423 0.00163768 0 0	4601000 0.57066456 0.471411576 0.11125313 4635 8.19479E-06 1.9122E-06 8.9888E-08 0	171863 0.009628863 0.00628949 0.00148432 0 0 0 0 0 0 0	Z51756 0.131718219 0.096412 0.0151384 720 1.17488E-06 2.7419E-07 1.287E-08 0 0	182792 0.04110167 0.00641708 23341 6.22714E-05 1.4534E-05 6.8309E-07 0 0	3756000 0.554084105 0.454437262 0.10724719 0 0 0		25545 2.09415E-08 1.36258E-08 3.413/E-09 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	190817 D.054144427 D.036009409 0.00849822 0 0 0 0	145709 0.0000888594 0.000157838 0.0001365 44465 0.000199249 4.6511E-05 2.186E-06 0 0		0 0 0 0 0.0254512082 0.02545120 0.021612 0.021612 0.021612 0.021620 0.021712 0.021712 0.021712 0.021712 0.02171	1971.4 UNEXCOMPAND UNERGY UNALING UNERGY UN	1220841 0.001473813 0.404554485 5142 9.18746E-06 2.1447E-06 1.0078E-07 0 0	190817 0.0277420 0.0277420 0.00819822 2200000 0.808165972 0.597462 0.02776507 2 0.537		165973 0.008105705 0.005291325 0.00124575 0 0 0 0 0 0 0 0	3769775 0.554354428 0.45712159 0.10731207 0 0 0 0 0 0		41580 6.584382-08 4.284522-08 1.0112-08 0 0 0 0 0 0	20104 7,99694E-09 5,2033E-09 1,228E-09 0 0 0 0 0 0	· 813986 2.67326E.06 1.733939E.06 4.105E.07 0 0 0 0 0 0 0	

Score	2.9E-09 0.00278	0.03192	0.04323	0.10896	0.00184	0.02266	0.00044	0.00146 3.5E-10	0.03871	0.00243	0.0969	74200 U	0.06205	0.35513	0.03511	2.2E-06	4.8E-06	0.00177	0.03456	100000	0.06247	0.37343	0.07611	0.06248	0.01042	0.00048	0.02438	3.5E-10	0	0.00153	0.00158	C1C70/0	0	0.00274	0.00118	2100'0	0.12949	0.07176	0.00162	0.04988	0.062	4.1E-08	2.5E-08	0.00556	0.23309	0.12165	0.17989	0.02448	2.6E-06	0.10725	0.1348	0.01556	0.07375	0.16392	0.11764	0.06727	0.10744	0.26323	0.25902	0.06698 0.14956	0.00598	1.2E-09	0.00401	
7	11	35	18	74	58		g :		56	51	1	6 9	50	12	68			87		ŧ¥	4	26	ļ	4		6	89 80	07		S	68 30	9		61	16	37	3	16	187	6				8	8	60		69			151	187		27		133	53	578	100	69	24	1	701 191	!
Score	0 6 0.002784	89 0.005623	84 0.001608	12 0.011439	78 0.001816	0	96 0.000440	0	66 0.009806	1 0.002428	CUCUMU:U 13	2007000 SZ	13 4.8439E-	03 0.002455	27 0.000406	e	0	47 0.001772	9 0 0055568	00000000 00 00	65 0.00047	71 0.000343	•	71 0.00047- 71 0.000343	0	28 0.000484	59 0.000494	0	0	07 0.001395	89 0.001399 00 0.001660	e e	• •	11 0.002744	31 0.001181	H7 0.001695	001001	127 0.009757	46 0.001617	43 0.016587	• •	• •	0	K5 0.00429	28 0.00244	121 0.005408	• •	49 0.002206	•		182 0.054554	0 172 0.007063	•	376 0.020268		172 0.004107	E31 0.052031	255 0.000968	02 0.000310	128 0.000455-	0 11 0.002974	0	334 0.004010	
Utility	0.04490	15 0.09070	5 0.02593	24 0.18451	15 0.02929	0	75 0.00710	0	58 0.15817	72 0.03916	24001.0 Class	25790'0 80	5 0.00078	G 0.03960	5 0.0656	•	•	5 0.02859	0 0 0 0 0	100012 Y	5 0.00765	75 0.00554	•	25 0.00765	Ð	5 0.00781	5 0.0079		Ð	0.02256	25 0.022 <i>5</i> 7	UK70'N		59 0.04426	5 0.01906	5 0.02734	0.01625	27 0.15738	5 0.02609	74 0.1062			•	77 0.06921 n	41 0.03951	84 0.08723		74 0.03545			22 0.87990	05 0.11391	•	12 0.32696	• •	09 0.0662-	15 0.83923	9 0.01567	0.00500	5 0.0078) 0	0.04800	0	24 0.06461	-
Score	0.04490459	0.09070613	0.025937	0.18450683	0.0292968	0	601/00'0	0	0.1581720	0.0391626	0.0049238	57TAN'8	0.0007812	0.0395989	0.006562	6	•	0.0285937	0 0070766	0.0016170.0	0.0076562	0.0055468	•	0.0076562	0	0.007812	0.0079687	0		0.0225	0.0225781	C70'0	•	0.0442596	0.019062	0.0273437	0.01625	0.1573792	0.0260937	0.1062397			•	0.0692042	0.0395115	0.0872293	• •	0.0354937	• •	• •	0.8799046	0	0	0.3269001	• •	0.0662451	0.839218-	0.01562	0.005	0.007812	9.0480025	0	0.022812	
MHRS	603	00	906	0	559	966 332	0	375	ŝ	•	16	005	1375	495	69	16	753	10	202	\$	• •	366	Ċ	3625	8	71	•	8 17		100	<u>5</u>	357	2	288	289	320	Ċ	552	244	350	0 000	1371	334	1075	0	0	0	4006	499	f o	0	452	, c	0	3500	1125	0	0		768 3275	006	£9	100	,
Score	0 O	0	• •		0 0		0	• •	0	0 0	0.062		0.062	¢	0	0	0	•			0.062	0	0	0.062	0	0	• •			0			>	Ð	ø	•	o	0.062	0	• •	0,060	0	0	0	0	0.062	0.062	0	0 0	• •	0.062	0.062	0,062	0,062	0,062	0.062	0	7en/n	0.062	0.062	e		• •	,
Score	0 0	0	• •		e (• •	0		o	۰ ہ				0	0	0	e	•				0	0		•	0	• •			0			•	¢	0	0	0	. –	ç			• •	0	o	•	-		- 0	0 0	• •	-	- 0	,	-	- 0	-	• •	- 0	-	- 0	c		0 0	,
TOTLOSS	• •	• •	0		0		Ð	• •	0		0	o -	- 0	•	0	0	0	0	• •	÷ -	- 0	0	-			0	0	ç	,	0	0	• •	5	0	0	Ð	~	. 0	0	0			• •	0	-	-	1		0		-	o -	• -	-	o -	. 0		• -		••	e		0 0	,
Score		0	•	0.00325		• •	0		0	•	0	2000.0	0	0.00325	0	0	0	e				0.00325	•	0.00175	•	•	• •	• •		0	ė e		•	8	0	ŧ	0.0065	0	•	•				0	0,00975	0	0	0	0		0.00325	• •	, e	0.00325	0.00325		•		0.0195	0.00325	-		• •	,
Score	• •	0	•	0.083333	•		•	• •	0	0 0	9	100001.0	0	0.083333	0	0	0	•	•	-	• •	0.083333	Đ	0 02333	0	•	•	• •		0	• •	• •	•	•	0	•	0.166667	0	Ð			-	-	0	0.25	0	•	0	•	• •	0.083333	• •	, e	0.083333	0.083333		÷.	5 5	0.5	0.083333	-	• •	• •	,
HYSPL				-								- 14	•	1								1		-													2	•							'n		•	-			1			-	-				4					
Score	• •	•	•	• •	•	• •	0		0	•	0	<2.600:0		7 0.0065	0	0	0	0	• •			3 0.00325	0	2 0,0065	0	0	0			0	0	3 0.00325	•	•	¢	0	c	• •	0	•			. 0	0	0.00975	0	0		0	• •	3 0.00325	• •	3 0.00325	3 0.00325	• •		• •		0.0195	• •	e		00	,
PL Score	• •	. 0	•	- 0	0		0		0	0	0 20	57 G	• •	0.16666	0	0	0	0	• •		• •	0.08333	0	0 16666	0	0	0 0			۰	0	0.08333	2	0	0	0	c		0	•			• •	•	0.25	0	0 1		0		0.08333	• •	0.08333	0.08333			• •	• •	0.5	• •	c		• •	ı
me OLS		_	_		_		_			_		- -		975 2	_	_	_	_	975			1 60	-			-	_			_	-	-	_	-	_	_	_		-					•	39 3	-			_		1 325			1 1				60	139 4	650				
are Sco			•		0		0							25 0.00	0	0	0	•	.25 0.00			1 0.0		0						0	•		-	0		•	•							-	1 0.0						83333 0.00	 		1 0.4				1	1					
FLSPL Se												'n		3					3			\$			1																				÷						1 0.0			s				w	n no	ю				
Score		00109375	0		0	• •	•	• •	. 0			• •		•	0	0	0	0	0.0021875			•	0	• •	. 0	0				0		• •	5	0	•	0	144725600	0	0	0	01769217			0.00125	02678807	0.0180765	0.00663306	0.0009375	0		0	• •	, .	0.01902452	0.00.1875	0	0.0192295	0		• •	e		• •	,
Score	• •	0.03125 (0		0	• •	•	• •	Þ	0			• •		0	•	0	0	0.0625		0 0	0	e	•		0	•		,	0		• •	5	0	•	0	91004573	0	0	0	505490483 (335714286	0 65373353	516471449	89516129 (U 126785714	0		0	0 0	, o	543557833	0	0	549414348	260080645			c		0 0	,
VREST	• •		0		0	• •	0	0 0	0	0	•	• •		•	0	0	0	0	ž -	÷ ,		. 0	0	• •		0	•		,	0	0	• •	0	0	0	0	0 015		0	0	0			8	770 0.	90 0.	23 0.	6 9 9	0		. 0			164 0.	• 1	. •	180 0.	5 82 0		• •	¢		۰ د	
DA DA			16963		0												0	0	3E-07	•		568762	0	• •	, o	0			,	•		•	-	0	•	e	196971	0	. 0	÷	• •			0	SUST	366352	93E-06		0	1000	. 0	•		254336	• •		766836	0						-
R S			61266 0.01		_		_		_	_	_					•	_	_	1E-05 3.91			34279 8.01	-			_	_			-		. .	~		•	•	10.0 29.030	1N/0 007104		_				_	10.0 73540	10'0 61180	11264959 4.76			10'n 67789				10.0 49553			275436 0.01	NE-05 3.9					ė :	-
DST Se			0.816-		-						-					-			2113	-		0.871	-			-											0.916	010'0							398.0	0.755	0.000			10/m				969'0			186'0 0	2.173						
DAVLO	•		138 73	••	•	• •				·	-	-			•	0	0	0	. 1	•		20 20 20	0	• •		e	Ð,	• •	•	¢	•	•	•	0	0	0	.r.	2 0		0	2	-		0	602 86	218 60	¥S I		•	5 - 3		•	, .	61 152	0 10		154 18(0 269		0 0		• •	• •	,
Score	• •	• •	38 0.004240	• •	¢	• •		•		0	•	• •			• •	0	0	0	0	0		01 0.02943	Ċ	• •	• •	0	0	• •	5	0	0	•		0	0	0	05200.0 44	44 U.UU/BU	. 0	0	0.0026			0	59 0.02032	74 0.00886	0		0	0.0026	• •	0 0	, o	31 0.00444	0	2000000 cm	87 0.00996	92 0.00282 0			d		• •	,
Score	• •		0.1009615		o	• •	• •	a a		0	÷	• •				0	0	ç	¢	¢ ·	- c	0.7008547	0	• •		0	0	0 0	•	0	0	•	•	0	0	0	0 1010007	0		0	0.0625		• •	0	1479743	0.2110042	0	• •	0	0.0020	• •	0 0	••	0.1057692	0	0	0.2371794	0.0673070			•		00	2
DAYHOSP	0 0	• •	6		0	• •		• •	• •	0	a	• •				• •	0	0	с ·	0		168	0	• •		0	0	• •	•	0	o	0	o .	0	0	0	۵¢	2 ¢	. 0	0	- •	•		o	110	3	0		0					2	0 •	n e	4	n c		• •	G		•	5

Score			0	¢ 0	• •	0	• •	0.006125	•	. 0	•	• •	•	• •			• •	•	÷		8	0		• •	9	•		•	• •				•		•	÷	• •	• •		•	• •	•	•			0	0	0	•		a a	
Score	0		¢	¢ (0	• •	0.4375	• •		•		. 6	-			• •		•		•	e		• •	•	•		e	¢ (• •		• •	e	• •	•	0	• •		• •	•		•	-		e	0	0	0	0	• •		
DISABILITY			0	•		0	0		0			• •		0					0		0		1			•			•				0		0	•	• •		• •	0		0	•		•	0	0	0	0	0 0	0 0	2
re PARTIAI					S .	_		105	_								_		_										_				•	•																		
r Sci			9		0.01	0		5 0.0			0	00	, 0	0		, ,	0.					5											Ū		Ū										_	-	-	-	_			
Y Scol	0	••	0	•	0	0	00	0.7	0 0		0	с с	¢	¢ i	••			• •	0		0	0		00	. 0	0			0			0 C	¢		0	0	•			0		0	00		Ģ	0			0		с с	
TOTAL DISABILIT	•		0		- 0	0			•		0	• •	• •	0	• •	• •			0	• •		0		• •	• •	•	• •		0 (o c	0	• •	0	0	0	0	0 0	ð		0	0 0		. 0	0	0	e	0	• •	0 0	,
Score	0	e/170	•	•	0.176	0	0,176	0.22	e (0.209		• •	• •	0	• •		• •		e	• •	•	0				e	0 10	0	•	• •		• •	•			•	• •			•		0	•	• •	• •	e	•	0.22	e	• •	0 0	
Score	•	9 o	c	•	5.0	•	0.5 2	0.625	¢	0.59375	¢	• •	• •	•	• •	• •	• •		0		•	0				•	0	0	-		,	• •	•	• •	¢	¢	• •	• •	• •	•	• •	•	•	• •		0	•	0.625	•		• •	;
TOTFTL			•	÷	- 0	0		. •	•		•	• •		0	•	• •	• •		6			•		• •		•		40	•	• •	•	• •	•	• •	0	0	•		• •		• •	0	•	• •		0	0	v		o o		•
Score	9	0.02548454		1.5914E-05	0.00370988	. 0	0,02346933	0.02605332	•	0.03067588	8.5498E-09	0 G	9.3407E-08	1.5071E-08	6.8337E-08				0	1.8013E-05	0	• •	0	0 0	. 0	0	0	0	0			1.0823E-05	0		0	00	0		9 9	• •	0 0	0 0						0.03542495	00	• • •		,
Utility	0	0.49966670	. 0	0.00033859	0.07893361	• •	0.49934744	0.55432589	0	0.65267838	1.8191E-07	• •	1.9874E-06	3.2067E-07	1.454E-06			• •	0	0.00038326	0	• •	0	0 0	. 0	0	0	0	0		. 0	0.0002302	. 0	• •	•	• •	0	• •	0 ¢	• •	• •	0 0						0 7537223	• •	00		,
Score	0	0.7497/0095	Ģ	0.001448271	0.246670866	. 0	0.749550019	0.785802121	0	0 0.84360198	7.79468E-07	00	8.51563E-06	1.37403E-06	6.23012E-06	a a	0 0		0	0.001638965	0	ð		• •		ð	0	0	0	0 0	2	0.000985506			•	a	0	• •		•	a a	0	0 1		• •	0	0	0 895557029	Ð	00	00	,
INJCOST	0	0		06868	783358		1100000	1780315	0	0 2865000	480	0 0	4800	840	3600				0	92886	0	0		• •	00	0	0	0	•	• •	2	80635		0 0	0	0	•	• •	00		c 0	0	0			0	0	1840000	0	• •	00	,
Score	1656E-08	11217201.0	.2579E-08	08655451	11124836 6361E-05	00631332	09681059	11946423	(-9389E-09	0.0338E-09	1.05E-06	1002160441	113815163	02124585	0.08605488	L1235E-09	5.0064E-10	CD-20/CD-	6.2039E-08	17450033	8EL960Z11	0 5.5178E-08	•	L2803E-10	5.6941E-10	L2968E-09	5.8965E-10	0103E-06	3.0032E-08	3.4875-10	0	1.13920721	L9656E-05	5.3084E-09 5.9175E-06	3.482E-10	0.02329159	3.5434E-09	1.35356-09	0.0241811	1001176091	3.482E-10 1.3633E-07	9.3512E-08	3.4364E-06	2.1985E-08 1.00020183	1.9975E-06	01115668	0 3.03568896 ^	0.236	3.482E-10 0	8.2258E-10 3.5809E-05	0 2.8721E-08 2.8935E-05	
Utility	4.93887E-08	0.44818845	1.38047E-07	0.3667564 0	0.471391354 0	0.026751374 0	0.410214356 0	2.11854E-08 4 0.506204352 0	1.66904E-08 3	8.87223E-09 2 0.380151022 0	4.44899E-06	0.345781395	1941/0/07/0	0.090024769	0.364639309	1.76352E-09	Z.12136E-09	0.000299049 7.63809E-09	1.3576E-07	0.73940819) 95922321570	0 2.33806F.07		1.81369E-09	2.41277F-09	5.49501E-09	2.49852E-09	1,28114E-06	3.39117E-07	3.39424E-09	Ð	0.589861054	0.000125661	1.40186E-08	1.47542E-09	0.09869317	1.50144E-08	S.73518E-09	0.102462273	0.304071667	1.47542E-09 5.77665E-07	3.96239E-07	0 1.45611E-05	9.31585E-08	8.46411E-06	0.047274056	0.151224426		1.47542E-09	3.4855E-09	0 1.21698E-07 0.000172607	
Scare	7.59055E-08	0.547923345	2.12164E-07	0,464710931	0.570646872	0.040447849	0.509812066	3.25598E-08 0.603978793	2.56515E-08	1.36357E-08 0.478787547	6.83762E-06	0.442342122	0.67661718	0.13117606	0.462471362	2.71036E-09 1.38289E-08	3.26031E-09	0.000459523	2.0865E-87	0.80678002	0.60991122	LU-AYLL62 L	instance of	2.78745E-09	3.70818F_09	8.44528E-09	3.83997E-09	0.463657238 6.57966E-06	S.21189E-07	5.21661E-09	60-300/077	0.68059611	0.000193113	2.154515-08	2.26758E-09	0.143104792	2.30756E-08	1.33317E-06 8.81439E-09	0.148255967	1336625754	2.26758E-09 8.87812E-07	6,08979E-07	2.23788E-05	1.43175E-07	1.30084E-05	0.070608334	0.213039757	-	1.26758E-09	5.35686E-09 0.000233174	1.87037E-07 0.000128431	TT-00TAMA'A
DAMCOST	13107	3442062	54249	961696	140600	187616	150000	34101	31615	25209 981812	92290	948158	10000001	247853	941146	21211	12552	138471	54067	16632795	6604397	6000		01110	1353-1	20609	13807	89816	64067	16314	(INCONT	10202757	128956	29817	10000	325000	30522	74356	353670	900752	00001	1///39	105302	1997U	81+£66	163731	6001-65	31001121	00001	16540 131025	52876	969971
LUNIA	8	89	8	80	8	8 2	8	8 8	8	22	3	\$ 3	i ș	\$	8	8 8	8	3 3	80	8	8 8	5	5	5	78	8	g 1	2 28	82	65	2	23	13	z :	ತ	58	8	8 8	58 a	3 28	2 X	8	*	98 98	8 28	88	68	2	2 22	5 S	23	z

Score	0.3649	0.12127	0.18617	0.40227	0.06883	0.30278	0.062	11£90'0	2.1E-09	0.38697	1091'0	0.11096	0.16009	73011.0	107900	2.1E-09	0.00466	7.1E-05	0.00049	0.2176	0.13432	0.12748	0.00012	0	4.3E-10	5 8P-05	4.8E-05	5.9E-10	0.35308	0.00145	0.0007	0.00063	0 0 1846	0.062	0.01237	0.0002	5.3E-05	0	0.02471	900070	1.4E-09	0.02437	0.07507	0.00115	1.4E-07	0	0.00014	0.06217	2E-06	001116	0	0.03627	0.58892	0.06233	8.2E-10	0.0	0.00033 0.00054
Utility Score	0.0246883 0.00153068	0.7982684 0.04949264 0.0	0.0373536 0.00231593	0 0	0 0.0318906 0.00197721	0 0	•	0.0178912 0.00110926	0 0	0.089899 0.00557374	0.0039064 0.0002422	0.0014063 8.719E-05	0.0066408 0.00041173	0.2423708 0.01502699	11162000 80969F0'0	0	0.0752248 0.00466394	•	0 0 0	0 0	0.0043751 0.00027126	0.0002344 1.4532E-05 0 0	0.0018751 0.00011625	0 0	0 0 0	0.0009375 5.8127E.05	0.0007813 4.8439E-05	0 0	•	0.0234383 0.00145317	0.0112504 0.00069752	0.0101566 0.00062971	• •	9	0.199001 0.01233806	0.0032814 0.00020344	0.0008594 5.3283E-05	•	0.0228133 0.00141442	0.0096878 0.00060065	0 0	0.0031251 0.00019376 A A	0.0533854 0.00330989	0.0185944 0.00115285	0 0 0 000000 0 00000 0 00000 0 00000 0000	0 0	0.0022657 0.00014047	u u u u u u u u u u u u u u u u u u u	0 0			0.0093753 0.00058127	0	0.0053908 0.00033423) — (0.0053908 0.00033423 0.0082034 0.00050861
Score	0.0246875	0.798262881	0.037352412		0.031889504	ø	0 0	9 0.017890625	0	0.089896258	0.00390625	0.00140625	0.006640625	0.242364496	F526569F0 0	8	0.075222469	¢	0 0.075	0	0.004375	0.000234375	0.001875		0	525600000	0.00078125	•	• •	0.0234375	0.01125	0.01015625	-	•	0.198995589	0.00328125	0.000859375		0.0228125	0.0096875	0	0.003125	0.05338365	0.01859375	0 D 0028175		0.002265625	0.999432411	0	-	•	0.009375	0	0.005390625	00	2	0.005390625 0.008203125
MHRS		754	767	• •	316	0	474		408	• •	• •	229	0	9 <u>6</u> 0	ÞŞ	2 22	85	1732	8 9	0	845	0	001		95 '	, t	1	0	• •	* 2	1 2	a	c	300	14	130			1563	7 S	=	8 Ę	69	124	0 ę	Ŧ	647	238	° %	2	6	7609	0	0	120	634	69 105
Score	• •	0.062	,	0.062	0.062	0	0.062	0.062	0	0 000	0.062	0.062	•	0.062	0.062	-	e	÷	• •	• •	0.062	0	0		•			0	0	0	• •	e	c	0.062	¢	• •	. 0				0	0	• •	0	• •		•	50	•	e	•	0	0	0.062	ç c	5	0 0
Score	• •		- c		- 0	. 0			0	• •		-	•			. 0	Ð	•				o	0		c 0			0			• •	0	0	-	¢	• •			00		0	• •	. 0	0	• •	5	0		0	e	5	¢	c	-	•	2	00
rotloss	•	•	ə -		0 C	·	• •	- 0	0				-	• •		• •	0	0	• •		ė	-	0		0			0	- •			0	-		0	•			• •	• •	0			o	•	5	0	0 0	0	0	þ	0	-	0	0	Ð	00
Score .	0.00325	• •	0.0065	0	• •		0	0	•	0.0065	0,0065	8	0.00325	0	0		0	0			0	•	•		•		. ¢	6	0.00325	• •		•	0.0035	•	0	• •			• •		0	• •	• •	0	• •	•	0		. 0	¢	2	0	0.0195	•	•	2	• •
Score	0.083333	•	0.166667	0	• •		0	0	0	0.166667	0.166667	0	0.083333	0	0000170		0	0	• •		•	•	8		•			0	0.083333			0	111110		•	•	• •		• •		0	• •		0	• •	•	Ð	ə c	. 6	-	•	0	0.5	0	• •	•	
IdSVH	-		•	•				-		~ ~	- 14	ı	-		4														-				-	•																			-				
Score	0.00975	0	0.00975	0.00975	0 0	7 0.0065	0	0.00325	Ŷ	7 0.0065	0.00975	6	0	0	C/ KNN'N		e	0	• •	• •	0	7 0.0065	0			• •		0	0.00975			0	c		Ð	•			• •		0	0	0	0	• •	•	0			-	•	o	0.039	c	0 0	•	• •
PL Score	0.25		- ¥	0.25	• •	0.16666	0	0.08333	0	0.16666	0.16666	0	0	0	67'N		0	¢	• •	• •	¢	0.16666	•		• •			0	0.25			0	c	• •	0	•			-		0	0	0	0	• •	2	0	э с		c	•	0	-	0	0	5	00
IS OLSI	96	_		• ••		7		 6	_	39			325	'	- -		_	_			66	1 2			_			_	e .			_	_		_				_			•	•				_				_	-	5 60			-	
ore Sco	, 0.0							0 0 0	9 0	1.0			13333 0.00				9 0	0	•		1 0.0	0							5.			0									0	•		0	•	-					•	0	1 0.0		0	•	
FLSPL Sc	ŵ			n				vo		v 5			1 0,05								s								•0																								ŝ	•			
Score	00268145	0	0	01767936	• •	• •	•	0.02258602	0	0	0.000625		01503024	00109375	6/060010.4	• •	0	0	•	02307284	°	0	0		•			0	•	5 e	• •	0	SOURCECON	0	0	0	, 0		0		0	• •		0	•	>	0	• •	÷¢	¢	Þ	0		•		0	00
Scare	076612903 (0	0	505124451	• •		0	0	0	0	017857143		(429435484 (0.03125	0 2385306452		0	0		0 1659224012	0	0	0		•	• •		0	•			0	202091405	0		•			0	• •	Ģ	• •		0	0	5	0	0 0	0	c	Þ	0	0	c	0	0	0 0
AYREST	- 2	0	- <u>1</u>	50	0 0	00	•	642 0	0	• •	40		40	~ ;	2 2 4		•	•	•	480	•	0	0		0	5 0		¢	0			0		i o	. 0	0			0		0	0	• •	0	•	•	0			c	•	0	0		0	0	00
Score D	0		0	30-380E	0674461 A			01604604	. 6	¢ (• =		0	•			•	•	• •	1366352	5031E-07	¢	0		6			•	•			•	2202110	0		•			•		0	•		0	•	•	0	• •			9	8	0	0		0	• •
Score	0 0		0	0572663 1.0	4700559 0.0		0	11446873 0.0	. 0	•	• •		0				•	Ð	0	1.0 0112005	1504E-05 9.4	0						0				0	0 22612121	no710401	. 0	0				• •	•	•		0	•	-	•	• •		ſ	-	0	0	0		Ð	8 8
TOST	1 0		0	- 0.00 6 0.00	17 0.37		•	98 0.85 0.85			• •		•					0		- - 	3 53				6			0				Đ				0						•	5 6	0			0	• •			0	0	0		8	Ð	• •
DAY	43269		10105	1375				042			0 0		0							37016		0				•		0	0			0				0					. 0			. 0		5	0			,	•	0	0	. 0	0	0	0 0
۶. ۲	0769 0.003	-	1000	10.0 6555			-	- - -	-	-		-	-	~		_	-	-		900.0 11034			-					-					200.0 10020	10070 16966						_	-	_	_		_	_		_			_				_	_	• -
P Scat	0.08173	0	0	0.27083	• •		0	~ -	. 0	•	00		0	0	с с			0	0	0 15117	0	0	c	•	0	•		0	0	00		0				Ō	• •		0		0	0		. 0	0	0	0	00	00		¢	¢	0	. 0	0	0	
DAYHOS	• •	¢	- :	2 K	• •	• •	0	248	• •	•	•	• •	•	•	•		• •	0	•	⇒ 7	: -	0	e	•	0	c (• •	¢			0	ŝ	3 0		0	• •	,	0	• •	• •	0		÷	0	Þ	0	• •		•	c	•	c	• •	0	•	0 C

Appendix G: Average Criteria Scores

	Unweighted Utility	Weight	Weinhled	Unweighted Utility	Weight	Weighted	Unweinhted Utility	Weight	Weighted	Unweighted Utility	Weight	Weighted	Total
EVENT	Damage to Equip Prop	Demace to Equip Prop	Dumase to Favip Prop	Degree Injury/Illness	Degree Injury/Illness	Degree Injury/Illness	Environmental Damage	Environmental Damage	Environmental Damage	Other MSN Eactors	Other MSN Factors	Other MSN Factors	Severity
5 8	0.360201	0.236	255700530.0	0.352426	14.0	11/17/11/1	0.418635	711.0	201045120.0	17608210	1110	10111100	1225594
5	000000	0.230		11/1/7-0			CIPATELE O	211.0		afathe n	22.10	ALPAACTAD O	219415.0
5	0.327101	962.0	121661770.0	0.290369	14:0	CULC/POLLU		211.0	U.U.D	0.356215	11.0	0.063049971	0.204771
7 8		766.0	0.061566127	0.167778	140	072670401	1061110	0.117	0.013126649	0.264932	0.177	0.04689303	0.200410
R 2	765794.0	962.0	0.117418768	0.003946	0.47	0.002813599	0.00000	0.117	0	0.350000	0.177	0.06195	0.152152
8	0.30250	0.236	0.077938898	0.077371	0.47	0.03636423	0.186489	0.117	0.021819246	0.204000	0.177	0.036108056	0.172230
-	164252-0	0.236	0.059573761	0.105731	0.47	0.049693339	0.097609	11.9	0.011420286	0.206312	11.0	0.03651719	0.157205
8 :	244226.0	962.0	1012010/070			0	1000000.0	0.117	0.017764494		111	001156120	1305210
7 8	0.10/020	912.0	0.0451/0641	010110	0.47	0.039991649	0.092903	0.117	0.010869603	0.181975	1110	0.03220949	0.128941
• #	0.326333	0.236	0.077014534	0.003763	0.47	0.001768616	0,038265	0.117	0.004549545	0.257592	0.177	0.045593863	0.128927
F	0.180173	0.236	0.042520744	0.026710	0.47	0.012586783	001111:0	0.117	0.0129917	0.274456	0.177	0.048578693	0.116685
5	0.092820	0.236	0.021905527	0,00000	0.47	8.40622E-09	0.083325	0.117	0 009749025	0.205919	F 10	0.016447699	0.068102
*	0.204301	0.236	0.048215044	0000000	14.0	4.670345-08	CTTT20.0	1110	C/964770000	UC164640'0	1110	Accuration of	035870.0
8 2	0.063259	967.0	7/0690410:0	107650'0	19.0	900/10/10/0	00104000	0.117	0.005022225	0.091047	0.177	0.016115346	0.051256
5 5	0.142197	912.0	0.057158554	0.00000	0.47	0	0,00000	0.117	0	0.00000	0.177	0	0.057159
1 3	0.140842	0.236	0.033238699	0,002780	0.47	0.001306376	0.124988	0.117	0.014623538	0.037926	0 177	0.006712903	0.055882
76	0.103165	0.236	0.024347051	0.002408	0.47	0,001151707	0.006172	0.117	0.00072215	0.128345	177	0.022717023	0.048918
2	0.157940	0.236	161672760.0	0.004593	0.47	0.002158103	0.00000	0.117	0	0.034382	171.0	0.006025632	0.045518
2	FE160.0	0.236	0.023206799	0.002119	0.47	0.001026955	0.004629	0.117	0.000541613	0.104614	1/1.0	201010101010	2121200
A 6	147660'0	967.0	2305530000	0.041601	140	0.021475604	0.044440	0,117	846615000	0.036641	6.17	0.0064854	0.041793
: 2	0.151224	0.236	0.035688965	0.00000	0.47	a	0.00000	0.117	0	0.003201	0.177	0.000520801	0.036270
; -	0.00000	0.236	1.15979E-08	0.074890	0.47	105351550.0	0.00000	0.117	0	0.000640	0.177	0.000113335	0.035312
7	0,090048	0.236	0.021251341	0000000	0,47	o	0.00000	0.117	0	0.077074	0.177	0.01364217	0.034294
5	0.062474	0.236	0.016159841	0.00000	0.47	1.60327E-08	0.009258	0.17	0.001083225	0.078505	0.177	0.013893466	4011000
R :	0.062440	0.236	0.014735735	0.012513	14/0 14/0	0,006035201	0.016634	0.117 0.117	0.0079401187	0.043648	0.177	2444/0800/D	1010104
1	190711.0	912.0	FT1940820.0	0.00000	0.47	4.31015E-09	0.00000	0,117	0	0.002541	0.177	0.000449814	0.029499
; •	0.012003	0.236	0.00283274	0.005923	0.47	0.002783945	00000010	0.117	Ð	0.121822	0.177	0.021562474	0.027179
•	0.00000	0,236	3.863215-10	0°0040E	0.47	0.004421672	0.00000	0.117	0	0,108905	0 177	0.019276122	0.02369K
¥	0.014387	0.236	0.003395269	0.036467	0.47	45651710.0	0.00000	0.117	•	0.016997	0.177	0.003001391	0.023543
1	0,094401	962.0	7652/72700		14.0	0.005480716		110	0.000770001	101210.0	110	0.006567283	0.021594
5	3017000	967-0		10761010	170	01/1411000	0.0003876	0117	0.000451441	1499E0.0	0.177	0.006426726	0.021356
8 7	0.041762	0.236	0.011507759	0.00000	0.47	0	0.00000	0.117	0	0.055482	0.177	0.009820332	0.02132E
2	0.026559	0.236	0.006268041	0.007649	0.47	231565600'0	0.012962	0.117	0.001516515	0.032596	0.177	0.005769478	0.017149
11	0.034702	0.236	0.008189664	968600'0	0.47	0.004650944	0.002209	0.117	0.000328619	0.019981	0.177	0.00153665	0.016706
201	0.000220	0.236	5.18218E-05	0.00000	0.47	0	0.00000	0.117	0 000051710	0.027649	11.0	0.00001355	0.0153066
3 5	1230300	91.6 0	0.01923417	0.00000	0.47	0	0.002778	0,117	0.000324968	0.005684	0.177	0.001006116	0.013255
. 7	12102010	0.236	0.004749948	0.00000	0.47		0,00000	0.117	0	0.040241	0.177	0.007122714	6/8110/0
=	0.047274	0.236	0.011156677	0.00000	0.47	0	0,000000	0.117	•	0.00000	0.177	0	0.011157
8:	0.00000	0,236	7.112135-10	0.000000	0.47	•	0.00000	0.117		0.007434	0.177	0.000190374	56E20070
3 6	262120:0	0.236	0.005166476	000000	0.47		0.006944	0.117	0.000812419	0.006132	0.177	0.001085371	0.007064
•	0.000011	0.236	2.61593E-06	0.000926	0.47	864564000.0	0.00000	0.117	o	0.025156	6.17	0.004452676	0.004891
N :	0.00028	0.236	6.4952E-06	0.00000	0.47	•	0.024851	0.117	0.002907604	126600.0	0.177	AFLAFAIDD.D	2004000
2 P	1 57000 0	912.0	17912C7n0'n	0000000	170	-	0.000000	0.117		0.011856	0.177	0.002098568	0.002269
: F	0,00000	962.0	3.164096-09	0.00000	0.47	• •	0.00000	0.117	0	872010.0	0.177	0.001872313	0.001872
Ģ	0.004367	0.236	0.001030606	0.00000	0.47	0	0,00000	0.117	9	0.001943	0.177	0.000343934	0.001375
6	0.000476	0.236	0.000112307	0.00000	0.47	0	D,00000	0.117		0.003117	1110	0.000551719	0.000664
7	0.000061	0.236	1.4482E-05	0,00000	60	•	0,00000	0.17		0.0012579	11.10	190174000.0	0.0004.50
m 9	00000000	967.0	7.776425-09	0,00000,0	0.47		0,00000	0.117		0.001384	0.177	0.000245025	0.000245
: ≃	000000	0.236	1.02544E-09	0,00000	640	-	0.00000	0.117	0	0.001367	0.177	0.000242	0.000242
=	0.00000	962.0	2.601691-09	0.00000	0.47	•	0 00000	0.117	0	0.001030	0.177	0.000112307	0.000182
	0,00000	0.236	5.51782E-08	0.00000	0.47	0	0.00000	0.117	•	0.000656	0.177	0.00011616	0.000116
-	0.00000	0.236	2.04002E-09	0.00000	14-0 1	• :	0,00000	0.117		0.0000.0	71.0	CII-370961 6	7600000
6	0.000076	962.0	1.79047E-05	0.00000.0	19.0		0.000000	0.117		0.00000	0.177		0.00000
5	000000	0.236	\$17852E-08	0.00000	0.47	. 0	0,00000	0.117		0.00000	0.177	0	0.00000

.

Appendix H: Sensitivity Analysis Graphs

H.1 General: This Appendix presents graphs developed for sensitivity analysis of criteria weights for accidents and hazards. These graphs were referred to but not presented in Chapter Four.

H.2 Accident Sensitivity Over the Full Range of Weights





Figure H-2 Accident Sensitivity: Property Damage Weight





Figure H-4 Accident Sensitivity: Other Mission Factors Weight

H.3 Hazard Sensitivity Over the Full Range of Weights



Figure H-5 Hazard Sensitivity: Injury Weight



Figure H-6 Hazard Sensitivity: Property Damage Weight



Figure H-7 Hazard Sensitivity: Environment Weight



Figure H-8 Hazard Sensitivity: Other Mission Factors Weight

Ż

BIBLIOGRAPHY

ANACAPA Sciences Inc. Aviation System Safety Risk Management Tool Analysis. <u>Volume 1.</u> Contract MDA 903-92-D-0025 (ADA273503), Fort Rucker, AL, October 1993.

Army Safety Center Web Site, safety.army.mil/stats/index.html

Bunn, Derek. Applied Decision Analysis. New York: McGraw-Hill, 1984.

Chief of Staff of the Army Statement on Risk Management. 27 July 1995.

Clemen, Robert T. <u>Making Hard Decisions: An Introduction to Decision Analysis</u>, 2nd Edition. Duxbury Press

Department of the Army. <u>Accident Reporting and Records</u>. Army Regulation 385-40. Washington DC: HQDA, 1 November 1994

Department of the Army. <u>Army Accident Investigating and Reporting</u>. Department of the Army Pamphlet 385-40. Washington DC: HQDA, 1 November 1994

Department of the Army. <u>Military Environmental Protection</u>. FM 20-400, September 1997 (draft)

Department of the Army. Risk Management. FM 100-14, April 1998.

- Department of the Army. <u>System Safety Engineering and Management</u>. Army Regulation 385-16. Washington DC: HQDA, 3 May 1990
- DPL for Windows 95. Versions 3.2, 1.44 MB, disk. Computer Software. ADA Decision Systems Advanced Version (Student Edition), 1995

Federal Aviation Administration. <u>The FAA Plan for Research, Engineering and</u> <u>Development. Volume 1</u>. Washington DC, Jan 1989 (AD-A221263)

Grelk, Brian, J. <u>A CERCLA-Based Decision Support System for Environmental</u> <u>Remediation Strategy Selection</u>. Thesis, Masters of Science, Air Force Institute of Technology, AFIT/GOR/ENS/97M-10. 1997.

Hunsaker, Oren. Personal Interview, 1998.

Jackson, Jack A. Jr., Jones, Brian, L., Lehnkuhl, Lee, J., An Operational Analysis for 2025, Maxwell AFB, AL: Air University, 1996.

- Keeney, Ralph L. <u>Value Focused Thinking: A Path to Creative Decision Making</u>. Cambridge Mass; Harvard University Press, 1992.
- Kerschus, Christine, E. <u>A CERCLA-Based Decision Model to Support Remedy</u> <u>Selection for an Uncertain Volume of Contaminants at a DOE Facility</u>, Thesis, Masters of Science, Air Force Institute of Technology, AFIT/GEE/GOR/97D-01. 1997.
- Kinzey, R.P. <u>System Safety Risk Assessment Manual</u>. Naval Safety Center, Norfolk, VA, Jan 1989 (AD-A205904)
- Kirkwood, C. W. <u>Strategic Decision Making: Multiobjective Decision Analysis With</u> <u>Spreadsheets.</u> California: Wadsworth Publishing Co, 1997.

Kloeber, Jack, M. Personal Interview.

- Logical Decisions for Windows 95, Version 5.0, 1.44 MB, disk. Computer Software. Logical Decisions, April 1998
- Merkhofer, Miley W., Rarilee Conway, and Robert G. Anderson. "Multiattribute Utility Analysis as a Framework for Public Participation in Siting a Hazardous Waste Management Facility," <u>Environmental Management</u>, 21: 831-839 (1997).

Moon, Tony. Personal Interview. 1998.

Risk Management Information Database, http://www.rmis.army.mil

U.S. Army Safety Center. Director of Army Safety's FY98 Strategic Plan for the U.S. Army Safety Center. Fort Rucker, AL, April 1998.

Warren, Colonel. Personal Interview. 1998.

Wilson, Helene, A. <u>A Risk Analysis of Remediation Technologies for a DOE</u> <u>Facility</u>, Thesis, Masters of Science, Air Force Institute of Technology, AFIT/GOR/ENS/98M-28. 1997.

Vita

Captain Brian K. Sperling was born on 4 October 1966 in Passaic, New Jersey. He graduated from Roxbury High School in New Jersey in 1985. He entered undergraduate studies at the United States Military Academy, West Point, N.Y. where he graduated with a Bachelor of Science degree in Mechanical Engineering Systems in May 1989.

His first assignment was attending the Officer Basic Course and Initial Entry Rotary Wing Course at Fort Rucker, Al. He was married to his wife, JoAnna, in September of 1990. He served as an AH-1 platoon leader for a cavalry squadron in Germany where he was deployed to Operation Desert Storm/Desert Shield. Upon returning stateside, CPT Sperling attended the Aviation Advanced Course and the AH-64 Apache transition course. He next moved to Fort Campbell, KY and served for 18 months as an Assistant Operations Officer in the 101st Aviation Brigade. In September of 1995 he accepted the command of B Company, 2nd Battalion of the 101st Aviation Brigade. In July 1997, he entered the Graduate Operations Research program at the Air Force Institute of Technology. Upon graduation he will be assigned to the Operations Research Division of the U. S. Army Safety Center.

Permanent Address:

5 Cynthia Drive Succasunna, N.J. 07876

REPOR	т рос	CUMENTATION PAGE			Form Approved OMB No. 0704-0188
Public reporting burden for this collection of in gathering and maintaining the data needed, ar collection of information, including suggestion Davis Highway, Suite 1204, Arlington, VA 22	nformation nd complet is for redu 2202-430	n is estimated to average 1 hour per ating and reviewing the collection of ucing this burden, to Washington Hea 2, and to the Office of Management a	response, including the time for re- nformation. Send comments reg- dquarters Services, Directorate for nd Budget, Paperwork Reduction	aviewing insta arding this bu or Information Project (0704	ructions, searching existing data sources, rrden estimate or any other aspect of this Operations and Reports, 1215 Jefferson 4-0188}, Washington, DC 20503.
1. AGENCY USE ONLY (Leave bla	ank)	2. REPORT DATE	3. REPORT TYPE AN	D DATES	COVERED
		March 1999		Master'	s Thesis
A VALUE FOCUSED APPROA HAZARDS IN ARMY AVIATI	ACH 7 ION	O DETERMINING THE	TOP TEN	J. TONE	
6. AUTHOR(S) Brian K. Sperling, CPT, USA					
7. PERFORMING ORGANIZATION Air Force Institute of Technolog	I NAME gy	(S) AND ADDRESS(ES)		8. PERF REPC	ORMING ORGANIZATION ORT NUMBER
WPAFB, OH 45433-7765				A	FIT/GOR/ENS/99M-16
9. SPONSORING/MONITORING A U.S. Army Safety Center Fort Rucker, Alabama 36362	GENCY	(NAME(S) AND ADDRESS(E	(S)	10. SPO AGE	NSORING/MONITORING NCY REPORT NUMBER
11. SUPPLEMENTARY NOTES Advisor: LTC Jack M. Kloeber	Jr., D	SN 785-6565 ext 4336. e	mail jkloeber@afit.af.n	l	
12a. DISTRIBUTION AVAILABILITY Distribution Unlimited	Y STAT	EMENT		12b. DIS	
13. ABSTRACT (Maximum 200 we The United States Army Safety of This research utilizes value-focu model designed to aid decision-r doctrinal manual (FM 100-14) a determines and rank orders the s of the relationship between the p Understanding this relationship was used to make recommendati probability. The model provide and the values of the current cha different leadership levels or sit protect the military force as a w	ords) Center 1sed th makers and has severit probab- is instr ions to s decis ain of o uations thole.	c (USASC) is challenged w inking and multiattribute p is in their analysis process. been tailored specifically y and risk for 65 categorie ility and severity of accide rumental in developing risl the USASC for developing sion-makers with a decisio command at the Army Saf s. Use of this model can r	with identifying the most preference theory conce The value model is ba for aviation-related access of accidents and 24 e ents and the risk of indive reduction controls. H ag controls to decrease I n analysis methodology ety Center. Furthermo educe the inherent risks	t severe h pts to pro sed on th cidents an xisting ha vidual has fence, the nazard/ac that is co re the mo s in Army	hazards in Army Aviation. boduce a decision analysis e Army's Risk Management ad hazards. The model azards. A thorough analysis zards was conducted. e information in this report ecident severity and consistent with Army doctrine odel can be adjusted for y Aviation and therefore
14 SUBJECT TERMS					15. NUMBER OF PAGES
Safety, Aviation, Helicopters, V Multi Attribute Value Analysis,	Value F Decis	Focused Thinking, Value N ion Analysis	Model, Hazards, Accide	ents,	200 16. PRICE CODE
17. SECURITY CLASSIFICATION	18. SE	ECURITY CLASSIFICATION	19. SECURITY CLASSIF	ICATION	20. LIMITATION OF ABSTRACT
Unclassified		Unclassified	Unclassified		SAR

Standard Form 298 (Rev. 2-89) (EG) Prescribed by ANSI Std. 239.18 Designed using Perform Pro, WHS/DIOR, Oct 94